FRANCHISE DISCLOSURE DOCUMENT



LG AS Franchisor LLC ("stayAPT") A Delaware limited liability company 10801 Monroe Road, Suite 200 Matthews, NC 28105 Telephone: 980.368.8100

Email: franchising@stayAPT.com

www.stayAPT.com

A stayAPT® Suites hotel (singularly "Hotel," collectively "Hotels") is an extended-stay hotel offering temporary housing on a weekly or monthly rental basis. stayAPT offers franchisees the right to develop a Hotel under a Franchise Agreement. Franchisee will provide hotel services under the name "stayAPT® Suites." We also offer multi-unit development rights to develop and operate multiple Hotels under individual franchise agreements.

The total investment necessary to begin operation of a stayAPT® Suites Hotel with 59 to 111 suites is from \$7,529,900 to \$12,904,400, excluding real estate costs. This includes between \$45,000 and \$47,500 that must be paid to us or an affiliate. There are no incremental initial investment costs if you become a multi-unit developer, but you will pay us a development territory fee of \$40,000 for each of the Hotel franchises you are required to develop if you sign a Multi-Unit Development Agreement.

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no governmental agency has verified the information contained in this document.**

You may wish to receive your disclosure document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact our VP of Franchise Operations at 10801 Monroe Road, Suite 200, Matthews, North Carolina 28105, (980) 368-8100.

The terms of your contract will govern your franchise relationship. Don't rely on the disclosure document alone to understand your contract. Read all of your contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or an accountant.

Buying a franchise is complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this disclosure document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: May 7, 2025

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about
	outlet sales, costs, profits or losses. You
	should also try to obtain this information from
	others, like current and former franchisees.
	You can find their names and contact
	information in Item 20 or Exhibit B.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to
	the franchisor or at the franchisor's direction.
	Item 7 lists the initial investment to open.
	Item 8 describes the suppliers you must use.
Does the franchisor have the financial	Item 21 or Exhibit C includes financial
ability to provide support to my business?	statements. Review these statements
	carefully.
Is the franchise system stable, growing, or	Item 20 summarizes the recent history of the
shrinking?	number of company-owned and franchised
WHI I I I I ADTO	outlets.
Will my business be the only stayAPT®	Item 12 and the "territory" provisions in the
Suites business in my area?	franchise agreement describe whether the
	franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal	Items 3 and 4 tell you whether the franchisor
history?	or its management have been involved in
nistory.	material litigation or bankruptcy proceedings.
What's it like to be stayAPT® Suites	Item 20 or Exhibit B lists current and former
franchisee?	franchisees. You can contact them to ask
	about their experiences.
What else should I know?	These questions are only a few things you
	should look for. Review all 23 Items and all
	Exhibits in this disclosure document to better
	understand this franchise opportunity. See the
	table of contents.

What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

<u>Business model can change</u>. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions</u>. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions</u>. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor</u>. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

Renewal. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends</u>. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit A.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About *This* Franchise

Certain states require that the following risk(s) be highlighted:

- 1. Out-of-State Dispute Resolution. The franchise agreement requires you to resolve disputes with the franchisor by mediation, arbitration and/or litigation only in the state where the franchisor has its headquarters (currently North Carolina). Out-of-state mediation, arbitration, or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to mediate, arbitrate, or litigate with the franchisor in the state where the franchisor has its headquarters (currently North Carolina) than in your own state. Certain states may require other risks to be highlighted.
- 2. **Short Operating History**. The franchisor is at an early stage of development and has a limited operating history. The franchise is likely to be a riskier investment than a franchise in a system with a longer operating history.
- 3. <u>Mandatory Minimum Payments</u>. You must make minimum royalty, advertising, or other payments, regardless of your sales levels. Your inability to make these payments may result in termination of your franchise and loss of your investment.
- 4. <u>Financial Condition</u>. The franchisor's financial condition, as reflected in its financial statements (see Item 21), calls into question the franchisor's financial ability to provide services and support to you.
- 5. <u>Unopened Franchises</u>. "The franchisor has signed a significant number of franchise agreements with franchisees who have not yet opened their outlets. If other franchisees are experiencing delays in opening their outlets, you also may experience delays in opening your own outlet."

Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

MICHIGAN DISCLOSURE

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU:

- (a) A prohibition on the right of a franchisee to join an association of franchisees.
- (b) A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- (c) A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provisions of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (d) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value, at the time of expiration, of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchised business are not subject to compensation. This subsection applies only if: (i) the term of the franchise is less than 5 years; and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least 6 months advance notice of franchisor's intent not to renew the franchise.
- (e) A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- (f) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state.
- (g) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:

- (i) the failure of the proposed franchisee to meet the franchisor's then current reasonable qualifications or standards.
- (ii) the fact that the proposed transferee is a competitor of the franchisor or subfranchisor.
- (iii) the unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- (iv) the failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- (h) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (c).
- (i) a provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000.00, the franchisor must, at the request of the franchisee, arrange for the escrow of initial investment and other funds paid by the franchisee until the obligations to provide real estate, improvements, equipment, inventory, training, or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

ANY QUESTIONS REGARDING THIS NOTICE SHOULD BE DIRECTED TO THE OFFICE OF THE ATTORNEY GENERAL, CONSUMER PROTECTION DIVISION, ATTN: FRANCHISE SECTION, G. MENNEN WILLIAMS BUILDING, 1ST FLOOR, LANSING, MICHIGAN 48933, (517) 373-7117.

TABLE OF CONTENTS

ITEM		PAGE
1	THE FRANCHISOR, AND ANY PARENTS, PREDECESSORS, AND AFFILIATES	3
2	BUSINESS EXPERIENCE	7
3	LITIGATION	9
4	BANKRUPTCY	9
5	INITIAL FEES	10
6	OTHER FEES	11
7	ESTIMATED INITIAL INVESTMENT	18
8	RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	32
9	FRANCHISEE'S OBLIGATIONS	35
10	FINANCING	36
11	FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING	36
12	TERRITORY	
13	TRADEMARKS	47
14	PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION	48
15	OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISED BUSINESS	49
16	RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL	50
17	RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION	51
18	PUBLIC FIGURES	59
19	FINANCIAL PERFORMANCE REPRESENTATIONS	60
20	OUTLETS AND FRANCHISEE INFORMATION	65
21	FINANCIAL STATEMENTS	69
22	CONTRACTS	69
23	RECEIPTS	69

EXHIBITS

- List of Agents to Receive Service of Process List of Franchises A.
- B.
- C. **Financial Statements**

- D. Franchise Agreement
- E. Franchise Standards Manual Table of Contents
- F. State Addenda
- G. Form Release of Claims
- H. Comfort Letter
- I. Multi-Unit Development Agreement
- J. Data Transfer Agreement
- K. Franchise Application
- L. Receipts

ITEM 1 THE FRANCHISOR, AND ANY PARENTS, PREDECESSORS, AND AFFILIATES

Franchisor

LG AS Franchisor LLC, the Franchisor, is referred to in this disclosure document as "stayAPT," "we," "us," or any similar reference. "You," "your," or any similar reference means the person or entity that buys the franchise, including the entity's owners.

stayAPT's principal business address is 10801 Monroe Road, Suite 200, Matthews, North Carolina 28105 and its telephone number is (980) 368-8100. stayAPT is a Delaware limited liability company formed on October 29, 2018 and conducts business under its entity name, the name stayAPT Suites and also under the name Affordable Suites of AmericaTM. stayAPT does not conduct any business apart from franchising extended-stay properties under the two brands StayAPT Suites and Affordable Suites of America and undertakes related activities such as enter into vendor contracts for marketing services and FF&E related to the brands. stayAPT does not own or operate the franchise being offered, but as discussed below, its affiliates do.

stayAPT's agents for service of process are disclosed in Exhibit A to this disclosure document.

Franchisor's Predecessors and Affiliates

stayAPT's parent is LGAS Brand Parent LLC. Its principal address is 10801 Monroe Road, Suite 200, Matthews, North Carolina 28105.

Affordable Suites of America, Inc. ("ASA Inc.") is our predecessor. It operated the Affordable Suites of America franchise system from 1998 until we acquired substantially all of its assets in January 2019. Its address was 117 West Perry Road, Myrtle Beach, SC 29579.

Our affiliate LG AS TRS LLC owns and operates corporate owned stayAPT hotels and Affordable Suites of America hotels since January 2019. Its principal address is 10801 Monroe Road, Suite 200, Matthews, North Carolina 28105.

Our affiliate AS Manager LLC manages both corporate owned and franchisee owned stayAPT hotels and Affordable Suites of America hotels since January 2019. Its principal address is 10801 Monroe Road, Suite 200, Matthews, North Carolina 28105.

Our affiliate LG AS Holdco LLC is a supplier of certain operating supplies for franchisee owned stayAPT hotels and Affordable Suites of America hotels since June 2022. Its principal address is 10801 Monroe Road, Suite 200, Matthews, North Carolina 28105.

We started offering StayAPT Suites franchises in March 2020. We have offered Affordable Suites of America extended stay hotel franchises since we acquired substantially all of the assets of ASA Inc. on January 10, 2019. ASA Inc. offered Affordable Suites of America franchises since January 1998, first under the name Affordable Efficiency Suites, and then from March 2000 under the name Affordable Suites of America. As of December 31, 2024, 30

Affordable Suites of AmericaTM hotels have opened, 18 of which were franchised, and 12 owned by our affiliates.

We do not have any affiliates that offer franchises in any other line of business.

Franchise Offered

stayAPT® Suites hotels (singularly "Hotel," collectively "Hotels") using the Marks and Business System (as defined below). stayAPT offers a franchise agreement ("Franchise Agreement") for development and operation of an individual Hotel. A copy of the current Franchise Agreement is attached to this disclosure document as Exhibit D. We may grant other qualified individuals rights to develop and operate multiple Hotels under our multi-unit development agreement ("Multi-Unit Development Agreement"). A copy of the current Multi-Unit Development Agreement is attached to this disclosure document as Exhibit I.

stayAPT® Suites Hotels are extended-stay hotels offering temporary housing on a weekly and monthly basis with rates generally lower than conventional hotels. The Hotel rooms are fully furnished with complete kitchens, separate bedrooms, cable television and weekly housekeeping service. The Hotels have laundry facilities on the premises. Franchised Hotels will typically be constructed with between 59 and 111 rooms.

Your Hotel will be characterized by a business system that includes distinctive exterior and interior design, decor, color scheme, and furnishings; uniform standards, specifications and procedures for operations; quality and uniformity of products and services offered; and advertising and promotional programs (the "Business System"). stayAPT may modify the Business System from time to time and will provide any new information and techniques to you through the Franchise Standards Manual.

The Business System is identified by means of certain trademarks, trade names, service marks, logos and commercial symbols (the "Marks"), including the mark stayAPT® Suites.

stayAPT may make changes in the Business System, Marks, operational standards, and facility, equipment, and fixture requirements. You may be required to make additional investments in the Hotel periodically during the term of the Franchise Agreement if such changes are made or if the Hotel's equipment or facilities wear out or become obsolete, or for other reasons.

Compliance with Laws

You must comply with a number of federal, state and local laws which apply generally to the establishment and operation of hotel businesses. The laws are subject to ongoing change and include those affecting zoning and construction, public accommodations, accessibility by persons with disabilities, health and safety, privacy, and labor. Many of the laws vary from jurisdiction to jurisdiction. It is your responsibility to apprise yourself of and comply with all applicable laws. stayAPT strongly recommends that you retain an attorney to advise you as to local, state and

federal laws that may impact or relate to your Hotel. Examples of applicable laws include, but are not limited to, the following:

Health and Sanitation. Most states have regulations or statutes governing the lodging business and related services. Many state and local authorities require licensing of lodging businesses to assure compliance with health and sanitation codes. Health-related laws affect the use of linens, towels and glassware, among other things.

Facility Operations. Lodging facilities are subject to innkeepers' laws that (i) allow innkeepers to impose liens against the possessions of guests who do not pay their bills; (ii) limit the liability of innkeepers regarding guests' valuables; (iii) require posting of house rules and suite rates in each suite or near the registration area; (iv) may require registration of guests and proof of identity at check-in, and retention of records for a specified period of time; (v) limit the rights of innkeepers to refuse lodging to certain guests; and (vi) may limit the right of innkeepers to evict guests in certain circumstances. Applicable federal and state civil rights laws prohibit discrimination in hotels on the basis of the race, creed, color or national origin. Some states prohibit "overbooking" and require innkeepers to find other accommodations if the guest has paid a deposit. Some states and municipalities have also enacted laws and regulations governing non-smoking areas and guest rooms.

Persons with Disabilities. The Americans with Disabilities Act ("ADA") requires public accommodations, such as lodging facilities, to (i) offer facilities without discriminating against persons with disabilities; (ii) offer auxiliary aids and services to persons with hearing, vision or speech disabilities who would benefit from those services unless doing so would result in an undue burden or in a fundamental alteration in the nature of the goods or services; and (iii) remove barriers to mobility or communication to the extent readily achievable. The U.S. Department of Justice has published "Accessibility Guidelines" ("ADAAG") that prescribe a specified number of handicapped accessible rooms, assistive devices for hearing, speech, and visually impaired persons, and general standards of design applicable to all areas of facilities, among other things. Any newly constructed Hotel (first occupied after January 26, 1993) must comply with ADAAG and be readily accessible to and useable by persons with disabilities. Alterations of existing facilities also may need to comply with ADAAG to the maximum extent feasible. The ADAAG specifies the minimum suite design and layout criteria for handicapped accessible rooms. You and your contractors are responsible for ADA and ADAAG compliance.

Fire Safety. The Hotel and Motel Safety Act of 1990 (P.L. 101-391) amends (among other laws) the Federal Fire Prevention and Control Act of 1974 to facilitate the creation of a master list of places of public accommodation that meet certain fire safety guidelines (including requirements for the installation of smoke detectors and sprinkler systems). Other state and local fire and life safety codes may require maps, lighting systems and other safety measures unique to lodging facilities.

OSHA Regulations. Like many other businesses, lodging facilities are subject to Occupational Safety and Health Administration ("OSHA") standards. State and local occupational safety laws and rules may also apply.

In addition to these laws, laws of general application may have special interest to hotels. Consult your attorney for more information on these laws.

Competition/General Market

The hotel business is developed and highly competitive with respect to price, location and quality of service and is often affected by changes in consumer tastes, economic conditions, population and traffic patterns. You must anticipate competing with other hotels offering a wide range of comparably priced products and services and a wide variety of service formats. Though there are a few competitors that offer corporate housing and short term housing, the businesses with which you should expect to compete include, in general, national or regional franchise systems and other chains and independently owned local hotels, motels, and apartments located in the area of your Hotel which offer similar facilities to the public. Your business will also be affected by its location, the locations of competing hotels, apartments, and other lodging businesses, your financial and managerial capabilities, availability of labor, interest rates, changes in traffic patterns, demographic or cultural conditions and other factors. There is also active competition for management and service personnel, as well as for attractive commercial real estate sites suitable for hotels. While we do not target any particular group, stayAPT believes, that stayAPT® Suites hotels will have a particular appeal to the transferred or temporarily assigned professional. In the majority of locations, occupancy is not seasonal.

ITEM 2 BUSINESS EXPERIENCE

President and CEO: Gary DeLapp

Mr. DeLapp has been the President and CEO of stayAPT and its affiliates since we began operations in January 2019. He works out of Matthews, North Carolina.

Chief Marketing Officer: Jennifer Kearney

Ms. Kearney has been our Chief Marketing Officer since we started doing business in January 2019. She works out of Matthews, North Carolina.

Chief Financial Officer: Sharon Zinser

Ms. Zinser has been our Chief Financial Officer since January 2024. Before that, she was the Chief Financial Officer of Greenlight Communities in Scottsdale, Arizona from September 2021 to December 2023, and Chief Financial Officer of Ray Fogg Corporation in Independence, Ohio from December 2019 to May 2021. She now works out of Matthews, North Carolina.

Chief Operations Officer: Brad Dew

Mr. Dew has been our Chief Operations Officer since January 2025. Before that, he was the Regional Director of Operations of Aimbridge Hospitality in Plano, Texas from April 2011 to January 2025. Mr. Dew now works out of Matthews, North Carolina.

Vice President, Franchise Operations: Adam Binder

Mr. Binder has been our Vice President of Franchise Operations since February 2021. Before that, he was the Director of National Accounts for Extended Stay America in Charlotte, North Carolina from January 2017 to January 2021. Mr. Binder works out of Matthews, North Carolina.

Vice President of Operations: Chris Rutherford

Mr. Rutherford has been our Vice President of Operations since January 2021. He was Chief Operating Officer for the Affordable Suites of America brand from April 2020 to December 2020. Mr. Rutherford works out of Matthews, North Carolina.

Regional Vice President, Franchise Development: Joseph Bickelmann

Mr. Bickelmann has been our Regional Vice President of Franchise Development since November 2022. Before that, he served as Regional Director for Best Western Hotels in Phoenix, Arizona from October 2021 to October 2022, Regional Director for Red Roof Inn in Toledo, Ohio from September 2020 to October 2021, and Regional Director for Best Western

Hotels in Phoenix, Arizona from April 2019 to September 2020. Mr. Bickelmann now works out of Matthews, North Carolina.

Director, Architecture & Design: William Wildes

Mr. Wildes has been our Director of Architecture & Design since June 2024. Before that, he was the Senior Director, Architecture & Design for Radisson Hotel Group Americas in St. Louis Park, Minnesota from November 2020 to April 2023, and Director of Design for Hyatt Hotels in Chicago, Illinois from June 2016 to April 2020. Mr. Wildes now works out of Matthews, North Carolina.

ITEM 3 LITIGATION

No litigation is required to be disclosed in this Item.

ITEM 4 BANKRUPTCY

In re: Binder, No. 17-31301, Western District of North Carolina, Charlotte Division. Prior to obtaining employment with stayAPT, our Director of Franchise Operations, Adam Binder, filed a personal bankruptcy petition under the liquidation provisions of Chapter 7 of the U.S. Bankruptcy Code on August 3, 2017, which was discharged on November 13, 2017. Mr. Binder's address and principal place of business is 10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105.

No other bankruptcy is required to be disclosed in this Item.

ITEM 5 INITIAL FEES

Franchise Agreement for Single Hotel

Initial Franchise Fee: You are required to pay stayAPT an Initial Franchise Fee of \$40,000 upon the execution of your Franchise Agreement. The Initial Franchise Fee is earned upon receipt and is nonrefundable.

Construction of the Hotel

Extension Fee: You must begin construction within 12 months of the Franchise Agreement effective date and open your Hotel within 24 months of the Franchise Agreement effective date. If you believe you will not be able to do so, you may request up to 4 30-day extensions. You must pay us a \$5,000 fee for each 30-day extension you request. If we do not approve your extension request, we will refund you the fee.

Site Visits: We will conduct 3 site visits to confirm your construction progress, and you must pay us a fee of \$2,500 per visit. If additional site visits are requested by you or required by us, you must pay us a fee of \$2,500 for each additional visit.

Initial Training

We will conduct an initial training program for you (or your owners, if you are an entity) and your Resident Manager at a cost of \$2,500 per attendee. You must also pay for the salaries, fringe benefits, travel costs and expenses, and related costs for each of your attendees. If you use a third-party management company, a representative of the management company will also have to attend and successfully complete the training.

Multi-Unit Development Agreement

If you are buying multi-unit development rights to develop an agreed number of Hotels, you will pay us a non-refundable multi-unit fee ("Multi-Unit Development Fee"). The Multi-Unit Development Fee depends on the total number of Hotels you are agreeing to open. It is \$40,000 per Hotel. If you pay a Multi-Unit Development Fee, you will not pay the Initial Franchise Fee for each Franchise Agreement you sign.

ITEM 6 OTHER FEES

Type of Fee	Amount (Note 1)	Due Date	Remarks
Royalty Fees	Greater of \$2,500 per month or 5% of your Gross Room Revenues.	On or before the 10th day of the month for the preceding month	stayAPT may require you to pay these fees by direct bank transfer to stayAPT's bank account, and require you to report Gross Room Revenues (Note 2)
Marketing Fees	Up to 3% of your Gross Room Revenues per month. Currently 2% of your Gross Room Revenues per month.	On or before the 10th day of the month for the preceding month	stayAPT may require you to pay these fees by direct bank transfer to stayAPT's bank account. (Note 3)
Independent Marketing Project Fee	Pro rata portion of incurred cost per marketing project.	As Incurred	In addition to or instead of the Marketing Fees, we may charge on a pro rata basis on marketing projects. This fee is intended to cover extraordinary marketing expenses not covered by the regular Marketing Fee.
Transfer Fee	\$10,000 or expenses if higher (Franchise Agreement)	Prior to consummation of transfer	(Note 4)
Initial Training for New Owner or New Resident Manager	\$2,500 per person	Before training	If you hire a new Resident Manager, you will pay the then-current training fee for training the new Resident Manager. Also, if the ownership in the franchisee is transferred, we will charge this fee for the new owner who must complete the Initial Training.
Additional Training Fee	Will vary under the circumstances	Before training	We may require you, your owners, and the Resident Manager to participate in additional training during the term of the Franchise Agreement. The fee for the training will be set to reimburse us for our expenses for organizing the training.

Type of Fee	Amount (Note 1)	Due Date	Remarks
National Meetings; Regional Seminars	The then-current meeting/seminar fee	As incurred	You must pay all travel and living expenses while attending these meetings. You will pay those charges directly to third parties.
Audit and Recordkeeping Costs	Will vary under the circumstances	After inspection or audit	Audits and inspections generally will be at our expense. However, if an audit is made necessary by your failure to furnish reports, financial statements, tax returns or schedules as required under the Franchise Agreement, or if any audit or inspection reveals that you have understated or underreported Gross Room Revenues, Royalty Fees, Marketing Fees or other amounts owed to us intentionally or by an amount greater than 2%, in addition to the amounts owed to stayAPT, you must reimburse stayAPT for the cost and out-of-pocket expenses of the inspection or audit.
Post-Opening Authorization to Open ("ATO") Work Inspection	Currently, \$910 per inspector per day, plus travel, food and lodging for the inspector(s)	On demand	Payable if we must make additional inspections of your hotel after it opens to ensure that you have completed additional work required in connection with the ATO.
Replacement Site Application Fee	\$5,000	As Incurred	If your initial site application is rejected, you will need to pay the fee for any subsequent site applications submitted to stayAPT.
Supplier/Product Testing Fee	Current fee or costs, if higher. Current fee is \$2,500.	As Incurred	Franchisee must reimburse Franchisor for its costs of review and testing the alternate product or service.
Interest Expenses	Will vary under the circumstances	When due	All amounts owed to stayAPT will bear interest at the maximum legal rate allowable, not to exceed 18%.
Property Improvement Plan Fee	\$5,000	As Incurred	(Note 5)
Re-inspection Fee Relicense/ Change of Ownership Fee	\$2,500 The then-current Initial Franchise Fee	As Incurred Upon execution of a successor Franchise Agreement	(Note 6) (Note 7)

Type of Fee	Amount (Note 1)	Due Date	Remarks
Technology, Internet, Email and Website Fee	Initial Website set-up fee is \$1,800 to \$2,500 for your microsite on our site Website maintenance and domain hosting is \$300-\$600 per year per location	As incurred	You will pay us this fee for our assistance in setting up your microsite on our website and your email accounts. Because of the rapid technology changes today we also reserve the right to charge a reasonable fee to off-set our cost in potential future technology support services that we will provide to you.
Kiosk	The then-current fee	As Incurred	We expect to implement kiosks in the lobby area for all stayAPT Suites hotels, which will allow guests to independently check-in and check-out, access mobile or printed keys, and obtain guest services support through a company agent. We anticipate that there will be a one-time cost to install and set up the kiosk, and then an ongoing maintenance and support fee. As of the issuance date of this FDD, the fees have not yet been established, but we expect the fee will be set so as to compensate us for expenses and costs we incur in connection with the roll-out and support of this program, including our overhead costs.
CoStar Subscription / STR Reports	Currently, \$1,000 to \$3,000 per hotel per year, based on the number of guest rooms at your hotel	As Incurred	Franchisee must reimburse Franchisor for its costs to provide the CoStar platform's Benchmark subscription, which provides property and portfolio level performance analytics, including weekly, monthly, and annual STR Reports for the Hotel.

Type of Fee	Amount (Note 1)	Due Date	Remarks
Reservation Fee	The then-current reservation fee and the cost of installing such reservation system	As Incurred	Reservations are currently processed through the Hotel's property management system, presently the Jonas Chorum PMS.
			If we implement a centralized reservation system or call center in the future, you will pay stayAPT the then-current reservation fee for each reservation delivered to you, including any pass-through costs such as service fees, travel agent commissions, and third-party distribution channel fees. In addition, you will, at your expense agree to purchase, install and maintain the reservation system equipment including all required computer hardware and software. As of the issuance date of this FDD the reservation fee has not been set.
Voice-of-Guest Program	Approximately \$1,500 per hotel, per year	As incurred	Our third-party voice-of-guest program allows hotels to solicit guest feedback. In addition to the annual subscription fee, hotels will be charged a one-time set-up fee of approximately \$150 per hotel.
Administrative Assistance Fee	A reasonable fee and stayAPT's costs incurred in connection with the services.	As Incurred	If you request our help with additions and administrative services, including assistance with finance closings or other transactions related to the Hotels, or negotiating agreements relative to the Hotel we will charge you this fee. The fee is intended to cover our expenses for providing the requested service.
Tax Reimbursement Fee	Will vary under the circumstances	As Incurred	You will indemnify us for any such taxes that are assessed or levied against stayAPT which arise or result from the Hotel.

Type of Fee	Amount (Note 1)	Due Date	Remarks
Insurance Procurement Fee	Will vary under the circumstances	As Incurred	If you do not maintain the required insurance coverage for the Hotel, we may obtain the insurance at your cost and expense. You will also be required to pay us a reasonable fee for our expenses in obtaining the insurance.
Indemnification	Will vary under the circumstances	As Incurred	You will indemnify us for any and all claims for any damages arising out of your operation of the Hotel under the Franchise Agreement.
De-Identification Fee	Will vary under the circumstances	As Incurred	Following termination or expiration, stayAPT or its agents may enter the Hotel to make such alterations to de-identify the Hotel at your sole risk and expense and without any responsibility for stayAPT for any actual or consequential damages to the property.
Temporary Management Fee	Will vary under the circumstances	As Incurred	In the event of death or incapacitation of you or a principal owner and the Hotel not being managed by an approved management company or if we determine it is not being properly managed, then we have the right to appoint a property management company and charge a reasonable management fee. There is no time limit for the period of time that the temporary manager will manage the property, but it is intended as a temporary solution.
Enforcement Fee	Will vary under the circumstances	As Incurred	In the event stayAPT incurs any expense to enforce an obligation against you, then we shall be entitled to recover all such expenses, legal fees and other related costs.

Type of Fee	Amount (Note 1)	Due Date	Remarks
Liquidated Damages	Average monthly Royalty Fees times the lesser of 36 months or the number of months remaining under the Franchise Agreement	At Termination	(Note 8)
	\$1,500 per guest suite (Termination prior to opening)		
	An amount equal to 2 times the annual salary of the employee involved plus all costs and attorneys' fees (Non-Solicitation)		

Notes:

- (1) Except where otherwise noted, all fees are imposed and collected by, and payable to stayAPT, are nonrefundable and are uniformly imposed.
- (2) "Gross Room Revenues" means the total revenues and receipts resulting from the rental of guest rooms, whether made for cash or credit and whether or not collected by the franchisee. "Gross Room Revenues" do not include charges for food and beverages, telephone, laundry, entertainment, security deposits or applicable sales tax. A report of your Gross Room Revenues is due to us by fax or email on or before the 4th day of each month for the preceding month. If you signed a Multi-Unit Development Agreement, you will sign the form of Franchise Agreement offered to Franchisees at the time you sign such Franchise Agreement and the calculation of Gross Room Revenues and your Royalty percentage may be different.
- (3) stayAPT will use the Marketing Fees for research, development, and preparation of national, regional, point of sale, and local direct sales advertising, marketing strategy materials, and the general promotion of Brand Hotels. stayAPT may increase the Marketing Fee upon not less than 30 days' notice, provided that Marketing Fees will not exceed 3% of monthly Gross Room Revenues over the term of the Franchise Agreement.
- (4) You must obtain stayAPT's written consent and pay a \$10,000 transfer fee per Hotel to stayAPT before you transfer any interest in the Franchise Agreement or Hotel. stayAPT does not charge a transfer fee for transfers to your spouse or child or to an entity which you control. See Item 17 for additional information regarding transfer requirements.
- (5) A Property Improvement Plan review ("PIP") may be elected by you or required by us for properties that have sustained evident and substantial wear and tear. During the PIP

- process, areas and items no longer meeting brand standards are identified, as well as items that need repair or cleaning to "like new condition," or replacement.
- (6) If your Hotel fails any PIP inspection or quality assurance review, you will be charged a fine. We typically re-inspect your Hotel after 60 to 90 days. If you haven't cured the default by then we can assess a new fine.
- (7) The Franchise Agreement is not renewable and generally not assignable. If we receive a franchise application for a change of ownership of the Hotel or a relicense upon expiration of the Franchise Agreement, we will review the application in accordance with our then-current standards, criteria and requirements. If the application is approved, a new Franchise Agreement will be executed on our then-current form and the then-current Initial Franchise Fee must be paid.
- (8) If the Franchise Agreement is terminated (other than by stayAPT's breach) you must pay stayAPT liquidated damages equal to the average Royalty Fees due during the past 12 months times the lesser of 36 months or the number of months remaining on the term of the Franchise Agreement. If Royalty Fees have been due for less than 12 months (i.e., because your Hotel has been open for less than 12 months), you must pay stayAPT liquidated damages equal to the average Royalty Fees since the Hotel opened times 36 months. If the Franchise Agreement is terminated (other than by stayAPT's breach) before the Hotel opens, you must pay stayAPT liquidated damages equal to \$1,500 per guest suite. If you violate the Non-Solicitation provision of the Franchise Agreement, you will owe the former employer an amount equal to 2 times the annual salary of the employee involved, plus all costs and attorneys' fees incurred by the former employer in connection with such default.

ITEM 7 ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT

59 to 111 Suite Prototype New Build

Type of Expenditure	Amount (Note 1)	Method of Payment	When Due	To Whom Payment Is To Be Made
Initial Franchise Fee For One Hotel	\$40,000 (Note 2)	Lump sum	Upon the execution of the Franchise Agreement	stayAPT
Real Estate	(Note 3)	(Note 3)	(Note 3)	(Note 3)
Project Soft Costs (Architectural, Engineering, Legal and Other Professional Services)	\$400,000 to \$557,000 (Note 4)	As incurred	Before opening	Suppliers (including Engineers, Architects and Other Professionals)
Insurance Costs	\$50,000 to \$94,000 (Note 5)	As incurred	Before opening, except for liability insurance which you must obtain before you commence construction of your hotel	Suppliers
Building Construction	\$6,170,200 to \$10,427,800 (Note 6)	As incurred	Before opening or as negotiated	Suppliers, Landowners or Landlord
Furniture, Fixtures & Equipment	\$552,700 to \$1,177,800 (Note 7)	As incurred	As incurred	Suppliers

Type of Expenditure	Amount (Note 1)	Method of Payment	When Due	To Whom Payment Is To Be Made
Operating Supplies & Equipment and Opening Inventory	\$76,500 to \$144,000 (Note 8)	As incurred	As incurred	Suppliers
Technology	\$135,500 to \$216,300 (Note 9)	As incurred	As incurred	Suppliers
Pre-Opening Training	\$5,000 to \$7,500 (Note 10)	As incurred	Before opening	stayAPT
Pre-Opening Marketing and Advertising	\$10,000 to \$15,000 (Note 11)	As incurred	Before opening	Suppliers
Additional Funds (3 Months)	\$90,000 to \$225,000 (Note 12)	N/A	As incurred	Employees and Suppliers
Total for One Hotel (Note 14)	\$7,529,900 to \$12,904,400, excluding real estate purchase costs and site work (Note 13)			

Notes:

- (1) General: The following are the estimated minimum requirements for beginning operations for your Hotel as of the issuance date of this franchise disclosure document. The estimated minimum requirements will vary depending on factors such as your financial condition, the arrangements and business decisions you make, and the number of guest rooms at your hotel. As of the issuance date of this franchise disclosure document it is uncertain if new U.S. tariffs will be enacted and how such tariffs may impact your initial investment. If significant tariffs are imposed, the cost of constructing and opening a Hotel may increase materially. Except where otherwise noted, all fees that you pay to stayAPT are nonrefundable. Third-party lessors and suppliers will decide if payments to them are refundable. These cost ranges are based on a prototypical new construction hotel with 59 to 111 guest suites. There are no incremental initial investment costs if you become a multi-unit developer.
- (2) **Initial Franchise Fee:** See Item 5 for additional information on the Initial Franchise Fee. Neither we nor our affiliates offer direct or indirect financing to franchisees for any items.
- (3) **Real Estate:** Most franchisees have purchased, contracted to purchase, or leased a suitable real estate site for the development of a Brand Hotel before signing a Franchise

Agreement with us. If you do not already own suitable real estate, you must purchase or lease the land for your Hotel at a price and on terms that are agreeable to you. The Hotel will require approximately 2 to 3 acres of land. The cost of purchasing unimproved land depends on many independent variables including land costs, Hotel location and lot size, building size, site improvement costs, soil and environmental conditions, federal, state and local building codes and fees, impact fees, health department requirements, local labor, materials and interest costs, union labor requirements, inflation and other factors. Land costs vary widely and are subject to significant market fluctuations. As a result, we are unable to provide a reasonable or reliable estimate for your land purchase price. Your Hotel could be located in many different types of locations, but generally it is good to choose a location in a well-populated area with proximity to local features, such as business hubs, military bases, or healthcare facilities, and easy access to daily amenities, such as grocery stores, pharmacies, and dry cleaners. In addition to the cost of unimproved land you may have to pay to have site work done. The necessary site work and related costs will vary significantly depending both on the natural conditions of the site and on zoning regulations in your location, and unusual site conditions or lack of adequate utilities available at the building site or adjacent thereto may significantly increase the amount of site work required and the related cost. You should consult with commercial realtors, brokers, real estate attorneys, and other professional advisors who are familiar with the local conditions and markets in which you intend to develop your Hotel and any work needed to prepare your specific property for construction.

- (4) **Project Soft Costs:** This estimate includes costs for engineering and architectural design services, due diligence costs such as third-party costs to prepare environmental and geotechnical reports, surveys, and soil analyses, closing costs, legal fees, and other professional services. Franchisor will provide to Franchisee one set of Franchisor's current standard prototype layout (not construction drawings) and specifications for the size Hotel selected, and Franchisee will, at its expense, complete the detailed building plans and specifications for the construction of the Hotel. Construction and development cost estimates assume that there are no unusual site conditions and that adequate utilities are available at the building site or adjacent thereto. This amount also includes miscellaneous opening costs, including security deposits and any unforeseen incidental expenses related to facilities deficiencies or equipment repairs.
- (5) Insurance: You must comply with all insurance requirements of any lease, mortgage, or deed of trust covering the Hotel as well as all stayAPT insurance requirements in the Franchise Agreement (Article 8). Insurance requirements are subject to change periodically. Your actual insurance cost will vary widely based on many factors, including the location of your Hotel (for example, depending on if it is in an area prone to national disaster) and the number of Hotels operated by you. The estimate assumes that your Hotel is located inland. Insurance rates for larger hotels and hotels located on the coast will be substantially higher. Premiums are payable in the amounts and at the times required by your insurers. You should consult with your insurance provider to determine the amount of your premiums. We make no representation or warranty with respect to the adequacy or sufficiency of the insurance required, and it is your sole responsibility to determine whether additional insurance or higher limits are appropriate. The estimate includes both the insurance required during the construction of the Hotel, and during the

first 3 months of operation of the Hotel. Flood and earthquake insurance, which may be required in some regions, is not included in the estimate.

At a minimum, you should have the following insurance coverage (in compliance with the detailed requirements and required amounts set forth in the Franchise Agreement) when you operate the Hotel: (a) Comprehensive or commercial general liability insurance; (b) Business auto liability; (c) Umbrella excess liability insurance; (d) Insurance on the Hotel (including FF&E, boiler, and machinery); (e) If the Hotel is located in whole or in part within an area identified by the Federal Government as having a special flood hazard, the maximum flood insurance available under the National Flood Insurance Program; (f) If the Hotel is located in an "earthquake prone zone" as determined by the U.S. Geological Survey, insurance in an amount not less than the probable maximum loss, including business interruption coverage; (g) Builder's risk property insurance; (h) Business interruption insurance covering loss of profits and necessary continuing expenses; (i) Employee dishonesty coverage on all Hotel employees; and (j) Worker's compensation insurance as required by applicable law, employer's liability insurance, and such other insurance as may be required by law.

During the construction of the Hotel (and other significant construction at the Hotel) you must also require the contractor to carry the insurance required under the Franchise Agreement, including, builder's risk insurance, commercial general liability insurance, business automobile insurance, workers' compensation and employer's liability insurance, and excess or umbrella liability insurance.

All policies of insurance (a) shall be specifically endorsed to provide that the coverages will be primary and non-contributory to insurance carried by any additional insured, and (b) shall contain a waiver of any rights of subrogation against the indemnitees under the Franchise Agreement. In addition, the Commercial General Liability, Business Automobile Liability, and Excess or Umbrella Liability insurance must name as additional insured the parties described in the Franchise Agreement. You must periodically supply to us certificates of insurance evidencing the insurance coverage and which includes a statement that coverage may not be canceled, altered, or permitted to lapse or expire without 30 days' advance written notice to stayAPT. Revised certificates of insurance will be forwarded to us each time you make any changes in coverages or your insurance carrier and/or upon renewal of expired coverages.

If you do not obtain and maintain the insurance coverage required under the Franchise Agreement or as otherwise required by stayAPT, we may, at our option, buy such insurance for you and you will repay all costs and expenses for such insurance including a reasonable fee for any expenses we may incur in obtaining such insurance, which be due to us immediately on demand.

(6) **Building Construction:** This estimate reflects the cost to construct your Hotel, including materials and labor, television, cable, water and electric hook-ups, exterior signage, and the Hotel's door locking system and security system. This estimate does not include fees that may be imposed by governmental authorities, such as tap and impact fees, which vary by location and cannot reasonably be estimated; you should contact the local

authorities having jurisdiction over your site to determine what fees may be assessed and how those fees are calculated. Your costs will depend on many variables, including the size, location, and condition of your property and variations in material and labor costs due to market fluctuations and regional differences. In most markets, we expect building construction costs to be within the stated range. If your site has unusual requirements, such as hurricane or earthquake requirements, special foundation requirements, or union labor requirements, your costs may be higher. Though not required, you may want to maintain a contingency fund of 5% to 10% of your estimated building construction costs.

- (7) **FF&E:** This estimate reflects the cost to purchase and install the furniture, fixtures and equipment ("FF&E") required for your Hotel's guest suites and public spaces. FF&E includes items such as casegoods, softgoods, decorative fixtures and lighting, mattresses and bedframes, artwork, mirrors, fitness equipment, and interior signage. Your costs will depend on many variables, including the number of suites at your Hotel and the suppliers and materials you choose. This estimate assumes that you will use our approved vendors and approved products; if you select alternate vendors or higher-end materials, your costs will be higher. This estimate does not include sales taxes and freight charges, which vary by location, vendor, source, and market fluctuations.
- (8) **OS&E** and **Opening Inventory:** This estimate reflects the cost to purchase and install the operating supplies and equipment (OS&E) and opening inventory required for your Hotel. OS&E includes items such as towels, linens, dishware, cookware, utensils, and small appliances, such as coffee makers and alarm clocks. Opening Inventory includes items such as guest suite amenities and supplies, housekeeping and cleaning supplies, and paper goods, which are necessary on the day your Hotel opens for business.
- (9) **Technology:** This estimate reflects the cost to purchase and install all technology needed to operate your Hotel on a day-to-day basis and includes the in-room entertainment system (with televisions), computer system, and telephone system. We estimate the cost for your Hotel's technology, including high-speed Internet access, to be approximately \$1,900 to \$2,100 per suite, which includes all required hardware and software, as well as wiring, installation, network access and email setup, and pre-opening service fees. Your costs will vary depending on the size and location of your Hotel and the hardware, software, and suppliers you choose. Computer system requirements are more fully described in Item 11.

You will also be required to use our designated property management system, currently the Jonas Chorum PMS (the "PMS"). The initial, one-time cost to buy, install, and activate the PMS software is approximately \$4,000 to \$4,500, depending on the payment gateway provider you select. Jonas Chorum also charges a one-time fee of \$2,100 for system training. We estimate the ongoing PMS subscription and service costs to be approximately \$5,300 to \$9,900 per year, depending on the number of suites at your Hotel and the payment gateway provider you select. Our estimate includes the PMS initial costs, training costs, and subscription and service costs for the first year of the Hotel's operation, which are paid in advance. After the first year, subscription and

service charges will be billed monthly, at the annual rate. The PMS is more fully described in Item 11.

(10) **Pre-Opening Training:** At least one of your owners and your Resident Manager are required to attend stayAPT's initial training at a cost of \$2,500 per attendee. You are also responsible for all travel and living expenses actually incurred by your training attendees. If you use a third-party management company, a representative of the management company will also have to attend and successfully complete the training. See Item 11 for additional information on training.

We may offer additional systems training and on-site pre-opening support during the week of your scheduled hotel opening. There is no cost for the on-site support, but you must reimburse us for our representatives' travel and living expenses.

- (11) **Pre-Opening Marketing & Advertising:** You are required to spend at least \$10,000 to \$15,000 for pre-opening marketing and Grand Opening advertising, depending on the number of suites at your Hotel, but we recommend you spend more on pre-opening advertising.
- (12) Additional Funds: This amount estimates your initial pre-opening and start-up expenses for a period of 3 months after the opening of your Hotel, including employee wages and taxes. The amounts are estimates based on stayAPT's estimate of average costs and market conditions prevailing as of the date of this disclosure document and our predecessor, ASA Inc.'s, experience in the hotel business since 1998. stayAPT cannot guarantee that you will not have additional expenses starting the business. Your costs will depend on factors such as your management skills and business experience, local economic conditions, the prevailing wage rate, competition, and the sales level reached during the initial period.
- (13) **Total Amount:** The estimate is exclusive of real estate costs and, if applicable, local impact fees. The total is based upon our estimate of average costs in developing stayAPT Suites and Affordable Suites of America hotels, market conditions prevailing as of the date of this disclosure document, our predecessor, ASA Inc.'s experience in the hotel business since 1998, and the experience of our management team in developing different extended-stay hotel brands. While construction costs may always change over time, the availability, cost, and lead times on construction materials are especially susceptible to change at this time. This may affect the total cost of the construction of your Franchised Business. These estimates also do not reflect finance charges, interest, or debt service obligations.
- (14) **Multi-Unit Development:** The stated ranges apply to the development and opening of a single Hotel under a single Franchise Agreement. If you sign a Multi-Unit Development Agreement with us, you will be developing and opening several Hotels. The number of Hotels you will open depends on the agreement between you and us. As an example, included below is a table showing the estimated investment for opening 3 Hotels, not taking into consideration inflation. To determine your estimated initial investment, you can multiply the estimated initial investment in the table above for 1 Hotel by the number

of Hotels you are agreeing to open. For example, if your Multi-Unit Development Agreement states that you will open 5 Hotels, your total estimated investment for all 5 Hotels is approximately \$37,649,500 to \$64,522,500, excluding the cost of land. We anticipate that most Multi-Unit Development Agreements will be for between 2 and 10 hotels. No matter how many Hotels you are opening, when estimating your initial investment you should consider that, due to inflation and potential System changes that may be made between the issuance date of this Franchise Disclosure Document and the time when you are developing those subsequent Hotels, your actual investment may be different from this estimate. Depending on the location of your Hotels and your development schedule, you may be able to realize certain cost efficiencies that reduce your total investment if, for example, you are able to contract with the same general contractor for all of your projects, purchase all of your project materials, FF&E and OS&E at one time, or negotiate a multiple policy discount with your insurer.

You will pay a Multi-Unit Development Fee of \$40,000 for each of the agreed upon number of Hotels you intend to develop. For example, if your Multi-Unit Development Agreement states that you will open 5 Hotels, your Development Fee will be \$200,000 (\$40,000 x 5). If you pay the Multi-Unit Development Fee when you sign the Multi-Unit Development Agreement, you will not pay the Initial Franchise Fee when you sign a Franchise Agreement for each hotel site you develop, up to the agreed upon number of Hotels. Neither we nor our affiliates offer direct or indirect financing to franchisees for any items.

If you sign a Multi-Unit Development Agreement with us, you may incur additional costs to identify Hotel sites and initiate real estate acquisitions within each of your Development Territories, including realtor and broker fees, costs to prepare business plans and development budgets, and costs to obtain market feasibility studies. A reasonable estimate for commercial realtor and broker fees is 5% to 7% of the property purchase price for each Hotel development site you purchase. In some cases, the property seller may pay all or a portion of the broker fees, depending on the location of your property, prevailing market terms, and terms of your purchase agreement. A reasonable estimate for a market feasibility study is \$8,000 to \$10,000 for each market you choose to evaluate. Your costs will depend on many factors, including your negotiation skills, the terms of your realtor/broker agreements, the terms of your purchase agreements, whether you have in-house real estate development and finance teams or hire external service providers, geography, market conditions, your desired level of due diligence investigation, and other factors.

Land costs vary widely based on a number of factors, including geography, land size, and development conditions, and are subject to significant market fluctuations. As a result, we are unable to provide a reasonable or reliable estimate for your land purchase price. In addition to the cost of unimproved land you may have to pay to have site work done. The necessary site work and related costs will vary significantly depending both on the natural conditions of the site and on zoning regulation in your location and unusual site conditions or lack of adequate utilities available at the building site or adjacent thereto may significantly increase the amount of site work required and the related cost. You should consult with commercial realtors, brokers, attorneys, contractors, and other

professional advisors who are familiar with the local conditions and markets in which you intend to develop your Hotels and any site work needed to prepare your specific properties for construction.

MULTI-UNIT DEVELOPMENT AGREEMENT – COST OF OPENING 3 FRANCHISED LOCATIONS – NOT ADJUSTED FOR INFLATION

Type of Expenditure	Amount (Note 1)	Method of Payment	When Due	To Whom Payment Is To Be Made
Initial Franchise Fee For One Hotel	\$120,000 (Note 2)	Lump sum	Upon the execution of the Franchise Agreement	stayAPT
Real Estate	(Note 3)	(Note 3)	(Note 3)	(Note 3)
Project Soft Costs (Architectural, Engineering, Legal and Other Professional Services)	\$1,200,000 to \$1,671,000 (Note 4)	As incurred	Before opening	Suppliers (including Engineers, Architects and Other Professionals)
Insurance Costs	\$150,000 to \$282,000 (Note 5)	As incurred	Before opening, except for liability insurance which you must obtain before you commence construction of your hotel	Suppliers
Building Construction	\$18,510,600 to \$31,283,400 (Note 6)	As incurred	Before opening or as negotiated	Suppliers, Landowners or Landlord
Furniture, Fixtures & Equipment	\$1,658,100 to \$3,533,500 (Note 7)	As incurred	As incurred	Suppliers
Operating Supplies & Equipment and Opening Inventory	\$229,500 to \$432,000 (Note 8)	As incurred	As incurred	Suppliers

Type of	Amount	Method of	When Due	To Whom
Expenditure	(Note 1)	Payment		Payment Is To
				Be Made
Technology	\$406,500 to	As incurred	As incurred	Suppliers
	\$648,900			
	(Note 9)			
Pre-Opening	\$15,000 to	As incurred	Before opening	stayAPT
Training	\$22,500			
	(Note 10)			
Pre-Opening	\$30,000 to	As incurred	Before opening	Suppliers
Marketing and	\$45,000			
Advertising	(Note 11)			
Additional	\$270,000 to	N/A	As incurred	Employees and
Funds (3	\$675,000			Suppliers
Months)	(Note 12)			
Total	\$22,589,700 to			
	\$38,713,300,			
	excluding real			
	estate purchase			
	costs and site			
	work (Note 13)			

Notes:

(1) General: This table shows the estimated initial expenses that you will incur if you sign a Multi-Unit Development Agreement requiring you to sign Franchise Agreements to open a total of 3 hotels. Each Multi-Unit Development Agreement is negotiated with respect to the number of locations you agree to open. If you are required to open fewer than 3 locations, or more than 3, you will have to adjust the estimated initial investment. The easiest way of doing so is to multiply the information in the table showing the estimated initial investment for one location by the number of locations you are agreeing to open. No matter how many locations you are agreeing to open (even if it's 3), you should consider that inflation is likely to impact the cost of most line items in the table and the total cost. The longer the time over which your development obligation is spread, the greater is the likely impact of inflation.

The factors that underlie this estimate can vary considerably depending on a number of variables, and the investment you may make may be lesser or greater than the estimates given. Unless otherwise indicated, all the payments listed in this Item are not refundable. The payments to us or our affiliates are non-refundable. Neither we, nor our affiliates, offer financing for your initial investment.

(2) **Initial Franchise Fee:** You will pay a Multi-Unit Development Fee of \$40,000 for each of the agreed upon number of Hotels you intend to develop. For example, if your Multi-Unit Development Agreement states that you will open 5 Hotels, your Development Fee will be \$200,000 (\$40,000 x 5). If you pay the Multi-Unit Development

Fee when you sign the Multi-Unit Development Agreement, you will not pay the Initial Franchise Fee when you sign a Franchise Agreement for each hotel site you develop, up to the agreed upon number of Hotels. Neither we nor our affiliates offer direct or indirect financing to franchisees for any items. See Item 5 for additional information on the Initial Franchise Fee. Neither we nor our affiliates offer direct or indirect financing to franchisees for any items.

- Real Estate: Most franchisees have purchased, contracted to purchase, or leased a (3) suitable real estate site for the development of a Brand Hotel before signing a Franchise Agreement with us. If you do not already own suitable real estate, you must purchase or lease the land for your Hotel at a price and on terms that are agreeable to you. The Hotel will require approximately 2 to 3 acres of land. The cost of purchasing unimproved land depends on many independent variables including land costs, Hotel location and lot size, building size, site improvement costs, soil and environmental conditions, federal, state and local building codes and fees, impact fees, health department requirements, local labor, materials and interest costs, union labor requirements, inflation and other factors. Land costs vary widely and are subject to significant market fluctuations. As a result, we are unable to provide a reasonable or reliable estimate for your land purchase price. Your Hotel could be located in many different types of locations, but generally it is good to choose a location in a well-populated area with proximity to local features, such as business hubs, military bases, or healthcare facilities, and easy access to daily amenities, such as grocery stores, pharmacies, and dry cleaners. In addition to the cost of unimproved land you may have to pay to have site work done. The necessary site work and related costs will vary significantly depending both on the natural conditions of the site and on zoning regulations in your location, and unusual site conditions or lack of adequate utilities available at the building site or adjacent thereto may significantly increase the amount of site work required and the related cost. You should consult with commercial realtors, brokers, real estate attorneys, and other professional advisors who are familiar with the local conditions and markets in which you intend to develop your Hotel and any work needed to prepare your specific property for construction.
- (4) **Project Soft Costs:** This estimate includes costs for engineering and architectural design services, due diligence costs such as third-party costs to prepare environmental and geotechnical reports, surveys, and soil analyses, closing costs, legal fees, and other professional services. Franchisor will provide to Franchisee one set of Franchisor's current standard prototype layout (not construction drawings) and specifications for the size Hotel selected, and Franchisee will, at its expense, complete the detailed building plans and specifications for the construction of the Hotel. Construction and development cost estimates assume that there are no unusual site conditions and that adequate utilities are available at the building site or adjacent thereto. This amount also includes miscellaneous opening costs, including security deposits and any unforeseen incidental expenses related to facilities deficiencies or equipment repairs.
- (5) **Insurance:** You must comply with all insurance requirements of any lease, mortgage, or deed of trust covering the Hotels as well as all stayAPT insurance requirements in the Franchise Agreement (Article 8). Insurance requirements are subject to change periodically. Your actual insurance cost will vary widely based on many factors,

including the location of your Hotels (for example, depending on if it is in an area prone to national disaster) and the number of Hotels operated by you. The estimate assumes that your Hotels are located inland. Insurance rates for larger hotels and hotels located on the coast will be substantially higher. Premiums are payable in the amounts and at the times required by your insurers. You should consult with your insurance provider to determine the amount of your premiums. We make no representation or warranty with respect to the adequacy or sufficiency of the insurance required, and it is your sole responsibility to determine whether additional insurance or higher limits are appropriate. The estimate includes both the insurance required during the construction of the Hotels, and during the first 3 months of operation of the Hotels. Flood and earthquake insurance, which may be required in some regions, is not included in the estimate.

At a minimum, you should have the following insurance coverage (in compliance with the detailed requirements and required amounts set forth in the Franchise Agreement) when you operate the Hotels: (a) Comprehensive or commercial general liability insurance; (b) Business auto liability; (c) Umbrella excess liability insurance; (d) Insurance on the Hotel (including FF&E, boiler, and machinery); (e) If the Hotel is located in whole or in part within an area identified by the Federal Government as having a special flood hazard, the maximum flood insurance available under the National Flood Insurance Program; (f) If the Hotel is located in an "earthquake prone zone" as determined by the U.S. Geological Survey, insurance in an amount not less than the probable maximum loss, including business interruption coverage; (g) Builder's risk property insurance; (h) Business interruption insurance covering loss of profits and necessary continuing expenses; (i) Employee dishonesty coverage on all Hotel employees; and (j) Worker's compensation insurance as required by applicable law, employer's liability insurance, and such other insurance as may be required by law.

During the construction of the Hotels (and other significant construction at the Hotel) you must also require the contractor to carry the insurance required under the Franchise Agreement, including, builder's risk insurance, commercial general liability insurance, business automobile insurance, workers' compensation and employer's liability insurance, and excess or umbrella liability insurance.

All policies of insurance (a) shall be specifically endorsed to provide that the coverages will be primary and non-contributory to insurance carried by any additional insured, and (b) shall contain a waiver of any rights of subrogation against the indemnitees under the Franchise Agreement. In addition, the Commercial General Liability, Business Automobile Liability, and Excess or Umbrella Liability insurance must name as additional insured the parties described in the Franchise Agreement. You must periodically supply to us certificates of insurance evidencing the insurance coverage and which includes a statement that coverage may not be canceled, altered, or permitted to lapse or expire without 30 days' advance written notice to stayAPT. Revised certificates of insurance will be forwarded to us each time you make any changes in coverages or your insurance carrier and/or upon renewal of expired coverages.

If you do not obtain and maintain the insurance coverage required under the Franchise Agreement or as otherwise required by stayAPT, we may, at our option, buy such

insurance for you and you will repay all costs and expenses for such insurance including a reasonable fee for any expenses we may incur in obtaining such insurance, which be due to us immediately on demand.

- (6) **Building Construction:** This estimate reflects the cost to construct your Hotels, including materials and labor, television, cable, water and electric hook-ups, exterior signage, and the Hotels' door locking system and security system. This estimate does not include fees that may be imposed by governmental authorities, such as tap and impact fees, which vary by location and cannot reasonably be estimated; you should contact the local authorities having jurisdiction over your site to determine what fees may be assessed and how those fees are calculated. Your costs will depend on many variables, including the size, location, and condition of your property and variations in material and labor costs due to market fluctuations and regional differences. In most markets, we expect building construction costs to be within the stated range. If your site has unusual requirements, such as hurricane or earthquake requirements, special foundation requirements, or union labor requirements, your costs may be higher. Though not required, you may want to maintain a contingency fund of 5% to 10% of your estimated building construction costs.
- (7) **FF&E:** This estimate reflects the cost to purchase and install the furniture, fixtures and equipment ("FF&E") required for your Hotels' guest suites and public spaces. FF&E includes items such as casegoods, softgoods, decorative fixtures and lighting, mattresses and bedframes, artwork, mirrors, fitness equipment, and interior signage. Your costs will depend on many variables, including the number of suites at your Hotels and the suppliers and materials you choose. This estimate assumes that you will use our approved vendors and approved products; if you select alternate vendors or higher-end materials, your costs will be higher. This estimate does not include sales taxes and freight charges, which vary by location, vendor, source, and market fluctuations.
- (8) **OS&E** and **Opening Inventory:** This estimate reflects the cost to purchase and install the operating supplies and equipment (OS&E) and opening inventory required for your Hotels. OS&E includes items such as towels, linens, dishware, cookware, utensils, and small appliances, such as coffee makers and alarm clocks. Opening Inventory includes items such as guest suite amenities and supplies, housekeeping and cleaning supplies, and paper goods, which are necessary on the day your Hotel opens for business.
- (9) **Technology:** This estimate reflects the cost to purchase and install all technology needed to operate your Hotels on a day-to-day basis and includes the in-room entertainment system (with televisions), computer system, and telephone system. We estimate the cost for your Hotels' technology, including high-speed Internet access, to be approximately \$1,900 to \$2,100 per suite, which includes all required hardware and software, as well as wiring, installation, network access and email setup, and pre-opening service fees. Your costs will vary depending on the size and location of your Hotels and the hardware, software, and suppliers you choose. Computer system requirements are more fully described in Item 11.

You will also be required to use our designated property management system, currently the Jonas Chorum PMS (the "PMS"). The initial, one-time cost to buy, install, and activate the PMS software is approximately \$4,000 to \$4,500, depending on the payment gateway provider you select. Jonas Chorum also charges a one-time fee of \$2,100 for system training. We estimate the ongoing PMS subscription and service costs to be approximately \$5,300 to \$9,900 per year, depending on the number of suites at your Hotel and the payment gateway provider you select. Our estimate includes the PMS initial costs, training costs, and subscription and service costs for the first year of the Hotels' operation, which are paid in advance. After the first year, subscription and service charges will be billed monthly, at the annual rate. The PMS is more fully described in Item 11.

(10) **Pre-Opening Training:** At least one of your owners and your Resident Manager are required to attend stayAPT's initial training at a cost of \$2,500 per attendee. You are also responsible for all travel and living expenses actually incurred by your training attendees. If you use a third-party management company, a representative of the management company will also have to attend and successfully complete the training. See Item 11 for additional information on training.

We may offer additional systems training and on-site pre-opening support during the week of your scheduled hotel opening. There is no cost for the on-site support, but you must reimburse us for our representatives' travel and living expenses.

- (11) **Pre-Opening Marketing & Advertising:** You are required to spend at least \$10,000 to \$15,000 for pre-opening marketing and Grand Opening advertising, depending on the number of suites at your Hotels, but we recommend you spend more on pre-opening advertising.
- (12) Additional Funds: This amount estimates your initial pre-opening and start-up expenses for a period of 3 months after the opening of your Hotels, including employee wages and taxes. The amounts are estimates based on stayAPT's estimate of average costs and market conditions prevailing as of the date of this disclosure document and our predecessor, ASA Inc.'s, experience in the hotel business since 1998. stayAPT cannot guarantee that you will not have additional expenses starting the business. Your costs will depend on factors such as your management skills and business experience, local economic conditions, the prevailing wage rate, competition, and the sales level reached during the initial period.
- (13) **Total Amount:** The estimate is exclusive of real estate costs and, if applicable, local impact fees. The total is based upon our estimate of average costs in developing stayAPT Suites and Affordable Suites of America hotels, market conditions prevailing as of the date of this disclosure document, our predecessor, ASA Inc.'s experience in the hotel business since 1998, and the experience of our management team in developing different extended-stay hotel brands. While construction costs may always change over time, the availability, cost, and lead times on construction materials are especially susceptible to change at this time. This may affect the total cost of the construction of your Franchised

Business. These estimates also do not reflect finance charges, interest, or debt service obligations.
[Remainder of page intentionally left blank.]

ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

Except as stated below, neither we, nor any of our affiliates, currently require you to purchase or lease from us our affiliates, designees, or approved suppliers, any goods, services, supplies, FF&E, inventory or computer hardware or software. To help assure a uniform image and uniform quality of products and services in all stayAPT® Suites hotels, you must maintain and comply with stayAPT's quality standards. Although you are not required to purchase or lease real estate from stayAPT, stayAPT must consent to the location of your Hotel. You must obtain stayAPT's consent to all plans and specifications before beginning construction or renovation of your Hotel and you must improve and equip the building from which you operate the Hotel in accordance with stayAPT's then-current approved design specifications and standards. In addition, it is your responsibility to ensure that your building plans comply with the Americans with Disabilities Act and all other federal, state or local laws. The Franchise Agreement requires you to sell or use in connection with the Business System only those products and services that stayAPT has approved in writing.

Approved Suppliers and Approved Products: You must purchase all products, supplies, uniforms, signs, equipment, computer hardware and software, fixtures, furnishings, stationery, and advertising, marketing and sales promotion materials for which we have specifications and/or standards ("Approved Products") from suppliers approved by stayAPT in writing. stayAPT is not currently, but may be a supplier of Approved Products from time to time. In particular, we expect that our affiliate LG AS Holdco LLC may become a supplier for the fabrics and coverlets you must use for your Franchised Business. At this time, we expect that franchisees will have the option to purchase these items from us or from the manufacturers. If they are purchased from LG AS Holdco LLC we may charge a relatively small percentage upcharge to compensate us for the cost of keeping the products in inventory, processing and handling your order, and similar expenses. Upon request, stayAPT will assist you in procuring Approved Products from approved suppliers though we may charge an administrative fee for the assistance. We are currently working on developing a procurement program and may require franchisees to use the program in the future.

Alternative Suppliers and Alternative Products: stayAPT will periodically provide you with a list of suppliers for Approved Products, and we may have only one approved supplier for certain Approved Products. You may submit a written request to stayAPT for the right to purchase Approved Products from alternative vendors, or to purchase products of equal quality to Approved Products. stayAPT charges a fee for the review. The fee is currently the higher of \$2,500 or its costs and expenses for the evaluation of alternative suppliers. Likewise, if you request to use alternative products or services to Approved Products, then we charge a fee for the cost involved in evaluating the alternative goods or services. You will pay us our then current fee to evaluate such goods or services or reimburse us for our costs if higher. Currently, the fee is \$2,500. stayAPT may grant approval of the item or the supplier if: the item complies with the applicable specifications and standards of stayAPT (including product quality, price and availability requirements); the supplier's facilities are adequate to meet the needs of franchisees; and the supplier and its facilities are accessible to periodic evaluation by stayAPT. For product reviews, stayAPT may also require you to prepare a sample room or space, incorporating the alternative product. stayAPT does not have the obligation to review requests for alternatives

suppliers for products that display stayAPT's Marks. stayAPT will notify you of its decision within 45 days of your request. stayAPT will provide you with 30 days' notice of any supplier revocation. If we revoke the approval of an alternative supplier, product, or service, you will be notified of the revocation in a manner we deem appropriate. We do not issue particular specifications and standards to Franchisees for approving alternate suppliers, products or services.

Modifications of Standards and Specifications: stayAPT may modify the standards and specifications for Approved Products from time to time. The modified standards and specification will become effective when we deem appropriate. stayAPT does not make these specifications and standards generally available to franchisees or suppliers, other than as stayAPT determines it is needed. stayAPT reserves the right to negotiate purchase arrangements from suppliers or service providers for the benefit of franchisees. Occasionally stayAPT is able to negotiate with third-party suppliers, so that the suppliers offer their goods or services to franchisees at favorable or discount prices. stayAPT reserves the right to receive rebates, but does not currently receive, commissions or other payments from suppliers based on sales of Approved Products to franchisees and from other service providers. stayAPT does not participate in, or is not aware of, any purchasing or distribution cooperatives.

Computer Hardware and Software: In addition to purchasing Approved Products, you must purchase certain computer hardware and software and (if required by stayAPT) Internet, email and Website access, for use at or in connection with your Hotel, including our designated property management system, currently the Jonas Chorum PMS.

Minimum Insurance Coverage: You must also purchase and maintain liability insurance in an aggregate amount that stayAPT designates periodically, as described in Item 7, and you must purchase and maintain any other insurance required by any agreement related to the Hotel business or by law. You must furnish copies of all insurance policies to stayAPT. You may use only marketing and promotional materials which stayAPT has approved. We also require that you carry certain minimum insurance coverage. At a minimum, you should have the following insurance coverage (in compliance with the detailed requirements and required amounts set forth in the Franchise Agreement) when you operate the Hotel: (a) Comprehensive or commercial general liability insurance; (b) Business auto liability; (c) Umbrella excess liability insurance; (d) Insurance on the Hotel (including FF&E, boiler, and machinery); (e) If the Hotel is located in whole or in part within an area identified by the Federal Government as having a special flood hazard, the maximum flood insurance available under the National Flood Insurance Program; (f) If the Hotel is located in an "earthquake prone zone" as determined by the U.S. Geological Survey, insurance in an amount not less than the probable maximum loss, including business interruption coverage; (g) Builder's risk property insurance; (h) Business interruption insurance covering loss of profits and necessary continuing expenses; (i) Employee dishonesty coverage on all Hotel employees; and (j) Worker's compensation insurance as required by applicable law, employer's liability insurance, and such other insurance as may be required by law.

During the construction of the Hotel (and other significant construction at the Hotel) you must also require the contractor to carry the insurance required under the Franchise Agreement, including, builder's risk insurance, commercial general liability insurance, business automobile

insurance, workers' compensation and employer's liability insurance, and excess or umbrella liability insurance.

All policies of insurance (a) shall be specifically endorsed to provide that the coverages will be primary and non-contributory to insurance carried by any additional insured, and (b) shall contain a waiver of any rights of subrogation against the indemnitees under the Franchise Agreement. In addition, the Commercial General Liability, Business Automobile Liability, and Excess or Umbrella Liability insurance must name as additional insured the parties described in the Franchise Agreement

If you do not obtain and maintain the insurance coverage required under the Franchise Agreement or as otherwise required by stayAPT, we may, at our option, buy such insurance for you and you will repay all costs and expenses for such insurance including a reasonable fee for any expenses we may incur in obtaining such insurance, which be due to us immediately on demand.

Revenue, Percentage of Required Purchases, No Material Benefits from Approved Suppliers: In 2024 we did not derive any revenues from required purchases or leases of products and services by our franchisees. In 2024 none of our affiliates derived any revenue from required purchases or leases of products and services by our franchisees.

In some instances, due to higher quality or improved standards, the cost of the Approved Products and services to you may be higher than the cost of other products and services available on the market serving the same purpose or function. stayAPT estimates that your purchase of equipment, building materials, products, supplies, and marketing materials which meet stayAPT's specifications and standards will represent approximately 80% to 90% of the cost to establish the franchise business, not including the real estate, and 30% to 60% of the cost to operate the franchise business on an ongoing basis.

Our President, Gary DeLapp, owns an interest in AS Manager LLC, a hotel management company that is an approved supplier of hotel management services to our franchisees, and LG AS Holdco LLC, a supplier of certain operating supplies for franchisee owned stayAPT hotels. No other officers currently own an interest in any of our approved suppliers.

Although stayAPT does not provide material benefits to you based on your use of designated or approved sources, stayAPT will consider a number of factors when determining whether you might qualify for an additional franchise. Among the factors considered will be your compliance with your Franchise Agreement and support of our programs and policies which would include compliance with our requirements and specifications.

ITEM 9 FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the franchise and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this disclosure document.

	Obligation	Section in Agreement ⁽¹⁾	Disclosure Document Item
a.	Site selection and acquisition/lease	Articles 1.1, 12.1, 12.2 of FA; Article 1.1 of MUDA	Items 1, 7, 11, and 12
b.	Pre-opening purchases/lease	Articles 7.19, 12.3 of FA	Items 5, 7, and 8
c.	Site development and other pre- opening requirements	Article 7.4, 12 of FA; Article 1.1, 3.3 of MUDA	Items 7 and 11
d.	Initial and ongoing training	Article 10 of FA	Items 6 and 11
e.	Opening	Articles 5.1, 10.3, 12.4 of FA; Article 3.3 of MUDA	Items 5 and 11
f.	Fees	Articles 4, 5.1, 5.2, 7.19, 7.20, 7.21, 10.2, 10.4 of FA; Articles 3.1, 3.2 of MUDA	Items 5, 6, and 7
g.	Compliance with standards and policies/Operations Manual	Articles 7, 9 of FA	Items 11 and 16
h.	Trademarks and proprietary information	Articles 3, 7.1-7.5, 9, 15.1 of FA Articles 1, 2 and 3 of Confidentiality and Non-Compete Agreement	Items 13 and 14
i.	Restrictions on products/services offered	Articles 7.8, 7.9 of FA	Items 8, 11 and 16
j.	Warranty and customer service requirements	Article 7.8 of FA	Item 16
k.	Territorial development and sales quotas	Not Applicable	Item 12
1.	Ongoing product/service purchases	Articles 7.1, 7.4, 7.8, 7.9 of FA	Items 8 and 11

	Obligation	Section in Agreement ⁽¹⁾	Disclosure Document Item
m.	Maintenance, appearance and remodeling requirements	Articles 7.6, 7.7 of FA	Items 6 and 11
n.	Insurance	Article 8 of FA	Items 6 and 8
0.	Advertising	Article 5, 7.19 of FA	Items 6, 7, and 11
p.	Indemnification	Article 20.2 of FA	None
q.	Owner's participation/ management/staffing	Articles 7.13, 7.14 of FA	Items 11 and 15
r.	Records/reports	Articles 6.1-6.4 of FA	Item 6
S.	Inspections/audits	Articles 6.4, 6.5, 7.15 of FA	Item 6
t.	Transfer	Articles 19 of FA; Article 8 of MUDA	Items 6 and 17
u.	Renewal	Article 2.2 of FA	Items 6 and 17
v.	Post-termination obligations	Article 15 of FA	Item 17
w.	Non-competition covenants	Article 16 of FA Article 5 of Confidentiality and Non- Compete Agreement	Item 17
х.	Dispute resolution	Article 17 of FA; Article 9 of MUDA	Item 17

(1) "FA" refers to the Franchise Agreement. "MUDA" refers to the Area Development Agreement.

ITEM 10 FINANCING

stayAPT does not offer direct or indirect financing to franchisees. stayAPT does not guarantee your loans, mortgages, or other obligations.

ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING

Except as listed below, stayAPT is not required to provide you with any assistance.

Pre-opening Assistance. Before you operate your Hotel, stayAPT will:

- (1) Grant to you a license to operate the Hotel from a defined location (Section 1.1 of the Franchise Agreement). All sites must be approved by us. We will not assist you with finding the location or negotiation for the site but refer you to real estate brokers in the desired area, for whose services we will provide no guarantee (Section 12.1 of the Franchise Agreement). You have 60 days from the Franchise Agreement effective date to submit your proposed site and we will have 60 days after the required information is received from you to approve or disapprove, at our sole discretion, the site as the location for the Hotel. We may require that you submit a site feasibility study with the request for approval or your proposed site. Our approval of the site does not in any way guarantee that the site will become a profitable Hotel, independent of whether you submitted a site feasibility study with your approval request. The feasibility study is only a tool to help in the determination of whether a site is appropriate for the operation of a Hotel. You should review thoroughly your suggested site and all information available to you about it (including the feasibility study) before you submit it to us for review. There are many factors that will determine a Hotel's profitability such as your management skills and business experience, local economic conditions, and the prevailing wage rate. Our review of your proposed site is primarily focused on ensuring that the site will meet our brand standards. A discussion of the selection of your site for the Hotel appears below in this Item 11 under the caption "Site Selection."
- (2) Provide you with a set of prototype plans (not construction drawings) and specifications for the exterior and interior design, color scheme and motif of the Hotel (Article 12 of the Franchise Agreement). If you request a set of prototype plans in CAD format, you must sign a Data Transfer Agreement in the form attached as Exhibit J before the plans will be provided. Since all properties vary in size and configuration you must have more detailed plans or blueprints prepared at your expense to take into account the location and local ordinances. It is up to you to make sure that the premises conform to local ordinance and buildings codes. We do not have the obligation to assist you to obtain any required permits and it is your responsibility to construct and design the Hotel according to the specifications (as well as remodel). You are responsible and liable for construction management and certification services.
- (3) Provide you with a list of the standard fixtures, equipment, signs, supplies and initial inventory to be used in the Hotel as well as a list of Approved Products (Sections 7.8 and 7.9 of the Franchise Agreement). We will provide you with specifications and/or names of approved suppliers for the supply of all Approved Products. stayAPT is not required to provide you with equipment, signs, fixtures and opening inventory and we do not deliver or install any of them.

- (4) Provide training as described below in this Item 11 (Section 10.1 of the Franchise Agreement). Apart from the pre-opening training program, we are not required to help train your employees and we do not help you hire employees.
- (5) Loan to you one copy of our Franchise Standards Manual for the business containing mandatory and suggested specifications, standards and operating procedures for the business (Section 11.1 of the Franchise Agreement).
- (6) Provide you with opening assistance for a maximum of 5 days, which may or may not be consecutive, to assist you with training employees, implementing the Business System and evaluating initial business operations (Section 10.3 of the Franchise Agreement). However, stayAPT is not involved in the hiring or firing of employees or any additional or ongoing training of the employees outside of what is disclosed in this Item 11.
- (7) If you are a multi-unit developer, our only obligation is to help you define your Development Territories and assign them to you. (Section 1.1 of the Multi-Unit Development Agreement)

Site Selection. You must select the site for your Hotel and secure fee or leasehold title for the site and the hotel. stayAPT may consult with you regarding site selection, but you are ultimately responsible for locating and obtaining an acceptable site. The general site selection criteria you should consider includes traffic patterns, ease of ingress and egress, size and cost of the property, off-road area for loading and unloading, customer parking, demographic surveys, types of businesses in the general vicinity and other similar factors. We recommend that you obtain a feasibility study of your proposed site, and we have the right to require you to submit one, as part of the site selection process. If we and you are unable to agree on the site for the Hotel, the opening of your Hotel may be delayed. (Section 12.1 of the Franchise Agreement).

Development Time. You must secure fee or leasehold title for the site of your hotel and commence construction of your Hotel within 12 months after you sign the Franchise Agreement and your Hotel must be open for business within 24 months after you sign the Franchise Agreement or your Franchise Agreement may be terminated. (Section 13.1(C)(2) of the Franchise Agreement) The length of time between the signing of the Franchise Agreement and the opening of your Hotel may be approximately 18 to 24 months. Factors affecting this length of time usually include arranging financing, construction, materials supply, weather, subcontractor scheduling, successfully completing training, and other possible factors. You should not expend funds or make any other commitment in connection with the franchise and should not resign from existing employment, relocate or take any similar action until stayAPT's approval of the site which stayAPT will specifically communicate to you in writing. (Section 13.1 of the Franchise Agreement) If you are a multi-unit developer, you must open your Hotels within the time specified in the Development Schedule to your Multi-Unit Development Agreement (Section 3.3 of the MUDA)

Ongoing Assistance. During the operation of your Business, stayAPT will:

- (1) Make available to you the Business System for a stayAPT® hotel (Article 9 of the Franchise Agreement).
- (2) Furnish you from time to time with updated and revised material for our confidential Franchise Standards Manual (Section 9.3 of the Franchise Agreement).
- (3) Inspect your Hotel periodically to ensure that you are complying with the stayAPT® System requirements and quality standards (Section 7.15 of the Franchise Agreement).
- (4) Sponsor periodic meetings for all franchisees (Section 7.21 of the Franchise Agreement).
- (5) Upon your request, consult with you on technical matters such as housekeeping, front office, maintenance and bookkeeping (Section 11.1 of the Franchise Agreement).
- (6) Provide additional administrative services to you (for a reasonable fee), including, assistance with closings of financing transactions, negotiations of documents and agreements with your lenders or prospective lenders, and other project-based tasks, as described in the Franchise Agreement (Section 4.3 of the Franchise Agreement).
- (7) Recommend room pricing for your Hotel including setting maximum prices for goods and services (Section 5.5 of the Franchise Agreement).
- (8) Provide Property Improvement Reviews and Property Improvement Plans at your request (Section 7.15 of the Franchise Agreement).

Advertising and Marketing Programs.

Requirements for All Marketing Materials and Local Promotion: Though not required under the Franchise Agreement, we recommend that you promote your Hotel locally. You are allowed to conduct your own local advertising, provided that all advertising must: (1) be factual, ethical, and in good taste; (2) conform to such standards and requirements as stayAPT may specify from time to time; and (3) be submitted to us for approval before implementation. stayAPT will not unreasonably withhold approval of any advertising materials if the materials are factually accurate and current, conform to the highest standards of ethical marketing, and are in good condition and accurately depict stayAPT's Marks. You must at all times comply with our instructions regarding the use of advertising materials, including modifying or ceasing to use those materials, whether or not we previously prepared or approved the materials. (Section 5.4 of the Franchise Agreement)

Grand Opening Advertising: You are required to conduct certain advertising and public relations activities in connection with the opening of the Hotel. Your minimum spend will be between \$10,000 and \$15,000, depending on the number of suites in your Hotel. The marketing spend must be on pre-approved types of activities. (Section 5.1 of the Franchise Agreement).

Marketing Fund: stayAPT has the right to require you to contribute Marketing Fees in the amount of 2% of Gross Rooms Revenue per month to the Marketing Fund, as described in Item 6, for marketing and promotion programs to promote the Business System and products. stayAPT may increase the Marketing Fees upon not less than 30 days' notice to you, provided

that Marketing Fees will not exceed 3% over the term of your Franchise Agreement. Existing franchisees either already contribute at the same rate or will do so in the near future. Affiliate owned locations contribute to the Marketing Fund at 2% of Gross Room Revenues. stayAPT reserves the right to charge Marketing Fees on a pro rata basis on marketing projects. stayAPT administers the Marketing Fund. stayAPT formulates, develops, and conducts marketing and promotion programs in a form and media stayAPT determines to be appropriate. The materials may be disseminated via radio, television, print media, online, through social media, or via outdoor advertising. stayAPT reserves the right to use the Marketing Fees to reimburse stayAPT for all costs that it incurs related to the marketing and promotion programs, including for (a) developing, implementing, maintaining and disseminating advertising, marketing, public relations, and promotional campaigns and materials including, without limitation, through radio, television, newspapers, magazines and online; (b) developing, maintaining and updating Business System and Mark-specific websites and social media presence, placing listings in national, regional or online hotel directories; and (c) conducting market research, developing training, and educational programs with respect to marketing, sales, advertising, customer service and public relations. The Marketing Fund is accounted for separately from our other funds. We will not use the Marketing Fund to defray any of our general operating expenses, except for administering the Marketing Fund, including maintaining a national sales and marketing staff, accounting expenses and the actual cost of salaries and fringe benefits paid to stayAPT employees engaged in the administration of the Marketing Fund. (Sections 5.2 and 5.3 of the Franchise Agreement).

stayAPT has no obligation to spend any amount on marketing in the area where you are located. The intention is to spend the entire Marketing Fund fee collected, provided, however, that excess Marketing Fees not spent in any fiscal year will be carried over for future use. The Marketing Fees are not placed in a trust or escrow account, and are the property of stayAPT. You will have no property rights of any kind with respect to the Marketing Fees. stayAPT has no fiduciary obligations to you or other franchisees regarding the Marketing Fees. stayAPT is not required to contribute Marketing Fees; however, any stayAPT® Suites hotels that are owned and operated by stayAPT will contribute Marketing Fees at 2% of its Gross Room Revenues. stayAPT is not contractually required to contribute Marketing Fees to the Marketing Fund for their stayAPT® Suites hotels. stayAPT is not required to audit the Marketing Fund. stayAPT will provide to you once each year upon request an unaudited accounting of expenditures of the Marketing Fees for the previous fiscal year. In 2024 we spent approximately 5% of the Marketing Fees on production, 95% on media placement, and 0% on administrative expenses. stayAPT does not intend to use the Marketing Fund principally to solicit new franchise sales.

stayAPT does not require you to participate in a local or regional advertising cooperative.

Apart from mandatory participation in the Marketing Fund, we do not require you to participate in any other advertising fund.

Development of Marketing Materials and Franchisee Advisory Council: As of the date of this disclosure document, stayAPT formulates, develops and conducts the marketing and promotion program in-house. However, stayAPT reserves the right to use a national or regional advertising agency for the marketing and promotion programs. stayAPT has not established, but has the right to establish and appoint franchisees to a Franchisee Advisory Council ("FAC") to

serve in an advisory capacity on marketing and promotion programs. stayAPT has the right to form, change or dissolve the FAC. When formed, stayAPT will appoint the members of FAC. The intent is for FAC to be representative of the franchisees, both with respect to small and large franchisees, and geography.

Technology Systems.

Computer System: You must install a computer system meeting our standards, as modified from time to time in our sole discretion in response to business, operations and marketing conditions (Articles 7.4 and 7.6 of the Franchise Agreement). As of the date of this disclosure document, the computer system includes a desktop computer running a current version of Windows®, a printer, a software system designed for use in hotel front desks, and software that enables you to perform payroll, receivables, and weekly and monthly financial reports. We estimate that the total cost to purchase or lease the computer system will be between \$5,000 and \$12,000 depending on the number of suites at your Hotel and the hardware, systems, and suppliers you select. However, the costs vary due to market and region.

Property Management System: You will be required to install and use our designated property management system, currently the Jonas Chorum PMS (the "PMS"), which will manage the Hotel's operations and reservations on a day-to-day basis. The initial, one-time cost to buy, install and activate the PMS is approximately \$4,000 to \$4,500, depending on the payment gateway provider you select. Jonas Chorum also charges a one-time fee of \$2,100 for system training. The ongoing subscription and service cost for the PMS is approximately \$5,300 to \$9,900 per year, depending on the number of suites at your Hotel and the payment gateway provider you select. If your Hotel has more than 111 suites, your costs may be higher. The initial system costs and first year subscription and service costs for the PMS are charged in advance. After the first year, subscription and service fees for the PMS are billed monthly, at the annual rate. You will also be billed monthly, in arrears, for the actual distribution channel transaction fees incurred by your Hotel. Transaction fees currently range from \$2.50 to \$14.40 per net reservation, depending on the channel (e.g. TopSail, Direct Connect, GDS, SiteMinder/AirBnB) used to book a reservation through the PMS.

Voice-of-Guest System: We have implemented a third-party voice-of-guest program that allows stayAPT Suites hotels to generate more guest reviews, leverage local search engines, and enhance the guest experience. You are required to install and use our designated system. The initial, one-time set-up cost is approximately \$150 per hotel. The ongoing subscription cost for the voice-of-guest system will be approximately \$1,500 per hotel, per year.

Internet, Website, and Data Access: stayAPT has the right to require you to have Internet access (including high-speed Internet access in each guest suite) and other on-line communication capabilities, and to participate on its website. As of the date of this disclosure document, stayAPT has independent access to the information contained in your computer system through the PMS. There are no contractual limitations on the data that stayAPT may extract from your computer system. For example, stayAPT may use the information when preparing its franchise disclosure document, or may share information data amongst other franchisees in the system. (Section 6.4 of the Franchise Agreement).

Maintenance, Updates, and Repairs. You will be responsible for your own maintenance of the computer hardware and software. You will be obligated to pay for any upgrades or updates with no contractual limitation on the frequency or cost of the obligation. At this time, we estimate that the annual cost of any optional maintenance, updating, upgrading or support is less than \$400 each year though the cost for repairs would be in addition to that amount. There are no restrictions on our right to require you to upgrade or update the computer system or video surveillance system, or on the frequency of how often we may require you to do so. We have no obligation to assist you in obtaining the computer goods and services described in this Item and neither stayAPT, nor any third party must provide ongoing maintenance, repairs, upgrades or updates to the computer system.

<u>Initial Training</u>. You must attend stayAPT's initial training program before you open your Hotel. Training is available to you, or one of your owners if you are an entity, and your designated Resident Manager. stayAPT's training program will be conducted at your Hotel, at stayAPT's headquarters, or at other locations designated by stayAPT. The training will generally be for 3 to 5 days and is scheduled approximately 60 to 90 days before you open your Hotel. We organize initial training as needed to train new franchisees. You and your Resident Manager must successfully complete (to stayAPT's satisfaction) the training program prior to operating the Hotel. If you are using a third-party management company, a representative of the management company will also have to attend and successfully complete the initial training. (Section 10.1 of the Franchise Agreement)

TRAINING PROGRAM

Subject ⁽¹⁾	Hours of "Classroom" Training	Hours of "Hands On" Training	Location
WELCOME AND INTRODUCTION	1	0	Matthews, North Carolina, or another location we designate
SALES	4	0	Matthews, North Carolina, or another location we designate
GENERAL OPERATION OVERVIEW & BRAND IMMERSION	8	0	Matthews, North Carolina, or another location we designate
HOUSEKEEPING/ MAINTENANCE/LINEN	4	16	Matthews, North Carolina, or another location we designate
MARKETING, SALES AND PROMOTION	2	0	Matthews, North Carolina, or another location we designate
GUEST SERVICE	2	2	Matthews, North Carolina, or another location we designate
QUALITY CONTROL	2	1	Matthews, North Carolina, or another location we designate
HUMAN TRAFFICKING	As required by applicable state law	0	Web-based

Subject ⁽¹⁾	Hours of "Classroom" Training	Hours of "Hands On" Training	Location
TOTALS	23	19	

(1) The instructional materials include the Franchise Standards Manual, lecture, instructional materials, hands-on demonstration and practice.

As of the date of this disclosure document, our franchise training is primarily conducted by Adam Binder, our Vice President of Franchise Operations, who has over 10 years of experience in the hospitality and tourism industries, including experience training franchisees and hotel managers. Mr. Binder joined us in February 2021.

Other members of our executive team may also present on topics that are relevant to their areas of expertise. Our presenters generally have between 5 and 25 years of management-level experience, or more, in the hospitality industry.

You must pay for the salaries, fringe benefits, travel costs and expenses, and related costs for all persons associated with you who attend the training program. If you are an entity and you want more than one of your owners to attend training or if you hire a new Resident Manager after your Hotel opens for business, you will be responsible for paying the then-current standard fee to train the new Resident Manager. Likewise, if the Hotel is sold or transferred, the new owner must also complete this training. As of the date of this disclosure document, the training fee is \$2,500 per person.

Ongoing Training and Conferences. In addition to the initial training program, we may require you, or your owners, or your Resident Manager to attend additional training. The duration and location of the training will be determined by us. You will pay our then-current training fee, as well as the salaries, fringe benefits, travel costs and expenses, and related costs for all persons associated with you who attend the training program. The amount of the training fee will be set to reimburse us for our expenses in connection with the training.

Consistent with the terms of your Franchise Agreement, if stayAPT develops a national conference, you, or your owners, and your Resident Manager must attend and you will be responsible for your costs of participation, such as any conference fees, travel costs and expenses, salaries and related cost for all of your participants.

Apart from those training programs, we are not required to help train your employees and we are not required to help you hire employees. However, if we require additional training, you must require your employees to successfully complete the training.

<u>Franchise Standards Manual</u>. The table of contents to our stayAPT Suites Franchise Standards Manual is attached as Exhibit E. It is 105 pages long, without exhibits.

ITEM 12 TERRITORY

Franchise Agreement

<u>Protected Area.</u> You are granted the right to operate one Hotel at a defined location. stayAPT will not build a company-owned stayAPT® Suites business, or allow another stayAPT® Suites franchisee to build a stayAPT® Suites business, without your permission within a certain radius around your hotel (the "Protected Area"). The radius depends on the population density around your Hotel. The Protected Area will typically have either a 1-mile, 3-mile, or 5-mile radius, or greater depending on population density. You are not required to achieve a certain sales volume or other contingency to maintain territorial rights.

Your Rights in the Protected Area. Because there are certain limitations to your Protected Area rights, your rights are not exclusive. You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

stayAPT's Reserved Rights: For example, stayAPT and other affiliates or related companies have the right: (i) outside of your Development Territory and/or Protected Area, to establish franchised or company-owned hotels (or other businesses) or to offer, sell or distribute products or services associated with the Business System under the Marks (as defined in Item 13) or any other trademarks or through any distribution channel or method; and (ii) within and outside of your Development Territory and/or Protected Area, to establish franchised or company-owned hotels (or other businesses) or to offer, sell or distribute products or services under trademarks other than the Marks (including the Affordable Suites of America brand hotels), or to offer, sell or distribute any products or services through channels of distribution other than franchised or company-owned hotels operating under the Marks, including through the Internet or other forms of electronic communication, all without compensation to any franchisee.

Acquisition of Other Hotel Brand: If stayAPT or any of its affiliates acquire any chain or system of properties, or multiple units (more than 4) of a chain or system, and such purchase includes properties in the Protected Area which stayAPT or the acquiring affiliate desires to convert to stayAPT® Suites Hotels (hereinafter, whether one or more, "Conversion Hotels"), stayAPT shall provide you with written notice within a reasonable time of its intent to convert the Conversion Hotels into stayAPT® Suites Hotels. You will have the right of first refusal to acquire such Conversion Hotels from us on the terms provided in the Franchise Agreement (Section 1.3(b) of the Franchise Agreement) if the sale is allowed by contract and applicable law. You will have 90 days after we provide you with written notice of the intent to convert the Conversion Hotels to provide written notification to us of your desire to purchase the Conversion Hotel. If you elect to purchase the Conversion Hotels, you must close on such purchase and execute the then current franchise agreement (which shall require payment of the initial franchise due under such franchise agreement) within 60 days from your written notice to us of your intent to purchase. The purchase price to be paid by you for the Conversion Hotels shall be the cash equivalent of the fair market value for each of the Conversion Hotels, as determined by an

independent appraiser selected and retained by stayAPT. In the event you do not elect to purchase and convert the Conversion Hotels, you shall have no further right or option to acquire such Conversion Hotels and we may sell such Conversion Hotels to another franchisee under the Business System or stayAPT or its affiliates may own and operate the Conversion Hotel under the Business System.

No Customer Restrictions: stayAPT does not impose any territorial or customer restrictions on your sales and you are not required to compensate and are not entitled to receive compensation from stayAPT or other franchisees due to any territorial or customer sales restrictions. You may not relocate your Hotel without stayAPT's written consent, which consent we can grant or withhold in our sole discretion.

No Options or Right of First Refusal: stayAPT does not grant options, rights of first refusal or similar rights to acquire additional franchises, but we do offer qualified franchises development rights under a Multi-Unit Development Agreement ("MUDA"). If you have not signed a MUDA and want to open additional Hotels, you will have to submit a new application, which will be reviewed according to the same criteria that apply to the new franchisees at that time. A copy of our current Franchise Application is attached to this disclosure document as Exhibit K.

Multi-Unit Development Agreement

<u>Development Territory</u>. Under a MUDA you are granted the right to enter into Franchise Agreements for multiple Hotels that must be located within one or more defined Development Territories. stayAPT will not build a company-owned stayAPT® Suites business, or allow another stayAPT® Suites franchisee to build a stayAPT® Suites business, without your permission within the Development Territories for as long as you maintain your rights to the Development Territory. Once a Franchise Agreement is signed for a location in a Development Territory your rights to that Development Territory would cease, and you would instead get certain protected rights under the Franchise Agreement for the Hotel. The Protected Area under the Franchise Agreement will typically be smaller than the Development Territory. You are not required to achieve a certain sales volume or other contingency to maintain territorial rights, but failure to comply with the Development Schedule is a breach of the MUDA that gives us the right, amongst other things, to reduce the Development Territory.

Your Rights in the Protected Area. Because there are certain limitations to your Development Territory rights, your rights are not exclusive. You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

stayAPT's Reserved Rights: For example, stayAPT and other affiliates or related companies have the right: (i) outside of your Development Territory and/or Protected Area, to establish franchised or company-owned hotels (or other businesses) or to offer, sell or distribute products or services associated with the Business System under the Marks (as defined in Item 13) or any other trademarks or through any distribution channel or method; and (ii) within and outside of your Development Territory and/or Protected Area, to establish franchised or company-owned hotels (or other businesses) or to offer, sell or distribute products or services

under trademarks other than the Marks (including the Affordable Suites of America brand hotels), or to offer, sell or distribute any products or services through channels of distribution other than franchised or company-owned hotels operating under the Marks, including through the Internet or other forms of electronic communication, all without compensation to any franchisee.

Acquisition of Other Hotel Brand: If stayAPT or any of its affiliates acquire any chain or system of properties, or multiple units (more than 4) of a chain or system, and such purchase includes properties in the Protected Area which stayAPT or the acquiring affiliate desires to convert to stayAPT® Suites Hotels (hereinafter, whether one or more, "Conversion Hotels"), stayAPT shall provide you with written notice within a reasonable time of its intent to convert the Conversion Hotels into stayAPT® Suites Hotels. You will have the right of first refusal to acquire such Conversion Hotels from us on the terms provided in the Franchise Agreement (Section 1.4) of the MUDA) if the sale is allowed by contract and applicable law. You will have 90 days after we provide you with written notice of the intent to convert the Conversion Hotels to provide written notification to us of your desire to purchase the Conversion Hotel. If you elect to purchase the Conversion Hotels, you must close on such purchase and execute the then current franchise agreement (which shall require payment of the initial franchise due under such franchise agreement) within 60 days from your written notice to us of your intent to purchase. The purchase price to be paid by you for the Conversion Hotels shall be the cash equivalent of the fair market value for each of the Conversion Hotels, as determined by an independent appraiser selected and retained by stayAPT. In the event you do not elect to purchase and convert the Conversion Hotels, you shall have no further right or option to acquire such Conversion Hotels and we may sell such Conversion Hotels to another franchisee under the Business System or stayAPT or its affiliates may own and operate the Conversion Hotel under the Business System.

No Customer Restrictions: stayAPT does not impose any territorial or customer restrictions on your sales and you are not required to compensate and are not entitled to receive compensation from stayAPT or other franchisees due to any territorial or customer sales restrictions. You may not relocate your Hotel without stayAPT's written consent, which consent we can grant or withhold in our sole discretion.

Our Other Brand: Other than operating and franchising hotels under the Affordable Suites of America brand, stayAPT or its affiliates have no current plans to operate or franchise a hotel business under a different trademark. It may, however, acquire additional brands. The Affordable Suites brand operates out of our main offices in Matthews, North Carolina, and we will provide classroom training from the same facilities. Though we don't currently expect territorial overlap between stayAPT Suites and Affordable Suites locations, Affordable Suites franchisees may solicit and accept reservations within your Protected Area and you may solicit and accept reservations within theirs protected areas. If there are conflicts regarding territory, customers, or franchisor support, we will resolve them on a case-by-case basis.

ITEM 13 TRADEMARKS

The Franchise Agreement licenses you to use the stayAPT® Suites trademark, as well as other trademarks, service marks, trade names, trade dress and commercial symbols (collectively, the "Marks").

Mark	Registration Number	Date	Principal/Supplemental Register
STAYAPT	6285073	March 2, 2021	Supplemental
STAY APT	6756013	June 7, 2022	Supplemental

stayAPT claims common law trademark rights for all of the Marks.

stayAPT has filed all required affidavits and renewals for the Marks.

Although stayAPT has a federal registration for the "stayAPT" service marks above, stayAPT does not currently have a federal registration for the stayAPT design mark, which is displayed on the FDD cover page. Therefore, stayAPT's trademark does not have as many legal benefits and rights as a federally registered trademark, which may increase your expense. If stayAPT's right to use the trademarks is challenged, you may have to change to an alternative trademark, which may increase your expenses.

There are no currently effective determinations of the Patent and Trademark Office, Trademark Trial and Appeal Board, the trademark administrator of any state or any court, or any pending infringement, opposition or cancellation proceeding, or any pending material litigation, involving the principal Marks, or any superior prior rights in the Marks or infringing uses actually known to stayAPT which could materially affect your use of the Marks. There are currently no agreements in effect that significantly limit the rights of stayAPT to use or license the use of any Marks in any manner material to the franchise.

stayAPT is not required to protect you against infringement or unfair competition claims arising out of your use of the Marks, or to participate in your defense and/or indemnify you. stayAPT reserves the right to control any trademark litigation and will be the sole judge as to whether suit will be brought or settled in any instance when any person or entity infringes the Marks. You must notify stayAPT promptly of any infringement or unauthorized use of the Marks of which you become aware and to cooperate with any action that stayAPT undertakes. If any party claims that its rights to use any of the Marks are superior and if stayAPT determines that the claim is valid, you must, at your expense, immediately make the changes and use the substitutions to the Marks as stayAPT requires. stayAPT may require that you modify or discontinue using any of the Marks. If that happens you will be required to do so and will have to pay for the necessary changes.

ITEM 14 PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

There are no patents, patent applications, or copyrights currently registered which are material to the franchise, although stayAPT does claim copyright protection for our Franchise Agreement, Franchise Standards Manual and for various sales promotional and other materials published from time to time. stayAPT will renew relevant copyright registrations.

You must notify stayAPT immediately if you learn about an infringement or challenge to your use of any patents or copyrights. However, stayAPT does not contract with you or other franchisees to protect patents or copyrights, to protect you against infringement or unfair competition claims arising out of your use of patents or copyrights, or to participate in your defense and/or indemnify you. stayAPT reserves the right to control any patent or copyright litigation and will be the sole judge as to whether suit will be brought or settled in any instance when any person or entity infringes stayAPT's patents or copyrights. You may not contest stayAPT's interests in patents or copyrights.

There are no currently effective determinations of the Copyright Office (Library of Congress), Patent and Trademark Office, Trademark Trial and Appeal Board, the trademark administrator of any state or any court, or any pending infringement, opposition or cancellation proceeding or any pending material litigation involving any patents or copyrights. There are currently no agreements in effect that significantly limit the rights of stayAPT to use or license the use of any patents or copyrights in any manner material to the franchise. There are no infringing uses actually known to stayAPT which could materially affect your use of the patents or copyrights.

You must keep confidential during and after the term of the Franchise Agreement all proprietary information, including the Franchise Standards Manual. Upon termination of the Franchise Agreement, you must return to stayAPT all proprietary information, including the Franchise Standards Manual and all other material copyrighted by stayAPT. You must notify stayAPT immediately if you learn about an unauthorized use of proprietary information. stayAPT is not obligated to take action, and will be the sole judge as to the appropriate response to any information regarding the unauthorized use of proprietary information.

Franchisee is obligated disclose to stayAPT all designs, products, services, procedures, ideas, plans, improvements, concepts, formulas, methods and techniques ("Inventions") relating to the Hotel or any similar business, whether conceived or developed by you or your employee, during the term of the Franchise Agreement. stayAPT and its affiliates will own and have the right to use the Inventions for any purpose in its sole discretion.

You must comply with all changes to the Franchise Standards Manual at your cost. In the future, the changes may include making the Franchise Standards Manual available on the Internet or other on-line or computer data transfer communications.

ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISED BUSINESS

stayAPT does not require that you personally operate the Hotel. You may operate the hotel or retain a professional management company to operate the hotel, in either case subject to our approval. stayAPT can withdraw its approval at any time. stayAPT may require you to retain a management company or to change your management company if we determine, in our sole discretion, that you or your management company are not financially capable or do not have adequate experience, skills, or operational capacity to operate or continue operating the Hotel. Any change in management company is subject to our approval. You must comply at all times with the Franchise Agreement even if you retain a management company. You must designate in writing to stayAPT the approved management company and the individual who will reside in the Hotel and will directly supervise and manage the operation of the Hotel (the "Resident Manager"). Any individual owner, principal owner, management company, and Resident Manager must complete stayAPT's training program. The management company and Resident Manager need not have an ownership interest in the franchisee entity; however, they may be required to sign a written agreement maintaining the confidentiality of proprietary information described in Item 14 and abiding by the non-solicitation covenants described in Item 17. State labor and employment laws may require that you have more than one resident manager.

If you are an entity, each of your "Principal Owners" (any person who directly or indirectly owns a 10% or greater interest in you, as further defined in Section 23.9 of the Franchise Agreement) must sign the Personal Guaranty attached to the Franchise Agreement. Any other person we designate, including any person who co-signed a loan or was involved in obtaining financing for the Hotel, must also sign the Personal Guaranty. These individuals agree to discharge all obligations of the franchisee under the Franchise Agreement and are bound by all its terms and conditions, including maintaining the confidentiality of proprietary information described in Item 14 and abiding by the noncompete covenants described in Item 17. We do not require spouses of your Principal Owners to sign the Personal Guaranty, but if your Principal Owners live in a community property state, the Principal Owner's spouse will have to consent to the Principal Owner signing the Personal Guaranty. In addition, all of your Principal Owners must sign the Acknowledgment Addendum attached to the Franchise Agreement. All owners of a multi-unit developer must sign the Personal Guaranty attached to the Multi-Unit Development Agreement.

ITEM 16 RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

stayAPT requires you to offer and sell only those products and services that stayAPT has approved (see Items 8 and 9) and you must offer all products and services that stayAPT requires. There are no limits on stayAPT's right to make modifications to the approved products and services from time to time as set forth in the Franchise Standards Manual or otherwise in writing. Any failure to comply with these standards may result in the termination of your Franchise Agreement (see Item 17).

You may not use your Hotel or the Hotel premises for any other purpose than the operation of the stayAPT® Suites hotel business. You may use only marketing and promotional materials that stayAPT has approved. You are not limited in the customers to whom you may sell products and services.

ITEM 17 RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists certain important provisions of the franchise and related agreements. You should read these provisions in the agreements attached to this disclosure document.

Franchise Agreement

	Provision	Section in Franchise or other Agreement	Summary
a.	Length of the franchise term	Article 2.1	20 years.
b.	Renewal or extension of the term	Article 2.1	The Franchise Agreement is not renewable. You may have the opportunity to obtain a then-current successor Franchise Agreement at the end of the 20-year term (which may contain materially different terms and conditions than your original agreement).
c.	Requirements for you to renew or extend	Article 2.1	The franchise agreement is not renewable. If you wish to remain as a franchisee after the initial term of your Franchise Agreement expires, you must file a relicense application with stayAPT. stayAPT will review the application in accordance with its then-current standards and requirements for new franchisees. If approved, you must sign stayAPT's then-current form of franchise agreement and pay a relicense fee, which is equal to the then-current Initial Franchise Fee. The new franchise agreement may contain materially different terms and conditions than your original

	Provision	Section in Franchise or other Agreement	Summary
			contract. You will be required to sign a general release as a condition of stayAPT's approval for a relicense (subject to state law, see below).
D.	Termination by you	Article 14	Subject to certain conditions, you may terminate the Agreement only for good cause with 60 days' advance notice of default to stayAPT and 60 days' opportunity for stayAPT to cure.
e.	Termination by stayAPT without cause	Not Applicable	Not Applicable
f.	Termination by stayAPT with cause	Article 13.1	stayAPT can terminate only if you default.
g.	"Cause" defined – curable defaults	Article 13.1	You have 10 days to cure non-payment of amounts due and owing and 30 days to cure defaults including failure to start or complete construction or renovation timely, starting to do business without our permission, failure to abide by stayAPT's standards and requirements for the Business, failure to meet stayAPT's requirements and specifications regarding products and services, and any other default not subject to immediate termination under the terms of the Agreement.
h.	"Cause" defined – non- curable defaults	Article 13.1	stayAPT may immediately terminate the Agreement for non-curable defaults: you are involved in conduct that reflects unfavorably on the Marks or Business System goodwill, you misuse confidential information, you willfully falsifies a report, insolvency, foreclosure is levied on the Hotel, you or your owners of directors/officers are convicted

	Provision	Section in Franchise	Summary
		or other Agreement	
			or plead guilty to any charge related to your business, abandonment, you or your owners transfers rights in violation of the Franchise Agreement, you commit an uncurable breach, repeated defaults within 12-month period, one of your affiliates defaults on any agreement with us or our affiliates (subject to state law). In the event of your bankruptcy the Franchise Agreement terminates without notice.
i.	Your obligations on termination/non-renewal	Articles 15, 16	Obligations include de-identification of the Hotel(s), payment of amounts due, stop using the Marks assigning business phone number, returning Franchise Standards Manual and proprietary information, offering Business Assets to stayAPT.
j.	Assignment of contract by stayAPT	Article 19.1	No restriction on stayAPT's right to assign. Assignee must fulfill stayAPT's obligations under the Agreement.
k.	"Transfer" by you – definition	Article 19	Includes any transfer (whether voluntary or involuntary) of the Agreement, the Hotel, any assets, revenues or profits of the Hotel (except in the ordinary course of the Business), or any interest in you.
1.	stayAPT's approval of transfer by franchisee	Article 19.6	stayAPT has the right to approve all transfers.
m.	Conditions for stayAPT's approval of transfer	Article 19.6	New franchisee qualifies, transferee not in default under the Agreement, transfer fee paid, all amounts owed by prior franchisee paid, training completed, non-compete agreements signed, guarantees or then-current franchise agreements signed, stayAPT approves the material

	Provision	Section in Franchise or other Agreement	Summary
			provisions of the transfer, and the terms of the transfer do not negatively impact the business' profits. You will be required to sign a general release as a condition of stayAPT's approval for transfer. (subject to state law, see below)
n.	stayAPT's right of first refusal to acquire your business	Article 18	stayAPT can match any offer for your business.
0.	stayAPT's option to purchase your business	Article 18	stayAPT has the option to purchase the business assets upon the expiration or termination of the Franchise Agreement.
p.	Your death or disability	Article 19.5	If Franchisee is an individual, Franchisee's ownership interest can transfer to third party approved by stayAPT upon death, disability or incapacity. Transfer conditions apply (see m, above).
q.	Non-competition covenants during the term of the franchise	Article 16.1	During the term of the Franchise Agreement, you may not hire any current employee or any former employee who is or was within the last 6 months employed in a management position by Franchisor or any other Brand Hotel franchisee (except your affiliates). Subject to state law.
r.	Non-competition covenants after the franchise is terminated or expires	Not Applicable	Not Applicable
S.	Modification of the agreement	Article 22.2, 22.6, 22.8	No modifications generally, but stayAPT may change the Franchise Standards Manual and list of Marks.

	Provision	Section in Franchise or other Agreement	Summary
t.	Integration/ merger clause	Article 22.6	Only the terms of the Franchise Agreement and other related written agreements are binding (subject to applicable state law). Any representations or promises outside of the disclosure document and Franchise Agreement may not be enforceable.
u.	Dispute resolution by arbitration or mediation	Article 17	Except for certain claims, all disputes must be mediated and/or arbitrated in Matthews, North Carolina (subject to state law).
v.	Choice of forum	Article 17.1	Litigation must be brought in Matthews, North Carolina (subject to state law, see below).
W.	Choice of law	Article 22.1	The law of the state of North Carolina will apply (subject to state law, see below).

Multi-Unit Development Agreement

	Provision	Section in Multi-Unit Development Agreement	Summary
a.	Length of the franchise term	Article 2.1	Ends at the earlier of the date we accept and execute the last Franchise Agreement required under the Development Schedule and the anniversary date agreed upon between us.
b.	Renewal or extension of the term	Article 2.1	The Multi-Unit Development Agreement is not renewable.
c.	Requirements for you to renew or extend	Article 2.1	The agreement is not renewable.
d.	Termination by you	Article 6	You may terminate if we are in breach of a material provision of

Provision		Section in Multi-Unit Development Agreement	Summary	
			the agreement and we do not cure the breach within 60 days of notice.	
e.	Termination by stayAPT without cause	Not Applicable	Not Applicable	
f.	Termination by stayAPT with cause	Article 5	stayAPT can terminate only if you default.	
g.	"Cause" defined – curable defaults	Article 5.1	(A) You violate any material provision of the agreement including failing to comply with the Development Schedule; (B) you fail to conform to the Business System, the standards of uniformity and quality for the goods and services or the policies and procedures promulgated by us in connection with the Business System; (C) you fail to timely pay any uncontested obligations or liabilities due and owing us, suppliers, banks, vendors, other creditors or any federal, state and municipal government (including, if applicable, federal and state taxes); (D) any Franchise Agreement or other agreement executed by you or your affiliates is (1) terminated by us or our affiliates or (is 2) wrongfully terminated by you.	
h.	"Cause" defined – non- curable defaults	Article 5.4	stayAPT may immediately terminate the agreement if you are adjudicated bankrupt, become insolvent, suffers a court-appointed receivership of substantially all of your property, make a general assignment for the benefit of creditors or suffer the filing of a voluntary or involuntary bankruptcy petition which is not dismissed within 90 days after	

	Provision	Section in Multi-Unit Development Agreement	Summary
			filing. We can also immediately terminate the agreement if you are involved in any act or conduct which materially impairs the goodwill associated with the Marks or the Business System; (B) you willfully and materially falsify any report, statement or other written data furnished to us; (C) a suit to foreclose any lien or mortgage against a Hotel's premises is instituted against you and not dismissed within 30 days; (D) you or any of your partners, Directors, officers or majority equity owners are convicted of, or plead guilty to or no contest to a charge of violating any law relating to your business, or any felony; (E) you cease to do business or otherwise abandon the business; (F) the nature of your breach makes it not curable; or (G) you fail to comply with one or more requirements of the Agreement 3 times in any one 12-month period even if the defaults are cured.
i.	Your obligations on termination/non-renewal	Article 7	You must pay all amount due under the agreement or any other agreement; you must comply with all other provisions of the agreement that continue after the termination.
j.	Assignment of contract by stayAPT	Article 8.1	No restriction on stayAPT's right to assign. Assignee must fulfill stayAPT's obligations under the Agreement.
k.	"Transfer" by you - definition	Article 8.2	Includes any transfer (whether voluntary or involuntary) of the Agreement, and an interest in you.
1.	stayAPT's approval of	Articles 8.2, 8.3, 8.4	stayAPT has the right to approve

	Provision	Section in Multi-Unit Development Agreement	Summary
	transfer by franchisee		all transfers.
m.	Conditions for stayAPT's approval of transfer	Articles 8.4, 8.5	All your monetary obligations due to us must be paid in full, and you must not be in default under the Agreement; you (and each Principal Owner, if applicable) has executed a written agreement in a form satisfactory to us agreeing to observe all applicable post-term obligations and covenants contained in the agreement; the has demonstrated to our satisfaction that they meet our managerial, financial, and business standards for new multi-unit developers, possesses; you have paid the transfer fee required under Article 8.5; transferee has executed our then-current standard multi-unit development agreement and each Principal Owner has executed the Personal Guaranty; and you (and each Principal Owner, if applicable) signs a general release, in a form and substance satisfactory to us.
n.	stayAPT's right of first refusal to acquire your business	Not applicable	Not applicable
0.	stayAPT's option to purchase your business	Not applicable	Not applicable
p.	Your death or disability	Article 8.3	If Franchisee is an individual, Franchisee's ownership interest can transfer to third party approved by stayAPT upon death, disability or incapacity. Transfer conditions apply (see m, above).
q.	Non-competition covenants during the term of the franchise	Not Applicable	Not Applicable
r.	Non-competition	Not Applicable	Not Applicable

Provision		Section in Multi-Unit Development Agreement	Summary
	covenants after the franchise is terminated or expires		
s.	Modification of the agreement	Articles 10.6,10.8	No change, modification, amendment or waiver of any of the provisions hereof, including by custom or usage of trade or course of dealing or performance, shall be effective and binding upon either party unless it is in writing, specifically identifies as an amendment hereto, and signed by the party to be charged.
t.	Integration/ merger clause	Article 10.6	Only the terms of the agreement and other related written agreements are binding (subject to applicable state law). Any representations or promises outside of the disclosure document and agreement may not be enforceable.
u.	Dispute resolution by arbitration or mediation	Article 9	Except for certain claims, all disputes must be mediated and/or arbitrated in Matthews, North Carolina (subject to state law).
v.	Choice of forum	Article 17.1	Litigation must be brought in Matthews, North Carolina (subject to state law, see below).
W.	Choice of law	Article 10.1	The law of the state of North Carolina will apply (subject to state law, see below).

There are state specific addenda attached as **Exhibit F** for the states of California, Hawaii, Illinois, Maryland, Minnesota, New York, North Dakota, Rhode Island, Virginia and Washington. The Michigan Addendum is attached following the state cover page.

ITEM 18 PUBLIC FIGURES

stayAPT does not use any public figure to promote the franchise. No public figure is involved in the actual management or control of stayAPT.

ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

The following tables set forth certain historical financial, performance, and operating information for established stayAPT Suites hotels (the "Established Hotels") for the trailing 12-month period ending February 28, 2025 (the "Measured Period"). Established Hotels are stayAPT Suites hotels that: (i) were open and operating under the Brand for at least 18 months as of February 28, 2025; (ii) reported data to us and to STR (defined below) for the Measured Period; and (iii) were not undergoing significant renovations that, during the Measured Period, caused more than 7% of total rooms to be out of order.

As of February 28, 2025, there were 36 stayAPT Suites operating in the United States, 25 of which were owned by our affiliates and 11 were owned by franchisees. Of these, 14 hotels were Established Hotels. 12 Established Hotels were owned and operated by our affiliates (the "Established Company-Owned Hotels"), and 2 Established Hotels were owned by franchisees (the "Established Franchised Hotels"), one of which was managed by our affiliate. The performance results for hotels managed by our affiliates do not materially differ from the performance results for other hotels included in this Item 19.

1. Average Occupancy, ADR, RevPAR, and RGI of Established Hotels

Performance of Established Hotels for the Measured Period					
	Number of Included Hotels	Average Occupancy Rate	ADR	Average RevPAR	RGI
All Established Hotels	14	76.8%	\$91.40	\$70.25	97.4
Established Company-Owned Hotels	12	74.7%	\$91.08	\$68.11	96.9
Established Franchised Hotels	2	89.0%	\$93.36	\$83.10	100.1

Table Notes:

- (1) "Occupancy Rate" means the percentage of occupied rooms for the described period.
- (2) "ADR" means the average daily room rate for the described period.

- (3) "RevPAR" means the revenue per available room for the described period.
- (4) "RGI" means revenue generation index or RevPAR index, which is a measurement of a hotel's RevPAR in relation to its competitive set for the described period.

This Item 19 does not include 12 company-owned hotels and 9 franchised hotels that were open for less than 18 months as of February 28, 2025, and 1 company-owned hotel that underwent a major room renovation during the Measured Period. All of the excluded franchised hotels opened in 2024.

Of the 14 Established Hotels for the trailing 12-month period ending February 28, 2025, 7 hotels or 50% had an average occupancy rate that met or exceeded the stated average, 7 hotels or 50% had an ADR that met or exceeded the stated average, and 5 hotels or 36% had an average RevPAR that met or exceeded the stated average; the median occupancy rate was 77.2%, the median ADR was \$89.91, and the median RevPAR was \$69.42. Of the 14 Established Hotels for the trailing 12-month period ending February 28, 2025, 8 hotels or 57% had an RGI that met or exceeded the stated average, and the median RGI was 98.2.

Of the 12 Established Company-Owned Hotels for the trailing 12-month period ending February 28, 2025, 7 hotel or 58% had an average occupancy rate that met or exceeded the stated average, 6 hotels or 50% had an ADR that met or exceeded the stated average, and 7 hotels or 58% had an average RevPAR that met or exceeded the stated average; the median occupancy rate was 75.9%, the median ADR was \$89.91, and the median RevPAR was \$68.89. Of the 12 Established Company-Owned Hotels for the trailing 12-month period ending February 28, 2025, 6 hotels or 50% had an RGI that met or exceeded the stated average, and the median RGI was 97.1.

Of the 2 Established Franchised Hotels for the trailing 12-month period ending February 28, 2025, 1 hotel or 50% had an average occupancy rate that met or exceeded the stated average, 1 hotel or 50% had an ADR that met or exceeded the stated average, and 1 hotel or 50% had an average RevPAR that met or exceeded the stated average; the median occupancy rate was 89.0%, the median ADR was \$93.36, and the median RevPAR was \$83.10. Of the 2 Established Franchised Hotels for the trailing 12-month period ending February 28, 2025, 1 hotel or 50% had an RGI that met or exceeded the stated average, and the median RGI was 100.1.

The "Average Occupancy Rate" was derived by dividing the total number of guest rooms reported as occupied during the trailing 12-month period ending February 28, 2025 by the total number of guest rooms reported as available during the same period. "ADR" was derived by dividing the Gross Room Revenues reported for the trailing 12-month period ending February 28, 2025 by the total number of guest rooms reported as occupied during the same period. "Average RevPAR" was derived by dividing the annual Gross Room Revenues by the total number of available rooms for the trailing 12-month period ending February 28, 2025. The median is the middle data point of the numbers used to calculate the corresponding averages.

The Average Occupancy Rate, ADR, and Average RevPAR are calculated based on information routinely reported to us on a monthly basis by individual hotels operating in the Business System. One Franchised Hotel that is not managed by our affiliate reported data on an annual basis only, and where required, that hotel's data was equally divided over twelve months

for the purposes of our calculations. ADR and Average RevPAR are calculated on the basis of Gross Room Revenues, which we define as the total revenues and receipts resulting from the rental of guest rooms, whether made for cash or credit and whether or not collected by the franchisee. "Gross Room Revenues" do not include charges for food and beverages, telephone, laundry, entertainment, security deposits or applicable sales tax.

RGI is a standard benchmarking metric used in the hospitality industry to measure a select hotel's RevPAR performance relative to its competitive set. A "competitive set" is an aggregated group of hotels that would generally compete for the same business and typically consists of at least 3 properties operating either independently or under different brands in the same area and in the same chain scale segment (e.g. luxury, upscale, midscale, etc.) as the select hotel. RGI is calculated using competitive set data routinely reported on a monthly basis by STR, LLC and its affiliates (collectively, "STR"), an independent research company that commonly provides performance data and benchmarking analytics for the hospitality industry. RGI is calculated by dividing the hotel's RevPAR by the aggregate RevPAR for its competitive set and multiplying the total by 100. An RGI of 100 indicates the hotel is capturing its fair share of available revenue for the market as compared to the performance of its competitive set. Similarly, an RGI above 100 (or below 100) indicates the hotel is capturing more (or less) than its fair share. STR receives performance data directly from hotel properties or hotel chains, and we have not independently verified or audited the information provided by STR.

2. Average Monthly Operating Results

The table below shows the average monthly operating results (expenses and gross profit), as a percentage of total revenue, for the 12 Established Company-Owned Hotels for the trailing 12-month period ending February 28, 2025:

	Average Monthly Expense/Profit as a Percentage of <u>Total Revenue</u>	Median Monthly Expense/Profit as a Percentage of <u>Total Revenue</u>
Operating Expenses:	35.4%	34.4%
Payroll & Related Expenses	13.4%	13.0%
Rooms Expenses	5.4%	5.5%
Maintenance Expenses	3.3%	3.3%
Sales & Marketing	2.6%	2.4%
General & Administrative	5.9%	5.9%
Cash/Credit Management	0.3%	0.3%
Utilities	4.5%	4.0%
GOP Margin:	64.6%	65.6%

The table below shows the average monthly operating results (expenses and gross profit), as a percentage of total revenue, for the 2 Established Franchised Hotels for the trailing 12-month period ending February 28, 2025:

	Average Monthly Expense/Profit as a Percentage of <u>Total Revenue</u>	Median Monthly Expense/Profit as a Percentage of <u>Total Revenue</u>
Operating Expenses:	30.7%	30.7%
Payroll & Related Expenses	11.3%	11.3%
Rooms Expense	4.6%	4.6%
Maintenance Expense	2.7%	2.7%
Sales & Marketing	2.3%	2.3%
General & Administrative	6.0%	6.0%
Cash/Credit Management	0.5%	0.5%
Utilities	3.3%	3.3%
GOP Margin	69.3%	69.3%

Table Notes:

- (1) "Operating Expenses" is the sum of monthly expenses in each of the listed categories.
- (2) "Payroll and Related Expenses" includes the salary and benefit costs for general managers, housekeeping staff, and contracted housekeepers.
- (3) "Rooms Expenses" includes costs for linens, guest room supplies, cleaning supplies, and guest laundry.
- (4) "Maintenance Expenses" includes costs for repairs and maintenance at the property, such as plumbing, carpet cleaning, landscaping, safety inspections, and trash removal services.
- (5) "Sales & Marketing" costs include pay-per-click fees, marketing ad spend, and associated marketing fees.
- (6) General & Administrative" costs include telecom, credit card fees, cable and in-room entertainment services, software licensing and support, uniforms, maintenance contracts, postage, and travel and entertainment.
- (7) "Cash/Credit Management" costs include chargebacks, cash, and write-offs.
- (8) "Utilities" costs include the hotel's cost for gas, electricity, water and sewage.
- (9) "GOP Margin" is derived by dividing the Gross Operating Profit by total revenue. "Gross Operating Profit" is derived by subtracting the monthly Operating Expenses from the monthly total revenue.

For the trailing 12-month period ending February 28, 2025, of the 12 Established Company-Owned Hotels, 4 hotels or 33% reported an average monthly GOP Margin that met or exceeded the stated average; and of the 2 Established Franchised Hotels, 1 hotel or 50% reported an average monthly GOP Margin that met or exceeded the stated average.

3. <u>Explanatory Notes</u>

The information presented in this Item 19 is derived from data reported to us on a monthly basis by company-owned and franchised hotels, including hotels that our affiliates manage. We have not independently verified or audited this information. Actual results vary between hotels and may be impacted by a number of factors, including market conditions, location, competition, customer base, geographic area, demand, and length of time the hotel has been operating.

You should conduct an independent analysis of this franchise offering, consult with attorneys, accountants, appraisers, and other professional advisors who are familiar with the

lodging industry, and prepare your own business plan and pro forma cash flow statement, balance sheet, and statement of operations. Franchisees or former franchisees, listed in this disclosure document, may be one source of information.

Some stayAPT Suites hotels have earned these amounts. Your individual results may differ. There is no assurance that you'll earn as much.

Written substantiation for the financial performance representations in this Item will be made available to you upon reasonable request.

Other than as described in the preceding financial performance representation, we do not make any representations about a franchisee's financial performance or the financial performance of company-owned or franchised outlets. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Adam Binder, Vice President of Franchise Operations, 10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, (980) 368-8100, the Federal Trade Commission, and the appropriate state regulatory agencies.

[Remainder of page intentionally left blank.]

ITEM 20 OUTLETS AND FRANCHISEE INFORMATION

TABLE NUMBER 1 Systemwide Hotel Summary For Years 2022 to 2024

Outlet Type	Year	Outlets at the Start of the Year	Outlets at the End of the Year	Net Change
Franchised*	2022	2	2	0
	2023	2	2	0
	2024	2	9	+7
Company-Owned	2022	4	11	+7
	2023	11	15	+4
	2024	15	23	+8
Total Hotels	2022	6	13	+7
	2023	13	17	+4
	2024	17	32	+15

TABLE NUMBER 2 Transfers of Hotels From Franchisee to New Owners (Other than the Franchisor) For Years 2022 to 2024

State	Year	Number of Transfers
TOTAL	2022	0
	2023	0
	2024	0

TABLE NUMBER 3 Status of Franchised Hotels For Years 2022 to 2024

State	Year	Outlets at the Start of the Year	Outlets Opened	Termina - tions	Non- Renewals	Reacquired by Franchisor	Ceased Operations / Other Reasons	Outlets at the End of the Year
Alabama	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0
	2024	0	1	0	0	0	0	1
Florida	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0
	2024	0	1	0	0	0	0	1
Georgia	2022	0	0	0	0	0	0	0

State	Year	Outlets at the Start of the Year	Outlets Opened	Termina - tions	Non- Renewals	Reacquired by Franchisor	Ceased Operations / Other Reasons	Outlets at the End of the Year
	2023	0	0	0	0	0	0	0
	2024	0	1	0	0	0	0	1
Indiana	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0
	2024	0	1	0	0	0	0	1
South Carolina	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2024	1	0	0	0	0	0	1
Tennessee	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0
	2024	0	2	0	0	0	0	2
Virginia	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2024	1	1	0	0	0	0	2
TOTAL	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	2024	2	7	0	0	0	0	9

TABLE NUMBER 4 Status of Company-Owned Hotels For Years 2022 to 2024⁽¹⁾

State	Year	Outlets at the Start of the Year	Outlets Opened	Outlets Reacquired From Franchisees	Outlets Closed	Outlets Sold to Franchisees	Outlets at the End of the Year
Alabama	2022	0	2	0	0	0	2
	2023	2	2	0	0	0	4
	2024	4	0	0	0	0	4
Arkansas	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2024	0	1	0	0	0	1
Florida	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2024	0	1	0	0	0	1
Georgia	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2024	0	1	0	0	0	1
North Carolina	2022	2	0	0	0	0	2
	2023	2	0	0	0	0	2
	2024	2	1	0	0	0	3
Pennsylvania	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2024	0	1	0	0	0	1
South Carolina	2022	1	2	0	0	0	3
	2023	3	1	0	0	0	4
	2024	4	0	0	0	0	4

State	Year	Outlets at the Start of the Year	Outlets Opened	Outlets Reacquired From Franchisees	Outlets Closed	Outlets Sold to Franchisees	Outlets at the End of the Year
Tennessee	2022	1	0	0	0	0	1
	2023	1	1	0	0	0	2
	2024	2	0	0	0	0	2
Texas	2022	0	3	0	0	0	3
	2023	3	0	0	0	0	3
	2024	3	3	0	0	0	6
TOTAL	2022	4	7	0	0	0	11
	2023	11	4	0	0	0	15
	2024	15	8	0	0	0	23

TABLE NUMBER 5 Projected Openings As of December 31, 2024

State	Franchise Agreements Signed But Business Not Opened(1)	Projected New Franchised Businesses in the Next Fiscal Year	Projected New Company-Owned Businesses in the Next Fiscal Year
Arizona	2	0	0
Florida	1	1	0
Georgia	1	1	1
Illinois	1	0	0
Indiana	3	0	0
Louisiana	1	1	0
New Mexico	1	0	0
North Carolina	0	0	1
Pennsylvania	1	0	0
Tennessee	1	0	1
Texas	1	1	1
Virginia	1	0	0
TOTAL	14	4	4

(1) Our "company owned" locations are owned by our affiliates and we execute franchise agreements with them. Those franchise agreements are not included in this column. As of December 31, 2024, we had 12 signed affiliate franchise agreements for company-owned locations that were not opened.

A list of all stayAPT® Suites franchisees is included in this disclosure document as Exhibit B, Part I. Franchisees who have had a franchise terminated, canceled or not renewed, or otherwise voluntarily or involuntarily ceased to do business under the Franchise Agreement during the most recently completed fiscal year, or has not communicated with stayAPT as of December 31, 2024 are listed in Exhibit B, Part II.

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

In some instances, franchisees may sign provisions restricting their ability to speak openly about their experience with stayAPT. You may wish to speak with current and former franchisees, but be aware that not all such franchisees will be able to communicate with you. Since stayAPT only started franchising as of the issuance date of this FDD there are no franchisees that are restricted in their ability to speak about their experience in the franchise system. stayAPT is not aware of any franchisee associations associated with the stayAPT Suites brand.

There is no trademark-specific franchisee organization associated with the franchise system.

[Remainder of page intentionally left blank.]

ITEM 21 FINANCIAL STATEMENTS

Attached as Exhibit C, Part I, to this franchise disclosure document are (i) the audited Financial Statements of LG AS Franchisor LLC as of and for the years ended December 31, 2024, and 2023, and the related statements of operations, changes in member's capital, and cash flows for the years ended December 31, 2024, 2023 and 2022 and the related notes to the financial statements.

If required by state law, attached as Exhibit C, Part II, to this franchise disclosure document are the unaudited balance sheet and related statements of operations, cash flows and changes in member's capital of LG AS Franchisor LLC as of a later date than the audited financial statements enclosed in Part I.

ITEM 22 CONTRACTS

The following agreements are attached to this disclosure document:

- stayAPT Franchise Agreement Exhibit D
- Form of Release of Claims Exhibit G
- Comfort Letter Exhibit H
- Multi-Unit Development Agreement Exhibit I
- Data Transfer Agreement Exhibit J
- Franchise Application Exhibit K

ITEM 23 RECEIPTS

The last pages of this disclosure document are a detachable acknowledgement of receipt.

EXHIBIT A

List of State Administrators and Agents to Receive Service of Process

STATE ADMINISTRATORS

Listed here are the names, addresses and telephone numbers of the state agencies having responsibility for franchising disclosure/registration laws.

California

Department of Financial Protection and Innovation State of California 320 West 4th Street Suite 750 Los Angeles, California 90013 (213) 576-7500 (866) 275-2677

Indiana

Franchise Section Indiana Securities Division Room E-111 302 West Washington Street Indianapolis, Indiana 46204 (317) 232-6681

Connecticut

Maryland

Office of the Attorney General Securities Division State of Maryland 200 St. Paul Place Baltimore, Maryland 21202 (410) 576-6360

Hawaii

Hawaii Commissioner of Securities					
Department of Commerce and					
Consumer Affairs					
Business Registration Division					
State of Hawaii					
P.O. Box 40					
Honolulu, Hawaii 96810					
(808) 586-2722					

Michigan

Consumer Protection Division
Franchise Section
Michigan Department of Attorney General
525 W. Ottawa Street
G. Mennan Williams Building, 1st Floor
Lansing, Michigan 48913
(517) 373-7117

Illinois

Franchise Bureau Illinois Attorney General 500 South Second Street Springfield, Illinois 62706 (217) 782-4465

Minnesota

Minnesota Department of Commerce Commissioner of Commerce 85 E 7th Place, Suite 280 St. Paul, Minnesota 55101-2198 (651) 539-1500

New York

Bureau of Investor Protection and Securities New York State Department of Law 28 Liberty Street, 21st Floor New York, New York 10005 (212) 416-8222

North Dakota

North Dakota Securities Department 600 East Boulevard Avenue State Capitol Fourteenth Floor, Dept 414 Bismarck, North Dakota 58505-0510 (701) 328-4712

Oregon

Department of Consumer and Business Services Division of Finance and Corporate Securities State of Oregon Labor and Industries Building Salem, Oregon 97310 (503) 378-4140

Rhode Island

Division of Securities State of Rhode Island 233 Richmond Street, Suite 232 Providence, Rhode Island 02903 (401) 222-3048

South Dakota

Division of Insurance Securities Regulation 124 S. Euclid Ave., Suite 104 Pierre, South Dakota 57501 (605) 773-3563

Virginia

State Corporation Commission Division of Securities & Retail Franchising Commonwealth of Virginia 1300 E. Main Street, 9th Floor Richmond, Virginia 23219 (804) 371-9051

Washington

Department of Financial Institutions Securities Division State of Washington P.O. Box 41200 Olympia, Washington 98504-1200 (360) 902-8760

Wisconsin

Division of Securities Department of Financial Institutions Wisconsin Commissioner of Securities P.O. Box 1768 Madison, Wisconsin 53701-1768 (608) 266-8559

AGENTS AUTHORIZED TO RECEIVE SERVICE OF PROCESS, BY STATE

California

Commissioner of Financial Protection and Innovation Department of Financial Protection and Innovation State of California Suite 750 320 West 4th Street Los Angeles, CA 90013-2344

Maryland

Maryland Securities Commissioner Office of the Attorney General Securities Division 200 St. Paul Place Baltimore, MD 21202-2020

Hawaii

Hawaii Commissioner of Securities Department of Commerce and Consumer Affairs Business Registration Division State of Hawaii 335 Merchant Street Room 204 Honolulu, Hawaii 96813

Michigan

Michigan Department of Commerce Corporation & Securities Bureau 6546 Mercantile Way Lansing, MI 48909

Illinois

Office of Attorney General State of Illinois 500 South Second Street Springfield, IL 62706

Minnesota

Commissioner of Commerce Minnesota Department of Commerce Franchise Section 85 7th Place East, Suite 280 St. Paul, MN 55101-2198

Indiana

Secretary of State State of Indiana 201 State House 200 W. Washington St. Indianapolis, IN 46204

New York

Secretary of State State of New York 41 State Street Albany, NY 12231

North Dakota

Securities Commissioner North Dakota Securities Department 600 East Boulevard Avenue State Capitol Fourteenth Floor, Dept 414 Bismarck, North Dakota 58505-0510

Virginia

Clerk of the State Corporation Commission Commonwealth of Virginia 1300 E. Main Street, 9th Floor Richmond, VA 23219

Oregon

Department of Consumer and Business Services Division of Finance and Corporate Securities State of Oregon 350 Winter Street, N.E., Room 21 Portland, OR 97310

Washington

Director of Financial Institutions Securities Division State of Washington 150 Israel Rd., S.W. Olympia, WA 98501

Rhode Island

Director of Business Regulation Department of Business Regulation Division of Securities State of Rhode Island 233 Richmond Street, Suite 232 Providence, RI 02903

Wisconsin

Commissioner of Securities Wisconsin Securities Commission 345 W. Washington Ave., 4th Floor Madison, WI 53703

South Dakota

Franchise Administration
Division of Securities
Department of Revenue and Regulation
State of South Dakota
124 S. Euclid Ave., Suite 104
Pierre, SD 57501-2000

EXHIBIT B

List of Franchisees and Franchisees Who Left the System

PART I

LIST OF FRANCHISEES

Following is the list of stayAPT Suites Franchisees as of December 31, 2024:

Operating Outlets:

ALABAMA

stayAPT Suites Madison-Greenbrier Stay Greenbrier, LLC 26800 Success Drive SW Madison, AL 35756 (256) 834-8852

FLORIDA

stayAPT Suites Pensacola-UWF/West Florida Hospital Area Pensacola Hotel Ventures, LLC 9229 University Parkway Pensacola, FL 32514 (850) 565-3600

GEORGIA

stayAPT Suites LaGrange Catarina of Tucson, LLC 112 Jameson Drive LaGrange, GA 30241 (706) 883-5700

INDIANA

stayAPT Suites Louisville North-Clarksville APT Ventures Louisville/Clarksville, LLC 1628 Leisure Way Clarksville, IN 47129 (812) 850-3100

SOUTH CAROLINA

stayAPT Suites Greenville-Greer-GSP Free Enterprises Greer LLC 1455 Highway 101 South Greer, SC 29651 (864) 473-5799

TENNESSEE

stayAPT Suites Cleveland Frontage Lodging, LLC 4415 Frontage Road, NW Cleveland, TN 37312 (423) 709-8666

stayAPT Suites Lebanon-Nashville Area Lebanon Partners, LLC 155 South Eastgate Court Lebanon, TN 37090 (615) 994-6330

VIRGINIA

stayAPT Suites Alexandria-Fort Belvoir Belvoir Development Group, LLC 8849 Richmond Highway Alexandria, VA 22309 (703) 349-0777

stayAPT Suites Suffolk-Chesapeake Harbor View Hospitality, LLC 5961 Harbour View Boulevard Suffolk, VA 23435 (757) 720-8899

Outlets Not Yet Open:

ARIZONA

(Location TBD) Modern Stay, LLC Scottsdale, AZ (864) 275-9528

(stayAPT Suites Yuma) Yuma 888 QOZB LLC Phoenix, AZ (415) 237-3310

FLORIDA

(stayAPT Suites Tallahassee-Capitol) Tallahassee Hotel Ventures, LP Matthews, NC (704) 246-5564

GEORGIA

(stayAPT Suites Atlanta-Lithia Springs) TGC Lithia, LLC Wichita, KS (316) 260-9088

ILLINOIS

(stayAPT Suites Bloomington) The Pines RV Park LLC Phoenix, AZ (415) 237-3310

INDIANA

(stayAPT Suites Camby) Heartland Lodging Associates Chattanooga, TN (423) 421-2086

(stayAPT Suites Indianapolis-Greenwood) APTVentures Greenwood LLC Wichita, KS (316) 841-2331

stayAPT Suites Indianapolis (Airport-West) Rockville Road Lodging Associates LLC Chattanooga, TN (423) 421-2086

LOUISIANA

(stayAPT Suites Lafayette) Farish Holdings, LLC Greenville, SC (864) 444-4501

NEW MEXICO

(stayAPT Suites Suites Rio Rancho) 9999 Irene LLC Albuquerque, NM (505) 508-4042

PENNSYLVANIA

(stayAPT Suites Lancaster)
Dutchland Realty Holding Corporation
Denver, PA
(717) 799-2542

TENNESSEE

(stayAPT Suites Murfreesboro) Rivermont Hospitality LLC Chattanooga, TN 37421 (423) 777-8997

TEXAS

(stayAPT Suites Waco) SPT Waco LLC Dallas, TX (972) 385-4182

VIRGINIA

(stayAPT Suites Hampton) Coliseum Central Hospitality, LLC Fredericksburg, VA 22405 (540) 371-7197

PART II

FRANCHISEES WHO LEFT THE SYSTEM

Following is the list of stayAPT Suites franchisees who have had a franchise terminated, canceled or not renewed, or otherwise voluntarily or involuntarily ceased to do business under the Franchise Agreement during the most recently completed fiscal year, or have not communicated with stayAPT as of December 31, 2024:

None

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

EXHIBIT C

Financial Statements

Audited Financial Statements

LG AS Franchisor LLC

Financial Statements as of December 31, 2024, and 2023, and for the Years Ended December 31, 2024, 2023, and 2022, and Independent Auditor's Report

TABLE OF CONTENTS

	Page
INDEPENDENT AUDITOR'S REPORT	1–2
FINANCIAL STATEMENTS AS OF DECEMBER 31, 2024 AND 2023, AND FOR THE YEARS ENDED DECEMBER 31, 2024, 2023, AND 2022:	
Balance Sheets	3
Statements of Operations	4
Statements of Changes in Member's Capital	5
Statements of Cash Flows	6
Notes to Financial Statements	7–13



INDEPENDENT AUDITOR'S REPORT

Deloitte & Touche LLP 650 South Tryon Street Suite 1800 Charlotte, NC 28202 USA

Tel: +1 704 887 1500 Fax: +1 704 887 1570 www.deloitte.com

To LG AS Franchisor LLC:

Opinion

We have audited the financial statements of LG AS Franchisor LLC (the "Company"), which comprise the balance sheets as of December 31, 2024, and 2023, and the related statements of operations, changes in member's capital, and cash flows, for each of the three years in the period ended December 31, 2024, and the related notes to the financial statements (collectively referred to as the "financial statements").

In our opinion, the accompanying financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2024, and 2023, and the results of its operations and its cash flows for each of the three years in the period ended December 31, 2024, in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Emphasis of Matter

As discussed in Notes 6 and 7 to the financial statements, LG AS Franchisor LLC, in the ordinary course of business, transacts with affiliated companies. Our opinion is not modified in respect to this matter.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for one year after the date that the financial statements are available to be issued.

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due to fraud
 or error, and design and perform audit procedures responsive to those risks. Such procedures include
 examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting
 estimates made by management, as well as evaluate the overall presentation of the financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that
 raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period
 of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

Deloitte & Touche LLP

May 7, 2025

BALANCE SHEETS AS OF DECEMBER 31, 2024 AND 2023

ASSETS	2024	2023
CASH AND CASH EQUIVALENTS	\$ 961,842	\$ 427,010
ROYALTY AND MARKETING FEE RECEIVABLES	625,591	569,083
DEVELOPMENT INCENTIVE NOTES RECEIVABLE	645,000	-
PREPAID EXPENSES AND OTHER ASSETS	91,150	79,479
OFFICE FURNITURE AND EQUIPMENT—Net	1,081	9,777
FRANCHISE AGREEMENTS—Net	1,038,191	1,158,087
TRADEMARKS—Net	583,532	728,528
TOTAL ASSETS	\$3,946,387	\$2,971,964
LIABILITIES AND MEMBER'S CAPITAL		
LIABILITIES: Accounts payable and accrued liabilities Deferred revenue Due to AS Manager LLC	\$ 418,871 3,353,429 81,147	\$ 412,077 2,128,345 284,804
Total liabilities	3,853,447	2,825,226
MEMBER'S CAPITAL	92,940	146,738
TOTAL LIABILITIES AND MEMBER'S CAPITAL	\$3,946,387	\$2,971,964

STATEMENTS OF OPERATIONS FOR THE YEARS ENDED DECEMBER 31, 2024, 2023, AND 2022

	2024	2023	2022
REVENUES:			
Royalties	\$ 2,983,473	\$ 2,216,963	\$ 1,691,771
Marketing fees	917,176	522,375	366,964
Other revenues	2,985	2,520	2,490
Total revenues	3,903,634	2,741,858	2,061,225
OPERATING EXPENSES:			
General and administrative expense			
reimbursement—AS Manager LLC	3,401,232	3,640,412	3,892,741
General and administrative	2,804,272	1,459,570	1,054,742
Brand development expenses	178,169	364,061	377,142
Depreciation and amortization	273,588	336,059	343,483
Total operating expenses	6,657,261	5,800,102	5,668,108
NET LOSS	\$(2,753,627)	\$(3,058,244)	\$(3,606,883)

See notes to financial statements.

FOR THE YEARS ENDED DECEMBER 31, 2024, 2023, 2022 AND 2021

MEMBER'S CAPITAL—December 31, 2021	\$ 979,559
Cash contributions	3,040,000
Cash distributions	(50,000)
Net loss	(3,606,883)
MEMBER'S CAPITAL—December 31, 2022	362,676
Cash contributions	2,962,306
Cash distributions	(120,000)
Net loss	(3,058,244)
MEMBER'S CAPITAL—December 31, 2023	146,738
Cash contributions	2,767,261
Cash distributions	(67,432)
Net loss	(2,753,627)
MEMBER'S CAPITAL—December 31, 2024	\$ 92,940

See notes to financial statements.

STATEMENTS OF CASH FLOWS FOR THE YEARS ENDED DECEMBER 31, 2024, 2023, AND 2022

	2024	2023	2022
CASH FLOWS FROM OPERATING ACTIVITIES: Net loss Adjustments to reconcile net loss to net cash used in operating activities:	\$ (2,753,627)	\$ (3,058,244)	\$ (3,606,883)
Depreciation and amortization Net change in assets and liabilities: Royalty and marketing fee receivables and	273,588	336,059	343,483
prepaid expenses	(68,179)	(418,265)	158,786
Development incentive notes receivable	(645,000)	-	-
Accounts payable and accrued liabilities	6,794	208,074	(35,794)
Deferred revenue	1,225,084	321,418	710,823
Due to AS Manager LLC	(203,657)	(826,206)	181,674
Net cash used in operating activities	(2,164,997)	(3,437,164)	(2,247,911)
CASH FLOWS FROM INVESTING ACTIVITIES— Office equipment additions			(2,317)
Net cash used in investing activities			(2,317)
CASH FLOWS FROM FINANCING ACTIVITIES:			
Cash contributions	2,767,261	2,962,306	3,040,000
Cash distributions	(67,432)	(120,000)	(50,000)
Net cash provided by financing activities	2,699,829	2,842,306	2,990,000
NET INCOPACE (DECORACE) IN CACH AND			
NET INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS	534,832	(594,858)	739,772
CASH AND CASH EQUIVALENTS—Beginning of year	427,010	1,021,868	282,096
CASH AND CASH EQUIVALENTS—End of year	\$ 961,842	\$ 427,010	\$ 1,021,868

NOTES TO FINANCIAL STATEMENTS AS OF DECEMBER 31, 2024 AND 2023 AND FOR THE YEARS ENDED DECEMBER 31, 2024, 2023, AND 2022

1. BUSINESS AND ORGANIZATION

General—LG AS Franchisor LLC ("Franchisor"), a Delaware limited liability company, was formed on October 29, 2018. Franchisor, having its principal office in Matthews, North Carolina, is a wholly owned subsidiary of LG AS Brand Parent LLC ("Brand Parent") whose majority members are investment funds sponsored and managed by Lindsay Goldberg & Company.

Franchisor was formed in anticipation of the January 10, 2019 acquisition by an affiliated entity of Franchisor, of certain assets of the Affordable Suites of America® (ASA) brand. Franchisor was assigned all of the rights, title, interest, duties, and obligations of 11 franchise agreements with unaffiliated third parties of ASA hotels located in North Carolina, Virginia, Georgia, and Indiana, one of which was under development as of December 31, 2019. Also, on January 10, 2019, Franchisor entered into 10 franchise agreements with LG AS TRS LLC ("TRS"), an affiliated entity of Franchisor, for ASA hotels located in North Carolina, South Carolina, and Virginia.

In January 2020, Franchisor announced the development of a new hotel brand, stayAPT® Suites ("stayAPT"), which offers fully furnished apartment-style suites that target long-term, value-conscious travelers. The hotels are expected to have an average room count of approximately 75 rooms. ASA hotels have been operating since 1998 and offer fully furnished apartment-style suites that target long-term, economy-minded travelers. Franchisor owns and licenses the brand and brand names to affiliated franchisees (owned by TRS) and to nonaffiliated third-party franchisees (see Note 7).

A summary of the execution of franchise agreements, including future locations, is as follows:

	Affi	liated	Nona	ffiliated	T	otal
	ASA	stayAPT	ASA	stayAPT	ASA	stayAPT
FRANCHISE AGREEMENTS—December 31, 2021	10	24	14	6	24	30
New agreements executed	2	8	-	10	2	18
FRANCHISE AGREEMENTS—December 31, 2022	12	32	14	16	26	48
New agreements executed		6	1	4	1	_10
FRANCHISE AGREEMENTS—December 31, 2023	12	38	15	20	27	58
New agreements executed Agreements terminated	- -		3 	5 1	3	5 3
FRANCHISE AGREEMENTS—December 31, 2024	12	36	18	24	30	60

In addition to the franchise agreements in the table above, as of December 31, 2024, Franchisor had also entered into development agreements related to 40 potential future franchise locations.

A summary of the number of properties on December 31, 2024 is as follows:

	Affiliated		Nonaf	filiated	
	Under		Under Un		
	Operational	Construction	Operational	Construction	Total
Affordable Suites of America™	12	-	18	-	30
stayAPT™ Suites	23	4	9	5	41

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Basis of Presentation—The accompanying financial statements of Franchisor have been prepared in accordance with accounting principles generally accepted in the United States of America ("US GAAP"), and include the financial position, results of operations, changes in member's capital and cash flows of Franchisor. For these financial statements, the statements of operations, statements of changes in member's capital, and the statements of cash flows are presented for the years ended December 31, 2024, 2023, and 2022.

Management anticipates that there may be future losses as well as payments under development incentive notes which will require further cash contributions to be made in the 12-month period from the date the accompanying financial statements were available to be issued. Franchisor has obtained a letter of support from Brand Parent that it will provide all necessary capital contributions through May 31, 2026 to allow the Company to meet its obligations as they come due.

Use of Estimates—The preparation of the accompanying financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, disclosures of contingent assets and liabilities at the date of the financial statements, and the reported amounts of revenues and expenses during the reporting period. Significant estimates include the estimated useful lives of tangible and intangible assets as well as the assessment of tangible and intangible assets for impairment. Actual results could differ from those estimates.

Cash and Cash Equivalents—Franchisor considers all cash on hand, demand deposits with financial institutions, credit card receivables, and short-term, highly liquid investments with original maturities of three months or less to be cash equivalents. Franchisor has deposits in excess of \$250,000 with financial institutions that are not insured by the Federal Deposit Insurance Corporation. Franchisor does not believe cash and cash equivalents expose it to significant credit risk.

Accounts Receivable and Allowance for Doubtful Accounts—Provision for doubtful accounts is made when collection of receivables is considered doubtful. Balances are considered past due when payment is not received by the contractual due date. No provision for doubtful accounts was recorded at December 31, 2024 or 2023. When management determines that receivables are uncollectible, they are written off against the allowance for doubtful accounts.

Office Furniture and Equipment—Office furniture and equipment additions are recorded at cost. Major improvements that extend the life or utility of office furniture and equipment are capitalized and depreciated over a period equal to the shorter of the estimated useful life of the improvement or the

remaining estimated useful life of the asset. Ordinary repairs and maintenance are charged to expense as incurred.

Depreciation and amortization are recorded on a straight-line basis over the following estimated useful lives:

Office furniture and equipment

3–7 years

Management assesses whether there has been impairment of the value of office furniture and equipment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. Franchisor did not recognize impairment charges related to office furniture and equipment for the years ended December 31, 2024, 2023, and 2022.

Intangible Assets and Liabilities—Intangible assets include trademarks and franchise agreements. Both the ASA trademarks and the ASA franchise agreements are considered finite-lived intangible assets and are amortized over their remaining expected useful lives. Finite-lived intangible assets are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount of the intangible asset may not be recoverable. At such time, Franchisor reassesses their useful lives.

No impairment charges related to intangible assets were recognized during the years ended December 31, 2024, 2023, and 2022.

Revenue Recognition—The primary source of revenue from franchising hotels is ongoing royalty fees, which are typically a percentage of gross room revenues of each franchised hotel. Franchisor recognizes royalty fee revenues over time as and when the underlying sales occur, and the franchisee derives value from the use of the brand name.

Franchisor's franchise agreements also require the payment of marketing fees, which are a percentage of gross room revenues of each franchised hotel, into a marketing fund which is intended to be used for the general promotion of the brand. Although Franchisor is generally contractually obligated to spend the marketing fees it collects from franchisees, there is not a separate performance obligation and therefore marketing fees are recognized as revenue over time when the underlying sales occur. Marketing costs are expensed as incurred.

Franchisor receives non-refundable initial franchise fees, which are initially recorded as deferred revenue, and are subsequently recognized as royalty revenues over the initial non-cancellable period of the franchise agreement, commencing when a franchised hotel opens for business. At December 31, 2024 and 2023, \$1,963,429 and \$1,938,345, respectively, of franchise fees received were recorded on the accompanying balance sheets. Upon termination of a franchise agreement, the initial franchise fee would be recognized as revenue. For the yar ended December 31, 2024, \$70,000 of initial franchise fees were recognized as royalty revenue from franchise agreements with affiliated entities which were terminated.

Franchisor also receives non-refundable development fees, which are initially recorded as deferred revenue, and are subsequently applied against initial franchise fees upon execution of the franchise agreement. Upon determination that an associated franchise agreement will not be entered into, the development fee would be recognized as revenue. At December 31, 2024 and 2023, \$1,390,000 and \$190,000, respectively, of development fees received were recorded on the accompanying balance sheets.

Advertising Costs—Advertising costs are expensed as incurred. For the year ended December 31, 2024, 2023, and 2022, advertising costs were approximately \$1,122,000, \$120,500, and \$274,000 respectively, and are included in general and administrative expenses in the accompanying statements of operations.

Fair Value of Financial Instruments—US GAAP establishes a three-level valuation hierarchy based upon observable and unobservable inputs for fair value measurement of financial instruments:

Level 1—Observable inputs, such as quoted prices in active markets at the measurement date for identical assets or liabilities

Level 2—Significant inputs that are observable, directly or indirectly, such as other quoted prices in markets that are not active or inputs which are observable, either directly or indirectly, for substantially the full term of the asset or liability

Level 3—Significant unobservable inputs for which there is little to no market data and for which Franchisor makes its own assumptions about how market participants would price the asset or liability.

Fair value is defined as the price that would be received when selling an asset or the price paid to transfer a liability in an orderly transaction between market participants at the measurement date (exit price). In instances where inputs used to measure fair value fall into different levels of the fair value hierarchy, the level in the fair value hierarchy within which the fair value measurement in its entirety has been determined is based on the lowest-level input significant to the fair value measurement in its entirety. Franchisor's assessment of the significance of a particular input to the fair value measurement in its entirety requires judgment and considers factors specific to the asset or liability.

Franchisor's financial instruments consist of cash and cash equivalents, royalty and marketing fee receivables, accounts payable and due to AS Manager LLC. The carrying value of cash and cash equivalents, accounts receivable, and accounts payable are representative of their fair value due to the short-term nature or frequent settlement of these instruments.

Income Taxes—No provision for income taxes has been reflected in the accompanying financial statements because Franchisor is a disregarded entity and is not subject to federal income taxes. For federal income tax purposes, the operating results of Franchisor are reportable by its member. Franchisor may be subject to state and local taxes in certain jurisdictions.

Subsequent Events—Franchisor has evaluated events and transactions occurring after December 31, 2024, through May 7, 2025, the date that the accompanying financial statements were available to be issued, for potential recognition or disclosure. Under a development incentive agreement with a non-affiliated franchisee, Franchisor paid \$750,000 to the franchisee in February, 2025 and issued a development incentive note in return as described in Note 7.

3. OFFICE FURNITURE AND EQUIPMENT

As of December 31, 2024 and 2023, office furniture and equipment—net consists of the following:

	2024	2023
Office furniture and equipment Less accumulated depreciation	\$162,471 (161,390)	\$162,471 (152,694)
Office furniture and equipment—net	\$ 1,081	\$ 9,777

4. INTANGIBLE ASSETS

Vears Ending

As of December 31, 2024, and 2023, intangible assets consist of the following:

2024	Estimated Useful Life	Gross Carrying Amount	Accumulated Amortization	Net Book Value
Definite-lived intangible assets: Trademarks Franchise agreements	10 years 3–20 years	\$1,450,000 1,890,000	\$ (866,468) (851,809)	\$ 583,532 1,038,191
Total intangible assets		\$3,340,000	\$(1,718,277)	\$1,621,723
2023	Estimated Useful Life	Gross Carrying Amount	Accumulated Amortization	Net Book Value
2023 Definite-lived intangible assets: Trademarks Franchise agreements	Useful	Carrying		

The weighted-average amortization period remaining for definite-lived intangible assets was approximately 8.1 years as of December 31, 2024. Estimated future amortization expense for intangible assets is as follows:

December 31	
2025	\$ 265,000
2026	251,000
2027	251,000
2028	251,000
2029	98,532
Thereafter	505,191
Total	\$1,621,723

5. EQUITY

Franchisor has one class of member interests authorized. Franchisor's board of managers may authorize other classes or series of units from time to time and determine the terms and conditions applicable to such new units.

6. AGREEMENTS WITH AFFILIATES

The following describes Franchisor's agreement related to expense reimbursements with an affiliate. Franchise agreements with affiliated companies are described in Note 7.

General and Administrative Expense Reimbursement Agreement—Effective January 10, 2019, Franchisor executed a general and administrative expense reimbursement agreement with an affiliate, AS Manager LLC ("Manager"). Under the terms of this agreement, Manager provides services for Franchisor's management and administrative functions, including operations, marketing, legal, finance, accounting, human resources, risk management, technology, and other related services. The agreement expires on December 31, 2029 but may be terminated by either party at any time for any reason upon notice. Fees charged to Franchisor under this agreement represent the actual cost incurred by Manager in performing these services for Franchisor.

Under the terms of this agreement, Franchisor recognized \$3,401,232, \$3,640,412, and \$3,892,741 of expense in the statements of operations for the years ended December 31, 2024, 2023, and 2022, respectively, and amounts due to Manager were \$91,545 and \$189,764 as of December 31, 2024, and 2023, respectively.

Franchise Agreements with Affiliated Individuals—Franchisor has franchise agreements with two ASA locations which are partially-owned by an affiliated individual. Franchisor recognized total revenue of \$73,516, \$67,149, and \$49,075 for the years ended December 31, 2024, 2023, and 2022, respectively, under the associated agreements. Franchisor also has franchise agreements with two stayAPT locations which are wholly-owned by an affiliated individual. Franchisor recognized revenue of \$6,331 from one location that began operating on June 24th 2024; the other remains under development. Revenue was recognized under terms consistent with other franchised locations.

7. FRANCHISE AGREEMENTS

Franchisor is the owner of the trademarks under the ASA and stayAPT names and other trademarks, trade names, service marks, logos, and commercial symbols (the "Marks") and has licensed the use of the Marks to affiliated and non-affiliated third-party franchisees under franchise agreements. Franchisor also enters into development agreements with potential franchisees which grant the right to develop franchises in specific locations or territories.

Franchisor recognizes royalty fees ranging from 1% to 5% of gross room revenues from both its affiliate (TRS) and non-affiliate 3rd party owners. For the years ended 2024, 2023, and 2022, Franchisor recognized affiliate revenues of \$2,006,675, \$1,401,343, and \$968,422, respectively, and non-affiliate 3rd party owner revenue of \$976,798, \$815,620, and \$723,349, respectively.

An incentive was offered to non-affiliated franchisees signing new agreements in 2024, and existing non-affiliated franchisees with anticipated hotel openings in 2024 and 2025. This incentive reduced the royalty fee percentage from 5% to 1% for 2024 and 2025, increasing 1% each year to 5% in 2029 and for the remainder of the franchise agreement.

Franchisor also recognized marketing fees ranging from 0% to 2% of gross rooms revenue. Franchisor is contractually obligated to expend marketing fees it collects from franchisees for marketing purposes such as advertising, promotion, and training for the respective franchisees, and is required to provide certain services to its franchisees, including technology and purchasing programs. Franchisees are obligated to operate their hotels under the Marks and conform to brand operating standards. For the years ended 2024, 2023, and 2022, Franchisor recognized affiliate revenues of \$666,885, \$464,886, and \$316,148, respectively, and non-affiliate 3rd party owner revenue of \$250,291, \$57,489, and \$50,816, respectively.

Franchisor has entered into development incentive agreements with certain non-affiliated franchisees, in accordance with which it may provide development incentive notes to assist with hotel development. Provided the franchisee is in compliance with the terms of the franchise agreement, the notes may be forgiven by Franchisor at 5% per year over the twenty-year life of the associated franchise agreement. Otherwise, the related principal is due and payable to Franchisor. Amounts forgiven are recorded as a reduction of royalty revenues.

At December 31, 2024 and 2023, \$645,000 and \$0, respectively, of development incentive notes receivable are recorded on the accompanying balance sheets. Under existing agreements, Franchisor, provided associated terms are met, expects to make future payments for development incentive notes of \$1,750,000 in 2025.

8. COMMITMENTS AND CONTINGENCIES

Legal Contingencies—Franchisor is not a party to any litigation or claims, other than routine matters arising in the ordinary course of business, that are incidental to the operation of the business of Franchisor. Franchisor believes that the results of all claims and litigation, individually or in the aggregate, will not have a material adverse effect on its business, financial position, or results of operations.

* * * * * *

Unaudited Financial Statements

These financial statements have been prepared without an audit. Prospective franchisees or Sellers of Franchises should be advised that no independent certified public accountant has audited these figures or expressed an opinion with regard to their content or form.

TOTAL LIABILITIES AND MEMBER'S CAPITAL

BALANCE SHEET AS OF FEBRUARY 28, 2025

Α	SS	Έī	ΓS

CASH AND CASH EQUIVALENTS	\$1,342,422
ROYALTY AND MARKETING FEE RECEIVABLES	418,262
DEVELOPMENT INCENTIVE NOTES RECEIVABLE	1,395,000
PREPAID EXPENSES AND OTHER ASSETS	80,750
OFFICE FURNITURE AND EQUIPMENT—Net	658
FRANCHISE AGREEMENTS—Net	1,018,119
TRADEMARKS—Net	559,366
TOTAL ASSETS	\$4,814,577
LIABILITIES AND MEMBER'S CAPITAL	
LIABILITIES: Accounts payable and accrued liabilities Deferred revenue Due to AS Manager LLC	\$ 360,038 3,693,659 160,164
Total liabilities	4,213,861
MEMBER'S CAPITAL	600,716

\$4,814,577

STATEMENT OF OPERATIONS FOR THE PERIOD JANUARY 1, 2025 TO FEBRUARY 28, 2025

REVENUES:	
Royalties	\$ 536,546
Marketing fees	194,574
Other revenues	671
Total revenues	731,791
OPERATING EXPENSES:	
General and administrative expense	
reimbursement—AS Manager LLC	622,117
General and administrative	338,398
Brand development expenses	17,231
Depreciation and amortization	44,662
Total operating expenses	1,022,408
NET LOSS	\$ (290,617)

STATEMENT OF CHANGES IN MEMBER'S CAPITAL FOR THE PERIOD JANUARY 1, 2025 TO FEBRUARY 28, 2025

MEMBER'S CAPITAL—December 31, 2024	141,333
Cash contributions	750,000
Cash distributions	-
Net loss	(290,617)
MEMBER'S CAPITAL—February 28, 2025	\$ 600,716

STATEMENT OF CASH FLOWS FOR THE PERIOD JANUARY 1, 2025 TO FEBRUARY 28, 2025

CASH FLOWS FROM OPERATING ACTIVITIES:	
Net loss	\$ (290,617)
Adjustments to reconcile net loss to net cash	
used in operating activities:	
Depreciation and amortization	44,662
Net change in assets and liabilities:	,
Royalty and marketing fee receivables and	
prepaid expenses	217,728
Development incentive notes receivable	(750,000)
Accounts payable and accrued liabilities	(58,833)
Deferred revenue	340,229
Due to AS Manager LLC	127,411
bue to no manager LLC	
Net cash used in operating activities	(369,420)
Net eash used in operating activities	(303,420)
CACLLELOVACE EDOMAINIVECTING A CTIVITIES	
CASH FLOWS FROM INVESTING ACTIVITIES—	
Net cash used in investing activities	_
Net cash used in investing activities	
CACLLELOVACE FROM FINIANICINIC ACTIVITIES.	
CASH FLOWS FROM FINANCING ACTIVITIES:	750,000
Cash contributions Cash distributions	750,000
Casil distributions	
Not each provided by financing activities	750,000
Net cash provided by financing activities	750,000
NET INCREASE (DECREASE) IN CASH AND	
CASH EQUIVALENTS	380,580
CASH AND CASH FOLINAL FAITS Degianing of ported	061.043
CASH AND CASH EQUIVALENTS—Beginning of period	961,842
CACILAND CACILEOUIIVALENTS. Fod of ported	ć 1 242 422
CASH AND CASH EQUIVALENTS—End of period	\$ 1,342,422

EXHIBIT D

Franchise Agreement



FRANCHISE AGREEMENT

BETWEEN

LG AS FRANCHISOR LLC

10801 Monroe Road Matthews, NC 28105 Telephone: 980-368-8100

AND

[Name of Franchisee(s)]

[Street Address] [City, State Zip] [Phone]

FOR

FRANCHISED LOCATION:

[Site Address]
[City, State Zip]

DATE OF FRANCHISE AGREEMENT ("Effective Date")
, 20

Hotel	#:		

LG AS FRANCHISOR LLC FRANCHISE AGREEMENT TABLE OF CONTENTS

ARTICLE 1 FRANCHISED LOCATION; GRANT OF FRANCHISE	1
ARTICLE 2 TERM OF FRANCHISE	4
ARTICLE 3 OWNERSHIP AND USE OF MARKS AND BUSINESS SYSTEM	4
ARTICLE 4 FEES	5
ARTICLE 5 ADVERTISING AND MARKETING	8
ARTICLE 6 FINANCIAL STATEMENTS, SALES REPORTS, FORMS AND ACCOUNTI 10	NG
ARTICLE 7 QUALITY CONTROL, UNIFORMITY AND STANDARDS REQUIRED OF T	
ARTICLE 8 INSURANCE AND INDEMNIFICATION	21
ARTICLE 9 CONFIDENTIAL OPERATIONS MANUAL AND OTHER INFORMATION	24
ARTICLE 10 INITIAL TRAINING PROGRAM; FRANCHISOR ASSISTANCE	27
ARTICLE 11 FRANCHISOR'S OBLIGATIONS	28
ARTICLE 12 SITE SELECTION; BUILDING PLANS	28
ARTICLE 13 FRANCHISOR'S TERMINATION RIGHTS	30
ARTICLE 14 FRANCHISEE'S TERMINATION RIGHTS	33
ARTICLE 15 FRANCHISEE'S OBLIGATIONS UPON TERMINATION	33
ARTICLE 16 NON-SOLICITATION	36
ARTICLE 17 DISPUTE RESOLUTION	36
ARTICLE 18 FRANCHISOR'S RIGHT OF FIRST REFUSAL TO PURCHASE	40
ARTICLE 19 <u>ASSIGNMENT</u>	41
ARTICLE 20 INDEPENDENT CONTRACTORS; INDEMNIFICATION; ATTORNEYS' FEES 45	
ARTICLE 21 EMINENT DOMAIN; DESTRUCTION OR DAMAGE	46
ARTICLE 22 GENERAL PROVISIONS	46
ARTICLE 23 DEFINITIONS	49
ARTICLE 24 ACKNOWLEDGMENTS AND COUNTERPARTS	51

EXHIBITS

- A. Personal Guaranty
- B.
- Ownership & Management Addendum
 Assignment of Telephone Numbers and Service C.
- Acknowledgment Addendum D.
- State Addendum (if applicable) E.

LG AS FRANCHISOR LLC FRANCHISE AGREEMENT

THIS FRANCE		ement") is made, entered into and effective LG AS FRANCHISOR LLC, a Delaware
	ny ("Franchisor"), and	, a, a
	WITNESSET	TH:
establish and operate standards, specificatio (as defined in Section service marks, logos a	extended-stay hotels of a disns, methods and procedures at 23.1) under the name StayAF	ts franchises to others for the right to stinctive character and quality utilizing and other aspects of the Business System of the Business System and other trademarks, trade names, nated to be part of the Business System rks"); and
human resources to d Marks and using the desires to acquire from and operate one Brand	evelop an extended-stay hotel Business System ("Brand Hot n Franchisor the right to use the	t it would take substantial capital and similar to the hotels operated under the els"), and as a consequence, Franchisee Marks and Business System and to own as the "Hotel") subject to and under the
this Agreement and to other personal advisor System, the financial i	be thoroughly advised of its tes, and has had sufficient time	adequate opportunity to read and review terms and conditions by legal counsel or to evaluate and investigate the Business the hotel industry and the business risks
		the mutual promises and covenants set uable consideration, the parties agree as
FDA	ARTICLE NCHISED LOCATION; GR	
this Agreement, a nor Franchisor's mandator procedures and technique changed, improved	EHISED LOCATION. Franchexclusive personal right to ory standards and specification ues prescribed in the Manual,	hisor grants to Franchisee, for the term of perate the Hotel in conformity with the ns, including the operational standards, and to use the Business System (as it may tranchisor) and the Marks in connection
Street		
City	State	Zip Code

Franchisee shall operate the Hotel at, and only at, the Franchised Location. If Franchisee's site application has not been approved at the time of this Agreement, this Article shall be automatically amended to list the Franchised Location upon the unconditional approval of the site application by Franchisor.

If Franchisee has not yet submitted a site application for a Franchised Location at the time this Agreement is executed, Franchisee must submit such application within 60 days of the Effective Date. If the initial site application is denied by Franchisor, or should the approved location not be acquired by Franchisee or become redundant for any reason, Franchisee may submit a replacement site application along with a non-refundable application fee of \$5,000. Franchisee will have no more than 60 days from the date that (a) Franchisor notifies Franchisee that the initial site application was denied; or (b) Franchisee first communicated to Franchisor its intention to abandon development of the previously approved site; to submit a replacement site application. Franchisor shall have 60 days after receipt of Franchisee's complete replacement site application (including all information and materials required by Franchisor for site review) to approve or disapprove the replacement site application for any reason. If a replacement site application is submitted, it will be approved or disapproved under the same terms and conditions as the initial site application. Franchisor shall have no obligation to consider more than two site applications under this Agreement. Franchisee's failure to timely submit a site application or obtain site approval may result in termination of this Agreement.

NONEXCLUSIVE. "Nonexclusive," for the purposes of this Agreement, means 1.2 that Franchisor, its affiliates or related companies with common ownership have the absolute and unconditional right: (a) outside of the Protected Area (as defined in Article 1.3 below), to establish franchised or company-owned hotels (or other businesses) or to offer, sell or distribute products or services associated with the Business System under the Marks or any other trademarks or through any distribution channel or method, all without compensation to any franchisee; (b) within and outside of the Protected Area, to establish franchised or company owned hotels (or other businesses) or to offer, sell or distribute products or services under trademarks other than the Marks, or to offer, sell or distribute any products or services through channels of distribution other than franchised or company-owned hotels operating under the Marks, including through the Internet or other forms of electronic communication; (c) acquire, be acquired by, sell all or substantially all of its assets, merge, go public, engage in a private placement, affiliate with or engage in any transaction with other businesses (whether competitive or not), with any hospitality business or other business, located anywhere or business conducted anywhere, undertake a refinancing, recapitalization, leveraged buyout or other economic or financial restructuring and (d) to any and all other rights not expressly granted to the Franchisee in this Agreement.

1.3 PROTECTED AREA.

(A) Notwithstanding the provisions of Articles 1.1 and 1.2 above, Franchisor will not open or operate, or grant to others the right to open or operate, a Brand Hotel, without the permission of Franchisee, within a _____ mile radius (the "Protected Area") from the Franchised Location, provided that nothing herein restricts the Franchisor or any other party to advertise one or more Brand Hotels or the Business

System within the Protected Area, or to open, operate, or grant others the right to open or operate Brand Hotels with protected areas that overlap the Protected Area, as long as no other Brand Hotels are located in Franchisee's Protected Area during the term of this Agreement.

- If Franchisor or any of its affiliates acquire any chain or system of properties, or multiple units (more than four) of a chain or system, and such purchase includes properties in the Protected Area which Franchisor or the acquiring affiliate desires to convert to Brand Hotels (hereinafter, whether one or more, "Conversion Hotels"), Franchisor shall provide written notice to Franchisee within a reasonable time of its intent to convert the Conversion Hotels into Brand Hotels. Such notice shall provide Franchisee with a right of first refusal to acquire such Conversion Hotels from Franchisor on the terms provided below if the sale by Franchisor of such properties to Franchisee is allowed by contract and applicable law. Subject to the foregoing, Franchisee shall have the right and option, exercisable within 90 days after receipt of such written notification, which notification shall include the proposed purchase price to provide written notice to Franchisor that Franchisee desires to purchase the Conversion Hotels and to convert all of such properties to Brand Hotels under the Business System. In the event Franchisee elects to purchase and convert the Conversion Hotels, Franchisee must close on such purchase and execute the Franchisor's then current franchise agreement for each of the Conversion Hotels (which shall require payment of the initial franchise fee due under such franchise agreement) within 60 days from the date of notice to Franchisor of Franchisee's election to purchase and convert. The purchase price to be paid by Franchisee for the Conversion Hotels shall be the cash equivalent of the fair market value for each of the Conversion Hotels, as determined by an independent appraiser selected and retained by Franchisor. In the event that Franchisee does not elect to purchase and convert the Conversion Hotels as provided in this paragraph, Franchisee shall have no further right or option to acquire such Conversion Hotels, and Franchisor may sell such Conversion Hotels to another franchisee under the Business System or Franchisor or its affiliates may own and operate the Conversion Hotels under the Business System.
- 1.4 FRANCHISOR'S RIGHT TO REJECT FRANCHISEE. Franchisor will have the unilateral right to reject Franchisee and declare this Agreement null and void if Franchisor determines that any financial, personal or other information provided by Franchisee to Franchisor is materially false, misleading, incomplete or inaccurate, or if Franchisee is not qualified or competent to properly operate the Hotel, or Franchisee has not successfully completed the training program.
- 1.5 <u>CONDITIONS TO FRANCHISE.</u> Franchisee undertakes the obligation to operate the Hotel under the Marks and Business System at the Franchised Location in strict compliance with the terms and conditions of this Agreement. The rights and privileges granted to Franchisee by Franchisor under this Agreement are applicable only to the Franchised Location, are personal in nature, and may not be used elsewhere or at any other location by Franchisee. Franchisee will not have the right to franchise, subfranchise, license, or sublicense its rights under this Agreement, or to exclude, control or impose conditions on Franchisor's development of future hotels at any time or at any location. No part of the Hotel may be offered

for sale, lease or otherwise as condominium or timeshare units, or otherwise used for any other purpose than the operation of a Brand Hotel.

ARTICLE 2 TERM OF FRANCHISE

2.1 TERM. The term of this Agreement will be for a period of 20 years, commencing on the Effective Date (the "Term"). This Agreement will not be considered executed and will not be enforceable until it has been signed by Franchisor and Franchisee, and, if Franchisee is an entity, the applicable Principal Owners sign a Personal Guaranty as described in Article 22.11. This Agreement expires on the last day of the Term, and Franchisee has no right to extend or renew this Agreement or the rights granted under it following its expiration.

If Franchisee continues to operate the Hotel using the Marks with Franchisor's express or implied consent following the expiration of the Term, as the case may be, such continued operation will be considered a month-to-month extension of this Agreement, unless otherwise agreed in writing ("Continuation Term"). All provisions of this Agreement, including all exhibits and each Personal Guaranty, will apply while Franchisee continues to operate the Hotel during the Continuation Term. The Continuation Term of this Agreement will be terminable by either party on thirty (30) days' written notice to the other party.

ARTICLE 3 OWNERSHIP AND USE OF MARKS AND BUSINESS SYSTEM

3.1 OWNERSHIP OF MARKS. Franchisor or its affiliates own all right, title and interest in and to the Marks and Business System and the goodwill symbolized by them. Franchisee expressly disclaims any right, title or interest therein or in any goodwill derived therefrom. Any and all improvements made by Franchisee relating to the Marks and Business System will become the property of Franchisor, who will have the exclusive right to register and protect all such improvements in its name in accordance with applicable law. Franchisee may not license, sublicense, or allow the Marks or Business System, or any part thereof, to be used by any other person, company or group without Franchisor's express, prior written consent.

3.2 USE OF MARKS.

(A) Franchisee will use the Marks and Business System only in the manner authorized and specified by Franchisor in the Manual (as defined in Section 23.7) or otherwise. Franchisee shall not, directly or indirectly, at any time during the Term or thereafter, do, cause, or suffer to be done any act or thing disputing, attacking, or in any way impairing or tending to impair the right, title, or interest of Franchisor in the Marks, the Business System or the goodwill therein. Any unauthorized use by Franchisee of the Marks and Business System will constitute an infringement of Franchisor's rights and will constitute a material default under this Agreement. Franchisee's right to use and identify with the Marks and Business System will exist concurrently with the Term. Franchisee will not allege any ownership in the Marks or Business System.

- (B) All use of the Marks in electronic commerce, which includes all forms of electronic or computer communication, must comply with the requirements set forth in the Manual. Franchisor may require that various types of marketing or advertising utilize a specific template or format. Franchisee must provide Franchisor with copies of all proposed uses of any of the Marks or any variation thereof for use in and for electronic commerce, including Franchisee's Internet or Website address and domain name. Franchisee must obtain Franchisor's prior written approval to file any such application, which Franchisor may withhold in its sole discretion. Upon expiration or termination of this Agreement, Franchisee agrees to transfer its Internet or Website addresses and domain names to Franchisor upon Franchisor's written request. Franchisee will not receive any compensation for such transfer.
- 3.3 ADVERSE CLAIMS TO MARKS. Franchisee shall immediately notify Franchisor in writing of all infringements or imitations of the Marks of which Franchisee becomes aware, and Franchisor shall exercise absolute discretion in deciding what action, if any, should be taken. Franchisee shall fully cooperate with Franchisor in the prosecution of any action to prevent the infringement, imitation, or illegal use of the Marks and agrees to be named as a party in any such action at Franchisor's request. Franchisor shall bear any and all legal expenses incident to Franchisee's participation, at Franchisor's request, in any action to prevent the infringement or illegal use of the Marks, except for the cost of any legal counsel separately retained by Franchisee. Except as expressly provided in this Article, Franchisor shall not be liable to Franchisee for any damages, costs, expenses, loss of profits or business opportunities, or incidental or consequential damages of any kind or nature whatsoever relating to any action involving the Marks. If there is a claim by any third party that its rights to any or all of the Marks are superior to those of Franchisor and if Franchisor is of the opinion that such claim is legally meritorious, or if there is an adjudication by a Court of competent jurisdiction that any third party's right to the Marks are superior to those of Franchisor, then upon receiving written notice from Franchisor, Franchisee will, at its expense, immediately make such changes and amendments to the Marks as may be specified by Franchisor. Franchisee will not make any changes or amendments whatsoever to the Marks or Business System unless specified by Franchisor in writing.
- **3.4 SUBSTITUTION OF MARKS**. Franchisor reserves the right to modify the Marks or to substitute different trademarks, trade names, service marks, logos, designs, and commercial symbols as the Marks used to identify the Brand Hotels. In that event, upon receipt of written notice from Franchisor, Franchisee will, at its expense, immediately make all modifications to the Marks specified by Franchisor.

ARTICLE 4 FEES

4.1 INITIAL FRANCHISE FEE. Franchisee will pay to Franchisor an Initial Franchise Fee of \$40,000. The Initial Franchise Fee will be due and payable on the date this Agreement is executed by Franchisee. The Initial Franchise Fee is fully earned upon receipt and is nonrefundable.

- **4.2 ROYALTY FEES.** Franchisee will, for the entire Term, pay to Franchisor monthly Royalty Fees equal to the greater of \$2,500 or 5% of Franchisee's monthly Gross Room Revenues, as defined in Section 23.5, which are received, billed or generated by, at, as a result of, or from the Hotel.
- 4.3 <u>ADMINISTRATIVE ASSISTANCE FEES.</u> If Franchisee requests and Franchisor or its affiliates provide additional administrative services to Franchisee, including, but not limited to: assistance with closings of financing transactions or other transactions relating to the Hotel; negotiations of comfort letters, non-disturbance agreements, and other instruments, documents and agreements with Franchisee's lenders or prospective lenders, lender's counsel, Franchisee's counsel, any other Franchisee representative, or third party; conducting research related to the Hotel and its operation; preparation of documents, instruments or agreements; and other project-based tasks, Franchisee agrees to pay to Franchisor a reasonable fee, as determined by Franchisor, for such services and to reimburse Franchisor and its affiliates for any costs (including attorney's fees) incurred in connection with the provision of such services.
- 4.4 TECHNOLOGY FEES. Franchisee will pay such set-up and maintenance fees as Franchisor charges from time to time, for web and email set-up, maintenance and hosting, and for such other technology support as the Franchisor may be providing to Franchisee. As of the Effective Date the initial web set-up fee is between \$1,800 and \$2,500 and the web maintenance and domain hosting fee is between \$300 and \$600, in each case depending on the scope of services provided to Franchisee. Because website, online and mobile technologies and their uses are constantly evolving, Franchisor reserves the right to require Franchisee's participation in other technologies being implemented as part of the Business System and charge reasonable fees therefor, in each case intended to off-set costs and expenses incurred by Franchisor or its affiliates in connection with such technologies and related services.
- 4.5 <u>CONSTRUCTION EXTENSION FEE</u>. Franchisee must complete construction of the Hotel within 24 months of the Effective Date. If franchisee believes it will be unable to do so, it may request up to four 30-day extensions for a fee of \$5,000 per extension. Any such extension will be granted only in Franchisor's sole discretion. Franchisee will pay Franchisor a fee of \$5,000 at the time the extension is granted, for each 30-day extension period requested. If the extension is not granted, the fee will be refunded.
- **4.6 PROPERTY IMPROVEMENT REVIEW FEE.** If Franchisee requests or Franchisor requires a property improvement review pursuant to Sections 7.7 and 7.15, Franchisee will pay Franchisor a fee of \$5,000 for such property improvement review. If the Hotel fails a property inspection conducted pursuant to Section 7.15 and Franchisor deems a re-inspection necessary, Franchisee will pay Franchisor a fee of \$2,500 for such re-inspection.
- **4.7** COSTAR SUBSCRIPTION/STR REPORT FEE. Franchisee must reimburse Franchisor for its cost to provide a CoStar platform benchmark subscription, inclusive of monthly and annual STR reports for the Hotel as incurred. The cost is currently between \$1,000 and \$3,000 per hotel per year depending on room count.
- **4.8 KIOSK FEE.** If Franchisor implements a check-in kiosk or similar technology to facilitate independent check-in or other independent guest access to services, Franchisee will pay

Franchisor or its designee an initial fee to install and set up the technology, as well as an ongoing maintenance and support fee for the technology.

- **4.9** OTHER FEES. Franchisee will pay such other fees that are set forth in other sections of this Agreement or otherwise imposed. Such fees shall be due as set forth in Section 4.9 of this Agreement, unless different payment terms are expressly stated for such fees at the time they are imposed or thereafter.
- terms to the contrary are expressly stated in this Agreement or otherwise, all payments required by Article 4, and all other payments due to Franchisor on a continuing basis ("Payments"), shall be due to Franchisor by the tenth (10th) day after the end of the calendar month in which such Gross Room Revenues were received by Franchisee, provided that, Franchisor may, upon notice to Franchisee, collect such payments more frequently than monthly. Franchisee's report of Gross Room Revenues required under Section 6.3 of this Agreement must be submitted to Franchisor on or before such date as Franchisor designates for the preceding month or reporting period, as the case may be. Franchisee acknowledges that nothing contained in this Article 4 shall constitute an agreement by Franchisor to accept such Payments after the same are due or a commitment by Franchisor to extend credit to, or otherwise finance Franchisee's operation of, the Hotel. Franchisee acknowledges that Franchisee's failure to pay all such amounts when due shall constitute grounds for termination of this Agreement, notwithstanding the provisions of this Article entitling Franchisor to interest on late Payments. The entitlement to such interest charges shall be in addition to any other remedies Franchisor may have.
- **METHOD OF PAYMENT**. All Payments shall be made by wire transfer, 4.11 electronic funds transfer or such other electronic transfer method as Franchisor may designate. In its sole discretion, Franchisor may collect Payments required by Article 4 by direct debit withdrawal by Franchisor from a designated bank account of Franchisee. Franchisee will, upon written request from Franchisor, execute such documents as Franchisor may request to provide Franchisee's authority and direction to its bank or financial institution authorizing and directing Franchisee's bank or financial institution to pay and deposit directly to the account of Franchisor, and to charge to the account of Franchisee, the amount of any Payment and other sums due and payable by Franchisee pursuant to this Agreement. The authorizations will be in the form prescribed by Franchisor's bank. Franchisee's authorizations will permit Franchisor to designate the amount to be debited or drafted from Franchisee's account and to adjust such amount from time to time, to the amount of the Payments and other sums then payable to Franchisor from Franchisee. Franchisee will, at all times during the term of this Agreement, maintain a balance in its account at its bank or financial institution sufficient to allow the appropriate amount to be debited from Franchisee's account for payment of the sums payable by Franchisee for deposit in the account of Franchisor.
- **4.12 INTEREST ON UNPAID AMOUNTS**. If Franchisee fails to pay any amounts owed to Franchisor by the due date, then the unpaid amounts due to Franchisor will bear interest at 18% per annum simple interest, or, if lower, at the maximum legal rate allowable in the state in which the Hotel is located. If Franchisee does not submit a report of Gross Room Revenues pursuant to Section 6.3, then Franchisor will have the right, in its sole judgment, to estimate the amount of the Royalty Fees, Marketing Fees and any other fees payable by Franchisee, and the

estimated unpaid monthly Royalty Fees, Marketing Fees and any other fees will bear interest at the rate set forth above.

4.13 NO RIGHT TO OFFSET. Franchisee will not, on grounds of the alleged nonperformance by Franchisor of any of its obligations under this Agreement, any other contract between Franchisor and its affiliates and Franchisee, or for any other reason, withhold payment of any Royalty Fees, Marketing Fees or any other fees or payments due to Franchisor under this Agreement or any other contract, promissory note or other obligation payable by Franchisee to Franchisor. Franchisee will not have the right to "offset" or withhold any liquidated or unliquidated amounts allegedly due to Franchisee from Franchisor against the Royalty Fees, Marketing Fees or any other payments due to Franchisor under this Agreement or any other contract, promissory note or other obligation payable by Franchisee to Franchisor.

ARTICLE 5 ADVERTISING AND MARKETING

- 5.1 PRE-OPENING ADVERTISING AND GRAND OPENING PROMOTION. Franchisee will conduct certain Franchisor approved advertising and public relations activities in connection with the opening of the Hotel. Franchisee shall obtain Franchisor's approval prior to conducting such advertising and public relations activities. Franchisee shall spend a minimum of \$10,000 for the Hotel if it has no more than 59 suites, and a minimum of \$15,000 if there are more than 59 suites.
- Franchisor nonrefundable monthly Marketing Fees of an amount equal to 2% of Franchisee's monthly Gross Room Revenues. The Marketing Fee will be deposited into a marketing fund ("Marketing Fund"). At Franchisor's discretion, the Marketing Fee may be reduced or increased upon not less than 30 days' notice to Franchisee, provided that the Marketing Fee will not exceed 3% of monthly Gross Room Revenues. Independent of the Marketing Fund, Franchisor may, at its option, charge Franchisee a reasonable fee on a project basis for marketing and advertising materials and projects prepared or undertaken for use by the Business System.
- **5.3** <u>USE OF MARKETING FEES</u>. Franchisor will use the Marketing Fund in its sole judgment for the payment of expenses incurred in connection researching, developing and preparing national, regional, point of sale, and local direct sales advertising, marketing strategy materials and the general promotion of the Brand Hotels, including:
 - (A) formulating, developing, implementing, maintaining and disseminating advertising, marketing, public relations, and promotional campaigns and materials including, without limitation, through radio, television, newspapers, magazines and online;
 - (B) developing, maintaining and updating Brand Hotel specific websites and social media presence, placing listings in national, regional or online directories of hotels:
 - (C) conducting market research, developing training, motivational and educational programs with respect to marketing, sales, advertising, customer service

and public relations and making such programs available to Brand Hotel franchisees and their employees; and

(D) administering the Marketing Fund, including maintaining a national sales and marketing staff, accounting expenses and the actual cost of salaries and fringe benefits paid to Franchisor's employees engaged in the administration of the Marketing Fund.

The methods of advertising, media employed, and contents, terms and conditions of advertising campaigns and promotional programs are within the sole judgment of Franchisor. Any Brand Hotels that are owned and operated by Franchisor will contribute Marketing Fees in the same manner as Brand Hotel franchisees; however, any hotel owned by an entity under common control with Franchisor will not be contractually required to contribute to the Marketing Fund. Franchisor is not required in its role as the franchisor to contribute to the Marketing Fund. The Marketing Fees paid by Franchisee are not refundable. Franchisee acknowledges that the intent of the Marketing Fund shall be to maximize general public recognition, direct sales programs, and acceptance of the Brand Hotels and Marks for the benefit of the Business System, and Franchisor shall have no obligation in administering the Marketing Fund, to make expenditures for Franchisee which are equivalent or proportionate to any payments by Franchisee, or to ensure that any particular Franchisee or any particular franchised location benefits directly or pro rata from advertising or promotion conducted under the Marketing Fund. The Marketing Fund is not a trust or escrow account, and Franchisor has no fiduciary obligation to franchisees with respect to the Marketing Fund; provided, however, that Franchisor will make a good faith effort to expend such fees in a manner that Franchisor determines is in the general best interests of the Business System. Franchisor has the right to advance monies to the Marketing Fund and subsequently obtain reimbursement of such advances out of Marketing Fund fees collected. Franchisor has the right to divide the Marketing Fund into two or more funds, by Brand Hotel or otherwise. If Franchisor operates more than one brand of hotels it has the right to combine the Marketing Fund with any marketing fund for such other brand or brands. If Franchisor operates more than one brand, for purposes of this Section, "Brand Hotel" will mean any hotel operated under the Marks and any hotel operated under such other brand.

- 5.4 <u>CONTENT AND FORMAT REQUIREMENTS</u>. All marketing and promotion by Franchisee shall be factual, ethical, and in good taste in the judgment of Franchisor. All advertising, marketing, and sales materials used by Franchisee in any medium shall be conducted in such manner, and shall conform to such standards and requirements, as Franchisor may specify from time to time. Franchisee must submit to Franchisor for its prior written approval samples of all advertising, marketing, and sales plans and materials and all other materials displaying the Marks that Franchisee desires to use which have not been prepared or previously approved by Franchisor.
- **5.5 ROOM PRICING.** Franchisee is responsible for setting the prices and rates for guest rooms and other products and services at the Hotel, and Franchisee will honor all pricing, rates, discounts, and terms as published at the time of booking. Franchisor may prohibit certain types of charges or billing practices that Franchisor determines are misleading or harmful to the Business System (whether or not they are contrary to applicable law in the location of the Hotel),

such as price-gouging, and may require that Franchisee set consistent pricing in all distribution channels. In no event will Franchisee set prices or rates, or use such pricing terms that are contrary to applicable law in the location of the Hotel. Franchisor and Franchisee recognize the value of pricing, promotions, and marketing programs that facilitate the marketing of the Business System, the goodwill, reputation, and uniformity of the Business System and consumer acceptance and recognition of Brand Hotels. Franchisee and Franchisor agree that, in order to better accomplish these objectives, Franchisor may from time to time in its sole judgment recommend: (a) that rates for lodging accommodations start at levels no higher than those determined by Franchisor; (b) mandatory maximum charges for the goods and services offered by Franchisee at the Hotel; (c) Franchisees offer some rooms in the Hotel at specified maximum prices; and (d) other rates and prices to the extent permitted by applicable law. Where prohibited by law Franchisor's recommendations will not be mandatory.

ARTICLE 6 FINANCIAL STATEMENTS, SALES REPORTS, FORMS AND ACCOUNTING

- 6.1 QUARTERLY AND ANNUAL FINANCIAL STATEMENTS. Franchisee will provide Franchisor with quarterly financial statements consisting of a balance sheet and income statement. Franchisee will also provide to Franchisor annual financial statements consisting of a balance sheet, income statement, statement of cash flows, and footnotes. The quarterly financial statements will present the financial results of the Hotel operations for both the current month and year to date. All financial statements provided to Franchisor for the Hotel will be presented in the exact form and format prescribed by Franchisor in writing, and will be formatted in accordance with the chart of accounts prescribed by Franchisor. Franchisee's quarterly financial statements will be delivered to Franchisor by the 20th day of the months of April, July, October and January of each year. Franchisee's annual financial statements will be delivered to Franchisor within 90 days of Franchisee's fiscal year end. Franchisee's annual financial statements must be certified by an independent certified public accountant, or, at the Franchisor's request audited by an independent certified public accountant.
- **6.2 INCOME TAX RETURNS**. Upon request from Franchisor, Franchisee will provide Franchisor with copies of all federal and state income tax returns filed by Franchisee and each of the owners of Franchisee.
- 6.3 REPORTS OF GROSS ROOM REVENUES. Franchisee will maintain an accurate written record of daily Gross Room Revenues for the Hotel and will remit a signed and verified statement of the monthly Gross Room Revenues generated by, at, as a result of, or in connection with the Hotel using such forms as Franchisor may prescribe in writing. The monthly or periodic statement of Gross Room Revenues is due to Franchisor at the same time as Franchisee's Royalty Fees for the applicable reporting period. In lieu of Franchisee providing Franchisor with any reports required by this Agreement, Franchisor has the right to require Franchisee's records and computer systems and applications as reasonably necessary for Franchisor to independently generate reports required by this Agreement on such dates that such reports become due. It is Franchisee's responsibility, whether Franchisee provides the reports or allows Franchisor access to generate them itself, to ensure that all data necessary to generate

such reports is fully and properly entered in such programs and systems as may be necessary to generate the reports.

- **6.4** ACCESS TO COMPUTER SYSTEM AND DATA. Upon Franchisor's request, Franchisee will provide Franchisor with electronic access at all times to Franchisee's financial statements and reports and all other data pertaining to the Hotel, whether maintained by Franchisee or by third parties.
- **FRANCHISOR'S AUDIT RIGHTS**. Within 3 days after receiving notice from Franchisor, Franchisee will make all of its financial records, books, ledgers, work papers, accounts, bank statements, tax returns (including without limitation all personal tax returns for each of the owners of Franchisee), sales tax returns, and other financial information pertaining to the Hotel ("Books and Records") available to Franchisor at all reasonable times for review and audit by Franchisor or its designee. The Books and Records for each fiscal year will be kept in a secure place by Franchisee and will be available for audit by Franchisor for at least 5 years after the end of such fiscal year. Further, in addition to or in lieu of an on-site audit, Franchisor has the right to require Franchisee to provide to Franchisor, at Franchisee's sole expense, copies of all Books and Records as requested by Franchisor. If an audit by Franchisor results in a determination that Franchisee's Gross Room Revenues were understated intentionally or by more than 2% for any accounting period, or if the audit is made necessary by Franchisee's failure to furnish reports, financial statements, or tax returns as required under Sections 6.1, 6.2 and 6.3, then Franchisee will, in addition to curing any deficiency in Royalty Fees or other amounts owed to Franchisor (plus interest as provided for herein), pay Franchisor for all costs and expenses, including salaries of Franchisor's employees, travel costs, room and board, and audit fees, that it has incurred as a result of the audit. The foregoing remedies shall be in addition to and not in lieu of all other remedies and rights of Franchisor hereunder or under applicable law, and Franchisor's right to audit will continue for 24 months following the termination or expiration of this Agreement. Franchisor has the right to use in any appropriate manner all reports, data and other information Franchisee submits to Franchisor or Franchisor obtains through review of Franchisee's books and records or by accessing Franchisee's computer system, which relate to the operation of the Hotel. Specifically, Franchisor has the right to share said reports, data and other information with third parties, including, but not limited to, consultants and existing and potential franchisees.
- 6.6 <u>DATA USAGE</u>. Franchisee agrees that Franchisor or its affiliates may, to the extent permitted by law, disclose to third parties data concerning and relating, directly or indirectly, to the Franchisee, the operations of Franchisee, and Franchisee's customers, including, but not limited to information about occupancy rates ("Operations Data"). Franchisee waives any notice in connection with the disclosure of Operations Data. Franchisor agrees that it, or its affiliates, will from time to time disclose to Franchisee such operations data as it deems appropriate regarding other Franchisees of Franchisor (Operations Data jointly with operations data of other Franchisees, "System Operations Data"). The Franchisor may, in its sole discretion, determine when and what System Operations Data will be disclosed, and may, without prior notice to, or consent from Franchisee, change the scope of the System Operations Data being disclosed to Franchisee or when it is disclosed. System Operations Data disclosed to Franchisee is disclosed solely for Franchisee's internal business purposes and to enable Franchisee to compare its results with those of other Franchisees of Franchisor. If Franchisor or

its Affiliates operate multiple brands, Franchisee agrees that Franchisor or its Affiliates may disclose System Operations Data to each other and to franchisees of such other brands. The disclosed Operations Data and System Operations Data remains confidential information of Franchisor. Franchisee may not disclose System Operations Data to other Franchisees of Franchisor, prospective Franchisees of Franchisor, competitors of Franchisor, prospective purchasers of Franchisee or any of Franchisee's assets, financial institutions, or any other third parties. The System Operations Data so disclosed will be based on information provided to Franchisor by its Franchisees. Such information will not be verified by Franchisor or any of its affiliates. Franchisor has no obligation to correct System Operations Data disclosed after it learns that it was incorrect or incomplete, or to inform Franchisee thereof.

ARTICLE 7 QUALITY CONTROL, UNIFORMITY AND STANDARDS REQUIRED OF THE FRANCHISEE

- 7.1 SYSTEM COMPLIANCE. From time to time, Franchisor will promulgate uniform standards of quality and service pertaining to the business operations of the Hotel so as to protect the distinction, valuable goodwill and uniformity represented and symbolized by the Marks and Business System. Any required standards exist to protect Franchisor's interests in the Business System and the Marks and not for the purpose of establishing any control or duty to take control over those matters that are reserved to Franchisee. Franchisee expressly acknowledges that adherence to each and every required or mandatory provision of the Business System is reasonable, necessary, and essential to establish and maintain the uniform image and favorable reputation of each Brand Hotel and the Business System and the success of Franchisor's franchise program.
- 7.2 <u>USE OF MARKS; IDENTIFICATION OF HOTEL</u>. Franchisee will operate its Hotel so that it is clearly identified and advertised under the StayAPTTM name or such other Mark as may be directed by Franchisor. However, the style and form of these words and the Marks in any advertising, marketing or promotional program or campaign must have the prior written approval of Franchisor. Whenever practical, Franchisee will use the Marks on all supplies, uniforms, FF&E, advertising, promotional materials, signs, and other articles in the manner prescribed by Franchisor in writing. Franchisee will, at its expense, comply with all notices of registration required by Franchisor and will, at its expense, comply with any other trademark, trade name, service mark, copyright, patent or other notice marking requirements required by Franchisor or applicable law.
- 7.3 FRANCHISEE'S BUSINESS NAME. Franchisee will not use the words "stayAPT" or any other Mark, or any combination, variation or abbreviation thereof or of any other Mark or words confusingly similar thereto in its corporate, limited liability company or partnership name. Franchisee will hold itself out to the public as an independent contractor operating its Hotel pursuant to a license from Franchisor. Whenever practical, Franchisee will clearly indicate on its business checks, stationery, purchase orders, business cards, receipts, promotional materials, and other written materials that Franchisee is a franchisee of Franchisor. Franchisee will display a sign in the front desk area of the Hotel which is clearly visible to the general public indicating that the Hotel is independently owned and operated as a franchised business. Franchisee will file for a certificate of assumed name in the manner required by

applicable state law so as to notify the public that Franchisee is operating its Hotel as an independent business pursuant to this Agreement.

- **COMPLIANCE WITH STANDARDS AND SPECIFICATIONS.** The Hotel will conform to Franchisor's prototype plans and specifications, exterior and interior design, décor, FF&E, signage, landscaping, color schemes and all other mandatory aspects of the Business System which must be approved by Franchisor prior to construction. Franchisee will not add any guest rooms or make any architectural, structural, design or decorating changes to the interior or exterior of the Hotel without Franchisor's prior written approval. Franchisee will purchase and install, at its expense, the fixtures, furnishings, equipment, furniture, telephone system, computer systems (including property management or related systems, frame relay, customer information system, reservation system), other systems and technology programs specified by Franchisor, signs, supplies and other items used in the operation of the Hotel ("FF&E"). The FF&E used in the Hotel must be installed and located in accordance with the floor plans and specifications approved by Franchisor, and must conform at all times to the quality standards and uniformity requirements established by Franchisor. Without limiting any other provision in this Agreement, Franchisor has and its affiliates own and have the perpetual right to use and to authorize other Brand Hotels to use, and Franchisee will fully and promptly disclose to Franchisor, all designs, products, services, procedures, ideas, plans, improvements, concepts, formulas, methods and techniques relating to the development or operation of an Brand Hotel or any similar business conceived or developed by Franchisee or Franchisee's employees during the term of this Agreement ("Inventions"). Franchisor will have no obligation to compensate Franchisee or its employees for any Inventions, it being understood and agreed that the benefit to Franchisee from the overall enhancement of the Business System is sufficient consideration for granting this right to Franchisor.
- 7.5 OWNERSHIP OF DEVELOPMENTS. If Franchisee develops any designs, products, services, procedures, or inventions deemed by Franchisor to be appropriate for use in other Brand Hotels or any franchise system of a Franchisor affiliate, it is understood and agreed that Franchisor shall be the owner of such designs, products, services, procedures, or inventions without obligation to compensate Franchisee, it being understood and agreed that the benefit to Franchisee from the overall enhancement of the Business System is sufficient consideration for granting this right to Franchisor, however Franchisee will be granted a license to use such designs, products, services, procedures, or inventions.
- **7.6** MAINTENANCE AND HOUSEKEEPING OF HOTEL. The Hotel and everything located on the Hotel premises shall be maintained by Franchisee in a clean, safe, orderly, and first-class condition in accordance with the standards specified in the Manual, and consistent with the image of a clean, sanitary, attractive, safe, and efficiently operated lodging accommodation. The Hotel shall be constructed, maintained, and operated in compliance with all applicable fire, safety, health, and sanitation laws, ordinances, and regulations, and Franchisee shall maintain the highest health standards and ratings applicable to the Hotel and otherwise maintain high moral and ethical standards at the Hotel. Franchisee will, at its expense, repair, paint and keep in a clean and sanitary condition the interior, exterior and grounds of the Hotel, and will replace all FF&E items, including computer systems, software, and signs as such items become worn-out, soiled or in disrepair. All mechanical equipment, including ventilation, heating, and air conditioning, will be kept in good working order by Franchisee at all times and

will meet Franchisor's quality standards. Any replacement FF&E items, including computer systems, software, and signs must comply with Franchisor's then-current standards and specifications.

- REMODELING AND REDECORATION OF HOTEL. At Franchisor's 7.7 request, Franchisee will be required to remodel, modernize, redecorate and renovate the Hotel, and to replace and modernize the FF&E and supplies used in the Hotel so that the Hotel will reflect the then-current image intended to be portrayed by the Franchisor, including, without limitation, such structural changes, remodeling, and redecoration and such modifications to existing improvements as may be deemed necessary by Franchisor. Franchisee acknowledges that the market place in which it operates may change frequently and quickly during the Term, and that some aspects of the Business System, such as computer systems and technology, may need frequent upgrades and changes. All remodeling, modernization, redecoration and renovation of the Hotel must be done in accordance with the standards and specifications as prescribed by Franchisor from time to time and with the prior written approval of Franchisor. All replacements for the equipment, decor items, furniture, fixtures and supplies must conform to Franchisor's then-current quality standards and must be approved by Franchisor in writing. Franchisee will commence remodeling, modernizing, redecorating and renovating the Hotel within 180 days of the date that Franchisee receives written notice specifying the required remodeling, modernization, redecoration and renovation from Franchisor. Franchisee will not be required to extensively remodel, modernize, redecorate or renovate the Hotel or to replace and modernize its furniture, fixtures, supplies and equipment more than once every 5 years during the term of this Agreement. Except as described above, Franchisee shall make no additions, alterations, or replacements to the Hotel or anything located on the Hotel premises without the prior written consent of Franchisor.
- 7.8 PRODUCTS AND SERVICES. Franchisee will sell only those products and services approved by Franchisor in writing and will offer for sale all products and services prescribed by Franchisor. Franchisee will have the right to sell all products and services at whatever prices and whatever terms it deems appropriate. Franchisee will conform to all customer service standards prescribed by Franchisor in writing.

7.9 <u>APPROVED SUPPLIERS AND APPROVED PRODUCTS.</u>

(A) Franchisor may require that Franchisee will purchase from suppliers approved in writing by Franchisor ("Approved Suppliers") any and all products, supplies, uniforms, signs, FF&E, computer hardware and software, advertising and marketing materials, or services relating to the operation of the Hotel ("Approved Products") to be used or sold at the Hotel which Franchisor determines meet the standards of quality and uniformity required to protect the valuable goodwill and uniformity symbolized by and associated with the Marks and Business System. If Franchisor has not designated one or more Approved Suppliers for an Approved Product Franchisee may purchase such Approved Product from any source, as long as the standards and specifications for such Approved Product are met. If Franchisee desires to purchase any Approved Products, other than Approved Products displaying any Mark, from unapproved suppliers, then Franchisee will submit a written request to Franchisor to determine whether the suppliers' products comply with Franchisor's standards and specifications. Franchisee

must submit whatever information, specifications, or samples Franchisor requires and reimburse Franchisor for its costs of review and testing the alternate supplier. Franchisor will respond in writing with its decision within 45 days of receiving Franchisee's written request and all other information, specifications and samples Franchisor deems reasonably necessary to evaluate the alternative supplier. Franchisor reserves the right to approve or disapprove proposed alternative suppliers in its sole discretion. Franchisor has the right to charge Franchisee a reasonable fee for the review of the alternative supplier, or request reimbursement for its costs of review and testing the alternate product or service, if such costs are higher than the then current fee. Franchisor may revoke an approval previously given at any time in its discretion, upon notice to Franchisee. Notice will be given in a manner that Franchisor deems appropriate.

- If Franchisee would like to use alternate products or services to Approved Products, Franchisee must first request in writing that Franchisor approve the alternate product or service. Franchisee must submit whatever information, specifications, or samples Franchisor requires (including a sample room or space using the alternate product), including standards and specifications for the alternative product, prepared by a qualified professional, which, if approved by Franchisor, shall not be materially modified without the prior written consent of Franchisor. As used in this Article the term "materially modified" shall mean any modification that would: (a) change the size or dimensions of any public areas, room, or amenities of the Hotel; (b) affect the appearance or design of any portion of the Hotel or the quality of the materials used therein; or (c) constitute a departure from the concept, standards or approved products or services of the Business System. Franchisor has the right to charge Franchisee a reasonable fee for the review of the alternative product or service, or request reimbursement for its costs of review and testing the alternate product or service, if such costs are higher than the then current fee. Franchisor reserves the right to approve or disapprove proposed alternative products or services in its sole discretion. Franchisor will notify Franchisee within a reasonable time of its approval or rejection of the alternate product or service. Franchisor may revoke an approval previously given at any time in its discretion, upon notice to Franchisee, for future purchases. Notice will be given in a manner that Franchisor deems appropriate.
- (C) FRANCHISOR AND ITS AFFILIATES MAKE NO WARRANTY WITH RESPECT TO ANY PRODUCTS, SERVICES, EQUIPMENT, SUPPLIES OR OTHER ITEMS FRANCHISOR APPROVES AND FRANCHISOR EXPRESSLY DISCLAIMS ALL WARRANTIES, EXPRESS AND IMPLIED, INCLUDING IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE, WITH RESPECT TO ANY SUCH PRODUCTS, EQUIPMENT (INCLUDING WITHOUT LIMITATION AND ANY REQUIRED COMPUTER SYSTEMS), SUPPLIES, OR OTHER APPROVED ITEMS.

7.10 <u>COMPLIANCE WITH APPLICABLE LAWS</u>.

(A) Franchisee will, at its expense, comply with all applicable federal, state, city, local, and municipal laws, ordinances, rules and regulations which apply to the construction, establishment and operation of the Hotel. The laws include those affecting zoning and construction, public accommodations, accessibility by persons with disabilities, employment, health and safety, data privacy and security and labor. In

addition, Franchisee will, at its expense, be responsible for determining the licenses and permits required by law for the Hotel, for obtaining and qualifying for all construction or all such licenses and permits, and for maintaining all such licenses and permits in full force and effect. Franchisee shall notify Franchisor in writing within 5 days of the commencement of any action, suit, or proceeding, and of the issuance of any inquiry, subpoena, order, writ, injunction, award, or decree of any court, agency, or other governmental instrumentality, arising out of, concerning, or which may affect the operation or financial condition of the Hotel, including, without limitation, any criminal action or proceeding brought by Franchisee against employees, customers, or other persons.

- (B) Franchisee and its owners understand the requirements of, and will abide by, all United States government economic sanctions requirements. Franchisee represents and warrants that neither it nor any of its direct or indirect owners, directors, officers, employees or agents is a person subject to trade restrictions under United States law, including (without limitation) the International Emergency Economic Powers Act, 50 U.S.C. §§ 1701 et seq., The Trading with the Enemy Act, 50 U.S.C. App. 1 et seq., or any Executive Orders or regulations promulgated thereunder (including Executive Order 13224 of September 24, 2001 Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism, and the Specially Designated Nationals and Blocked Persons List) ("Anti-Terrorism Laws"). Franchisee and its owners may not engage in any activity that would expose Franchisor or its affiliates to a risk of criminal or civil penalties under applicable United States law. Any violation of the Anti-Terrorism Laws by Franchisee or its owners, or any blocking of Franchisee's or its owners' assets under the Anti-Terrorism Laws, shall constitute good cause for immediate termination of this Agreement.
- **7.11 PAYMENT OF OBLIGATIONS**. Franchisee will timely pay all of its noncontested and liquidated obligations and liabilities due and payable to Franchisor, and to suppliers, lessors and creditors of Franchisee. Franchisee's failure to comply with this provision will be deemed a material breach of this Agreement.
- 7.12 PAYMENT OF TAXES. Franchisee will be responsible and liable for the prompt payment of all taxes payable in connection with the Hotel. Franchisor will have no liability for any taxes which arise or result from the Hotel, and Franchisee will indemnify Franchisor for any such taxes that may be assessed or levied against Franchisor which arise or result from the Hotel. It is expressly agreed by the Personal Guarantors to this Agreement that their personal guaranty applies to the prompt payment of all taxes which arise or result from the Hotel.
- **7.13 STAFFING.** Franchisee shall employ or retain qualified management personnel as prescribed in the Manual. All personnel employed or retained by Franchisee in the position of Resident Manager or general manager shall attend and successfully complete, to Franchisor's satisfaction, Franchisor's manager training program and be certified by Franchisor. The initial Resident Manager/general manager shall complete the initial training at least 60 days before the Hotel opens. Any replacement Resident Manager/general manager must complete the entire initial training for property/general managers within 90 days of employment. All other

personnel, including any subsequent property manager, shall sign up for training within 14 days of employment and complete their field-based on-the-job training within 90 days of employment. These periods may be extended if space in the applicable training program is not available to Franchisee's personnel during the specified periods. Franchisee will require its employees to wear the standard attire or uniforms approved by Franchisor and will comply with such programs as may from time to time be promulgated by Franchisor to promote the common business image.

7.14 <u>BUSINESS HOURS; RESIDENT MANAGER</u>. The Hotel will be open for business on such days and for such hours as Franchisor may designate from time to time. Franchisee will, at all times, employ a "Resident Manager" who will reside in the Hotel and who will be responsible for supervising the employees and the day-to-day operations of the Hotel. Franchisee will notify Franchisor in writing of the identity of the Resident Manager in Section 2.B of the Ownership & Management Addendum attached hereto as Exhibit B. Franchisee will have a sufficient number of adequately trained and competent personnel on duty to guarantee efficient service to the guests at the Hotel. Franchisee is exclusively responsible for the terms of employment, compensation, scheduling, benefits, disciplining and all other personnel decisions respecting Hotel employees without any influence or advice from Franchisor.

FRANCHISOR'S INSPECTION RIGHTS. Franchisor's representatives will 7.15 have the right to inspect the Hotel at any reasonable time to examine representative samples of goods and products sold or used at the Hotel, and to examine and evaluate the quality of services provided by Franchisee to its guests. Franchisee will cooperate fully with Franchisor's agents during the inspections. Franchisee shall provide all information requested by Franchisor for the purpose of Franchisor's conducting guest satisfaction audits and surveys. If a vacancy exists, Franchisee will provide Franchisor's representatives with a room free of charge at the Hotel during such inspection. Franchisor will have the right to photograph and film the interior and exterior of the Hotel. Franchisor will have the right to use all photographs and recordings of the Hotel for such purposes as Franchisor deems appropriate including, but not limited to, use in advertising, promotional materials, training, marketing, and public relations. Franchisee will not be entitled to, and hereby expressly waives, any right that it may have to be compensated by Franchisor, its advertising agencies, and other franchisees for using photographs or recordings in the manner described herein. Franchisee will take such steps as may be reasonably necessary to correct any deficiencies detected during such an inspection, upon the written request of Franchisor or its agents, within such reasonable time as may be specified therein. If Franchisee fails an inspection for any health or safety reason that Franchisor, in its discretion, deems to constitute a danger to the health or safety of the public, employees at the Property, or guests or visitors to the Hotel, Franchisee shall, immediately upon Franchisor's request, take such action as required by Franchisor, including closing all or part of the Hotel, or not allowing any new guests, until the dangerous conditions have been remedied to Franchisor's satisfaction. Nothing in this Section shall limit or restrict Franchisor's rights under Article 13, or any other Article of this Agreement. In addition, if the Hotel has sustained substantial wear and tear, Franchisee may request, or Franchisor may require, that a property improvement review ("PIR") be performed. The PIR will identify areas of the Hotel or items that do not satisfy the then current System standards as set forth in the Manual, and identify steps required to be taken to remedy identified deficiencies. Franchisee will pay Franchisor the then current fee for performing the PIR. The fee will be payable upon request for the PIR to be performed.

- **7.16 SECURITY INTEREST**. This Agreement and the license granted to Franchisee hereunder may not be the subject of a security interest, lien, levy, attachment or execution by Franchisee's creditors or any financial institution, except with the prior written approval of Franchisor.
- 7.17 NOTICES. Franchisee will immediately deliver to Franchisor a copy of any notice of default from any mortgagor, landlord, or trustee under any deed of trust with respect to the Hotel, and copies of all notifications of any lawsuits, contract breaches, consumer claims, federal or state administrative or agency proceedings or investigations, or other civil or governmental claims, actions or proceedings relating to the Hotel. Upon request from Franchisor, Franchisee will provide additional information as may be required by Franchisor regarding the alleged default, lawsuit, claim or proceeding or any subsequent action or proceeding in connection with the alleged default, lawsuit, claim or proceeding.
- 7.18 <u>CREDIT CARDS</u>. Franchisee will honor all credit, charge, courtesy or cash cards or other credit devices required or approved by Franchisor, and will enter into all necessary credit card arrangements. Franchisee must obtain the written approval of Franchisor prior to honoring any unapproved credit, charge, courtesy or cash cards or other credit devices. Franchisor will respond with its decision to Franchisee's request to approve a credit, charge, courtesy or cash cards, or other credit devices within 30 days of receiving Franchisee's written request for approval. Franchisee shall, at its expense, comply with all information security standards published by the Payment Card Industry Security Standards Council, including the Payment Card Industry Data Security Standard (commonly known as "PCI Compliance" or "PCI-DSS"), and any successor thereto, as well as any and all applicable laws intended to protect personal information and to ensure security when transactions are processed using a payment card.
- 7.19 WEBSITE, ONLINE PRESENCE, MOBILE APPLICATIONS AND DATA Upon 90 days' prior written notice, Franchisor may require Franchisee, at Franchisee's expense, to participate in a brand-specific website ("Website") or other online presence, including social media and mobile applications. Franchisor will determine the content and use of the Website and any other brand-specific online presence, will establish the rules under which franchisees may or will participate in such Website and online presence. Franchisor will retain all rights relating to the Website and may alter or terminate the Website upon written notice to Franchisee. Franchiser shall have the only brand-specific website and Franchisee may not have any individual Website for the Hotel. Franchisee's general conduct on the Website or in or through other online presence and specifically its use of the Marks on the Website or through other online presence will be subject to the provisions of this Agreement. Franchisee acknowledges that certain information obtained through its participation in the Website may be considered Confidential Information, including access codes and identification codes. Franchisee's right to participate in the Website or otherwise use the Marks or System on the Internet and other online presence will terminate when this Agreement expires or terminates. Franchisee is solely responsible for protecting itself from disruptions, internet access failures, internet content failures, and attacks by hackers and other unauthorized intruders and Franchisee waives any and all claims Franchisee may have against Franchisor as the direct or indirect result of such disruptions, failures, or attacks. Franchisee must comply with all laws and regulations related to privacy and data protection and breach response policies Franchisor may periodically

establish. Franchisee must notify Franchisor immediately of any suspected data breach at or in connection with the Hotel. Because website, online and mobile technologies and their uses are constantly evolving, Franchisor reserves the right to require Franchisee's participation in other technologies being implemented as part of the Business System, such as social media and mobile applications, other mobile marketing and payment technologies.

7.20 HOTEL MANAGEMENT; OPERATORS OF HOTEL FACILITIES.

- Starting at least four months before the scheduled opening of the Hotel, (A) either Franchisee or a professional hotel management company (the "Management Company") will oversee the Hotel operations and the Resident Manager, in either case only with Franchisor's prior written approval. Franchisee will engage only a qualified and experienced Management Company and will require the Management Company to operate and maintain the Hotel in the same manner as if Franchisee were operating it directly pursuant to this Agreement. Franchisor may review any management agreement between Franchisee and its proposed Management Company. Franchisor may withhold its approval of any proposed Management Company or manager of the Hotel if Franchisor determines, in its sole discretion, that such entity or individual is not financially capable or does not have adequate experience, skills, or operational capacity to operate the Hotel in accordance with the Manual and this Agreement. Franchisor may condition its approval of Franchisee or any Management Company upon the satisfactory completion of certain training programs. If Franchisee or its proposed Management Company is not approved, Franchisor may require the Hotel to be managed by another management company approved by Franchisor. The approved Management Company will be identified in Section 2.A of the Ownership & Management Addendum attached hereto as **Exhibit B**.
- (B) For any change in Management Company during the Term, Franchisee must obtain Franchisor's prior written approval, which approval may be withheld in Franchisor's sole discretion. In each case, the proposed Management Company must demonstrate that it has the financial capability, experience, skills, and operational capacity to oversee the Hotel operations and Resident Manager in accordance with the Manual and pursuant to this Agreement. Any change in Management Company will be documented by execution of a revised Ownership & Management Addendum in the form attached hereto as **Exhibit B**.
- (C) Franchisee will at all times remain solely responsible for the management, direction, and control of the Hotel and for complying with the obligations of this Agreement, whether or not a Management Company is retained. Franchisor reserves the right to revoke its approval of the Management Company if, in its sole discretion, the Management Company is no longer qualified to manage the Hotel, and may, upon 30 days' notice, require Franchisee to terminate its agreement with the Management Company and retain a replacement Management Company. The management agreement between Franchisee and the Management Company shall at all times be subject and

subordinate to this Agreement and, in the event of any conflict between the management agreement and this Agreement, the controlling contract shall be this Agreement.

7.21 <u>FRANCHISEE MEETINGS</u>. During the term of this Agreement, Franchisor may conduct meetings and seminars for Brand Hotel franchisees at such times and at such locations as Franchisor deems appropriate. Franchisee (or, as designated by Franchisor, the Resident Manager) may be required to attend (at Franchisor's then-current fee) a minimum of one annual conference of all Brand Hotel franchisees each year and such regional meetings or seminars as may be arranged by Franchisor from time to time. All expenses incurred by Franchisee, its Resident Manager, or any of its employees in connection with attending the meetings or seminars will be paid by Franchisee.

7.22 RESERVATION SYSTEM AND RESERVATION DATA.

- (A) During the term of this Agreement, if Franchisor establishes a reservation system to facilitate the making of reservations at Franchisee's Hotel (hereinafter referred to as the "Reservation System"), Franchisee will be required to participate in such Reservation System and observe all terms and conditions of participation specified by Franchisor. Franchisee agrees to furnish to Franchisor such information as Franchisor or its designee may request for that purpose. Franchisee will pay Franchisor a reservation fee equal to the then-current reservation fee for each reservation delivered to Franchisee. In addition to the Reservation Fee, Franchisee at its expense agrees to purchase, install and maintain the Reservation System equipment including all required computer hardware and software.
- (B) Reservation data, guest personal data, and credit card information are all subject to the general privacy and data protection provisions in this Agreement and must be protected and safeguarded at all times to ensure privacy, protection from theft, piracy, or unauthorized use. Franchisee must comply with all laws and regulations related to privacy and data protection. Franchisee may not sell, transfer, publish, or otherwise use reservation data, guest personal data, or credit card information for any purpose other than in connection with the lawful ownership and operation of the Hotel during the Term and in accordance with this Agreement. Franchisee will inform all of its employees of their obligations concerning this requirement. Franchisor must be notified immediately upon discovery of any prohibited use or disclosure of confidential or proprietary information or any breach of these obligations; and will cooperate fully to prevent further prohibited use.
- 7.23 Privacy and Data Protection. Franchisee must: (i) comply with all applicable international, national, federal, provincial, state, or local laws, codes or regulations that regulate the processing of information that can be used (alone or when used in combination with other information Franchisee controls) to identify, locate or contact an individual or pertains in any way to an identified or identifiable individual, including without limitation names, addresses, telephone numbers, e-mail addresses, employee identification numbers, signatures, passwords, financial information, credit card information, biometric or health data, government-issued identification numbers and credit report information ("Personal

Information") in any way, including, but not limited to, national data protection laws, laws regulating marketing communications and/or electronic communications, information security regulations and security breach notification rules ("Privacy Laws"); (ii) comply with all standards, specifications, requirements, criteria, and policies that have been and are in the future developed and compiled by us that relate to Privacy Laws and the privacy and security of Personal Information; (iii) implement all administrative, physical and technical safeguards necessary to protect any Personal Information; (iv) refrain from any action or inaction that could cause Franchisor to breach any Privacy Laws; (v) do and execute, or arrange to be done and executed, each act, document and thing Franchisor deems necessary in Franchisor's business judgment to keep Franchisee in compliance with the Privacy Laws; and (vi) immediately report to Franchisor the theft or loss of Personal Information (other than the Personal Information of Franchisee's own officers, directors, owners, employees or service providers). It is entirely Franchisee's responsibility (even if Franchisor provides Franchisee any assistance or guidance in that regard) to confirm that the safeguards Franchisee uses to protect Personal Information comply with all applicable laws and industry best practices related to the collection, access, use, storage, disposal and disclosure of Personal Information.

ARTICLE 8 INSURANCE AND INDEMNIFICATION

- **8.1 REQUIRED INSURANCE COVERAGE**. During the Term, Franchisee shall comply with all insurance requirements of any lease, mortgage, or deed of trust covering the Hotel as well as all insurance requirements of Franchisor as set forth in the Manual or as otherwise communicated by Franchisor. All insurance shall be procured at the earliest possible time that Franchisee has an insurable interest with respect thereto, and shall be written by insurance companies with an A.M. Best rating of AVI or greater. At a minimum, Franchisee shall maintain the following:
 - (A) Comprehensive or commercial general liability insurance, for any claims or losses arising or resulting or pertaining to the Hotel or its operation, provided that, if the Hotel is managed by a property management company, business auto liability insurance and worker's compensation insurance must be carried by such property management company. If the general liability coverage contains a general aggregate limit, it shall apply in the aggregate to this Hotel only. Such insurance shall be on an occurrence policy form and shall include premises and operations, independent contractors, blanket contractual, products and completed operations, advertising injury, employees as additional insureds, broad form property damage, personal injury, severability of interests, and explosion, collapse and underground coverage during any construction.
 - (B) Business auto liability including owned, non-owned, and hired vehicles.
 - (C) Umbrella excess liability insurance.
 - (D) Insurance on the Hotel (including FF&E, boiler, and machinery) against such risks as Franchisor may specify, including fire, lightning, vandalism, malicious

mischief, and all other risks covered by the special extended coverage endorsements, in an amount equal to full replacement value thereof.

- (E) If the Hotel is located in whole or in part within an area identified by the Federal Government as having a special flood hazard, flood insurance in an amount not less than the maximum coverable available under the National Flood Insurance Program and excess flood coverage with reasonable limits including business interruption coverage in an amount not less than that set forth in paragraph H. below.
- (F) If the Hotel is located in an "earthquake prone zone" as determined by the U.S. Geological Survey, insurance in an amount not less than the probably maximum loss less any applicable deductibles, including business interruption coverage in an amount not less than set forth in Paragraph H below.
 - (G) Builder's risk property insurance of not less than the full contract price.
- (H) Business interruption insurance covering loss of profits and necessary continuing expenses, including coverage for payments of royalty fees and contributions to the Marketing Fund, for any interruption in Franchisee's business operations, as well as the cost of conducting a pre-opening review before reopening of the business in the event of closure for repairs or rebuild.
 - (I) Employee dishonesty coverage on all Hotel employees.
- (J) Worker's compensation insurance as required by applicable law, employer's liability insurance, and such other insurance as may be required by law.
- (K) During construction at the Hotel (including the initial construction of the Hotel), Franchisee shall cause the general contractor to maintain, with an insurer carrying an A.M. Best rating of AVI or greater, the following insurance:
 - (1) Comprehensive or commercial general liability insurance, for any claims or losses arising or resulting or pertaining to the Hotel or its operation. If the general liability coverage contains a general aggregate limit, it shall apply in the aggregate to this Hotel only. Such insurance shall be on an occurrence policy form and shall include premises and operations, independent contractors, blanket contractual, products and completed operations, advertising injury, employees as additional insureds, broad form property damage, personal injury, incidental medical malpractice, severability of interests, and explosion, collapse and underground coverage during any construction.
 - (2) Workers' compensation as required by applicable law, employer's liability insurance, and such other insurance as may be required by law.
 - (3) Umbrella excess liability insurance.
 - (4) Business auto liability insurance including owned, non-owned and hired vehicles.

(5) Any other insurance required by state or federal law.

Franchisor does not represent or warrant that any insurance that Franchisee is required to purchase, or which Franchisor procures on Franchisee's behalf, will provide adequate coverage for Franchisee. Franchisee should consult with its own insurance agents, brokers, attorneys or other insurance advisors to determine the level of insurance protection it needs and desires, in addition to the coverage and limits required herein. Franchisor's review and verification of certain elements of the franchisee's insurance does not in any way reduce or eliminate Franchisee's obligations to fully comply with all insurance requirements. It is Franchisee's sole obligation to fully comply with these insurance requirements and it is Franchisee's sole obligation to confirm with its insurance providers that its policies are in compliance.

- 8.2 **REQUIRED POLICY TERMS.** All policies of insurance (a) shall be written on a fully insured basis with no deductibles in excess of \$50,000 nor any self-insured retentions, (b) shall name Franchisor, its owners, its affiliated companies, and each of their respective managers, agents, representatives, officers, directors and employees as additional insureds, except with respect to workers compensation insurance and other employee liability insurance, (c) shall be specifically endorsed to provide that the coverages will be primary and that any insurance carried by any additional insured, including the Franchisor and its owners, its affiliated companies, and each of their respective owners, managers, agents, representatives, officers, directors and employees (collectively, the "Indemnitees"), shall be excess and non-contributory, (d) shall contain a waiver of any rights of subrogation against the Indemnitees, and (e) shall contain a severability of interest provision in favor of the Indemnitees. At all times during the Term, Franchisee will furnish to Franchisor certificates of insurance evidencing the term and limits of coverage in force, names of applicable insurers, and persons insured, and a statement that coverage may not be canceled, altered, or permitted to lapse or expire without 30 days' advance written notice to Franchisor. Revised certificates of insurance shall be forwarded to Franchisor each time a change in coverage or insurance carrier is made by Franchisee and/or upon renewal of expired coverages. At Franchisor's option, Franchisee may be required to provide certified insurance policy copies. Franchisor may increase the minimum protection or coverage requirements of any policy required under this Article 8, and may require different or additional kinds of insurance at any time to reflect inflation, identification of special risks, changes in law or standards of liability, higher damage awards, or other relevant changes in Franchisee acknowledges and understands that Franchisor makes no circumstances. representation or warranty with respect to the adequacy or sufficiency of the insurance required under this Article, and that Franchisee shall have the sole responsibility to determine whether additional insurance or higher limits are appropriate.
- **8.3 FRANCHISOR'S RIGHT TO ACQUIRE INSURANCE**. If Franchisee does not obtain and maintain the insurance coverage required by this Agreement, as revised by the Manual or otherwise in writing, Franchisor may, but shall not be obligated to, procure such insurance, and the cost or expense thereof, together with a reasonable fee for Franchisor's expenses in so acting, shall be payable by Franchisee immediately upon demand.
- **8.4 INDEMNIFICATION**. Franchisee shall indemnify, hold harmless, and promptly reimburse the Indemnitees for, from and against any and all fines, damages, legal fees, costs, expenses, and other liabilities suffered or incurred by the Indemnitees by reason of any actual or

threatened claim, demand, lawsuit, tax, penalty, investigation, or other proceeding (even where Indemnitee's negligence or other wrongful conduct is alleged) arising directly or indirectly from, as a result of, or in connection with (a) any application submitted to Franchisor by Franchisee, (b) the development, construction, operation, condition, use, occupancy, or sale of the Hotel, (c) any occurrence at or on the Hotel premises, (d) any environmental matters of any kind pertaining to the Hotel, (e) any breach of any terms and provisions of this Agreement by Franchisee, (f) any offering of securities, units, or other ownership interests of Franchisee, including, without limitation, the violation of any federal and/or state securities laws, and (g) any allegation that the Franchisor is liable under federal or state employment or labor laws as the employer of Franchisee's employees; (h) any allegation that the Franchisor is liable under federal or state data privacy laws; and (i) any breach by Franchisee of data privacy laws. Upon Franchisor's request, Franchisee shall defend Indemnitees against all such matters. In any event, Franchisor shall have the right, through counsel of its choice, to control any matter to the extent Franchisor reasonably determines that such matter may have a significantly adverse effect on Indemnitees. Franchisee shall also indemnify and promptly reimburse Franchisor for all expenses reasonably incurred by Franchisor to protect itself from, or to remedy, any breach of this Agreement by Franchisee. Franchisee's indemnity obligations under this Agreement shall survive the expiration or other termination of this Agreement and shall be in addition to all other rights and remedies of Franchisor. Franchisee's obligations to indemnify Franchisor under this Article shall not be limited in any way by reason of any insurance which may be maintained by Franchisor, nor shall Franchisee's performance of its obligation to maintain insurance relieve Franchisee of liability under this indemnity provision or be construed to be a limitation on the amount of Franchisee's indemnity obligations. The right of the Indemnitees to indemnity under this Agreement shall arise notwithstanding that joint or concurrent liability may be imposed on the Indemnitees by statute, ordinance, regulation, or other law.

8.5 Franchisee shall notify Franchisor in writing within 5 days of receipt of notice or knowledge of any claim, dispute, loss, or damage in the amount of \$25,000 or more, real or alleged, arising from Franchisee's activities in, at, or around the Hotel, whether or not such claim names Franchisor. A quarterly notice will be provided by Franchisee to Franchisor regarding any claim, dispute, loss, or damage. Franchisee has no authority to, and shall not, accept any service of process on behalf of Franchisor, any of its affiliated companies, or the Indemnitees.

ARTICLE 9 CONFIDENTIAL OPERATIONS MANUAL AND OTHER INFORMATION

9.1 <u>COMPLIANCE WITH MANUAL</u>. In order to protect the reputation and goodwill of Franchisor and to maintain uniform operating standards under the Marks and Business System, Franchisee agrees to at all times to comply with each and every requirement of the Business System and to operate the Hotel in strict compliance with such mandatory standards, specifications, methods, and techniques (as opposed to any best practices and other optional advice that may be included in the Manual) as Franchisor may, prescribe in the Manual, as the same may be modified from time to time by the Franchisor in its sole discretion. Such modifications and supplementations may relate to, without limitation, changes in the business, authorized products and services, FF&E requirements, quality standards, operating procedures, compliance with any requirements for installing frame relay, customer information systems, reservations systems and other systems or technology programs, and to pay any fees or charges

associated with any such Business System modifications or supplementations and any other changes reflected in the Manual. Franchisee shall refrain from deviating therefrom and from otherwise operating in any manner which adversely reflects on the Business System, the Marks, the goodwill associated therewith, or Franchisor's rights therein. Notwithstanding the foregoing, Franchisor is not entitled to exert control over Franchisee's employees working conditions, except to the extent of controlling the quality of the Business System/brand and the quality of the products and services that Franchisee offers. Franchisor will provide Franchisee with 1 copy of the Manual which Franchisee will have on loan from Franchisor for the term of this Agreement. Franchisee will keep the Manual in the Hotel at all times. Franchisee acknowledges and agrees that in the future the Manual and other system communications may only be available on the Internet or other on-line or computer data transfer communications. Franchisee further acknowledges and agrees that any required standards set forth in the Manual exist to protect Franchisor's interest in the Business System and the Marks and not for the purpose of establishing any control or duty to take control over those matters that are reserved to Franchisee.

- 9.2 <u>CONFIDENTIALITY OF MANUAL</u>. During the term of this Agreement and thereafter, Franchisee will treat the Manual, any other manuals created for or approved for use in the operation of the Hotel, and the information contained therein as secret and confidential, and Franchisee will use all reasonable means to keep such information secret and confidential. Franchisee will not make any copy, duplication, record, or reproduction of the Manual, or any portion thereof, available to any unauthorized person.
- 9.3 REVISIONS TO MANUAL. The Manual will at all times during the term of this Agreement and thereafter, remain the property of Franchisor. Franchisor reserves the right to revise the Manual at any time during the term of this Agreement and Franchisee expressly agrees to operate its Hotel in accordance with all such revisions. Franchisee will at all times keep the Manual current and up-to-date, and in the event of any dispute regarding the Manual, the terms of the master copy of the Manual maintained by Franchisor will be controlling in all respects.
- **9.4** <u>CONFIDENTIALITY OF OTHER INFORMATION</u>. Franchisee acknowledges and agrees that:
 - (A) Franchisor possesses certain Trade Secrets, and in general, methods, techniques, formats, specifications, programs, procedures, information systems, and knowledge, in the operation and franchising of Properties and other lodging concepts. "Trade Secrets" shall mean confidential information, including, without limitation, (i) the design for Brand Hotels, (ii) methods of service and operations at Brand Hotels, (iii) knowledge of sales and profit performance at any one or more Brand Hotels, (iv) knowledge of test programs, concepts, or results relating to operating, new advertising and promotional programs, (v) sources of suppliers of equipment, (vi) advertising, promotion, and marketing techniques, (vii) methods and information regarding the selection and training of managers and other employees for Brand Hotels; and (viii) the Manual.
 - (B) Franchisor will be disclosing and providing to Franchisee certain confidential and proprietary information concerning the Business System and Franchisor

has a proprietary interest in all such information. "Confidential Information may include by way of example, Trade Secrets, standard plans for Brand Hotels, the Manual, data, know-how, processes, procedures, technology, methods of operation, designs, sketches, photographs, plans, drawings, specifications, reports, financial information, customer lists, pricing information, studies, findings, inventions, and ideas. During the Term or thereafter, Franchisee will not communicate or divulge to any other person or entity any Confidential Information which may be communicated to Franchisee in the Manual or otherwise, or by virtue of this Agreement. Franchisee will divulge Confidential Information only to its employees that must have access to it in order to operate the Hotel and may only use Confidential Information for the purpose of operating the Hotel. Franchisee shall not at any time, without Franchisor's prior written consent, copy, duplicate, record, or otherwise reproduce the materials or information, in whole or in part, nor otherwise make the same available to any unauthorized person.

- (C) Franchisee shall acquire no interest in the Trade Secrets, other than the right to utilize them in the development and operation of the Hotel during the Term. The use or duplication of the Trade Secrets in any other business will constitute an unfair method of competition. The Trade Secrets are proprietary and are disclosed to Franchisee in confidence and solely on the condition that Franchisee agrees, and Franchisee hereby agrees that Franchisee (i) will not use the Trade Secrets in any other business or capacity; (ii) will maintain the absolute confidentiality of the Trade Secrets during and after the Term; (iii) will not make unauthorized copies of any portions of the Trade Secrets disclosed in written form, including, without limitation, any plans, the Manual; and (iv) will operate and implement all reasonable procedures prescribed by Franchisor to prevent the unauthorized use and disclosure of the Trade Secrets. Franchisee shall immediately notify Franchisor of any unauthorized use of disclosure of the Manual or any of the Trade Secrets or if the Manual or any other materials containing any Trade Secrets are lost or stolen.
- (D) The foregoing restrictions on Franchisee's disclosure and use of Trade Secrets shall not apply to information, processes, or techniques that are or become generally known and used by other similar hotels or lodging concepts, other than through disclosure (whether deliberate or inadvertent) by Franchisee, and disclosure of Trade Secrets in judicial or administrative proceedings to the extent that Franchisee is legally compelled to disclose such information, provided, Franchisee shall have used Franchisee's best efforts, and shall have afforded Franchisor the opportunity, to obtain an appropriate protective order or other assurance satisfactory to Franchisor of confidential treatment for the information required to be so disclosed.
- (E) Unless the context otherwise requires, the term "Franchisee" as used in this Article 9 shall include, individually and collectively, all partners, officers, directors, and managers of Franchisee, and owners or holders, directly or indirectly (and any partners, officers, directors, and managers of any such holder), of 5% or more of the beneficial interest in Franchisee.
- (F) In the event any provision of this Article is deemed by a court of competent jurisdiction to be more restrictive than permissible at law or equity, then

Franchisee agrees that the provisions hereof may be reformed and modified and enforced by such court to the maximum extent permissible under applicable law and principles of equity. Franchisee agrees that specific performance and injunctive relief are necessary and appropriate remedies for violations of this Article and agrees to the enforcement of such remedies, but without prejudice to the right of Franchisor to recover money damages, which are in no event a full and adequate remedy for such violations.

9.5 CONFIDENTIALITY AGREEMENTS. Franchisee will require all of Franchisee's employees and independent contractors who have access to the Manual or other confidential information to execute an agreement in a form satisfactory to Franchisor, where the employees and independent contractors agree to maintain the confidentiality of all information designated by Franchisor as confidential. During the course of their employment or relationship with Franchisee and thereafter, copies of all executed agreements will be submitted to Franchisor upon request. At Franchisor's request, Franchisee shall also require and obtain execution of a confidentiality agreement in a form acceptable to Franchisor from any or all of the following persons: (a) all officers, directors, and holders of a beneficial interest of 5% or more in the Franchisee or any entity that is directly or indirectly the owner of Franchisee. Failure by Franchisee to obtain execution of the confidentiality agreement required by this Article, or to deliver such confidentiality agreement to Franchisor, shall constitute a material breach of this Agreement.

ARTICLE 10 INITIAL TRAINING PROGRAM; FRANCHISOR ASSISTANCE

INITIAL TRAINING. Franchisor will provide an initial training program for Franchisee and Franchisee's Resident Manager at such location as may be designated by Franchisor to educate, familiarize and acquaint Franchisee and Franchisee's Resident Manager with the operations of Brand Hotels. Franchisee and Franchisee's Resident Manager must successfully complete the initial training program and be certified by Franchisor prior to commencing to operate the Hotel. If Franchisee is using a management company to manage the Hotel, a representative of the management company must also successfully complete the initial training program and be certified by Franchisor prior to starting to manage the Hotel. As of the Effective Date, the period of the initial training program is generally between 4 and 5 days and will be scheduled approximately 60 days before the opening of the Hotel. In the event Franchisor determines that Franchisee or Franchisee's Resident Manager has failed to successfully complete the initial training program, then he or she will not be permitted or authorized to participate in the management of the Hotel. If this Agreement is transferred the new owner of the Hotel will be required to complete the initial training program within 90 days of the date of the transfer ("New Owner Training"). If the Resident Manager is replaced the new Resident Manager will be required to successfully complete the initial training program within 90 days of his or her start date ("New Resident Manager Training"). Franchisor will charge Franchisee its then-current training fee (currently \$2,500) for each person who participates in New Owner Training or New Resident Manager Training. The initial training program is subject to change. If the mandatory initial training program for owners, Resident Managers, or other personnel changes between the Effective Date and date Franchisee and its owners and personnel participate in the program, such

persons will be required to complete the then-current mandatory initial training program and pay such fees associated therewith at the time of the training.

- 10.2 <u>PAYMENT OF EXPENSES</u>. During the initial training program, including any New Owner Training and New Resident Manager Training, Franchisee will provide and pay for the room and board for all persons attending the initial training program on behalf of Franchisee. Franchisee will pay the salaries, fringe benefits, payroll taxes, unemployment compensation, workers' compensation insurance, travel costs and all other expenses for the Franchisee's Resident Manager and other persons sent to the initial training program by Franchisee, and will comply with all applicable state and federal laws pertaining to employees.
- 10.3 OPENING ASSISTANCE. After Franchisee and its Resident Manager have successfully completed the initial training program, Franchisor may furnish a representative to Franchisee for a maximum of 5 days, which may or may not be consecutive, who will provide opening assistance to Franchisee as deemed necessary and appropriate by Franchisor including, but not limited to, assistance with implementing the Business System and evaluating initial business operations. Franchisee will reimburse Franchisor for the room and board, travel costs, and related expenses incurred by its representatives to provide opening assistance.
- **ADDITIONAL TRAINING.** During the term of this Agreement, Franchisee, its owners, or its Resident Manager may (in addition to the meeting/seminar attendance requirements set forth in Section 7.21) be required by Franchisor to attend additional training at such location and time as may be designated by Franchisor. Franchisee will pay the then-current training fee and the room and board, travel costs, salaries, fringe benefits, payroll taxes, unemployment compensation, workers' compensation insurance, and all other expenses for the persons who attend additional training on behalf of Franchisee.
- 10.5 <u>CHANGES IN MANAGEMENT PERSONNEL</u>. Franchisee will immediately notify Franchisor in writing of any personnel changes in the management positions of the Hotel. In the event Franchisee hires a new Resident Manager who has not successfully completed Franchisor's training program, then the new Resident Manager must successfully complete the training program prior to assuming the duties of Resident Manager of the Hotel. Franchisee must pay Franchisor the then-current training fee for each new Resident Manager (or any other person) who attends Franchisor's training program.
- 10.6 <u>TRAINING OF FRANCHISEE'S EMPLOYEES</u>. Except as set forth above in this Article, it is Franchisee's obligation to ensure that its employees have adequate training to perform any tasks assigned to them by Franchisee, and it is Franchisee's obligation to monitor that such employees perform their tasks in compliance with the Business System.

ARTICLE 11 FRANCHISOR'S OBLIGATIONS

11.1 BUSINESS SYSTEM GUIDANCE. Franchisor or its authorized representative will:

- (A) provide Franchisee with one set of prototypical building plans and specifications for the exterior and interior design, color scheme and motif of the Hotel;
- (B) provide Franchisee with advice pertaining to the plans and specifications for the Hotel;
- (C) provide Franchisee with a written schedule of all furniture, fixtures, supplies and equipment necessary and required for the operation of the Hotel;
- (D) provide Franchisee with the Manual and any continuing supplements and modifications to the Manual; and
- (E) inspect the Hotel as often as Franchisor deems necessary and render written reports to Franchisee as deemed appropriate by Franchisor.

ARTICLE 12 SITE SELECTION; BUILDING PLANS

- 12.1 SITE SELECTION. Franchisee will be solely responsible for selecting the site for the Hotel and for purchasing the real estate or otherwise acquiring possession of the site. Accordingly, no provision of this Agreement will be construed or interpreted to impose any obligation upon Franchisor to locate a suitable site for the Hotel, to assist Franchisee in the selection of a suitable site, or to provide any assistance to Franchisee in the purchase of the site. Franchisor recommends that Franchisee, at Franchisee's expense, retain an experienced commercial real estate broker or salesperson to assist Franchisee with the selection of a site for the Hotel and completes a market study for the selected site. Franchisee will not purchase or otherwise acquire the site until the site has been reviewed in writing by Franchisor to evaluate accessibility, visibility, traffic flows and other demographic information. Franchisee acknowledges that neither Franchisor's review of the proposed site for the Hotel nor any assistance that may be provided by Franchisor in the selection or development of the site constitutes a representation, warranty or guaranty by Franchisor regarding the potential financial success of the Hotel operated at that site, and Franchisee assumes all business and economic risks associated with the site.
- 12.2 STANDARD LAYOUT AND SPECIFICATIONS. The Hotel must be constructed or renovated to conform to Franchisor's standard prototype layout and specifications for exterior and interior design and fixtures. Franchisor will provide to Franchisee one set of Franchiser's current standard prototype layout (not construction drawings) and specifications and Franchisee will, at its expense, cause complete and detailed building plans and specifications for the construction or renovation of the Hotel including, but not limited to, site plans, to be prepared and certified by a licensed architect and to be submitted to Franchisor for written approval. All variances from Franchisor's construction standards and specifications and Franchisee's building plans and specifications must have the prior written approval of Franchisor. Any unauthorized variances from Franchisor's construction standards and specifications and the building plans and specifications that have not been approved by Franchisor in writing will be deemed to be a material breach of this Agreement. Franchisee will be responsible for ascertaining that the Hotel is constructed or renovated according to

Franchisor's construction standards and specifications and the building plans and specifications approved in writing by Franchisor. Franchisee will be solely responsible for acquiring all licenses and building permits required in connection with the construction or renovation of the Hotel and for ensuring that the Hotel is in compliance with all applicable local, state, and federal laws and with the Americans with Disabilities Act.

- 12.3 SIGNS. Franchisee will, at its expense, either lease or purchase the standard free-standing sign and the standard driveway directional sign and any other sign that the Franchisor deems necessary (hereinafter collectively referred to as the "signs") which will be displayed at the Hotel. Franchisee will pay for all costs incurred in connection with the installation of the signs, including all electrical work, construction of the base and foundation, relocation of power lines and all required soil preparation work. The signs must conform to Franchisor's standard sign plans and specifications and must be installed at the Hotel in the place, location and manner specified by Franchisor in writing. Franchisee must maintain the signs in good working order and condition at all times during the term of this Agreement. Franchisee will, at its expense, be responsible for obtaining and paying for any and all permits, licenses, repairs, maintenance, utilities, insurance, taxes, assessments and levies in connection with the signs. Franchisee will comply with all federal, state, and local laws, regulations, building codes and ordinances relating to the installation, maintenance, and use of the signs. Franchisee may not alter, remove, change, modify, redesign the signs or add additional signs unless approved by Franchisor in writing.
- **12.4 OPENING OF HOTEL**. Franchisee is required to obtain the prior written approval of Franchisor before the Hotel opens for business. Franchisee will provide Franchisor with all copies of necessary licenses, permits or other information requested prior to Franchisor determining if the Hotel is ready to open for business.

ARTICLE 13 FRANCHISOR'S TERMINATION RIGHTS

- 13.1 <u>GROUNDS FOR TERMINATION</u>. In addition to the other rights of termination contained in this Agreement, Franchisor will have the right (but not the obligation) to terminate this Agreement:
 - (A) immediately upon written notice to Franchisee without providing Franchisee the opportunity to correct the breach if:
 - (1) Franchisee is involved in any act or conduct which reflects materially and unfavorably on the operation or reputation of the Business System or the goodwill associated with the Marks or the Business System;
 - (2) Franchisee misuses or discloses Confidential Information in violation of this Agreement;
 - (3) Franchisee willfully and materially falsifies any report, statement or other written data furnished to Franchisor;

- (4) Franchisee becomes insolvent, has a receiver appointed, or makes a general assignment for the benefit of creditors;
- (5) a suit to foreclose any lien or mortgage against the Hotel premises is instituted against Franchisee and not dismissed within 30 days;
- (6) Franchisee or any of its partners, Directors, officers or majority stockholders are convicted of, or plead guilty to or no contest to a charge of violating any law relating to Franchisee's business, any crime involving moral turpitude, consumer fraud, any felony, or any crime that Franchisor believes is likely to have an adverse effect on Franchisee's ability to carry out the duties imposed by this Agreement or to have an adverse effect on the Business system and the goodwill associated therewith;
- (7) Franchisee ceases to do business, otherwise abandons the business, or forfeits the legal right to do business in the jurisdiction in which the Hotel is located:
- (8) Franchisee or one of its owners transfers any of its rights or obligation in violation of the terms of Article 19 of this Agreement;
 - (9) the nature of Franchisee's breach makes it not curable;
- (10) Franchisee fails to comply with one or more requirements of this Agreement 3 times in any one 12-month period even if the defaults are cured;
- (11) any affiliate of Franchisee defaults under any franchise agreement or other agreement with Franchisor or with any affiliate of Franchisor which is not curable, or if such default is curable, has not cured such default within the applicable cure period.
- At Franchisor's option, without notice, in the event Franchisee shall (B) become bankrupt or become subject to a proceeding under any chapter of the United States Bankruptcy Code, unless Franchisee shall: (i) timely undertake to reaffirm the obligations under the Agreement, (ii) timely comply with all conditions as legally may be imposed by Franchisor upon such an undertaking to reaffirm the Agreement, and (iii) timely comply with such other conditions and provide such assurance as may be legally required in or under relevant provisions of the United States Bankruptcy Code; provided, however, that the parties acknowledge that this Agreement constitutes a personal services contract made in reliance on the qualifications and personal characteristics of Franchisee and its directors, officers, managers, or owners, as the case may be, and in the expectation of a material degree of personal involvement in the management and operation of the Hotel, and consequently, the parties agree that any attempt by any other party, including a trustee in bankruptcy or any other third party, to assume or accept a transfer or assignment of this Agreement shall be void, and that in no event shall this Agreement or any rights or duties of Franchisee hereunder, be transferred to any individual or entity who does not comply with all requirements for transfer specified in this Agreement.

- (C) Effective upon the expiration of 30 days' notice (10 days in the case of non-payment of any payment obligation arising under this Agreement) in the event Franchisee defaults, and does not cure to Franchisor's reasonable satisfaction within the 30 day (or 10 day) notice period, in the performance of any other covenant or provision of this Agreement, including without limitation:
 - (1) Franchisee is unable to obtain site approval or secure fee or leasehold title for the Franchised Location;
 - (2) construction or renovation of the Hotel has not commenced within 12 months after the Effective Date, or within such longer time granted by Franchisor in writing;
 - (3) the Hotel has not opened for business within 24 months after the Effective Date, or within such longer time granted by Franchisor in writing;
 - (4) Franchisee opens and commences operations of its Hotel without the written permission of Franchisor;
 - (5) Franchisee fails to conform to the Business System or the standards of uniformity and quality for the products and services promulgated by Franchisor, including through the Manual; or
 - (6) Franchisee fails to timely pay any of its obligations or liabilities due and owing to Franchisor, Franchisor's affiliates, suppliers, banks, vendors, other creditors or any federal, state or municipal government (including, if applicable, federal and state taxes).
 - (7) Franchisee is in default of its obligations under this Agreement.
- 13.2 NOTICE OF BREACH. If Franchisee commits one or more of the defaults set forth in Section 13.1 of this Agreement and does not cure such defaults to Franchisor's reasonable satisfaction within 30 days (10 days in the case of non-payment of any payment obligation arising under this Agreement) of Franchisor's written notice to Franchisee of such defaults, then this Agreement shall terminate effective upon the expiration of the notice period stated in Franchisor's notice. In the event that applicable law specifies a different notice period, then the notice period in Franchisor's notice shall be automatically amended to the notice period stated in applicable law.
- 13.3 <u>DAMAGES</u>. Termination of this Agreement shall not relieve Franchisee of any unfulfilled obligations to Franchisor created hereunder unless it is so agreed by Franchisor in express writing. In the event this Agreement is terminated by Franchisor pursuant to this Article, or if Franchisee breaches this Agreement by a wrongful termination or a termination that is not in accordance with the terms and conditions of Article 15 of this Agreement, then Franchisor will be entitled to seek recovery from Franchisee for all damages that Franchisor has sustained. Without limiting the preceding sentence, Franchisee will pay Franchisor an amount equal to the average monthly Gross Room Revenues due over the past 12 months (or for the period since the Franchisee started operating, should it be less than 12 months) times

5%, times the lesser of 36 months or the number of months remaining under this Agreement. If this Agreement is terminated for any reason (other than as a result of Franchisor's breach) before the Hotel opens, Franchisee will pay Franchisor \$1,500 per projected guest suite at the Hotel, and, if not already paid in full, the outstanding balance of the Initial Franchise Fee. Franchisee acknowledges the difficulty of ascertaining damages to Franchisor resulting from an early termination of this Agreement. The parties acknowledge that a precise calculation of the full extent of the damages which Franchisor will incur if this Agreement is terminated as a result of Franchisee's default is extremely difficult, and agree that the lump sum payment provided under this Section 13.3 is reasonable in light of such premature termination and is not a penalty or in lieu of any other payments required under this Agreement. Franchisee's payment of liquidated damages will be in addition to amounts due to Franchisor or our affiliates under this Agreement. Except as otherwise provided by applicable law, Franchisee's payment of liquidated damages under this Section 13.3 will not affect Franchisor's right to obtain appropriate, equitable relief and remedies, nor will it affect Franchisor's right to pursue any other remedies.

- 13.4 <u>OTHER REMEDIES</u>. Nothing in this Article will preclude Franchisor from seeking damages or other remedies under state or federal laws, common law or under this Agreement against Franchisee including, but not limited to, attorneys' fees, damages and injunctive relief.
- 13.5 <u>NO WAIVER</u>. No forbearance of Franchisor from asserting any default or giving any permitted notice of termination shall constitute a waiver of such default or right to terminate or an estoppel against such right as to any continuing default or subsequent occurrence of a default, whether similar or dissimilar in nature to the prior default. The rights of Franchisor to terminate this Agreement are in addition to, and not in lieu of, other remedies available at law or equity for defaults by Franchisee in the payment and performance of its obligations hereunder.

ARTICLE 14 FRANCHISEE'S TERMINATION RIGHTS

- 14.1 <u>CONDITIONS OF BREACH</u>. Franchisee will have the right and privilege to terminate this Agreement, as provided herein, if Franchisor violates any material provision, term or condition of this Agreement.
- 14.2 NOTICE OF BREACH. Franchisee will not have the right to terminate this Agreement or to commence an action or lawsuit against Franchisor for breach of this Agreement, injunctive relief, violation of any state, federal or local law, violation of common law (including allegations of fraud and misrepresentation), rescission, general or punitive damages, or termination, unless and until written notice setting forth the alleged breach or violation in detail has been delivered to Franchisor by Franchisee; and Franchisor fails to correct the alleged breach or violation within 60 days after receipt of the written notice by personal service or prepaid registered or certified mail. If Franchisor fails to correct the alleged breach or violation as provided herein within 60 days after receiving written notice, then this Agreement may be terminated by Franchisee provided that Franchisee complies with the post-term obligations described in Articles 15 and 16.

- 14.3 <u>NOTICE OF TERMINATION</u>. If Franchisee has complied with the notice provisions of this Article and if Franchisor has not corrected the alleged breach set forth in the written notice within the time period specified in this Agreement, then Franchisee will have the right to terminate this Agreement by giving Franchisor written notice that this Agreement is terminated, and in that event the effective date of termination of this Agreement will be the day the written notice is received by Franchisor.
- 14.4 <u>WAIVER</u>. Franchisee must give Franchisor immediate written notice of an alleged breach or violation of this Agreement after Franchisee has knowledge of, believes, determines, or is of the opinion that there has been an alleged breach or violation of this Agreement by Franchisor. If Franchisee fails to give written notice to Franchisor of an alleged breach or violation of this Agreement within one year from the date that Franchisee has knowledge of, believes, determines or is of the opinion that there has been an alleged breach or violation by Franchisor, then the alleged breach or violation will be deemed to be condoned, approved and waived by Franchisee, the alleged breach or violation will not be deemed to be a breach or violation of this Agreement by Franchisor and Franchisee will be barred from commencing any action against Franchisor for that alleged breach or violation.

ARTICLE 15 FRANCHISEE'S OBLIGATIONS UPON TERMINATION

- **15.1 TERMINATION OF USE OF MARKS; OTHER OBLIGATIONS**. In the event this Agreement expires or is terminated for any reason, all rights granted hereunder to Franchisee will terminate and Franchisee will immediately (unless another time frame is stated below):
 - (A) cease to operate the Hotel and shall not thereafter, directly, or indirectly, represent itself to the public or hold itself out as a franchisee of Franchisor, except if Franchisee operates other hotels under the Business System;
 - (B) cease to use the Business System, including the Manual and any other operating and training manuals or aids, intranet, advertising and promotional materials, and all confidential materials delivered to Franchisee pursuant to this Agreement.
 - (C) and the Marks, all similar names and marks, or any other designation or mark indicating or tending to indicate that Franchisee is or was a franchisee of Franchisor. Franchisee shall promptly amend or terminate any filings or registrations with any governmental authorities containing or pertaining to the use of Franchisor's name and Proprietary Marks, or any confusingly similar marks. Franchisee shall not promote or advertise the fact that it was formerly a franchisee of Franchisor;
 - (D) pay all amounts due and owing to the Franchisor and its affiliates. Such sums shall include payment of all damages that may be due in connection with the termination of this Agreement, and all costs, expenses, including reasonable attorneys' fees, incurred by Franchisor in connection with seeking or obtaining injunctive or other relief for enforcement of any provisions of this Agreement;

- (E) deliver to Franchisor the Manual, customer and supplier lists, brochures, advertising, marketing and promotional materials and all other printed materials pertaining to the operation of the Hotel and shall retain no copy or record of the foregoing, except for Franchisee's copy of this Agreement and any correspondence between the parties and any other documents which Franchisee reasonably needs for compliance with applicable law;
- (F) discontinue all advertising as a franchisee of Franchisor, including but not limited to removal of all signs and other identifying marks and colors, and shall destroy or surrender to Franchisor any letterheads, forms, printed matter and advertising containing Franchisor's Proprietary Marks and any similar names, marks or designations tending to indicate that Franchisee is or was an authorized franchisee of Franchisor; and
- (G) comply with all terms and conditions that by their terms or by implication survive the termination or expiration of this Agreement.

To the extent Franchisee is a franchisee of other Brand Hotels under other franchise agreements with Franchisor, which agreements are not terminated at the time of termination or expiration of this Agreement, this Article only applies to the Hotel subject to this Agreement.

ALTERATION OF FRANCHISED LOCATION. If this Agreement is terminated for any reason or if the Franchised Location ever ceases to be used as a Brand Hotel, then Franchisee will, at its expense, alter, modify and change, both the exterior and interior appearance of Hotel so that it will be easily distinguished from the standard appearance of a Brand Hotel. At a minimum, such changes and modifications to the Hotel will include removal of all distinctive physical features identifying the Hotel as being a Brand Hotel such as removing all exterior and interior signs and emblems and altering the lobby so that it will be easily distinguished from Brand Hotels. Franchisee agrees that for 90 days following termination or expiration, Franchisor or its designated agents may enter the Property and adjacent areas, and hereby grants Franchisor an irrevocable license and permit to go upon the Property premises for such purposes, at any time to make such alterations, at Franchisee's sole risk and expense, without responsibility for any actual or consequential damages to the property of Franchisee or others. Franchisee acknowledges that such actions by Franchisor are authorized and permitted and shall not be deemed a violation of any civil or criminal law or any basis for an action under such laws by Franchisee or others. Franchisee expressly acknowledges that its failure to make such alterations will cause irreparable injury to Franchisor, and consents to entry, at Franchisee's expense, of an ex parte order by any court of competent jurisdiction authorizing Franchisor or its agents to take such action, if Franchisor seeks such an order.

15.3 TRANSFER OF TELEPHONE DIRECTORY LISTINGS AND ONLINE ACCOUNTS AND PROFILES. As set forth in the Assignment of Telephone Numbers and Service Addendum attached to this Agreement as Exhibit C, upon the termination or expiration of this Agreement, Franchisor will have the right to notify the telephone company and all listing agencies of the termination or expiration of Franchisee's right to use all telephone numbers and all classified and other directory listings for the Hotel or otherwise placed under the name StayAPTTM, and to authorize the telephone company and all listing agencies to transfer to Franchisor or its assignee all telephone numbers and directory listings

for the Hotel. Franchisee acknowledges that Franchisor has the right and interest in and to all telephone numbers and directory listings associated with the Marks, and Franchisee authorizes Franchisor to direct the telephone company and all listing agencies to transfer all of Franchisee's telephone numbers and directory listings to Franchisor or its assignee if this Agreement expires or is terminated for any reason. The telephone company and all listing agencies shall accept this Agreement as evidence of the exclusive rights of Franchisor to such telephone numbers and directory listings. This Agreement will constitute Franchisee's authorization for the telephone company and listing agencies to transfer the telephone numbers and directory listings for the Hotel to Franchisor, and will constitute a release of the telephone company and listing agencies by Franchisee from any and all claims, actions and damages that Franchisee may at any time have the right to allege against them in connection with this Article. Furthermore, Franchisee shall take such action as may be required to transfer ownership and control of any websites, online accounts and profiles used in the operation and promotion of the Hotel to the Franchisor.

15.4 RIGHT OF FRANCHISOR TO PURCHASE BUSINESS ASSETS. Upon the expiration or termination of this Agreement, Franchisor will have the right, but not the obligation, to purchase the signs and the then-usable supplies, inventory, FF&E, and all other assets that are required by Franchisor for a standard Brand Hotel and owned by Franchisee in its Business (the "Business Assets"). Franchisee must give Franchisor written notice listing the cost of each one of the Business Assets in detail and the fair market value for each Business Asset within 5 business days after Franchisee ceases to do business as a Brand Hotel, or after this Agreement expires or is terminated by either party. If Franchisee fails to give Franchisor written notice containing all the requested information, or if Franchisor and Franchisee cannot agree on the fair market value of the Business Assets, then either party will have the right to demand that the fair market value of the Business Assets be determined by a mutually appointed independent appraiser. If the Franchisor and Franchisee cannot agree, each party shall appoint an independent appraiser who shall jointly determine the fair market value of the Business Assets. Franchisor will have the right, but not the obligation, to purchase any or all of the Business Assets from Franchisee for cash within 15 business days after the fair market value of the Business Assets has been agreed to, or established by the independent appraisers, as the case may be. Franchisee shall not before or during such 15 business day period remove from the Hotel, or transfer, assign, hypothecate, pledge, or otherwise encumber such Business Assets. Nothing in this Article will prohibit Franchisor from enforcing the terms and conditions of this Agreement.

ARTICLE 16 NON-SOLICITATION

16.1 NON-SOLICITATION. During the term of this Agreement or any subsequent franchise agreement, Franchisee shall not, either directly or indirectly, individually, or through, or on behalf of, or in conjunction with any person, persons, or entity employ or seek to employ any person who is or was within the immediate past 6 months employed in a management position by Franchisor or any other Brand Hotel franchisee (except for Brand Hotel franchisees who are affiliates of the Franchisee) or induce or seek to induce any such person to leave his or her employment. Franchisor shall not employ or seek to employ any person who is or was within the immediate past 6 months employed by Franchisee in a management position or induce or seek to induce any such person to leave his or her employment. Franchisor and Franchisee agree

that the restrictions in this Section are intended to enable Franchisor and Franchisee to invest in the training of its management employees. Nothing in this provision is intended to prohibit any employee from independently seeking employment with the Franchisor or the Franchisee, or any other Brand Hotel franchisee. Any party violating the provisions of this Section 16.3 shall pay to the former employer as liquidated damages (which the parties agree are difficult of ascertainment) an amount equal to 2 times the annual salary of the employee involved, plus all costs and attorneys' fees incurred by the former employer in connection with such default. The parties hereto agree that each current and future Franchisee in the Business System shall be a third-party beneficiary of the provisions of this Section 16.1, and shall be entitled to enforce the provisions hereof. Franchisor shall have no obligation to enforce the provisions of this Section 16.1 for the benefit of any current or future Franchisee in the Business System of for the benefit of any franchisee of another brand or franchise system operated and managed by Franchisor or any of its affiliates. This section only applies where permitted by applicable law.

16.2 <u>INJUNCTIVE RELIEF</u>. Franchisee, Franchisee's Principal Owners, and any other owners or partners of Franchisee agree that in case of any alleged breach or violations of this Article, Franchisor will have the right to seek injunctive relief in addition to all other remedies that may be available to Franchisor at equity or law and Franchisee shall pay to Franchisor all costs, expenses, including reasonable attorneys' fees, incurred by Franchisor in connection with seeking or obtaining injunctive or other relief for enforcement of any alleged breach of this Article.

ARTICLE 17 DISPUTE RESOLUTION

- 17.1 <u>DISPUTES NOT SUBJECT TO MEDIATION OR ARBITRATION</u>. The following disputes, claims and controversies between Franchisor and Franchisee will not be subject to mediation or arbitration:
 - (A) the payment of fees by Franchisee to Franchisor;
 - (B) any dispute involving the Marks, including the ownership, validity, use of, or right to use or license the Marks;
 - (C) any dispute involving immediate termination of this Agreement by Franchisor pursuant to Section 13.1(A) of this Agreement;
 - (D) any dispute involving enforcement of the confidentiality provisions set forth in Sections 9.4 and 9.5 of this Agreement, including to prevent threatened disclosure or misuse of confidential information; and
 - (E) any dispute involving enforcement of the non-solicitation provisions set forth in Article 16 of this Agreement.

Any and all court proceedings arising from or relating in any manner to any dispute, claim or dispute excluded in this Article from mediation and arbitration shall, at Franchisor's option, be brought in, the United States District Court for the district in which the Franchisor's headquarters are located at the time the complaint is filed. No individual shall be joined as a

party to such proceedings if such joinder has the effect of destroying federal court jurisdiction unless that individual or entity is a necessary party to the proceeding as a matter of law. If the United States District Court does not have jurisdiction over the dispute, the proceeding may be initiated in, and only in, the state court for the location in which in which the Franchisor's headquarters are located at the time the complaint is filed. In either case, Franchisor and Franchisee hereby consent and submit to the exercise of jurisdiction by such courts.

THE PARTIES AGREE THAT ALL DISPUTES, CLAIMS AND CONTROVERSIES ADMITTED TO THE COURT PURSUANT TO THIS PROVISION SHALL BE TRIED TO THE COURT SITTING WITHOUT A JURY, NOTWITHSTANDING ANY STATE OR FEDERAL CONSTITUTIONAL OR STATUTORY RIGHTS OR PROVISIONS.

17.2 **MEDIATION**. Except as provided in Sections 13.1 (A) (Grounds for Immediate Termination) and 17.1 (Disputes not Subject to Mediation or Arbitration), all disputes, controversies and claims between Franchisor, its officers, Directors, owners and affiliates, and Franchisee, its officers, Directors, owners and Personal Guarantors, arising under, as a result of, or in connection with this Agreement, or relating to Franchisee's development or operation of the Hotel (each a "Dispute") will be subject to non-binding mediation. The mediator will be appointed in accordance with the Commercial Mediation Rules of the American Arbitration Association unless Franchisor and Franchisee agree on a mediator in writing within 10 days after either party gives written notice of mediation. All mediation hearings will take place exclusively in such city and state where the Franchisor has its headquarters at the time, and will be held within 20 days after a request for mediator has been made. The mediation shall be private, voluntary, and nonbinding. The hearing will be informal and the mediator will have the right to hear and review all testimony and evidence presented by either Franchisor or Franchisee. Any party may withdraw from the mediation at any time before signing a settlement agreement upon written notice to each other party and to the mediator. The cost of the mediator will be paid equally by Franchisor and Franchisee. The mediator shall be disqualified as a witness, consultant, expert or counsel for either party with respect to the matters in Dispute and any related matters. Unless the parties agree otherwise, the entire mediation process shall be confidential and without prejudice. The parties and the mediator shall not disclose any information, documents, statements, positions, or terms of settlement. Nothing said or done or provided by the parties in the course of mediation shall be reported or recorded or, except as ordered by a court of competent jurisdiction, placed in any legal proceeding or construed for any purpose as an admission against interest. Nevertheless, evidence otherwise discoverable or admissible is not excluded from discovery or admission as a result of its use in mediation. The commencement of any mediation shall not act to prevent either party from instituting or proceeding with any action which may be the subject of the Dispute.

17.3 ARBITRATION. If a Dispute is not resolved in accordance with the provisions for mediation set forth in such Article, then the Dispute shall be submitted to binding arbitration in accordance with the Commercial Rules and Regulations of the American Arbitration Association. All arbitration hearings will take place exclusively in such city and state where the Franchisor has its headquarters at the time. The authority of the arbitrator will be limited to making a finding, judgment, decision and award relating to the interpretation of or adherence to the written provisions of this Agreement. The arbitrator will not have the authority or right to add to, delete, amend or modify in any manner the terms, conditions and provisions of this

Agreement, nor have the right or authority to award punitive damages to Franchisor or Franchisee or their officers, directors, or owners, and Franchisee's Personal Guarantors. Franchisor and Franchisee and their respective officers, directors, owners, and Franchisee's Personal Guarantors expressly waive their rights to plead or seek punitive damages. All findings, judgments, decisions and awards by the arbitrator will be final and binding on Franchisor, Franchisee, and their respective officers, directors, owners, and Franchisee's Personal Guarantors. The written decision of the arbitrator will be deemed to be an order, judgment and decree and may be entered as such in any court of competent jurisdiction by either party. Unless required to be disclosed by law, the written decision of the arbitrator will be confidential. Franchisor and Franchisee acknowledge that a summary of such decisions may have to be disclosed in Franchisor's future franchise disclosure documents.

INJUNCTIVE RELIEF. Notwithstanding any provisions to the contrary, Franchisee recognizes that its Hotel is one of a larger number of hotels identified by the Marks and Business System and the failure on the part of a single franchisee to comply with the terms and conditions of this Agreement could cause irreparable damage to Franchisor and/or to other Brand Hotel franchisees. Therefore, it is mutually agreed that in the event of a breach of threatened breach of any of the terms of this Agreement by Franchisee or any dispute, claim and controversy regarding the matters set forth in Section 17.1 of this Agreement, then Franchisor will be entitled to an injunction restraining such breach and/or to a decree of specific performance, without showing or proving actual damage, together with recovery of reasonable attorneys' fees and other costs incurred in obtaining said equitable relief, until such time as a final and binding determination is made by the arbitrator, or court, as the case may be. Similarly, it is mutually agreed that in the event of a breach or threatened breach of any of the terms of this Agreement by Franchisor, Franchisee will be entitled to an injunction restraining such breach and/or to a decree of specific performance, without showing or proving any actual damage, together with recovery of reasonable attorneys' fees and other costs incurred in obtaining said equitable relief, until such time as a final and binding determination is made by the arbitrator. The foregoing equitable remedies shall be in addition to, and not in lieu of, all other remedies or rights which the parties might otherwise have by virtue of any breach of this Agreement by the other party.

herein, all arbitration findings and awards expressly made by the arbitrator will be final and binding on Franchisor and Franchisee and their officers, Directors, or owners, and Franchisee's Personal Guarantors; however, such arbitration findings and awards may not be used to collaterally stop either party from raising any like or similar issues, claims or defenses in any other or subsequent arbitration, litigation, court hearing or other proceeding involving third parties or other franchisees. Any disagreement between Franchisee and Franchisor (and Franchisor's affiliates, officers, directors, or owners) will be considered unique as to its facts and must not be brought as a class action, and Franchisee waives any right to proceed against Franchisor (and Franchisor's affiliates, officers, directors, or owners) by way of class action, or by way of a multi-plaintiff, consolidated or collective action. No party except Franchisor, Franchisee and their officers, Directors, owners or partners and Franchisee's Personal Guarantors will have the right to join in any arbitration proceeding arising under this Agreement, and, therefore, the arbitrator will not be authorized to permit or approve class actions or to permit any

person or entity that is not a party to this Agreement to be involved in or to participate in any arbitration hearings conducted pursuant to this Agreement.

Any disagreement between Franchisee and Franchisor (and Franchisor's affiliates, officers, directors, or owners) will be considered unique as to its facts and must not be brought as a class action, and Franchisee waives any right to proceed against Franchisor (and Franchisor's affiliates, officers, directors, or owners) by way of class action, or by way of a multi-plaintiff, consolidated or collective action

- 17.6 NO PUNITIVE OR EXEMPLARY DAMAGES. NO PUNITIVE OR EXEMPLARY DAMAGES SHALL BE AWARDED AGAINST EITHER FRANCHISOR OR FRANCHISEE OR ANY AFFILIATES OF EITHER OF THEM, IN ANY PROCEEDING, AND ALL CLAIMS TO SUCH DAMAGES ARE HEREBY WAIVED.
- 17.7 <u>BUSINESS JUDGMENT</u>. The parties hereto recognize, and any mediator or judge is affirmatively advised, that certain provisions of this Agreement describe the right of Franchisor to take (or refrain from taking) certain actions in the exercise of its discretion based on its assessment of the overall best interest of the network and/or franchise program. Where such discretion has been exercised, and is supported by the business judgment of Franchisor, neither a mediator nor a judge shall substitute his or her judgment for the judgment so exercised by Franchisor.
- 17.8 <u>CONFIDENTIALITY</u>. All evidence, testimony, records, documents, findings, decisions, judgments and awards pertaining to any arbitration hearing between Franchisor and Franchisee will be secret and confidential in all respects. Franchisor and Franchisee will not disclose the decision or award of the arbitrator and will not disclose any evidence, testimony, records, documents, findings, orders, or other matters from the arbitration hearing to any person or entity except as required by law.
- 17.9 ENFORCEMENT. Any claim arising out of or relating to this Agreement, the relationship of the parties, Franchisor's operation of the Business System, or Franchisee's operation of the Hotel will be barred unless filed before the expiration of the earlier of: (1) the time period for bringing an action under any applicable state or federal statute of limitations; (2) 1 year after the date upon which a party discovered, or should have discovered, the facts giving rise to an alleged claim; or (3) 2 years after the first act or omission giving rise to an alleged claim. Notwithstanding the foregoing, where state or federal law mandates or makes possible by notice or otherwise a shorter period, such shorter period shall apply in all cases, in lieu of the time specified above. Claims of Franchisor for the underreporting of Gross Room Revenues, for indemnification, or for claims related to Franchisor's rights under any of the Trademarks shall be subject only to the applicable state or federal statute of limitations.
- 17.10 ENFORCEMENT COSTS. In the event Franchisor incurs legal fees or costs or other expenses to enforce any obligation of Franchisee hereunder, or to defend against any claim, demand, action or proceeding by reason of Franchisee's failure to perform or observe any obligation imposed upon Franchisee by this Agreement, then Franchisor shall be entitled to recover from Franchisee the amount of all legal fees, costs and expenses, including reasonable attorneys' fees, whether incurred prior to, or in preparation for or contemplation of the filing of

any claim, demand, action, or proceeding to enforce any obligation of Franchisee hereunder or thereafter or otherwise.

ARTICLE 18 FRANCHISOR'S RIGHT OF FIRST REFUSAL TO PURCHASE

RESTRICTIONS. Franchisee and its owners, if Franchisee is an entity as defined in Section 23.2, will not sell, assign, trade, transfer, lease, sublease, or otherwise dispose of any interest in or any part of the Hotel or this Agreement to any party without first offering the same to Franchisor in a written offer that contains all material terms and conditions of the proposed sale or transfer. Within 30 business days after receipt by Franchisor of Franchisee's written offer, Franchisor will give Franchisee written notice which will either waive its right of first refusal to purchase, or will state an interest in purchasing the Hotel according to the proposed terms. If Franchisor commences negotiations to purchase the Hotel as set forth herein, then Franchisee may not sell the Hotel to a third party for at least 60 days or until Franchisor and Franchisee agree in writing that the negotiations have terminated, whichever occurs first. If Franchisor waives its right to purchase, then Franchisee will have the right to complete the sale or transfer of the Hotel according to the terms set forth in the written notice to Franchisor; however, any such sale, transfer or assignment to a third party is expressly subject to the terms and conditions set forth in Article 19 of this Agreement. Franchisee's obligations to comply with the terms and conditions of this Agreement including, but not limited to, its obligations to pay the Royalty Fees and all other fees and to operate the business as a Brand Hotel, will in no way be affected or changed because of Franchisor's non-acceptance of Franchisee's written offer to purchase the Hotel, and as a consequence, the terms and conditions of this Agreement will remain in full force and effect.

CORPORATE FRANCHISEE. If the Franchisee is an entity as defined in Section 23.2, then an ownership interest of Franchisee ("ownership interest") may not be sold, pledged, assigned, transferred, traded or otherwise disposed of by any of Franchisee's owners until the ownership interest has first been offered to Franchisor in writing under the same terms and conditions as those offered to any third party. Franchisor will have 20 business days within which to accept any owner's offer to sell, assign, trade, transfer or dispose of the ownership interest. Notwithstanding the terms of this Article, Franchisee's initial owners may bequeath, sell, assign, trade or transfer their ownership interest to their spouse or children or to the other initial owners of Franchisee's entity without first offering it to Franchisor, however, Franchisee must provide Franchisor with prior written notice of all such transactions and the proposed transferee owners must agree to be personally liable under this Agreement. Franchisee acknowledges and agrees that it will obtain from each of its owners signed copies of all necessary governing documents, with appropriate assignment and transfer restrictions, to allow Franchisor to exercise its rights under Articles 18 and 19 of this Agreement, and in accordance with the Manual, as periodically modified by Franchisor. Franchisor will have the right, upon request, to review and approve the governing documents described in the preceding sentence.

ARTICLE 19 ASSIGNMENT

19.1 ASSIGNMENT BY FRANCHISOR. This Agreement may be unilaterally assigned and transferred by Franchisor without the approval or consent of Franchisee, and will inure to the benefit of Franchisor's successors and assigns. The assignee will be required to fully perform Franchisor's obligations under this Agreement and will be solely responsible for all obligations of Franchisor under this Agreement from the date of assignment and transfer. Franchisee waives any and all rights that it may have or allege against Franchisor resulting from the assignment of this Agreement. Franchisor and its owners also have the right to sell, pledge, assign, transfer, trade or otherwise dispose of their ownership interest in Franchisor without the approval or consent of Franchisee.

19.2 LIMITATIONS ON ASSIGNMENT AND TRANSFER BY FRANCHISEE.

This Agreement is not transferable by Franchisee except as permitted herein. The rights and duties set forth in this Agreement are personal to Franchisee and are granted in reliance on the individual and collective business skill, financial capacity, and personal character of Franchisee and its owners. Accordingly, neither this Agreement, nor the franchise granted hereunder, nor any part or all of any owner's direct or indirect ownership interest in Franchisee, the Hotel, nor a substantial portion of, or all of, the Hotel's assets (collectively, the "Franchised Interests"), may be transferred by Franchisee without Franchisor's prior written approval, and then only in accordance with the provisions of this Agreement. Any purported Transfer by Franchisee or any owner of any Franchised Interest, by operation of law or otherwise, which is not permitted hereunder, shall be null and void and shall constitute a material breach of this Agreement, for which Franchisor may then terminate without opportunity to cure. Any references in this Article 19 to a "transfer" shall mean the voluntary, involuntary, direct or indirect assignment, sale, gift, or other transfer of any Franchised Interest, including, without limitation, the following events: (i) the transfer of ownership of the stock or partnership or limited liability company ownership interest of Franchisee; (ii) any merger, reorganization, consolidation, or issuance of additional securities representing a direct or indirect ownership interest in Franchisee or the Hotel; (iii) any sale of a controlling interest in Franchisee in a single transaction or a related series of transactions; (iv) transfer of a Franchised Interest by declaration, division, or otherwise in a divorce, insolvency, corporate or partnership dissolution proceeding or otherwise by operation of law; (v) transfer of a Franchised Interest in the event of Franchisee's death or the death of one of its owners with a controlling interest, by will, declaration of or transfer in trust, or under the laws of intestate succession; (vi) any change in ownership or control of any or all of the Franchised Interest by sale, gift, assignment, or otherwise; (vii) if Franchisee or any owner with a controlling interest is a trust, any change in the trustees or the beneficial owners of the trust; or (viii) a pledge, hypothecation, or encumbrance of any Franchised Interest intended as security for an obligation.

19.3 <u>SECURITY INTERESTS</u>. Franchisee shall grant no security interest, lien, mortgage, or deed of trust on any or all of the Franchised Interests without prior written notice to Franchisor and then only if Franchisee agrees to request that the lender enter into a comfort letter as to such Franchised Interests in form and substance reasonably acceptable to Franchisor.

19.4 ASSIGNMENT BY INDIVIDUAL FRANCHISEE TO ENTITY. If Franchisee is an individual or several individuals, this Agreement may be transferred or assigned by Franchisee to an entity which is wholly owned by Franchisee, provided that the Principal Owners of the assignee entity sign or have signed the Owner's Acknowledgement to this Agreement and the Personal Guaranty in the form attached as Exhibit A to this Agreement, and further provided that Franchisee furnishes prior written proof to Franchisor substantiating that the entity will be financially able to perform all of the terms and conditions of this Agreement. Franchisee will give Franchisor 15 days' written notice prior to the proposed date of assignment or transfer of this Agreement to an entity wholly owned by Franchisee; however, the transfer or assignment of this Agreement will not be valid or effective until Franchisor has received the legal documents which its legal counsel deems necessary to properly and legally document the transfer or assignment of this Agreement to the entity as provided for herein.

19.5 ASSIGNMENT IN EVENT OF DEATH OR PERMANENT DISABILITY.

- (A) If, upon Franchisee's death or Incapacity, or, if Franchisee is a corporation, partnership, or limited liability company, upon the death of a Principal Owner or upon the determination by Franchisor that a Principal Owner of Franchisee is Incapacitated, Franchisee's or such owner's executor, administrator, conservator, guardian, or other legally appointed personal representative must transfer Franchisee's interest in this Agreement or the Principal Owner's interest in Franchisee to a third party. However, the assignment of this Agreement to the transferee, assignee or beneficiary of Franchisee will be subject to the applicable provisions of Section 19.6, and will not be valid or effective until Franchisor has received the properly executed legal documents which its legal counsel deems necessary to properly and legally document the transfer, assignment or bequest of this Agreement, and until each of the Principal Owners of the transferee, assignee or beneficiary agrees to be unconditionally bound by the terms and conditions of this Agreement and to personally guarantee the performance of Franchisee's obligations under this Agreement, and signs the Personal Guaranty, as applicable, in the form attached as Exhibit A to this Agreement. As used in this Agreement, "Incapacitated" or "Incapacity" means, the inability of Franchisee, or its Principal Owner if an entity, to operate the Hotel in the ordinary course of business for 30 days or more in any consecutive 90-day period.
- (B) Such disposition of the Agreement or the interest in Franchisee of a Principal Owner (including, without limitation, transfer by bequest or inheritance) must be completed within a reasonable time, not to exceed one year from the date of death or Incapacity. A failure to transfer Franchisee's interest in this Agreement or the interest of a Principal Owner in Franchisee within this period of time constitutes a breach of this Agreement. Adequate provision must be made, in the sole discretion of Franchisor, for management of the Property during such period. Franchisee's interest in this Agreement or any Principal Owner's interest in Franchisee which is an entity may, with Franchisor's consent, which will not be unreasonably withheld, be transferred to the decedent's spouse, parent, sibling, or direct descendant or to spouse's direct descendant.
- (C) If, upon Franchisee's death or Incapacity, or upon the death or Incapacity of a Principal Owner, the Hotel is not being managed by an approved management

company, Franchisee's or the owner's executor, administrator, conservator, guardian, or other legally appointed personal representative must within a reasonable time, not to exceed 30 days from the date of death or declaration of Incapacity, appoint an approved management company to operate the Hotel. Such an approved management company may be appointed only with Franchisor's prior written approval and will be required to complete training at Franchisee's expense. Pending the appointment of an approved management company as provided above or if, in Franchisor's reasonable judgment, the Hotel is not being managed properly at any time after Franchisee's death or declaration of Incapacity or after the death or declaration of Incapacity of an owner of a Controlling Interest in Franchisee, Franchisor has the right, but not the obligation, to appoint a property manager or management company for the Hotel. All funds from the operation of the Hotel during the management by Franchisor's appointed property manager or management company will be kept in a separate account, and all expenses of the Hotel, including compensation, other costs, and travel and living expenses incurred by the management company, will be charged to this account. Franchisor also has the right to charge a reasonable management fee (in addition to the Royalty Fees and Marketing Fee payable under this Agreement) during the period that Franchisor's appointed property manager or management company manages the Hotel. Operation of the Hotel during any such period will be on the transferee's behalf, provided that Franchisor only has a duty to utilize commercially reasonable efforts and will not be liable to Franchisee or its owners for any debts, losses or expenses, or obligations incurred by the Hotel or to any creditors for any products, materials, supplies, or services the Hotel purchases during any period it is managed by Franchisor's appointed property manager or management company. The transferee will remain solely responsible for maintaining the Hotel during any period in which Franchisor's appointed property manager or management company is managing the Hotel on the transferee's behalf.

- 19.6 <u>CONDITIONS TO TRANSFER OR ASSIGNMENT</u>. Neither Franchisee, nor its owners, will transfer, whether voluntary or involuntary, assign or otherwise dispose of, in one or more transactions, of any Franchised Interests, including this Agreement, the Hotel, any substantial portion of the Hotel's assets, revenues or profits (except in the ordinary course of business), or any interest in Franchisee unless Franchisee obtains Franchisor's prior written consent (except as provided in Section 19.4). Franchisor will not unreasonably withhold its consent to any such transfer, assignment or sale, provided Franchisee and/or the transferee franchisee comply with the following conditions:
 - (A) All of Franchisee's monetary obligations due to Franchisor have been paid in full, and Franchisee is not otherwise in default under this Agreement; and
 - (B) Franchisee (and each Principal Owner, if applicable) has executed a written agreement in a form satisfactory to the Franchisor, in which Franchisee and each Principal Owner agrees to observe all applicable post-term obligations and covenants contained in this Agreement; and
 - (C) The transferee franchisee has demonstrated to Franchisor's satisfaction that he, she or it meets Franchisor's managerial, financial, business and other standards and criteria for new franchisees, possesses a good business reputation and credit rating,

and possesses the aptitude and ability to operate the Hotel in an economic and businesslike manner (as may be evidenced by prior related business experience or otherwise); and

- (D) The transferee franchisee has executed Franchisor's then-current standard Franchise Agreement and other ancillary agreements required by Franchisor, and each Principal Owner of the transferee franchisee, as applicable, has executed the Personal Guaranty attached to the Franchise Agreement and signed the Owner's Acknowledgement to the Franchise Agreement, all as required by Franchisor; and
- (E) Prior to the date of transfer, the transferee franchisee and its Resident Manager will have successfully completed the training program prescribed by the Franchisor and will have paid all then-current training fees; and
- (F) Franchisee or transferee at its expense upgrades the Hotel to conform to the then-current Business System standards and specifications, and completes the upgrading and other requirements within the time specified by Franchisor; and
- (G) The transferee franchisee will have purchased the real estate and building or otherwise have a leasehold interest acceptable to Franchisor in the Franchised Location; and
- (H) Franchisee (and each Principal Owner, if applicable) signs a general release, in a form and substance satisfactory to us, of any and all claims against Franchisor and its affiliates, officers, directors, employees and agents, except to the extent limited or prohibited by applicable law; and
- (I) Franchisor approves the material provisions of the transfer, assignment or sale of assets, which transfer, assignment or sale cannot permit Franchisee to retain a security interest in this Agreement or any other intangible asset; and
- (J) Franchisee agrees to continue to be bound by, and remain liable for all of the obligations to Franchisor in connection with the Hotel that arose prior to the effective date of the transfer, and any covenants that survive the termination or expiration of this Agreement, and shall execute any and all instruments reasonably requested by Franchisor to evidence such liability; and
- (K) The purchase price and terms of the transfer, assignment or sale may not negatively impact the capability of the Hotel to profit after the transfer, assignment or sale; and
- (L) Either Franchisee or the transferee franchisee, as applicable, will have paid the fee required in Section 19.7.

Franchisor's consent to a transfer by Franchisee or by the owners of Franchisee of any Franchised Interests, shall not constitute a waiver of any claims Franchisor may have against the transferring party, nor shall it be deemed a waiver of Franchisor's right to demand exact compliance with any of the terms of this Agreement by the transferee.

19.7 TRANSFER FEE.

- (A) If Franchised Interests are assigned, transferred or bequeathed as a result of the death or incapacity of Franchisee or its owners, or if Franchisee's direct or indirect owners transfer less than 50% of their ownership interest to another person or entity (provided that no change in control of Franchisee or the Hotel occurs as a result of the transfer), then Franchisee will pay to Franchisor a transfer fee of \$10,000, or, if the Franchisor's expenses incurred in connection with the transfer exceed such amount, the actual expenses incurred by Franchisor. The expenses to be considered in calculating the Franchisor's expenses for purposes of the preceding sentence will include Franchisor attorneys' fees, accountants' fees, out-of-pocket expenses, and administrative expenses incurred in connection with determining whether the transfer and transferee satisfy the conditions to transfer or assignment set forth in Section 19.6, as well as the execution of the transfer.
- (B) In the event of any transfer of Franchised Interests not addressed in Section 19.7 (A), then the transferee franchisee will pay a transfer fee equal to Franchisor's then-current Initial Franchise Fee.

ARTICLE 20 INDEPENDENT CONTRACTORS; INDEMNIFICATION; ATTORNEYS' FEES

- **INDEPENDENT CONTRACTORS.** Franchisor and Franchisee are each 20.1 independent contractors, and as a consequence, Franchisor and Franchisee will not be deemed or considered joint venturers, partners, agents, servants, subsidiaries, employees, fiduciaries or representatives of each other for any purpose whatsoever. Franchisee is not authorized to, and will not, make any agreements, representations or warranties in the name of or on behalf of Franchisor or represent that their relationship is other than that of franchisor and franchisee. Neither Franchisor nor Franchisee will be obligated by or have any liability to the other under any agreements or representations made by the other to any third parties. Nothing in this Agreement or in its performance, or otherwise is intended to make Franchisor the employer or joint employer of Franchisee, or of any of Franchisee's employees, agents or independent contractors. Franchisee agrees that Franchisor is not in a position to, and does not undertake to: exercise control over the employment, supervision, or discharge of Hotel employees and has no right to do so, other than to the extent of controlling the quality of the Business System/brand and the quality of the products and services that Franchisee offers; Hotel maintenance; guest safety and health; or other matters arising out of or affecting Hotel operations, which are within the responsibility of Franchisee as a qualified independent business operator.
- **20.2 INDEMNIFICATION**. Franchisor will not be obligated to any person or entity for damages arising out of Franchisee's operation of the Hotel. Franchisee will indemnify and hold Franchisor harmless from and against all claims, lawsuits, damages, obligations, liability, actions and judgments alleged by any person or entity against Franchisor arising out of, as a result of or in connection with Franchisee's operation of the Hotel regardless of cause or any concurrent or contributing fault or negligence of Franchisor. Further, in any such action,

Franchisee will indemnify and reimburse Franchisor for all costs incurred by Franchisor in the defense of any claim brought against it or in any action in which it is named as a party including, without limitation, reasonable attorneys' fees, investigation expenses, court costs, deposition expenses, and travel and living expenses.

ARTICLE 21 EMINENT DOMAIN; DESTRUCTION OR DAMAGE

- 21.1 EMINENT DOMAIN. Franchisee will give Franchisor notice of any proposed taking or sale in lieu thereof by eminent domain in the United States as soon as Franchisee becomes aware of the possibility. If the Hotel or a substantial part thereof is taken or sold and this renders it impractical or impossible for Franchisee to continue to operate the Hotel, Franchisor may terminate this Agreement by notice to Franchisee. If the Hotel is able to continue to operate after a partial taking, Franchisee will perform whatever construction is necessary to bring the Hotel into compliance with this Agreement. Franchisee will pursue on Franchisor's behalf a claim for compensation which includes Franchisor's lost Royalty Fees income and pay any amount as recovered to Franchisor. If Franchisor is allowed to pursue such a claim on its own behalf, Franchisee will cooperate with Franchisor with respect to this effort. If this Agreement is terminated pursuant to this Article, and Franchisee was not in default of this Agreement, Franchisor may grant Franchisee a franchise for a new hotel provided that Franchisee files an application with Franchisor. Franchisor will review the application in accordance with its then-current standards, criteria and requirements. If the application is approved, Franchisor and Franchisee will enter into a new then-current Franchise Agreement.
- **21.2 DESTRUCTION OR DAMAGE**. If the Hotel is destroyed or suffers damage by fire, wind, rain, vandalism or other casualty, Franchisee will promptly perform whatever construction is necessary to bring the Hotel into compliance with this Agreement and recommence business as soon as reasonably possible.

ARTICLE 22 GENERAL PROVISIONS

- **22.1** GOVERNING LAW. Except to the extent governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. §1051 et seq.), or modified by the Federal Arbitration Act of the United States of America (9 U.S.C. § 1 et seq.), this Agreement and the relationship between Franchisor and Franchisee will be governed by the laws of the state of North Carolina, without reference to its choice of law provisions. The provisions of this Agreement which conflict with or are inconsistent with applicable governing law will be superseded and/or modified by such applicable law only to the extent such provisions are inconsistent. All other provisions of this Agreement will be enforceable as originally made and entered into upon the execution of this Agreement by Franchisee and Franchisor.
- **22.2 SEVERABILITY**. All provisions of this Agreement are severable, and this Agreement will be interpreted and enforced as if all completely invalid or unenforceable provisions were not contained herein and partially valid and enforceable provisions will be enforced to the extent valid and enforceable. If any applicable law or rule of any jurisdiction requires a greater prior notice of the termination of or refusal to renew this Agreement than is

required hereunder or the taking of some other action not required hereunder, or if under any applicable law or rule of any jurisdiction, any provision of this Agreement or any specification, standard or operating procedure prescribed by the Franchisor is invalid or unenforceable, the prior notice or other action required by such law or rule will be substituted for the notice requirements hereof, or such invalid or unenforceable provision, specification, standard or operating procedure will be modified to the extent required to be valid and enforceable. Such modifications to this Agreement will be effective only in such jurisdiction and will be enforced as originally made and entered into in all other jurisdictions.

- **YAIVER**. Franchisor and Franchisee may, by written instrument signed by Franchisor and Franchisee, waive any obligation of or restriction upon the other under this Agreement. No delay, waiver, omission, or forbearance on the part of Franchisor to exercise any right, option, duty, or power arising out of any breach or default by Franchisee, or any other franchisee, of any of the terms, provisions, covenants, or conditions hereof shall constitute a waiver by Franchisor to enforce any such right, option, duty, or power as against Franchisee, or as to subsequent breach or default by Franchisee. Subsequent acceptance by Franchisor of any payment by Franchisee or late performance of any other obligation under this Agreement shall not be deemed a waiver by Franchisor of any preceding breach by Franchisee.
- **22.4** <u>CUMULATIVE RIGHTS</u>. The rights of Franchisor hereunder are cumulative and no exercise or enforcement by Franchisor of any right or remedy hereunder will preclude the exercise or enforcement by Franchisor of any other right or remedy hereunder or which Franchisor is entitled by law to enforce.
- **22.5 BINDING AGREEMENT**. This Agreement is binding upon the parties hereto and their respective executors, administrators, heirs, permitted assigns and permitted successors in interest.
- ENTIRE AGREEMENT. This Agreement, together with any other documents or agreement executed by the parties contemporaneously hereto, supersedes and terminates all prior agreements, either oral or in writing, between the parties involving the franchise relationship and therefore, representations, inducements, promises or agreements between the parties not contained in this Agreement or not in writing signed by Franchisor and Franchisee will not be enforceable. The preambles are a part of this Agreement, which, together with any other documents or agreement executed by the parties contemporaneously hereto, constitutes the entire agreement of the parties, and there are no other oral or written understandings or agreements between Franchisor and Franchisee relating to the subject matter of this Agreement. Nothing in the Agreement is intended to disclaim the representations Franchisor made in the Franchise Disclosure Document that Franchisor furnished to Franchisee. No change, modification, amendment or waiver of any of the provisions hereof, including by custom or usage of trade or course of dealing or performance, shall be effective and binding upon either party unless it is in writing, specifically identifies as an amendment hereto, and signed by the party to be charged.

- **22.7 JOINT AND SEVERAL LIABILITY**. If Franchisee consists of more than 1 individual, then the liability of all such individuals under this Agreement will be deemed to be joint and several.
- **22.8 NO ORAL MODIFICATIONS**. No oral modification, change, addition, rescission, release, amendment or waiver of this Agreement and no approval, consent or authorization required by any provision of this Agreement may be made except by a written agreement signed by Franchisor.
- 22.9 <u>HEADINGS</u>; <u>TERMS</u>. The headings of the Articles and Sections are for convenience only and do not define, limit or construe the contents of such Articles and Sections. The term "Franchisee" as used herein is applicable to one or more persons, a corporation, a limited liability company, a partnership, or other legal entity, as the case may be, and the singular usage includes the plural, the masculine usage includes the neuter and the feminine, and the neuter usage includes the masculine and the feminine. References to "Franchisee," "assignee" and "transferee" which are applicable to an individual or individuals will mean the principal owner or owners of the equity or operating control of Franchisee or any such assignee or transferee if Franchisee or such assignee or transferee is a legal entity. The term "including" means "including, but not limited to."
- **22.10** NOTICES. All notices to required or permitted under this Agreement will be in writing and will be made by personal service, certified or registered mail, postage prepaid, return receipt requested, sent via a nationally recognized overnight delivery service, or sent by e-mail (with a confirming copy mailed or sent by overnight delivery service) at the following addresses designated by the parties or such other address as has been designated by written notice to the other party:

Notices to Franchisor:

Attn:

10801 Monroe Road, Suite 200

Matthews, NC 28105

Email:

@stayAPT.com

Notices to Franchisee:

Attn.:

Email:

Any notice shall be deemed to have been given at the earlier of actual receipt or 3 business days after mailing by certified or registered mail, or 1 business day after sending by e-mail or overnight delivery service.

- 22.11 PERSONAL GUARANTY. All of Franchisee's Principal Owners (if Franchisee is an entity) will sign the Personal Guaranty in the form attached to this Agreement as Exhibit A (the "Personal Guaranty"). Any person or entity that at any time after the Effective Date becomes a Principal Owner of Franchisee under the provisions of Article 19 or otherwise will, as a condition of becoming a Principal Owner, sign the Personal Guaranty. In addition, any other person Franchisor designates, including without limitation any person who co-signed a loan or was involved in obtaining financing for the Hotel, must also sign the Personal Guaranty. Any person or entity that signs the Personal Guaranty is referred to herein as a "Personal Guarantor." In connection with the Personal Guaranty, Franchisor has the right to require Franchisee to provide Franchisor with additional information, including without limitation loan documents. Further, all of Franchisee's Principal Owners must sign the Acknowledgment Addendum attached as Exhibit D.
- **22.12 NO THIRD-PARTY BENEFICIARIES**. This Agreement is made solely for the benefit of the parties hereto and their respective successors and permitted assigns, and nothing herein shall create any right to rely upon the terms hereof in favor of any third party nor confer any right or remedy upon any third party, except as specifically provided in Section 16.3 of this Agreement.
- **22.13 TIME IS OF THE ESSENCE**. Time is of the essence of this Agreement and all provisions hereof shall be so interpreted. Any provision of this Agreement which imposes an obligation after termination or expiration of this Agreement shall survive such termination or expiration.

ARTICLE 23 DEFINITIONS

For purposes of this Agreement, the following words will have the following definitions:

- **23.1 BUSINESS SYSTEM**. "Business System" means the distinctive hotel services which are associated with Franchisor's Marks, copyrights, interior and exterior design, decor, FF&E, signs, logos, commercial symbols and color combinations, standards, specifications, programs and procedures for operations; programs and procedures for quality control; training and assistance; and advertising, direct sales, and promotional programs developed by Franchisor for the operation of a Brand Hotel under the Marks utilizing the Trade Secrets. Franchisor may operate multiple franchise systems under different Proprietary Marks in which case, for purposes of this Agreement, "Business System" only encompasses the above to the extent applicable to the franchise system being licensed hereunder to Franchisee.
- **23.2 ENTITY**. "Entity" means a corporation, general partnership, joint venture, limited partnership, limited liability partnership, limited liability company, trust, estate or other business entity.
- **23.3 FINANCIAL STATEMENTS**. "Financial Statements" means a balance sheet, profit and loss statement, statement of changes in financial condition and footnotes prepared in accordance with generally accepted accounting principles applied on a consistent basis and any other schedules or forms that may be required by Franchisor.

- **23.4 FRANCHISED LOCATION**. "Franchised Location" means the actual principal location of the Hotel.
- 23.5 GROSS ROOM REVENUES. "Gross Room Revenues" means the total revenues and receipts resulting from the rental of guest rooms, whether made for cash or credit and whether or not collected. "Gross Room Revenues" will not include charges for food and beverages, telephone, laundry, entertainment, security deposits or any sales, use or gross receipts tax imposed by any governmental authority directly upon sales, if: (A) the amount of the tax is added to the selling price and is expressly charged to the customer; (B) a specific record is made at the time of each sale of the amount of such tax; and (C) the amount thereof is paid to the appropriate taxing authority by Franchisee.
- **23.6** <u>HOTEL</u>. "Hotel" means the real property and all improvements located or to be located at the Franchised Location. The improvements include all buildings, facilities, appurtenances, landscaping, furniture, furnishings, fixtures, equipment and signs and all entry, exit and parking areas.
- **23.7** MANUAL. "Manual" means collectively, the documentation of the Business System standards, methods, procedures, techniques and specifications, whether contained in written, machine readable, electronic, or any other form, as they may be modified, amended or supplemented by Franchisor in its sole discretion. The Manual includes any operations and systems manual, newsletters and other communications from the Franchisor to Franchisee conveying any aspect of the Business System.
- **23.8** OCCUPANCY RATE. "Occupancy Rate" means the ratio of the average number of occupied Hotel rooms to the number of available Hotel rooms.
- 23.9 PRINCIPAL OWNER. "Principal Owner" means any person or entity who directly or indirectly owns a 10% or greater interest in Franchisee. If any entity other than a partnership is a Principal Owner, a "Principal Owner" also will mean an owner of a 10% or greater interest in such entity. If a partnership is a Principal Owner, a "Principal Owner" also will mean each general partner of such partnership and, if such general partner is an entity, each owner of a 10% or greater interest in such general partner. If Franchisee is one or more individuals, each individual will be deemed a Principal Owner of Franchisee. Franchisee must have at least one Principal Owner. Franchisee's Principal Owner(s) is identified in Section 1 of the Ownership & Management Addendum attached hereto as Exhibit B. As used in this Agreement, any reference to Principal Owner includes all Principal Owners.
- **23.10 RESIDENT MANAGER**. "Resident Manager" means the individual who resides at the Hotel and is responsible for the overall management of the Hotel including, but not limited to, basic operations, marketing, community relations, recordkeeping, employee staffing and training, inventory control, and maintenance of the Franchised Location.

ARTICLE 24 ACKNOWLEDGMENTS AND COUNTERPARTS

- **24.1** <u>INDEPENDENT INVESTIGATION</u>. Franchisee acknowledges that it has conducted an independent investigation of the Hotel business licensed hereunder, and recognizes that the business venture contemplated by this Agreement involves business and economic risks.
- 24.2 <u>NO INCOME OR REFUND WARRANTIES</u>. Franchisee acknowledges that Franchisor does not warrant or guarantee that: (A) Franchisee will derive income or profits from the Hotel; or (B) Franchisor will refund all or any part of the Initial Franchise Fee or the price paid for the Hotel or repurchase any of the furniture, fixtures, products, equipment, supplies or chattels supplied by Franchisor or an approved supplier if Franchisee is unsatisfied with its Hotel.
- **24.3** <u>RECOMMENDATION TO OBTAIN LEGAL COUNSEL</u>. Franchisee acknowledges that Franchisor has strongly recommended that Franchisee should retain legal counsel to review this Agreement and Franchisor's Franchise Disclosure Document and to advise Franchisee as to the terms and conditions of this Agreement and the potential economic benefits and risks of loss relating to this Agreement and the Hotel business.
- **24.4 OTHER FRANCHISEES**. Franchisee acknowledges that other franchisees of Franchisor have or will be granted franchises at different times and in different situations, and further acknowledges that the terms and conditions of such franchises may vary substantially in form and in substance from those contained in this Agreement.
- **24.5 NO WAIVERS IN VIOLATION OF STATE LAW.** No acknowledgement contained in this Article 24 is intended to act as a waiver in violation of applicable state law.
- **24.6** <u>COUNTERPARTS</u>. This Agreement may be executed in counterparts, which together shall constitute one agreement of the parties.

{Signatures appear on the following page}

IN WITNESS WHEREOF, Franchisor and Franchisee have respectively signed and sealed this Franchise Agreement as of the day and year first above written.

	"Franchisor"
	LG AS FRANCHISOR LLC
	By
	Its:
	"Franchisee"
	By
	Its:
OWNERS' AC	KNOWLEDGEMENT
Acknowledgment, not as a party, but only	ranchisee. Each following party is signing this to the extent necessary to indicate its, his or her by, the specific rights and duties of an owner of
sign here if the owner is an entity	sign here if the owner is an individual
By:	Print name:
Name:	
Its:	
Percentage Ownership:	_

sign here if the owner is an entity	sign here if the owner is an individual
By:	
Name:	Percentage Ownership:
Its:	
Percentage Ownership:	
sign here if the owner is an entity	sign here if the owner is an individual
By:	Print name:
Name:	Percentage Ownership:
Its:	<u></u>
Percentage Ownership:	

EXHIBIT A

PERSONAL GUARANTY AND AGREEMENT TO BE BOUND PERSONALLY BY THE TERMS AND CONDITIONS OF THE FRANCHISE AGREEMENT

In consideration of the execution of this Franchise Agreement by the Franchisor, and for other good and valuable consideration, the undersigned, for themselves, their heirs, successors, and assigns, do jointly, individually and severally hereby become surety and guaranty for the payment of all amounts and the performance of the covenants, terms and conditions in this Franchise Agreement, to be paid, kept and performed by the Franchisee.

Further, the undersigned, individually and jointly, hereby agree to be personally bound by each and every condition and term contained in this Franchise Agreement and agree that this Personal Guaranty should be construed as though the undersigned and each of them executed a Franchise Agreement containing the identical terms and conditions of this Franchise Agreement, including without limitation the dispute resolution, confidentiality and non-solicitation provisions contained herein.

If any default should at any time be made therein by the Franchisee, then the undersigned, their heirs, successors and assigns, do hereby, individually, jointly and severally, promise and agree to pay to Franchisor all monies due and payable to the Franchisor under the terms and conditions of this Franchise Agreement.

In addition, if the Franchisee fails to comply with any other terms and conditions of this Franchise Agreement, then the undersigned, their heirs, successors and assigns, do hereby, individually, jointly and severally, promise and agree to comply with the terms and conditions of this Franchise Agreement for and on behalf of the Franchisee.

In addition, should the Franchisee at any time be in default on any obligation to pay monies to the Franchisor or any subsidiary or related company of the Franchisor, whether for products, supplies, furniture, fixtures, equipment or other goods purchased by the Franchisee from the Franchisor, or any subsidiary or affiliate of the Franchisor, or for any other indebtedness of the Franchisee to the Franchisor, or any subsidiary or affiliate of the Franchisor, then the undersigned, their heirs, successors and assigns, do hereby individually, jointly and severally promise and agree to pay all such monies due and payable from the Franchisee to the Franchisor, or any subsidiary or affiliate of the Franchisor.

Each of the undersigned waives: (i) notice of demand for payment of any indebtedness or nonperformance of any obligations hereby guaranteed; (ii) protest and notice of default to any party respecting the indebtedness or nonperformance of any obligations hereby guaranteed; and any right the undersigned has to require that an action be brought against Franchisee or any other person as a condition of liability.

It is further understood and agreed by the undersigned that the provisions, covenants and conditions of this Personal Guaranty will inure to the benefit of the successors and assigns of the Franchisor.

In the Presence of:	PERSONAL GUARANTORS		
	Individually Address		
	Individually		ually
	Address		
	City	State	Zip Code
To be signed by spouses of guarantors rest Nevada, New Mexico, Texas, Washington,	_	Arizona, Califo	rnia, Idaho, Louisiana,
CONSENT OF S	SPOUSE OF GU	JARANTOR	
The undersigned spouse of	agrees to be bo spouse and furth not be construct	und thereby to er agrees that the d as an agreement	the extent of his/her heir community assets ent by the undersigned
	By:		
	Name:		

EXHIBIT B

LG AS FRANCHISOR LLC ADDENDUM TO FRANCHISE AGREEMENT OWNERSHIP & MANAGEMENT ADDENDUM

Name	Percentage of Interest	Home Address	
information contain	shall immediately notify Franchi ed in Section 1 of this Addendum am containing the correct information preement.	and, at Franchisor's request, pre-	epare and
information containsign a new Addendarticle 19 of the Ag	ed in Section 1 of this Addendum um containing the correct information	and, at Franchisor's request, proon. Any change is subject to the	epare and terms of
information containsign a new Addend Article 19 of the Ag	ed in Section 1 of this Addendum containing the correct information reement. El Management. The following individual(s) or m Franchisee and approved by I	and, at Franchisor's request, proon. Any change is subject to the	epare and terms of

	Agreement.		
<u>Name</u>	Address	Phone	<u>Email</u>
information contained	in Section 2 of this Ad containing the correct i	ldendum and, at Franchi	of any change in the sor's request, prepare and e is subject to the terms of
3. <u>Effe</u> , 20		ndum is effective as of th	is day of
"Franchisee"		"Franchisor"	
		LG AS FRANCHISO	OR LLC
		Ву:	
		Its:	

Franchisee represents and warrants to Franchisor that the following

individual(s) is the Resident Manager of Franchisee under the Franchise

B.

EXHIBIT C

ASSIGNMENT OF TELEPHONE NUMBERS AND SERVICE

THIS ASSIGNMENT is made and entered into this _ day of	, 20, by
and between ("Franchisee")	and LG AS
Franchisor LLC ("Franchisor").	
BACKGROUND	
A. Franchisor and Franchisee are parties to an LG AS Franchisor L. Agreement (the "Franchise Agreement") dated	_, whereby
B. Section 15.3 of the Franchise Agreement requires Franchisee to ass to each of the telephone numbers for Franchisee's Hotel to Franchisor to be e termination or expiration of the Franchise Agreement.	
C. Franchisee's Hotel has been assigned the following phone number(s Numbers") by , the local telephone service provider ("Telephone Company	
() () ())
() () ())

AGREEMENTS

Franchisor and Franchisee agree as follows:

Franchisee hereby assigns to Franchisor all right, title and interest to the Telephone Numbers to Franchisor, such assignment to be effective upon termination or expiration of the Franchise Agreement.

Franchisee agrees to pay all amounts due and payable to the Telephone Company for services rendered, including amounts accrued and unbilled, as of the date of this Assignment and all amounts that the Telephone Company may require in connection with such transfer or otherwise. Franchisee agrees to indemnify and hold harmless the Telephone Company and Franchisor from all claims, actions and damages in connection with this and any other assignment and any other matters or disputes related in any way hereto.

The Telephone Company is authorized and requested to honor this Assignment upon notice by Franchisor that such Franchise Agreement is terminated or has expired.

Upon transfer of the Telephone Numbers to Franchisor, Franchisor may, in its sole judgment, continue service with the Telephone Company or cancel such service. Franchisor will not be obligated or liable for any arrears or charges for services rendered by the Telephone Company to Franchisee prior to such transfer.

IN WITNESS WHEREOF, each of the undersigned hereby acknowledges having read this Assignment, understands and consents to be bound by all of its terms.

"Franchisee"	"Franchisor"
	LG AS FRANCHISOR LLC
	By:
	Its:

EXHIBIT D

ACKNOWLEDGMENT ADDENDUM

(<u>NOT FOR USE</u> IN THE FOLLOWING STATES: CA, HI, IL, IN, MD, MI, MN, NY, ND, RI, SD, VA, WA, WI)

As you know, you and we are entering into a Franchise Agreement for the operation of a Brand Hotel franchise. The purpose of this Acknowledgment Addendum is to determine whether any statements or promises were made to you that we have not authorized or that may be untrue, inaccurate or misleading, and to be certain that you understand the limitations on claims that may be made by you by reason of the offer and sale of the franchise and operation of your business. Please review each of the following questions carefully and provide honest responses to each question.

Acknowledgments and Representations*.

1.	Did you receive a copy of our disclosure document (and all exhibits and attachments) at least 14 calendar days prior to signing the Franchise Agreement? Check one: () Yes () No. If no, please comment:
2.	Have you studied and reviewed carefully our disclosure document and Franchise Agreement? Check one: () Yes () No. If no, please comment:
3.	Except for fill in the blank provisions and changes made as a result of negotiations that you initiated, did you receive a copy of the Franchise Agreement at least 7 calendar days prior to the date on which the Franchise Agreement was executed? Check one: () Yes () No. If no, please comment:
4.	Did you understand all the information contained in both the disclosure document and Franchise Agreement? Check one: () Yes () No. If no, please comment:
5.	Was any oral, written or visual claim or representation made to you which contradicted the disclosures in the disclosure document? Check one: () Yes () No. If yes, please state in detail the oral, written or visual claim or representation:
6.	Did any employee or other person speaking on behalf of LG AS Franchisor LLC make any oral, written or visual claim, statement, promise or representation to you that stated, suggested, predicted or projected sales, revenues, expenses, earnings, income or profit levels at any Brand Hotel location or business, or the likelihood of success at your

	franchised business? Check one: () Yes () No. If yes, please state in detail the oral, written or visual claim or representation:
7.	Did any employee or other person speaking on behalf of LG AS Franchisor LLC make any statement or promise regarding the costs involved in operating a franchise that is not contained in the disclosure document or that is contrary to, or different from, the information contained in the disclosure document. Check one: () Yes () No. If yes, please comment:
8.	Do you understand that that the franchise granted is for the right to operate a business at the franchised location only and, except as expressly provided in Article 1 of the Franchise Agreement (or, if applicable, in the Area Development Agreement) includes no territorial protection, and that we and our affiliates have the right to issue franchises or operate competing businesses for or at locations, as we determine, near your protected area? Check one: () Yes () No. If no, please comment:
9.	Do you understand that the Franchise Agreement contains the entire agreement between you and us concerning the franchise for the franchised business, meaning that any prior oral or written statements not set out in the Franchise Agreement will not be binding? Check one: () Yes () No. If no, please comment:
10.	Do you understand that the success or failure of your Hotel will depend in large part upon your skills and experience, your business acumen, the hours you work, your location, the local market for hospitality services under the STAYAPT TM trademarks, interest rates, the economy, inflation, the number of employees you hire and their compensation, competition, lease terms and other economic and business factors? Further, do you understand that the economic and business factors that exist at the time you open your Hotel may change? Check one: () Yes () No. If no, please comment:

YOU UNDERSTAND THAT YOUR ANSWERS ARE IMPORTANT TO US AND THAT WE WILL RELY ON THEM. BY SIGNING THIS ADDENDUM, YOU ARE REPRESENTING THAT YOU HAVE CONSIDERED EACH QUESTION CAREFULLY AND RESPONDED TRUTHFULLY TO THE ABOVE QUESTIONS. IF MORE SPACE IS NEEDED FOR ANY ANSWER, CONTINUE ON A SEPARATE SHEET AND ATTACH.

<u>NOTE</u>: IF THE RECIPIENT IS A CORPORATION, PARTNERSHIP, LIMITED LIABILITY COMPANY OR OTHER ENTITY, EACH OF ITS PRINCIPAL OWNERS MUST EXECUTE THIS ACKNOWLEDGMENT.

Signed:	Signed:
Print Name:	Print Name:
Date:	Date:
Signed:	
Print Name:	
Date:	

(<u>NOT FOR USE</u> IN THE FOLLOWING STATES: CA, HI, IL, IN, MD, MI, MN, NY, ND, RI, SD, VA, WA, WI)

EXHIBIT E

STATE ADDENDUM

[Insert State Addendum to Franchise Agreement here, if applicable]

EXHIBIT E

Franchise Standards Manual Table of Contents

stayAPT Suites Brand Standards Manual

Gene	eral Information		6		
	Welcome to stayAPT				
	ces!				
	6				
	Trademark Disclosure	. 7			
	Introduction	7			
	Confidentiality	8			
	Revisions to the Manual		. 8		
	Franchise Feedback	8			
Fran	chise Compliance	9			
	System Compliance and Waivers		9		
	Compliance with Legal Requirements for	Guests	with		
Disa	abilities 10				
	Independent Contractor and Indemnifica	tion			11
	Compliance with Laws				
	Property Visitation				
	1 1				
Pre-	Opening Timeline and Training		13		
	Franchisor Standard Prototype				
	Timeline				
	Licenses and Permits		1.3		
	Hiring of Staff				
	Pre-Opening Training				
	Soft Opening				
	2010 Sp 0112119				
Kev	stayAPT Suites Operations Standards			14	
01	Staffing Model		•		
	Service and Maintenance Vendors		17		
	Days and Hours of Operations				
	Guest Room Housekeeping Schedule				
	Commercial Laundry		••••••	10	
	Furniture, Fixtures, Equipment, Lighti		Sunnl	ies	
		119, 4110	Duppi	105	
•••••	Facilities for Persons with Disabiliti	A S			1 9
	Signage		••••••	• • • • • • • • • • • • • • • • • • • •	• 10
	Digitage	20			
Dror	perty Exterior	20			
101	Hotel Building				
	Exterior Signage		20		
	Flags and Banners				
	Entry Drive				
	Parking Lot				
	Landscape				
	Walkwavs	22			

Main Hotel Office Entrance, Main Hotel Entrance, and Additional Entrances 23

Public Areas	24
Hotel Office	24
Stairways and Stairwells	26
Exterior and Interior Corridors ,	/ Hallways
26	-
Interior Courtyard	
Designated Smoking Areas	
Public Area Restroom	
Gaming Machines and Tickets	30
Guest Laundry	
Equipment	
Costs	
Maintenance	
Other Supplies / Set-up	
ADA Legal Requirements Compliance	
Jane 100 Jan	
Guest Fitness Center	31
Equipment	
Maintenance	
Other Supplies / Set-up	
ADA Legal Requirements Compliance	
Tibil begai nequilements compilation	•
Front Desk	32
Appearance and Organization	32
Front Desk Signage	
Jonas Chorum Cloud PMS Software	
Reservations	33
Payment	
Check-in	
After Hours Check In	40
Dishonored Reservations	
Room and Tax Charges and Posting	
Rate Setting and Rate Gouging	
Children Rate	
Check-out	42
Fees	43
Room Payment Refunds	
Petty Cash	
Telephone Call Handling	
Guest Towel and Linen Exchange, I	
Guest Mail and Packages	45
Lost and Found	
Privacy and Personal Information	
Requirements	
<u>. </u>	

ADA Legal Requirements		48	
Guest Rooms	47		
Typical Guest Room Sizes		4.8	
ADA Legal Requirements			
Bed Types		40	
Non-Smoking Rooms		- 7man	:+:00
Standard Room Furnishings, Fixtures,	Fdarbuei	it, Amen.	ittes,
Supplies 49	4.0		
Flooring			
Bedding		F.0	
Guest Room Telephones			
Guest Room Case Goods and Amenities	_		50
Trashcans	52		
Housekeeping	. 53		
Hotel and Guest Room Upkeep		53	
Housekeeping Closets			
Processing of Outsourced Laundry			55
Housekeeping Staff Room Clean Assigr			
Guest Room Cleaning Requirements			
Guest Room Cleaning Procedures			
Property Maintenance		7	
Day to Day			
Guest Room, Public Areas, Property B		and Inte	rior
Maintenance and Repair Issues			
Preventative Maintenance		79	
Pest Control			
Maintenance Contractors / Techniciar			80
Landscape Maintenance	{	30	
Snow and Ice Preparation and Removal		•	81
Uniforms and Appearance	81	L	
Safety and Security	\$	21	
Franchisee Responsibility			
Staff Safety and Security Training			82
Workplace Safety			02
Bank Deposits		,	
		=	
Guest Room Security)	
Key Control		07	
Hot Water Supply Temperature		8 /	
Fire Safety			0.0
Hazardous and Flammable Materials St	corage		89
Hotel Technology			
Room Data Integrity		90	

Guest Data Security	90	
Sales and Marketing	92 93	
Human Resources	94	
Finance and Accounting Financial Reporting to Franchisor Franchisee Payments to Franchisor Insurance Requirements Insurance Certificate Example Hotel Accounting		96 96
Customer Service Customer Service - General Guest Requests Handling Guest Complaints Medical Assistance	101	
Franchisee Training and Meetings	Training	• • • • • • • • • • • • • • • • • • • •
105 Franchise Meetings		9
Appendices		
Appendix A - stayAPT Suites Interior Signal Appendix B - Design & Construction Manual Appendix C - Material Specification Manual Appendix D - Prototype Manual Appendix E - Millwork Appendix F - Technology Appendix G - Emergency Disaster Preparedness	-	

EXHIBIT F

State Addenda

State Addenda to Franchise Disclosure Document

MULTI-STATE ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT (FOR THE FOLLOWING STATES: CA, HI, IL, IN, MD, MI, MN, NY, ND, RI, SD, VA, WA, WI)

This Addendum pertains to franchises sold in the state that have adopted as law the NASAA Statement of Policy Regarding the Use of Franchise Questionnaires and Acknowledgements (the "SOP") and is for the purpose of complying with the statutes and regulations of such states. For franchises sold in such states, this franchise disclosure document is amended by adding the following section at the end of Item 9:

"No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise."

ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT PURSUANT TO THE CALIFORNIA FRANCHISE INVESTMENT LAW

The registration of this franchise offering by the California Department of Financial Protection and Innovation does not constitute approval, recommendation, or endorsement by the commissioner.

THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE FRANCHISE DISCLOSURE DOCUMENT AT LEAST 14 DAYS PRIOR TO EXECUTION OF AGREEMENT.

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination, transfer or nonrenewal of a franchise. If the franchise agreement contains a provision that is inconsistent with the law, the law will control.

The Franchise Agreement and Multi-Unit Development Agreement each provide for termination upon bankruptcy. These provisions may not be enforceable under Federal bankruptcy law (11 U.S.C.A. Sec. 101 et. seq.).

Section 31125 of the Franchise Investment Law requires us to give to you a disclosure document approved by the Commissioner of Business Oversight before we ask you to consider a material modification of your Franchise Agreement.

You must sign a general release of claims if you renew or transfer your franchise. California Corporations Code Section 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code Sections 31000 through 31516). Business and Professions Code Section 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code Sections 20000 through 20043).

Neither the Franchisor nor any person listed in Item 2 of this Franchise Disclosure Document is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. 78a *et seq.*, suspending or expelling such persons from membership in such association or exchange.

Section 17.9 of the Franchise Agreement and Section 9.9 of the Multi-Unit Development Agreement each limit the statute of limitations to 1 year from the date the complaining party knew or should have known of facts giving rise to the claim. This provision is void to the extent it is inconsistent with the provisions of Corporations Code 31303-31304. Corporations Code Section 31512 provides that "Any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with any provision of this law or any rule or order is void."

The Franchise Agreement and Multi-Unit Development Agreement each require application of the laws of North Carolina. This provision may not be enforceable under California law.

The Franchise Agreement and Multi-Unit Development Agreement each require binding arbitration. The arbitration will occur at North Carolina, or in such city and state where the Franchisor is headquartered, with the costs being borne by both parties unless Franchisor incurs expense to enforce any obligation of Franchisee, then Franchisor shall be entitled to recover all legal fees and expenses from Franchisee.

Prospective franchisees are encourage to consult private legal counsel to determine the applicability of California and federal laws (such as Business Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a franchise agreement restricting venue to a forum outside of the State of California.

OUR WEBSITE IS WWW.STAYAPT.COM. OUR WEBSITE HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION. ANY COMPLAINTS CONCERNING THE CONTENT OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION AT WWW.DFPI.CA.GOV.

The Franchise Agreement and Multi-Unit Development Agreement each provide for mediation. The mediation will occur in the city and state where we have our principal place of business with the costs of the mediation service being borne equally by both parties.

The Franchise Agreement contains a liquidated damages clause. Under California Civil Code Section 1671, certain liquidated damages clauses are unenforceable.

Item 5 is amended by adding the following: "The Department of Financial Protection and Innovation requires that the franchisor defer the collection of all initial fees from California franchisees until the franchisor has completed all its pre-opening obligations and franchisee is open for business."

Any provision of a franchise agreement, franchise disclosure document, acknowledgment, questionnaire, or other writing, including any exhibit thereto, disclaiming or denying any of the following shall be deemed contrary to public policy and shall be void and unenforceable.

- (a) Representations made by the franchisor or its personnel or agents to a prospective franchisee.
- (b) Reliance by a franchisee on any representations made by the franchisor or its personnel or agents.
- (c) Reliance by a franchisee on the franchise disclosure document, including any exhibit thereto.
- (d) Violations of any provisions of this division.

ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT PURSUANT TO THE HAWAII FRANCHISE INVESTMENT LAW

THESE FRANCHISES HAVE BEEN FILED UNDER THE FRANCHISE INVESTMENT LAW OF THE STATE OF HAWAII. FILING DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE DIRECTOR OF THE DEPARTMENT OF COMMERCE AND CONSUMER AFFAIRS OR A FINDING BY THE DIRECTOR OF THE DEPARTMENT OF COMMERCE AND CONSUMER AFFAIRS THAT THE INFORMATION PROVIDED HEREIN IS TRUE, COMPLETE AND NOT MISLEADING.

THE FRANCHISE INVESTMENT LAW MAKES IT UNLAWFUL TO OFFER OR SELL ANY FRANCHISE IN THIS STATE WITHOUT FIRST PROVIDING TO THE PROSPECTIVE FRANCHISEE, OR SUBFRANCHISOR, AT LEAST SEVEN (7) DAYS PRIOR TO THE EXECUTION BY THE PROSPECTIVE FRANCHISEE, OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST SEVEN (7) DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION BY THE FRANCHISEE OR SUBFRANCHISOR, WHICHEVER OCCURS FIRST, A COPY OF THE FRANCHISE DISCLOSURE DOCUMENT, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE.

THIS FRANCHISE DISCLOSURE DOCUMENT CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT. THE CONTRACT OR AGREEMENT SHOULD BE REFERRED TO FOR A STATEMENT OF ALL RIGHTS, CONDITIONS, RESTRICTIONS AND OBLIGATIONS OF BOTH THE FRANCHISOR AND THE FRANCHISEE.

No release language included in the Franchise Agreement shall relieve the franchisor or any other person, directly or indirectly, from liability imposed by the laws concerning franchising in the State of Hawaii.

ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT PURSUANT TO THE ILLINOIS FRANCHISE DISCLOSURE ACT

Notwithstanding anything to the contrary set forth in the Franchise Disclosure Document, the following provisions shall supersede and apply to all franchises offered and sold in the State of Illinois:

1. Item 17 of the Franchise Disclosure Document is amended by the addition of the following language at the beginning thereof:

"Notice Required By Law

THE TERMS AND CONDITIONS UNDER WHICH YOUR FRANCHISE CAN BE TERMINATED AND YOUR RIGHTS UPON NON-RENEWAL MAY BE AFFECTED BY ILLINOIS LAW, 815 ILCS 705/19 - 705/20."

- 2. Illinois law shall supersede any provisions of the Franchise Agreement or North Carolina law which are in conflict with Illinois law.
- 3. The provisions of Section 27 of the Act supersede the provisions of Section 17.9 of the Franchise Agreement and Section 9.9 of the Multi-Unit Development Agreement that set a limitation period of 1 year to the extent that claims are brought under Section 26 of the Act.
- 4. Nothing in Section 22.1 of the Franchise Agreement or Section 10.1 of the Multi-Unit Development Agreement waives any rights you may have under Section 41 of the Illinois Franchise Disclosure Act of 1987.
- 5. The provisions of Section 4 of the Act supersede Sections 17.2 and 17.3 of the Franchise Agreement and Sections 9.1 and 9.2 of the Multi-Unit Development Agreement which provide for venue in a forum outside of Illinois.

ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT PURSUANT TO THE MARYLAND FRANCHISE REGISTRATION AND DISCLOSURE LAW

- 1. Item 5. Based upon the franchisor's financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by franchisees shall be deferred until the franchisor completes its preopening obligations under the franchise agreement. In addition, all development fees and initial payments by area developers shall be deferred until the first franchise under the development agreement opens.
- 2. Item 17.c ("Requirements for franchisee to renew or extend") and Item 17.m ("Conditions for franchisor approval of transfer") of the Franchise Disclosure Document are amended to provide that "The requirement that you provide a general release of all claims against us in order to renew or transfer your franchise shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law."
- 3. Item 17.g ("Cause defined curable defaults") of the Franchise Disclosure Document is amended to provide that "Termination upon filing of a bankruptcy petition against you or any shareholder may not be enforceable under federal bankruptcy law."
- 4. Item 17.v ("Choice of forum"), of the Franchise Disclosure Document is amended to provide that "Franchisee may sue in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law."
- 5. Any claim arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT PURSUANT TO THE MINNESOTA FRANCHISE INVESTMENT LAW

Notwithstanding anything to the contrary set forth in the franchise disclosure document and/or Franchise Agreement, as applicable, the following provisions shall supersede and apply to all franchises offered and sold in the State of Minnesota:

- 1. MINN. STAT. SECTION 80C.21 and MINNESOTA RULES 2860.4400(J) prohibit the franchisor from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial, or requiring the franchisee to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the Franchise Disclosure Document or agreement(s) can abrogate or reduce (1) any of the franchisee's rights as provided for in MINN. STAT. CHAPTER 80C or (2) franchisee's rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction.
- 2. With respect to franchises governed by Minnesota law, the franchisor will comply with MINN. STAT. SECTION 80C.14 SUBD. 3-5, which require (except in certain specified cases)
 - (i) that a franchisee be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the franchise agreement and
 - (ii) that consent to the transfer of the franchise will not be unreasonably withheld.
- 3. Minnesota considers it unfair to not protect the franchisee's right to use the trademarks. Refer to MINN. STAT. SECTION 80C.12 SUBD. 1(G). The franchisor will protect the franchisee's rights to use the trademarks, service marks, trade names, logotypes, or other commercial symbols or indemnify the franchisee from any loss, costs, or expenses arising out of any claim, suit, or demand regarding the use of the name.
- 4. <u>MINNESOTA RULES 2860.4400(D)</u> prohibits a franchisor from requiring a franchisee to assent to a general release.
- 5. The franchisee cannot consent to the franchisor obtaining injunctive relief. The franchisor may <u>seek</u> injunctive relief. See <u>MINNESOTA RULES 2860.4400(J)</u> also, a court will determine if a bond is required.
- 6. The Limitations of Claims section must comply with MINN. <u>STAT. SECTION</u> 80C.17 SUBD. 5.

ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT PURSUANT TO THE NEW YORK FRANCHISE LAW

1. The cover page of the Franchise Disclosure Document is amended to add the following statement:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT A OR YOUR PUBLIC LIBRARY FOR SERVICES OR INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THIS FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE APPROPRIATE STATE OR PROVINCIAL AUTHORITY. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS WHICH ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. Item 3 of the Franchise Disclosure Document is amended by deleting the last paragraph and substituting the following:

"Neither we, our predecessor, nor a person identified in Item 2, or an affiliate offering franchises under our principal trademark:

- A. Has an administrative, criminal or civil action pending against that person alleging: a felony; a violation of a franchise, antitrust or securities law; fraud, embezzlement, fraudulent conversion; misappropriation of property; unfair or deceptive practices or comparable civil or misdemeanor allegations. Moreover, there are no pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature or financial condition of the franchise system or its business operations.
- B. Has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the ten year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antitrust or securities law; fraud, embezzlement, fraudulent conversion or misappropriation of property, or unfair or deceptive practices or comparable allegations.
- C. Is subject to a currently effective injunction or restrictive order or decree relating to the franchise, or under a federal, State or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the

Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunction or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent."

3. Item 4 of the Franchise Disclosure Document is amended by deleting the last paragraph and substituting the following:

"Neither we, our affiliate, our predecessor nor our officers during the 10 year period immediately before the date of the Franchise disclosure document: (a) filed as debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code; (b) obtained a discharge of its debts under the bankruptcy code; or (c) was a principal officer of a company or a general partner in a partnership that either filed as a debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code or that obtained a discharge of its debts under the U.S. Bankruptcy Code during or within one year after the officer or general partner of the franchisor held this position in the company or partnership."

4. Item 5 of the Franchise Disclosure Document is amended by adding the following to the subsection entitled "Franchise Fee For Single Hotel":

"We will use the Initial Franchise Fee to cover our costs associated with fulfilling its obligations under the Franchise Agreement and to cover other overhead costs and expenses."

5. Item 17 of the Franchise Disclosure Document is amended by adding the following statement to the summary column (c), titled "Requirements for franchisee to renew or extend," and Item 17(m), entitled "Conditions for franchisor approval of transfer":

"However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687(4) and 687(5) be satisfied."

6. Item 17 of the Franchise Disclosure Document is further amended by adding the following statement to the summary column (d) entitled "Termination by You":

"You can terminate upon any grounds available by law."

7. Item 17 of the Franchise Disclosure Document is further amended by adding the following statement to the summary column (w) indicating the choice of law:

"The foregoing choice of law should not be considered a waiver of any right conferred upon either the Franchisor or upon the Franchisee by the General Business Law of the State of New York."

FRANCHISOR REPRESENTATION

THE FRANCHISOR REPRESENTS THAT THIS PROSPECTUS DOES NOT KNOWINGLY OMIT ANY MATERIAL FACT OR CONTAIN ANY UNTRUE STATEMENT OF MATERIAL FACT.

ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT PURSUANT TO THE NORTH DAKOTA FRANCHISE LAW

Notwithstanding anything to the contrary in the Franchise Disclosure Document, the following provisions shall supersede and apply to all franchises offered and sold in the State of North Dakota:

- 1. The Commissioner has determined that the requirement for a franchisee to consent to a waiver of trial by jury to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. Accordingly, the jury waiver in Section 17.1 of the Franchise Agreement is deleted.
- 2. Waiver of exemplary and punitive damages is prohibited by law in the State of North Dakota. Accordingly, the waiver of exemplary and punitive damages in Sections 17.3 and 17.6 of the Franchise Agreement and Section 9.6 of the Multi-Unit Development Agreement is deleted.
- 3. North Dakota prohibits a provision that the franchisee shall pay all costs and expenses incurred by Franchisor in enforcing the Franchise Agreement. Accordingly, Section 17.10 of the Franchise Agreement and Section 9.10 of the Multi-Unit Development Agreement are each restated to read as follows:

"In the event either party incurs legal fees or costs or other expenses to enforce any obligation of the other party hereunder, or to defend against any claim, demand, action or proceeding by reason of the other party's failure to perform or observe any obligation imposed upon that party by this Agreement, then the prevailing party shall be entitled to recover from the other party the amount of all legal fees, costs and expenses, including reasonable attorneys' fees, whether incurred prior to, or in preparation for or contemplation of the filing of any claim, demand, action or proceeding to enforce any obligation of the other party hereunder or thereafter or otherwise."

- 4. The laws of the State of North Dakota supersede any provisions of the Franchise Agreement, and the other agreements or North Carolina law if those provisions are in conflict with North Dakota law.
- 5. Except to the extent the laws of the State of North Dakota are preempted by the Federal Arbitration Act, any provision in the Franchise Agreement which designates jurisdiction or venue, or requires the Franchisee to agree to jurisdiction or venue, in a forum outside of North Dakota, is deleted from any Franchise Agreement and Multi-Unit Development Agreement issued in the State of North Dakota.
- 6. The Commissioner has determined that requiring a franchisee to sign a general release upon renewal of the franchise agreement is unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. Accordingly, all references to the Franchisee signing a general release upon renewal of the Franchise Agreement

in Item 17(c) of the Franchise disclosure document, and Section 8.4 of the Multi-Unit Development Agreement are deleted.

- 7. The Commissioner has determined that the requirement for a franchisee to consent to termination or liquidated damages to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. Accordingly, all references to the Franchisee paying termination or liquidated damages in Item 17(i) of the Franchise Disclosure Document and in Sections 15 and 16 of the Franchise Agreement are deleted.
- 8. The Commissioner has held that covenants restricting competition are contrary to Section 9-08-06 of the North Dakota Century Code, and are unfair, unjust, or inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. Covenants not to compete such as those mentioned above are generally considered unenforceable in the State of North Dakota. To the extent any covenants restricting competition in the Franchise Agreement are contrary to Section 9-08-06 of the North Dakota Century Code or Section 51-19-09 of the North Dakota Franchise Investment Law they are deleted and Item 17(r) of the Franchise Disclosure Document is amended to state so.
- 9. "No statement, questionnaire, or acknowledgment signed or agreed to by a franchise in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise"

ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT PURSUANT TO THE RHODE ISLAND FRANCHISE DISCLOSURE ACT

Section 19-28.1-14 of the Rhode Island Franchise Investment Act, as amended by the laws of 1991, provides that "A provision in a Franchise Agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act."

ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT PURSUANT TO THE VIRGINIA RETAIL FRANCHISING ACT

Notwithstanding anything to the contrary in the Franchise Disclosure Document, the following provisions shall supersede and apply to all franchises offered and sold in the State of Virginia.

The following paragraph is added at the end of Item 17:

Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any grounds for default or termination stated in the franchise agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

"The design mark on the cover page, mark no. 88658375, is not currently registered with the United States Patent and Trademark Office ("USPTO"). Therefore, the trademark does not have as many legal benefits and rights as a federally registered trademark, which may increase your expense. If stayAPT's right to use the trademarks is challenged, you may have to change to an alternative trademark, which may increase your expenses."

Estimated Initial Investment. The franchisee will be required to make an estimated initial investment ranging from \$7,895,300 to \$13,410,300. This amount exceeds the franchisor's stockholders' equity as of December 31, 2022, which is \$362,676.

WASHINGTON ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT, THE FRANCHISE AGREEMENT, AND ALL RELATED AGREEMENTS

The provisions of this Addendum form an integral part of, are incorporated into, and modify the Franchise Disclosure Document, the franchise agreement, and all related agreements regardless of anything to the contrary contained therein. This Addendum applies if: (a) the offer to sell a franchise is accepted in Washington; (b) the purchaser of the franchise is a resident of Washington; and/or (c) the franchised business that is the subject of the sale is to be located or operated, wholly or partly, in Washington.

- 1. Conflict of Laws. In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, chapter 19.100 RCW will prevail.
- 2. Franchisee Bill of Rights. RCW 19.100.180 may supersede provisions in the franchise agreement or related agreements concerning your relationship with the franchisor, including in the areas of termination and renewal of your franchise. There may also be court decisions that supersede the franchise agreement or related agreements concerning your relationship with the franchisor. Franchise agreement provisions, including those summarized in Item 17 of the Franchise Disclosure Document, are subject to state law.
- 3. Site of Arbitration, Mediation, and/or Litigation. In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.
- 4. General Release. A release or waiver of rights in the franchise agreement or related agreements purporting to bind the franchisee to waive compliance with any provision under the Washington Franchise Investment Protection Act or any rules or orders thereunder is void except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2). In addition, any such release or waiver executed in connection with a renewal or transfer of a franchise is likewise void except as provided for in RCW 19.100.220(2).
- 5. Statute of Limitations and Waiver of Jury Trial. Provisions contained in the franchise agreement or related agreements that unreasonably restrict or limit the statute of limitations period for claims under the Washington Franchise Investment Protection Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.
- **6. Transfer Fees.** Transfer fees are collectable only to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.
- 7. Termination by Franchisee. The franchisee may terminate the franchise agreement under any grounds permitted under state law.

- **8.** Certain Buy-Back Provisions. Provisions in franchise agreements or related agreements that permit the franchisor to repurchase the franchisee's business for any reason during the term of the franchise agreement without the franchisee's consent are unlawful pursuant to RCW 19.100.180(2)(j), unless the franchise is terminated for good cause.
- 9. Fair and Reasonable Pricing. Any provision in the franchise agreement or related agreements that requires the franchisee to purchase or rent any product or service for more than a fair and reasonable price is unlawful under RCW 19.100.180(2)(d).
- 10. Waiver of Exemplary & Punitive Damages. RCW 19.100.190 permits franchisees to seek treble damages under certain circumstances. Accordingly, provisions contained in the franchise agreement or elsewhere requiring franchisees to waive exemplary, punitive, or similar damages are void, except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2).
- 11. Franchisor's Business Judgement. Provisions in the franchise agreement or related agreements stating that the franchisor may exercise its discretion on the basis of its reasonable business judgment may be limited or superseded by RCW 19.100.180(1), which requires the parties to deal with each other in good faith.
- 12. Indemnification. Any provision in the franchise agreement or related agreements requiring the franchisee to indemnify, reimburse, defend, or hold harmless the franchisor or other parties is hereby modified such that the franchisee has no obligation to indemnify, reimburse, defend, or hold harmless the franchisor or any other indemnified party for losses or liabilities to the extent that they are caused by the indemnified party's negligence, willful misconduct, strict liability, or fraud.
- 13. Attorneys' Fees. If the franchise agreement or related agreements require a franchisee to reimburse the franchisor for court costs or expenses, including attorneys' fees, such provision applies only if the franchisor is the prevailing party in any judicial or arbitration proceeding.
- 14. Noncompetition Covenants. Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provision contained in the franchise agreement or elsewhere that conflicts with these limitations is void and unenforceable in Washington.
- 15. Nonsolicitation Agreements. RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any

such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

- 16. Questionnaires and Acknowledgments. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 17. Prohibitions on Communicating with Regulators. Any provision in the franchise agreement or related agreements that prohibits the franchisee from communicating with or complaining to regulators is inconsistent with the express instructions in the Franchise Disclosure Document and is unlawful under RCW 19.100.180(2)(h).
- 18. Advisory Regarding Franchise Brokers. Under the Washington Franchise Investment Protection Act, a "franchise broker" is defined as a person that engages in the business of the offer or sale of franchises. A franchise broker represents the franchisor and is paid a fee for referring prospects to the franchisor and/or selling the franchise. If a franchisee is working with a franchise broker, franchisees are advised to carefully evaluate any information provided by the franchise broker about a franchise.
- 19. Amendment to Item 5. cItem 5 is amended to add the following: All initial fees and payments owed by franchisees described in this Item 5 are deferred until the franchisor completes its preopening obligations under the Franchise Agreement and the franchisee is open for business.

State Addenda to Franchise Agreement

MULTI-STATE AMENDMENT TO FRANCHISE AGREEMENT (FOR THE FOLLOWING STATES: CA, HI, IL, IN, MD, MI, MN, NY, ND, RI, SD, VA, WA, WI)

This Amendment pertains to franchises sold in the states that have adopted as law the NASAA Statement of Policy Regarding the Use of Franchise Questionnaires and Acknowledgements (the "SOP") and is for the purpose of complying with the statutes and regulations of such states. Signing this Amendment where the SOP, because applicable jurisdictional requirements are not met, does not apply, does not subject the parties to the provisions of the SOP. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

, 20, will be amende	ed as follows:
1. The following languag Agreement:	ge is added immediately before the signature block of the Franchise
to by a franch franchise related claims under a the inducement by any franchise of the franchised document exection.	questionnaire, or acknowledgment signed or agreed isee in connection with the commencement of the ionship shall have the effect of (i) waiving any ny applicable state franchise law, including fraud in t, or (ii) disclaiming reliance on any statement made for, franchise seller, or other person acting on behalf or. This provision supersedes any other term of any uted in connection with the franchise."
this Amendment as of the date	e set forth above.
Franchisor:	Franchisee:
LG AS Franchisor LLC	
a Delaware limited liability of	ompany
By:	By:
Name:	Name:
Title:	Title:

CALIFORNIA AMENDMENT FRANCHISE AGREEMENT

This Amendment pertains to franchises sold in the State of California that are subject to the California Franchise Investment Law (the "Act") and is for the purpose of complying with California statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

•	ning which may be contained in the body of the nent is amended to include the following:
The parties to the Franchise Agreement of that the Franchise Agreement will be amended a	lated,, hereby agree s follows:
1. Section 4.1 of the Franchise Agree	ement is amended by adding the following:
<u> </u>	d all other initial fees is postponed until after all nplete and Franchisee is open for business."
2. The following language is add Agreement:	ded as new Section 22.14 of the Franchise
California Corporations Code Section franchise disclosure document, and any writing, disclaiming or denying: (a) repror agents to Franchisee before entering Franchisee on any representations made reliance by Franchisee on the franchise	ary in this Agreement, to the extent required by 31512.1, any provision in this Agreement, the other acknowledgement, questionnaire, or other resentations made by Franchisor or its personnel into the Franchise Agreement; (b) reliance by the by Franchisor, or its personnel or agents; (c) the disclosure document; or (d) violations of any thise Investment Law; is void and will not be
Dated:	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:

Title: _____

ILLINOIS AMENDMENT TO FRANCHISE AGREEMENT

This Amendment pertains to franchises sold in the State of Illinois that are subject to the Illinois Franchise Disclosure Act (the "Act") and is for the purpose of complying with Illinois statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

Subject to the foregoing, notwithstanding any Franchise Agreement to the contrary, the Agreement	ything which may be contained in the body of the ement is amended to include the following:
The parties to the Franchise Agreement that the Franchise Agreement will be amended	at dated, 20, hereby agreed as follows:
Illinois law shall supersede and Carolina law which are in conflict with the law	y provisions of the Franchise Agreement or North w.
	Franchise Agreement waives any rights Franchisee achise Disclosure Act of 1987, which provides:
person acquiring any franchise this Act or any other law of prevent any person from entering a general release regarding a provisions of this Act, nor sh pursuant to the provisions of T	ation, or provision purporting to bind any to waive compliance with any provision of this State is void. This Section shall not any into a settlement agreement or executing potential or actual lawsuit filed any of the tall it prevent the arbitration of any claim itle 9 of the United States Code."
	Agreement is amended to provide that venue shall al jurisdiction or United States District Court in
Dated:, 20	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:

MARYLAND AMENDMENT TO FRANCHISE AGREEMENT

This Amendment pertains to franchises sold in the State of Maryland that are subject to the Maryland Franchise Registration and Disclosure Law (the "Act") and is for the purpose of complying with Maryland statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

	Franchisor and Franchisee hereby agree that the Franchise Agreement dated _	
20	, will be amended as follows:	

1. The following paragraph has been deleted in its entirety from the Whereas Section of the Franchise Agreement:

"WHEREAS, Franchisee has had a full and adequate opportunity to read and review this Agreement and to be thoroughly advised of its terms and conditions by legal counsel or other personal advisors, and has had sufficient time to evaluate and investigate the Business System, the financial requirements, the economics of the hotel industry and the business risks associated with owning and operating a hotel."

2. The following language is added to Section 4.1 of the Franchise Agreement:

"Based upon Franchisor's financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by Franchisee shall be deferred until Franchisor completes its preopening obligations under the Agreement. In addition, all development fees and initial payments by area developers shall be deferred until the first franchise under the development agreement opens."

3. The following language is added to Sections 14.4 and 22.6 of the Franchise Agreement:

"Representations requiring prospective franchisees to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law."

4. The following language is added to the end of Section 17.1 of the Franchise Agreement:

"Franchisee may bring a lawsuit in Maryland for claims arising out of the Maryland Franchise Registration and Disclosure Law."

5. The following language is added to the end of Section 17.9 of the Franchise Agreement:

"all claims arising under the Maryland Franchise Registration and Disclosure Law shall be commenced within three (3) years after the grant of the franchise."

6. The following language is added to the end of Section 19.6(H) of the Franchise Agreement:

", except that the general release provisions shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law."

7. Section 24.1 through 24.3 of the Franchise Agreement of the Franchise Agreement is deleted.

IN WITNESS WHEREOF, Franchisor and Franchisee have duly executed and delivered this Amendment as of the date set forth above.

Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:

MINNESOTA AMENDMENT TO FRANCHISE AGREEMENT

This Amendment pertains to franchises sold in the State of Minnesota that are subject to the Minnesota Franchise Act (Minn. Stat. Sec. 80C.1 et seq., the "Act") and is for the purpose of complying with Minnesota statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

1. Section 3.3 of the Franchise Agreement is amended to add the following language.

"The Minnesota Department of Commerce requires the Franchisor to indemnify Minnesota franchisees against liability to third parties resulting from claims by third parties that the Franchisee's use of the tradename infringes trademark rights of the third party. Franchisor indemnifies Franchisee against the consequences of Franchisee's use of the tradename in accordance with the requirements of the license, and, as a condition to indemnification, Franchisee must provide notice to Franchisor of any such claims within ten (10) days and tender the defense of the claim to Franchisor. If Franchisor accepts the tender of defense, Franchisor has the right to manage the defense of the claim including the right to compromise, settle or otherwise resolve the claim, and to determine whether to appeal a final determination of the claim."

2. Section 13.2 of the Franchise Agreement are amended to read as follows:

"At the election of Franchisor, Franchisor may terminate the Agreement effective upon the expiration of 90 days after giving of written notice in the event Franchisee defaults, and does not cure to Franchisor's reasonable satisfaction within the 60-day notice period, in the performance of any other covenant or provision of this Agreement, including without limitation, the obligation to pay when due any financial obligation to Franchisor, the obligation to make reports and provide information when due hereunder, or failure to maintain any of the standards or procedures prescribed for the Franchised Business in this Agreement, the Manual or otherwise; provided, however, that Franchisee shall be entitled to notice and opportunity to cure any such default only once in any six month period, and any subsequent occurrence of the same or substantially similar default within such six month period shall entitle Franchisor, at its option,

to terminate this Agreement effective immediately upon the giving of notice and without opportunity to cure."

and

"With respect to franchises governed by Minnesota law, Franchisor will comply with Minn. Stat. Sec. 80C.14, Subds. 3, 4, and 5 which require, except in certain specified cases, that a Franchisee be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the Franchise Agreement."

- 3. Franchisor will protect the Franchisee's right granted hereby to use the Marks or will indemnify the Franchisee from any loss, costs or expenses arising out of any claim, suit or demand regarding the use of the Marks.
- 4. The second-to-last paragraph of Section 17.1 of the Franchise Agreement is amended as follows:

"Minn. Stat. §80C.21 and Minn. Rule 2860.4400J prohibit us from requiring litigation to be conducted outside Minnesota. In addition, nothing in the Franchise Disclosure Document or agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum, or remedies provided for by laws of the jurisdiction."

5. Minn. Rule 2860.4400J. prohibits waiver of a jury trial. Accordingly, the last paragraph of Section 17.1 of the Franchise Agreement is amended as follows:

"Nothing contained herein shall limit Franchisee's right to submit matters to the jurisdiction of the courts of Minnesota to the full extent required by Minn. Rule 2860.4400J."

- 6. Minn. Rule 2860.4400J. prohibits requiring a franchisee to consent to liquidated damages.
- 7. Minn. Rule 2860.4400D prohibits a franchisor from requiring a franchisee to assent to a general release.
- 8. Section 17.9 of the Franchise Agreement is amended to add the following to the end of the Section:

"Any claims pursuant to Minn. Stat. Sec. 80C.17 may be commenced within the time period provided in Minn. Stat. Sec. 80C.17, subd. 5."

IN WITNESS WHEREOF, the parties hereto have duly executed and delivered this Amendment to the Franchise Agreement as of the day and year set forth above.

Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

NEW YORK AMENDMENT TO FRANCHISE AGREEMENT

This Amendment pertains to franchises sold in the State of New York that are subject to the New York Franchise Act (New York State General Business Law, Article 33, Sec. 680 et seq., the "Act") and is for the purpose of complying with New York statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

notwithstanding anything which may be contained contrary, the Agreement is amended to include the	ed in the body of the Franchise Agreement to the
The parties to the Franchise Agreement the Franchise Agreement will be amended as follows:	dated, 20, hereby agree that ows:
	e a resident of the State of New York and your
2. The following is added as a new S	Section 14.5 of the Franchise Agreement:
"Franchisee may terminate this Ag	greement upon any grounds available at law."
3. The following is added to the end	of Section 22.1 of the Franchise Agreement:
	red a waiver of any right conferred upon you by w York State General Business Law and the
IN WITNESS WHEREOF , the parties day and year first above written.	have executed and delivered this Rider on the
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:

NORTH DAKOTA AMENDMENT TO FRANCHISE AGREEMENT

This Amendment pertains to franchises sold in the State of North Dakota that are subject to the North Dakota Franchise Investment Law (the "Act") and is for the purpose of complying with North Dakota statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

The parties to the Franchise Agreement dated	, 20, hereby agree that the
Franchise Agreement will be amended as follow	ws:

- 1. Section 13.3 of the Franchise Agreement is amended as follows:
 - "Remedies Upon Termination. If the Franchise is terminated, and in addition to the obligations of the Franchisee as otherwise provided herein, Franchisor shall retain the full amount of any fees heretofore paid to Franchisor and Franchisee shall continue to remain liable to Franchisor for any and all damages which Franchisor has sustained or may sustain by reason of such default or defaults and the breach of the Franchise Agreement on the part of the Franchisee for the unexpired Term of the Franchise Agreement."
- 2. Sections 15 and 16 of the Franchise Agreement are amended by deleting any obligation of Franchisee to consent to paying termination or liquidation damages.
- 3. Section 17.1 of the Franchise Agreement is deleted.
- 4. Section 17.6 of the Franchise Agreement is deleted.
- 5. Section 17.10 of the Franchise Agreement is amended as follows:

"In the event either party incurs legal fees or costs or other expenses to enforce any obligation of the other party hereunder, or to defend against any claim, demand, action or proceeding by reason of the other party's failure to perform or observe any obligation imposed upon that party by this Agreement, then the prevailing party shall be entitled to recover from the other party the amount of all legal fees, costs and expenses, including reasonable attorneys' fees, whether incurred prior to, or in preparation for or contemplation of the filing of any claim, demand, action or proceeding to enforce any obligation of the other party hereunder or thereafter or otherwise."

6. Section 19.6(H) of the Franchise Agreement is deleted.

- 7. The laws of the State of North Dakota supersede any provisions of the Franchise Agreement (including Section 22.1 of the Franchise Agreement), the other agreements or North Carolina law if such provisions are in conflict with North Dakota law.
- 8. Any provision in the Franchise Agreement (including Sections 17.1-3 of the Franchise Agreement) which designates jurisdiction or venue, or requires the Franchisee to agree to jurisdiction or venue, in a forum outside of North Dakota, is deleted from any Franchise Agreement issued in the State of North Dakota.
- 9. Section 7.11 of the Franchise Agreement is amended by deleting "and liquidated obligations" in the first sentence.
- 10. "No statement, questionnaire, or acknowledgment signed or agreed to by a franchise in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise"

Dated:, 20	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:

RHODE ISLAND AMENDMENT TO FRANCHISE AGREEMENT

This Amendment pertains to franchises sold in the State of Rhode Island that are subject to the Rhode Island Franchise Investment Act (the "Act") and is for the purpose of complying with Rhode Island statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

The parties to the Franchise Agreement d Franchise Agreement will be amended as follows	ated, 20 hereby agree that the s:
1. Section 19-28.1-14 of the Rhode I the laws of 1991, provides that "A provision in a venue to a forum outside this state or requiring the with respect to a claim otherwise enforceable und	ne application of the laws of another state is void
Dated:, 20	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
a Delaware infined hability company	
By:	By:
Name:	Name:
Title:	Title:

VIRGINIA AMENDMENT TO FRANCHISE AGREEMENT

This Amendment pertains to franchises sold in the State of Virginia that are subject to the Virginia Retail Franchising Act (the "Act") and is for the purpose of complying with Virginia statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

Franchise Agreement to the contrary, the Agr	nything which may be contained in the body of the element is amended to include the following:
The parties to the Franchise Agreement that the Franchise Agreement will be amende	nt dated, 20, hereby agree ed as follows:
1. Section 13.1 of the Franchise language to the beginning of the Section:	e Agreement is amended by adding the following
"§13.1-564 of the Virginia Re a franchisor to cancel a franchise with	etail Franchising Act provides that it is unlawful for nout reasonable cause."
Dated:, 20	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:

WASHINGTON ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT, THE FRANCHISE AGREEMENT, AND ALL RELATED AGREEMENTS

The provisions of this Addendum form an integral part of, are incorporated into, and modify the Franchise Disclosure Document, the franchise agreement, and all related agreements regardless of anything to the contrary contained therein. This Addendum applies if: (a) the offer to sell a franchise is accepted in Washington; (b) the purchaser of the franchise is a resident of Washington; and/or (c) the franchised business that is the subject of the sale is to be located or operated, wholly or partly, in Washington.

- 1. Conflict of Laws. In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, chapter 19.100 RCW will prevail.
- 2. Franchisee Bill of Rights. RCW 19.100.180 may supersede provisions in the franchise agreement or related agreements concerning your relationship with the franchisor, including in the areas of termination and renewal of your franchise. There may also be court decisions that supersede the franchise agreement or related agreements concerning your relationship with the franchisor. Franchise agreement provisions, including those summarized in Item 17 of the Franchise Disclosure Document, are subject to state law.
- 3. Site of Arbitration, Mediation, and/or Litigation. In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.
- 4. General Release. A release or waiver of rights in the franchise agreement or related agreements purporting to bind the franchisee to waive compliance with any provision under the Washington Franchise Investment Protection Act or any rules or orders thereunder is void except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2). In addition, any such release or waiver executed in connection with a renewal or transfer of a franchise is likewise void except as provided for in RCW 19.100.220(2).
- 5. Statute of Limitations and Waiver of Jury Trial. Provisions contained in the franchise agreement or related agreements that unreasonably restrict or limit the statute of limitations period for claims under the Washington Franchise Investment Protection Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.
- **6. Transfer Fees.** Transfer fees are collectable only to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.
- 7. **Termination by Franchisee.** The franchisee may terminate the franchise agreement under any grounds permitted under state law.

- 8. Certain Buy-Back Provisions. Provisions in franchise agreements or related agreements that permit the franchisor to repurchase the franchisee's business for any reason during the term of the franchise agreement without the franchisee's consent are unlawful pursuant to RCW 19.100.180(2)(j), unless the franchise is terminated for good cause.
- 9. Fair and Reasonable Pricing. Any provision in the franchise agreement or related agreements that requires the franchisee to purchase or rent any product or service for more than a fair and reasonable price is unlawful under RCW 19.100.180(2)(d).
- 10. Waiver of Exemplary & Punitive Damages. RCW 19.100.190 permits franchisees to seek treble damages under certain circumstances. Accordingly, provisions contained in the franchise agreement or elsewhere requiring franchisees to waive exemplary, punitive, or similar damages are void, except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2).
- 11. Franchisor's Business Judgement. Provisions in the franchise agreement or related agreements stating that the franchisor may exercise its discretion on the basis of its reasonable business judgment may be limited or superseded by RCW 19.100.180(1), which requires the parties to deal with each other in good faith.
- 12. Indemnification. Any provision in the franchise agreement or related agreements requiring the franchisee to indemnify, reimburse, defend, or hold harmless the franchisor or other parties is hereby modified such that the franchisee has no obligation to indemnify, reimburse, defend, or hold harmless the franchisor or any other indemnified party for losses or liabilities to the extent that they are caused by the indemnified party's negligence, willful misconduct, strict liability, or fraud.
- 13. Attorneys' Fees. If the franchise agreement or related agreements require a franchisee to reimburse the franchisor for court costs or expenses, including attorneys' fees, such provision applies only if the franchisor is the prevailing party in any judicial or arbitration proceeding.
- 14. Noncompetition Covenants. Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provision contained in the franchise agreement or elsewhere that conflicts with these limitations is void and unenforceable in Washington.
- 15. Nonsolicitation Agreements. RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any

such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

- 16. Questionnaires and Acknowledgments. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 17. Prohibitions on Communicating with Regulators. Any provision in the franchise agreement or related agreements that prohibits the franchisee from communicating with or complaining to regulators is inconsistent with the express instructions in the Franchise Disclosure Document and is unlawful under RCW 19.100.180(2)(h).
- 18. Advisory Regarding Franchise Brokers. Under the Washington Franchise Investment Protection Act, a "franchise broker" is defined as a person that engages in the business of the offer or sale of franchises. A franchise broker represents the franchisor and is paid a fee for referring prospects to the franchisor and/or selling the franchise. If a franchisee is working with a franchise broker, franchisees are advised to carefully evaluate any information provided by the franchise broker about a franchise.
- **19. Amendment to Section 4.1.** The second sentence of Section 4.1 of the Franchise Agreement is deleted in its entirety and replaced by the following sentence: "The Initial Franchise Fee will be due and payable on the date that Franchisor has completed its preopening obligations under the Agreement and Franchisee opens the Hotel for business."

The undersigned parties do hereby acknowle	dge receipt of this Addendum.
Dated thisday of	, 20
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:

State Addenda to Multi-Unit Development Agreement

MULTI-STATE AMENDMENT TO MULTI-UNIT DEVELOPMENT AGREEMENT (FOR THE FOLLOWING STATES: CA, HI, IL, IN, MD, MI, MN, NY, ND, RI, SD, VA, WA, WI)

This Amendment pertains to franchises sold in the states that have adopted as law the NASAA Statement of Policy Regarding the Use of Franchise Questionnaires and Acknowledgements (the "SOP") and is for the purpose of complying with the statutes and regulations of such states. Signing this Amendment where the SOP, because applicable jurisdictional requirements are not met, does not apply, does not subject the parties to the provisions of the SOP. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Multi-Unit Development Agreement to the contrary, the agreement is amended to include the following:

amended to include the following:	
Franchisor and Franchisee hereby ag dated, 20, will be amended	ree that the Multi-Unit Development Agreement das follows:
1. The following language is added in Multi-Unit Development Agreement:	immediately before the signature block of the
to by a franchisee in connecti franchise relationship shall he claims under any applicable sta the inducement, or (ii) disclaim by any franchisor, franchise sel	or acknowledgment signed or agreed on with the commencement of the ave the effect of (i) waiving any ate franchise law, including fraud in ning reliance on any statement made aller, or other person acting on behalf on supersedes any other term of any on with the franchise."
IN WITNESS WHEREOF, Franchison this Amendment as of the date set forth above	r and Franchisee have duly executed and delivered
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
, ,	
By:	By:
Name:	Name: Title:
THE.	HUC.

FA-Multistate-1

CALIFORNIA AMENDMENT MULTI-UNIT DEVELOPMENT AGREEMENT

This Amendment pertains to franchises sold in the State of California that are subject to the California Franchise Investment Law (the "Act") and is for the purpose of complying with California statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Multi-Unit Development Agreement to the contrary, the Franchise Agreement is amended to include the following:

	e contrary, the Franchise Agreement is amended
	opment Agreement dated, velopment Agreement will be amended as follows:
1. Section 3.1 of the Multi-Unit following:	Development Agreement is amended by adding the
	nd all other initial fees is postponed until after all of omplete under the first Franchise Agreement signed business."
2. The following language is Development Agreement:	added as new Section 11.7 of the Multi-Unit
California Corporations Code Section franchise disclosure document, and a writing, disclaiming or denying: (a) to or agents to Franchisee before enter Franchisee on any representations may reliance by Franchisee on the franchisee on the franchisee on the franchisee.	ntrary in this Agreement, to the extent required by on 31512.1, any provision in this Agreement, the any other acknowledgement, questionnaire, or other representations made by Franchisor or its personnel ring into the Franchise Agreement; (b) reliance by nade by Franchisor, or its personnel or agents; (c) hise disclosure document; or (d) violations of any ranchise Investment Law; is void and will not be
Dated:	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:

FA-Multistate-1

ILLINOIS AMENDMENT TO MULTI-UNIT DEVELOPMENT AGREEMENT

This Amendment pertains to franchises sold in the State of Illinois that are subject to the Illinois Franchise Disclosure Act (the "Act") and is for the purpose of complying with Illinois statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Multi-Unit Development Agreement to the contrary, the Agreement is amended to include the following:

Multi-Unit Dev the following:	relopment Agreement to the cont	rary, the Agreement is amended to include
		t Agreement dated, 20, greement will be amended as follows:
	linois law shall supersede any porth Carolina law which are in con	provisions of the Multi-Unit Development flict with the law.
	_	ti-Unit Development Agreement waives any he Illinois Franchise Disclosure Act of 1987,
tl p a p	erson acquiring any franchise to whis Act or any other law of this strevent any person from entering in general release regarding a poten	or provision purporting to bind any aive compliance with any provision of State is void. This Section shall not to a settlement agreement or executing stial or actual lawsuit filed any of the prevent the arbitration of any claim of the United States Code."
	ue shall be in an appropriate Illinoi	Unit Development Agreement is amended to is court of general jurisdiction or United States
Dated:	, 20	
Franchisor: LG AS Franchi a Delaware lim		Franchisee:
Ву:		By:
Name:		Name:
Title		Title:

MARYLAND AMENDMENT TO MULTI-UNIT DEVELOPMENT AGREEMENT

This Amendment pertains to franchises sold in the State of Maryland that are subject to the Maryland Franchise Registration and Disclosure Law (the "Act") and is for the purpose of complying with Maryland statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Multi-Unit Development Agreement to the contrary, the Agreement is amended to include the following:

	Franchisor a	and	Franchisee	hereby a	agree	that the	Multi-Unit	Development	Agreement
dated		20_	_, will be an	mended a	as foll	ows:			

1. The following language has been deleted in its entirety from the Whereas Section of the Franchise Agreement:

"WHEREAS, Franchisee has had a full and adequate opportunity to read and review this Agreement and to be thoroughly advised of its terms and conditions by legal counsel or other personal advisors, and has had sufficient time to evaluate and investigate the Business System, the financial requirements, the economics of the hotel industry and the business risks associated with owning and operating hotels."

2. The following language is added to Section 3.1 of the Multi-Unit Development Agreement:

"Based upon Franchisor's financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by Franchisee shall be deferred until Franchisor completes its preopening obligations under the first Franchise Agreement to be signed hereunder. In addition, all development fees and initial payments by area developers shall be deferred until the first franchise under the development agreement opens."

- 3. The following language is added to Section 8.4 of the Multi-Unit Development Agreement:
 - "Representations requiring prospective franchisees to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law."
- 4. The following language is added to the end of Section 8.4 of the Multi-Unit Development Agreement:

- ", except that the general release provisions shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law."
- 5. The following language is added to the end of Section 9.2 of the Multi-Unit Development Agreement:

"Franchisee may bring a lawsuit in Maryland for claims arising out of the Maryland Franchise Registration and Disclosure Law."

- 6. The following language is added to the end of Section 9.9 of the Multi-Unit Development Agreement:
 - "all claims arising under the Maryland Franchise Registration and Disclosure Law shall be commenced within three (3) years after the grant of the franchise."
- 7. Section 11.1 through 11.3 of the Multi-Unit Development Agreement are deleted.
- 8. Section 11.5 of the Multi-Unit Development Agreement is deleted.

IN WITNESS WHEREOF, Franchisor and Franchisee have duly executed and delivered this Amendment as of the date set forth above.

Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:	
By:	By:	
Name:	Name:	
Title:	Title:	

MINNESOTA AMENDMENT TO MULTI-UNIT DEVELOPMENT AGREEMENT

This Amendment pertains to franchises sold in the State of Minnesota that are subject to the Minnesota Franchise Act (Minn. Stat. Sec. 80C.1 et seq., the "Act") and is for the purpose of complying with Minnesota statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Multi-Unit Development Agreement to the contrary, the Agreement is amended to include the following:

The parties to the Multi-Unit Development Agreement dated ________, 20____, hereby agree that the Multi-Unit Development Agreement will be amended as follows:

1. New Section 5.8 of the Multi-Unit Development Agreement is added and is to read as follows:

"At the election of Franchisor, Franchisor may terminate the Agreement effective upon the expiration of 90 days after giving of written notice in the event Franchisee defaults, and does not cure to Franchisor's reasonable satisfaction within the 60-day notice period, in the performance of any other covenant or provision of this Agreement, including without limitation, the obligation to pay when due any financial obligation to Franchisor, the obligation to make reports and provide information when due hereunder, or failure to maintain any of the standards or procedures prescribed for the Franchised Business in this Agreement, the Manual or otherwise; provided, however, that Franchisee shall be entitled to notice and opportunity to cure any such default only once in any six month period, and any subsequent occurrence of the same or substantially similar default within such six month period shall entitle Franchisor, at its option, to terminate this Agreement effective immediately upon the giving of notice and without opportunity to cure."

and

"With respect to franchises governed by Minnesota law, Franchisor will comply with Minn. Stat. Sec. 80C.14, Subds. 3, 4, and 5 which require, except in certain specified cases, that a Franchisee be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the Franchise Agreement."

2. Franchisor will protect the Franchisee's right granted hereby to use the Marks or will indemnify the Franchisee from any loss, costs or expenses arising out of any claim, suit or demand regarding the use of the Marks.

3. The second-to-last paragraph of Section 9.2 of the Multi-Unit Development Agreement is amended as follows:

"Minn. Stat. §80C.21 and Minn. Rule 2860.4400J prohibit us from requiring litigation to be conducted outside Minnesota. In addition, nothing in the Franchise Disclosure Document or agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum, or remedies provided for by laws of the jurisdiction."

- 4. Minn. Rule 2860.4400D prohibits a franchisor from requiring a franchisee to assent to a general release.
- 5. Section 8.4 of the Franchise Agreement is amended to add the following to the end of the Section:

"Any claims pursuant to Minn. Stat. Sec. 80C.17 may be commenced within the time period provided in Minn. Stat. Sec. 80C.17, subd. 5."

IN WITNESS WHEREOF, the parties hereto have duly executed and delivered this Amendment to the Franchise Agreement as of the day and year set forth above.

Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	_ Title:
Date:	Date:

NEW YORK AMENDMENT TO MULTI-UNIT DEVELOPMENT AGREEMENT

This Amendment pertains to franchises sold in the State of New York that are subject to the New York Franchise Act (New York State General Business Law, Article 33, Sec. 680 et seq., the "Act") and is for the purpose of complying with New York statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are

notwithstanding anything which may be contained in the body of the Multi-Unit Development Agreement to the contrary, the Agreement is amended to include the following:
The parties to the Multi-Unit Development Agreement dated, 20 hereby agree that the Multi-Unit Development Agreement will be amended as follows:
1. Franchisor and Franchisee are parties to that certain Multi-Unit Development Agreement dated
2. The following is added at the end of Section 6.1 of the Multi-Unit Development Agreement:
"Franchisee may terminate this Agreement upon any grounds available at law."
3. The following is added to the end of Section 10.1 of the Multi-Uni Development Agreement:
"This section shall not be considered a waiver of any right conferred upon you by the provisions of Article 33 of the New York State General Business Law and the regulations issued thereunder."
IN WITNESS WHEREOF , the parties have executed and delivered this Rider on the day and year first above written.
Franchisor: Franchisee: LG AS Franchisor LLC a Delaware limited liability company
By: By:
Name: Name:

Title:	Title
11tte	1 Itic

NORTH DAKOTA AMENDMENT TO MULTI-UNIT DEVELOPMENT AGREEMENT

This Amendment pertains to franchises sold in the State of North Dakota that are subject to the North Dakota Franchise Investment Law (the "Act") and is for the purpose of complying with North Dakota statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Multi- Unit Development Agreement to the contrary, the Agreement is amended to include the following:

The parties to the Multi-Unit Development Agreement dated ______, 20___, hereby agree that the Multi-Unit Development Agreement will be amended as follows:

- 1. Section 17.6 of the Multi-Unit Development Agreement is deleted.
- 2. Section 9.10 of the Multi-Unit Development Agreement is amended as follows:

"In the event either party incurs legal fees or costs or other expenses to enforce any obligation of the other party hereunder, or to defend against any claim, demand, action or proceeding by reason of the other party's failure to perform or observe any obligation imposed upon that party by this Agreement, then the prevailing party shall be entitled to recover from the other party the amount of all legal fees, costs and expenses, including reasonable attorneys' fees, whether incurred prior to, or in preparation for or contemplation of the filing of any claim, demand, action or proceeding to enforce any obligation of the other party hereunder or thereafter or otherwise."

- 3. The laws of the State of North Dakota supersede any provisions of the Franchise Agreement, the other agreements or North Carolina law if such provisions are in conflict with North Dakota law.
- 4. Any provision in the Multi-Unit Development Agreement which designates jurisdiction or venue, or requires the Franchisee to agree to jurisdiction or venue, in a forum outside of North Dakota, is deleted from any Multi-Unit Development Agreement issued in the State of North Dakota.

Dated:, 20	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:

Name:	Name:		
Title:	Title:		

RHODE ISLAND AMENDMENT TO MULTI-UNIT DEVELOPMENT AGREEMENT

This Amendment pertains to franchises sold in the State of Rhode Island that are subject to the Rhode Island Franchise Investment Act (the "Act") and is for the purpose of complying with Rhode Island statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Multi-Unit Development Agreement to the contrary, the Agreement is amended to include the following:

U	
The parties to the Multi-Unit Development that the Multi-Unit Development	elopment Agreement dated, 20 at Agreement will be amended as follows:
the laws of 1991, provides that "A provision	de Island Franchise Investment Act, as amended by in a Franchise Agreement restricting jurisdiction or ag the application of the laws of another state is void under this Act."
Dated:, 20	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:

VIRGINIA AMENDMENT TO MULTI-UNIT DEVELOPMENT AGREEMENT

This Amendment pertains to franchises sold in the State of Virginia that are subject to the Virginia Retail Franchising Act (the "Act") and is for the purpose of complying with Virginia statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act.

Subject to the foregoing, notwithstanding an	ything which may be contained in the body of the contrary, the Agreement is amended to include
The parties to the Multi-Unit Development hereby agree that the Multi-Unit Development	oment Agreement dated, 20 nt Agreement will be amended as follows:
1. Section 7.1 of the Franchise language to the beginning of the Section:	e Agreement is amended by adding the following
"§13.1-564 of the Virginia Re a franchisor to cancel a franchise with	etail Franchising Act provides that it is unlawful for hout reasonable cause."
Dated:, 20	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:

WASHINGTON AMENDMENT TO MULTI-UNIT DEVELOPMENT AGREEMENT AND RELATED AGREEMENTS

This Amendment pertains to franchises sold in the State of Washington that are subject to the Washington Franchise Investment Protection Act (the "Act") and is for the purpose of complying with Washington statutes and regulations. Signing this Amendment where the jurisdictional requirements of the Act are not met does not subject the parties to the provisions of the Act. Subject to the foregoing, notwithstanding anything which may be contained in the body of the Multi-Unit Development Agreement to the contrary, the Agreement is amended to include the following:

The parties to the Multi-Unit Development Agreement dated _______, 20__ hereby agree that the Multi-Unit Development Agreement will be amended as follows:

- 1. The State of Washington has a statute, RCW 19.100.180, which may supersede the Multi-Unit Development Agreement in your relationship with Franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the Multi-Unit Development Agreement in your relationship with Franchisor including the areas of termination and renewal of your license.
- 2. In the event of a conflict of law, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW shall prevail.
- 3. A release or waiver of rights executed by a Franchisee shall not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial may not be enforceable.
- 4. Transfer fees are collectable to the extent that they reflect the Franchisor's reasonable estimated or actual costs in effecting a transfer.
- 5. Section 3.1 of the Multi-Unit Development Agreement is revised by adding the following at the end of the Section:
 - "The Development Fee for each Brand Hotel will be due and payable on the date that such Brand Hotel is opened for business."
- 6. Section 11.1 of the Multi-Unit Development Agreement is amended by deleting it in its entirety and replacing it with the following:
 - "BUSINESS RISK; NO FINANCIAL PROJECTIONS. Franchisee acknowledges that it has conducted an independent investigation of the Hotel business licensed hereunder, and recognizes that the business venture contemplated by this Agreement involves

business and economic risks and that the success of its Hotel will be primarily dependent upon the personal efforts of Franchisee, its management, and its employees."

- 7. In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.
- 8. Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.
- 9. RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.
- 10. The second sentence of Section 6.4 (Waiver) of the Multi-Unit Development Agreement is deleted.
- 11. The first sentence of Section 9.9 (Enforcement) of the Multi-Unit Development Agreement is deleted.

The undersigned hereby acknowledges receipt of this amendment.

Dated:, 20	
Franchisor: LG AS Franchisor LLC a Delaware limited liability company	Franchisee:
By:	By:
Name:	Name:
Title:	Title:

EXHIBIT G

Form Release of Claims

FORM RELEASE OF CLAIMS

THIS IS A CURRENT RELEASE FORM THAT GENERALLY WILL BE USED WITH OR INCORPORATED INTO A SEPARATE AGREEMENT. THIS FORM IS SUBJECT TO CHANGE OVER TIME.

Franchisor LLC ("stayAPT"), _____ ("Franchisee") and ____ ("Guarantors") enter into this Release of Claims ("Agreement") as of this ____ day of

For and in consideration of the Agreements and covenants described below, LG AS

	RECITALS
A.	stayAPT and Franchisee entered into an stayAPT Franchise Agreement dated,
В.	[NOTE: Describe the circumstances relating to the release.]
	<u>AGREEMENTS</u>
1.	Consideration. [NOTE: Describe the consideration paid.]
2-3.	[NOTE: Detail other terms and conditions of the release.]
4.	Release of Claims.
	A. Definitions.
attorneys, aff	1. stayAPT Parties: Franchisor, and each of its officers, directors, filiates, agents, employees, shareholders, successors and assigns.
	2. Franchisee Parties: Franchisee and each of the Guarantors, and heirs, executors, administrators, trustees, agents, partners, employees, affiliates, ecessors and assigns.
and other	B. The Franchisee Parties release and forever discharge the stayAPT Parties d from any and all past, present, and future claims, demands, obligations, actions causes of action at law or in equity, whether arising by statute, common law, or wise, including claims for negligence, that they may now have, hereafter have, or to have, that arise out of or relate to the franchise relationship, the development or

operation of any franchised store, the sale of any franchise, or any franchise or

this release of claims covers all of their claims, including those known and unknown

The Franchisee Parties hereto specifically and expressly contemplate that

development agreement between Franchisor and Franchisee.

claims for known and unknown injuries and/or damages, and those for expected and unexpected consequences.

5. **General**. No amendment to this Agreement or waiver of the rights or obligations of either party shall be effective unless in writing signed by the parties. This Agreement is governed by the laws of the State of North Carolina without regard to conflicts of laws principles. If any provision of this Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Agreement will remain in full force and effect. This Agreement contains the entire agreement and understanding of the parties concerning the subject matter of this Agreement. [NOTE: Detail other miscellaneous provisions.]

[Add for Washington Franchisees: The Release is inapplicable with respect to claims arising under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted thereunder.]

FRANCHISEE:	LG AS FRANCHISOR LLC
BY:	BY:
NAME:	NAME:
TITLE:	TITLE:
DATE:	DATE:
PERSONAL GUARANTORS:	
BY:	
NAME:	
TITLE:	
DATE:	<u> </u>
DV	
BY:	_
NAME:	<u> </u>
TITLE:	

DATE:		
DILLE.		

EXHIBIT H

Comfort Letter

COMFORT LETTER

Name and
Address
Of Bank
To Whom It May Concern:
This Comfort Letter Agreement ("Agreement") is made on, 20,
by and between LG AS Franchisor LLC, a Delaware limited liability company, ("Franchisor"),
and ("Lender"). Franchisor and Lender may be referred to collectively
herein as the "Parties".
This letter references a certain agreement between Franchisor and
("Franchisee" and/or "Borrower"), dated, 20 (the "Franchise Agreement"),
which grants Franchisee the right to operate an extended-stay hotel (the "Franchised Location")
at (the "Approved Location"), and a [Loan
Agreement - INSERT CORRECT TITLE] (together with any collateral agreements thereto, the
'Loan Agreement"), dated20, between Lender and Franchisee, pursuant to
which, Lender is lending to Franchisee the principal sum of \$ (the "Loan") to
acquire real property and construct the Location at the Approved Location, which Location will
be operated as a Franchised Location under the Franchise Agreement.

At Franchisee's request, and in order to induce Lender to enter into the Loan Agreement and the Loan, Franchisor and Lender agree to the terms and conditions set forth herein.

1. Notice of Default and/or Termination.

- **A. By Franchisor.** At the same time and in the same manner that written default notices are sent to Franchisee under the Franchise Agreement, Franchisor shall give Lender written notice of any event of default or breach under the Franchise Agreement (a "Default Notice"). Franchisor shall also provide Lender with any termination notice, at the same time and in the same manner that a termination notice is sent to Franchisee under the Franchise Agreement, (a "Termination Notice").
- **B.** By Lender. At the same time and in the same manner that Lender is required to give written default notices to the Franchisee, under the Loan Agreement (a "Lender Default Notice"), Lender shall also give Franchisor written notice of any event of default or breach or termination under the Loan Agreement.

2. Default Under the Loan/Lender Foreclosure.

A. Lender Acquisition and Assumption.

- 1. Franchisor hereby agrees that, if Lender acquires the Property from the current Franchisee through foreclosure, a deed in lieu of foreclosure, or through any other exercise of its rights under the Loan Agreement ("Lender Acquisition"), and if the Franchise Agreement has not yet otherwise been terminated by Franchisor, and if the Lender desires to continue to operate the property as a Franchised Location, Lender shall assume Franchisee's rights and obligations under the Franchise Agreement ("Lender Assumption"), subject to the terms of this Agreement, by giving written notice to Franchisor of its intention to assume Franchisee's rights and obligations under the Franchise Agreement (the "Assumption Notice"), together with the payment of a Three Thousand Dollar \$3,000.00 processing fee.
- 2. The Assumption Notice shall be delivered to Franchisor no later than ten (10) calendar days after Lender Acquisition. Franchisor may require the submission by Lender of its organizational documents, ownership structure and evidence of compliance with Franchisor' then-current insurance standards, in connection with the submission of the Assumption Notice. For purposes of this Agreement, the Property shall be deemed to be acquired by the Lender effective immediately upon the execution of a deed in lieu of foreclosure, order of a court, or any other act of transferring ownership of the Property to Lender.
- **B.** Management. Within ten (10) days of Lender's assumption of the Franchise Agreement, rather by delivery of the Assumption Notice or by its Deemed Assumption (defined below), Lender must select a management company to manage the Property from a list recommended by Franchisor or engage Franchisor' affiliate, _________, or such other management company as Franchisor may approve in its reasonable discretion, to operate the Property. Lender shall be responsible for the fees and costs associated with the placement of qualified management company. Such management company, or a successor reasonably acceptable to Franchisor, shall continue to manage the Property during such time as Lender is in possession of and operating the Property under the Franchise Agreement or any successor franchise agreement entered into in accordance with the provisions of this Agreement, as applicable.

C. Default Under the Loan/Lender Foreclosure in conjunction with Default Under the Franchise Agreement.

1. In addition to the requirements set forth above in this Paragraph 2, as a condition to the consummation of a Lender Assumption or Deemed Assumption, Lender will be required to: (i) promptly cure (subject to Paragraph 3 D) any uncured breach or default under the Franchise Agreement existing at the time of Lender Assumption, relating to quality, service or other deficiency in the Franchisee's performance or obligations under the Franchise Agreement (excluding defaults that are personal to the Franchisee and cannot be cured by the Lender, as defined below); and (ii) pay any and all past due royalties or funds due the Franchisor and its affiliates.

D. Non-monetary Defaults Requiring Possession of Premises to Cure.

- 1. In the event of a breach or default under the Franchise Agreement which the Lender is unable to cure absent its possession, control, or ownership or of the Property, Lender shall have such an additional time period to cure said default as Franchisor may, in its reasonable discretion, grant to Lender. In no event shall the additional time period exceed one hundred twenty (120) days after Lender's acquisition of possession, control, or ownership of the Property, and it shall further be subject to that:
 - a) Lender immediately commences, and diligently prosecutes to completion, proceedings to obtain possession, control or ownership of the Property, as may be necessary to enable Lender to cure Franchisee's default under the Franchise Agreement; and
 - b) following its acquisition of possession, control or ownership of the Property, Lender shall diligently proceed to cure all defaults, except for personal and non-curable defaults (as defined below) within the time period granted by Franchisor, and shall thereafter perform all obligations of the Franchisee under the Franchise Agreement.
- 2. "Diligently proceed to cure" shall include the physical renovation work necessary at the Property to comply with Franchisor requirements under the Franchise Agreement and/or Franchisor's brand standards. A "Personal and Non-Curable" default shall include a default that occurred prior to the Lender's acquisition of possession, control, or ownership of the Property that is personal to the Franchisee (i.e. bankruptcy, failure to provide adequate notice, moral turpitude, failure to maintain company status, etc.) and is unrelated to operation of the Property.

E. Failure to Assume.

Lender's failure to timely deliver an Assumption Notice shall be deemed an assumption of the Franchise Agreement and all of its terms and conditions with the exception of Article 15.12 regarding liquidated damages (the "Deemed Assumption"), and entitle Franchisor to proceed in the ordinary course of business to pursue its legal rights and remedies under the Franchise Agreement and this Agreement. Such rights include, but are not limited to Franchisor' right, but not the obligation, to immediately terminate the Franchise Agreement. Lender shall be immediately required to begin performing the obligations of Franchisee under the Franchise Agreement.

F. Rejection

Should Lender acquire possession, control or ownership of the Property but elect not to exercise its right to assume the Franchise Agreement, Lender shall immediately comply with Franchisor' request to perform the obligations of the Franchisee under the Franchise Agreement with regard to the de-identification and/or de-branding of the Property and shall indemnify, defend and hold Franchisor and its affiliates harmless from and against any loss, claim or other liability of any kind arising from or in connection with the operation of the Property from the

Comfort Letter

time of Lender's rejection until such de-identification and de-branding has been completed. Lender's obligations under this Paragraph shall survive any termination of this Agreement and nothing in this Agreement shall limit Franchisor's rights, if any, to pursue its legal rights and remedies for any unauthorized use of Franchisor's trademarks, service marks, or systems.

G. Management During Receivership.

In the event that Lender gains any right to control the Property or revenues from the Property via the appointment of a Receiver or otherwise, Lender shall comply with all terms and provisions of the Franchise Agreement, including all monetary obligations arising under the Franchise Agreement, specifically, but not limited to the payment of royalties and outstanding fees due Franchisor and/or its affiliates.

3. Assignment and Transferability.

- A. Lender and Franchisee represent, warrant and covenant to the Franchisor that that the Franchise Agreement has been collaterally assigned, pledged, or granted as a security interest or otherwise transferred to the Lender or its affiliates. Franchisee further represents and covenants that the granting of the Loan Agreement will not cause the Franchisee to violate any financial covenants contained in the Franchise Agreement. Neither the Lender nor the Franchisee shall have the right or authority to sell, convey, assign or in any manner transfer any rights hereunder or under the Franchise Agreement without the written consent of Franchisor, and in accordance with the provisions of the Franchise Agreement, including its confidentiality provisions.
- **B.** Notwithstanding anything to the contrary contained herein, Lender may transfer its rights under the Franchise Agreement to a wholly owned subsidiary, as approved in writing by Franchisor, for the purposes of operating and maintaining the Property pursuant to the requirements herein and under the Franchise Agreement and for a period of time as designated in Paragraph 6 below.
- C. Notwithstanding anything to the contrary contained herein, this Agreement may be assigned to any subsequent holder or holders of the Loan without Franchisor's consent; provided the Loan is assigned to a commercial bank, investment bank, pension fund, finance company, insurance company, or other entity engaged in the business of making loans and any fund managed by any of the foregoing.
- **4.** <u>Immediate Termination of Franchise Agreement</u>. Notwithstanding any provision to the contrary contained herein, Franchisor reserves its right to immediately terminate the Franchise Agreement if, subsequent to a Lender Acquisition, Lender does not comply with the terms and provisions of this Agreement.
- 5. <u>Transition of Control of the Property</u>. Lender, Franchisee and Franchisor hereby agree to fully cooperate with one another in the event of change of control of the Property as contemplated herein. Said transition shall be conducted efficiently without inconvenience to the

guests or employees of the Property and shall be in compliance with any applicable federal or state law or regulation.

6. <u>Intent of this Agreement and Subsequent Sale of the Property.</u>

- A. The parties hereto acknowledge that a Lender Assumption or Deemed Assumption, and any subsequent acquisition of the Franchise Agreement by Lender is intended to serve as temporary remedy to lessen the negative impact of a Franchisee/Borrower Default under the Loan on both the Lender and the Franchisor. It is not the Franchisor's desire to maintain a long term franchise relationship with the Lender, and the Lender is not in the business of, and does not intend to be in the business of, operating Franchised Locations.
- **B**. Notwithstanding anything to the contrary herein, in the event of a Lender Acquisition and subsequent Lender Assumption or Deemed Assumption the Lender shall, within eighteen (18) months from the date of the Lender Assumption or Deemed Assumption ("Transfer Period"), in conjunction with the sale, assignment or transfer of the Property, relinquish its rights to the Franchise Agreement to a transferee acceptable to Franchisor.
- C. Lender agrees to diligently pursue a qualified third party to affect such transfer and Franchisor agrees that it shall use its resources within the Franchisor's franchise system to aid the Lender in finding a qualified replacement, but makes no guarantee that it will be able to do so.
- **D.** Such Transfer shall be upon all of the terms and conditions of the Franchise Agreement and shall require the proposed new transferee to enter into a new, then current form of Franchise Agreement with Franchisor and to pay to Franchisor the then current Initial Franchise Fee.
- **E.** In the event that Lender would fail to transfer the Franchise Agreement within eighteen (18) months from the date of the Lender Acquisition, Franchisor shall have the right to terminate the Franchise Agreement effective immediately upon written notice to the Lender. Franchisor may extend the Transfer Period, at its sole discretion, upon written request of the Lender, received by Franchisor prior to the expiration of the Transfer Period.
- 7. Default Under the Franchise Agreement Lender's Right to Cure. Upon receipt of a default notice, Lender shall have the right, but not the obligation, to cure Franchisee's default or breach, during the notice period granted by Franchisor to Franchisee, if any. Franchisor shall extend, upon Lender's prior written request, Lender's right to cure beyond the cure period established in such default notice (not to exceed an additional five (5) days for a monetary default or ten (10) days for a non-monetary default. The Lender's inability or failure to cure shall not prevent Franchisor from exercising is legal rights and remedies under the Franchise Agreement.
- **8.** <u>Notices</u>. All notices, requests, demands and other communications under this Agreement, including any Notice of Default or Assumption Notice, shall be in writing and shall be served either by personal delivery, by certified mail return receipt requested, by receipted overnight

courier service or by facsimile (with a confirming copy mailed if served only by facsimile), properly addressed as follows:

If to Franchisee:	
	Attn:
	Facsimile: ()
If to Lender:	Bank of
	Attn:
	Facsimile: ()
If to Franchisor:	
	Attn:
	Facsimile: ()

Any party may change its address for purposes of this paragraph by giving the other parties written notice of the new address in the manner provided above. Notices shall be deemed received at the earlier of receipt or three business days after mailing by certified mail or one business day after sending by overnight courier service.

- **9.** Franchisee's and Guarantor's Obligation's. Franchisee and any guarantor(s) of the Franchise Agreement shall not be released from any obligations under the Franchise Agreement or guaranty upon any assumption of the Franchise Agreement by Lender.
- 10. <u>Subordination</u>. Franchisor hereby acknowledges and agrees that any lien or encumbrance on or any security interest in the Property in favor of Franchisor or its affiliates arising out of or in connection with the Franchise Agreement shall be subordinate to Lender's rights under or with respect to the Loan, including without limitation any lien or encumbrance on the Property securing any such rights of Lender. The foregoing subordination is a subordination of liens, encumbrances and/or security interests, as the case may be, and is not intended to be, and shall not be construed as, a subordination by Franchisor of its right to payments from Franchisee under the Franchise Agreement. Without limiting the generality of the foregoing, it is acknowledged and agreed by the parties hereto that the subordination set forth in this Section 10 shall not hinder or delay Franchisor from pursuing any of its rights under the Franchise

Comfort Letter

Agreement against Franchisee to collect royalties, fees or other amounts due Franchisor under the Franchise Agreement.

Lender hereby acknowledges that its rights under this agreement are subordinate to 1) the rights granted to any other Lender or other third party holding a prior lien against the property which is the subject of this Comfort Letter, and for which Franchisor has executed a Comfort Letter to facilitate the loan; or 2) to any Comfort Letter executed prior to the date hereof, for as long as that Comfort Letter is in full force and effect, by and between Franchisor and another Lender or third party.

- 11, <u>Confidentiality and Non-Disclosure</u>. The provisions of this letter agreement shall not be disclosed by Lender or Franchisee to any third party, excepting Franchisee's or Lender's respective employees, directors, officers, agents or legal and financial representatives on a need-to-know basis, and/or unless as required by law or as mutually agreed to by the parties, and/or as part of any due diligence performed as part of the sale or participation of the Loan by Lender. As part of such disclosure, Lender must ensure that third parties are advised of, and agree to be bound by, the terms of this confidentiality provision. Franchisee and Lender further agree not to copy, reproduce or otherwise make available in any form whatsoever to any other person, firm, corporation or business the provisions of this letter agreement.
- **12.** <u>Automatic Termination of Comfort Letter Agreement</u>. This Agreement shall immediately terminate and Lender shall have no rights hereunder if:
 - a) Lender has been taken over in any manner by any state or federal agency or is in receivership, conservatorship, reorganization, or liquidation, or Lender or any of its officers or directs has entered into or is subject to a cease and desist order or any other formal or informal written agreement with a federal or state agency;
 - b) Lender no longer holds a valid mortgage or security deed with respect to the Property (other than as the result of a Lender Acquisition); or
 - c) the Franchise Agreement expires at the end of its term.
- **13.** General. The provisions contained herein are applicable to the Property and the parties hereto only. Issuance and execution of this Agreement or the granting of any conditions provided herein does not constitute and obligation on Franchisor's part to provide the same at any future date.
- 14. Governing Law and Enforceability. This Agreement shall be construed in accordance with, and governed by, the internal laws of the State of North Carolina, without regard to its conflict of laws rules. The invalidity or unenforceability of any provision of this Agreement shall not invalidate or affect the enforceability of any other provision of this Agreement. This Agreement shall be binding on, and inure to the benefit of, the parties hereto and their respective heirs, legal representatives, successors and permitted assigns.

- **15. Estoppel**. As of the date of this Agreement, Franchisee is in reasonable compliance with the terms and conditions of the Franchise Agreement, there is no pending Notice of Default to the Franchisee and the Franchise Agreement is in full force and effect.
- 16. No Representations or Warranties. In no event shall this Agreement or any other circumstances surrounding the financing by Lender be construed to involve (i) any representation by Franchisor that it endorses, approved, recommends or otherwise concurs in the financing; (ii) any guarantee or assurance by Franchisor that it or any other party to the Loan will be able to repay the Loan in accordance with its terms; (iii) any endorsement, approval, recommendation or concurrence in any financial projections submitted to Lender in connection with the Loan; or (iv) any endorsement, approval or recommendation of the Franchisee's character or reputation.
- 17. <u>Entire Agreement</u>. This is the entire agreement among the parties relating to the subject matter of this Agreement and it supersedes any previous oral or written agreements. This Agreement may be amended only by a writing signed by all parties. This agreement may be executed in counterparts with the same force and effect as if the parties had executed one instrument, and each such counterpart shall constitute an original hereof.
- 18. <u>Effectiveness</u>. This Agreement shall become effective when, and not before, Franchisor receives a fully executed original of the Agreement. If a fully executed original of this Agreement is not received by Franchisor within thirty (30) calendar days of the date of this Agreement this Agreement shall be deemed null and void.

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed on the date set forth above.

Lender:	
By:Name:	
Title:	
Franchisor: LG AS Franchisor LLC	
By:	
Agreement Approved by:	

Franchisee		
By:		
Name:		
Title:		

EXHIBIT I

Multi-Unit Development Agreement

STAYAPT® SUITES MULTI-UNIT DEVELOPMENT AGREEMENT

THIS MULTI-UNIT DEVELOPMENT AGREEMENT ("Agreement"), is made, entered into and effective this day of, 20, ("Effective Date") by and between LG AS Franchisor LLC, a Delaware limited liability company ("Franchisor"), and ("Franchisee").
WITNESSETH:
WHEREAS, Franchisor operates and grants franchises to others for the right to establish and operate extended-stay hotels providing distinctive hotel services which are associated with Franchisor's Marks, copyrights, interior and exterior design, decor, FF&E, signs, logos, commercial symbols and color combinations, standards, specifications, programs and procedures for operations; programs and procedures for quality control; training and assistance; and advertising, direct sales, and promotional programs developed by Franchisor for the operation of a Brand Hotel under the Marks utilizing the Trade Secrets (hereinafter referred to as the "Business System") under the name stayAPT® Suites and other trademarks, trade names, service marks, logos and commercial symbols designated to be part of the Business System (collectively, as amended from time to time, the "Marks"); and
WHEREAS, Franchisee acknowledges that it would take substantial capital and human resources to develop an extended-stay hotel similar to the hotels operated under the Marks and using the Business System ("Brand Hotels") and as a consequence, Franchisee desires to acquire from Franchisor the right to use the Marks and Business System and to own and operate Brand Hotels (each hereinafter referred to as a "Hotel" and collectively as the "Hotels") subject to and under the terms and conditions set forth in this Agreement; and
WHEREAS, Franchisee has had a full and adequate opportunity to read and review this Agreement and to be thoroughly advised of its terms and conditions by legal counsel or other personal advisors, and has had sufficient time to evaluate and investigate the Business System, the financial requirements, the economics of the hotel industry and the business risks associated with owning and operating hotels.
NOW, THEREFORE , in consideration of the mutual promises and covenants set forth in this Agreement and for other good and valuable consideration, the parties agree as follows:
ARTICLE 1 RIGHTS TO DEVELOPMENT BRAND HOTELS
1.1 GRANT AND OPTION.
(A) <u>Initial Grant.</u> Franchisor hereby grants to Franchisee, pursuant to the
terms and conditions of this Agreement, the non-exclusive right to obtain franchises to establish and operate one (1) Brand Hotel under the Business System in each of the ()

territories set forth under the heading "Initial Grant" on Exhibit A hereto (each territory a "Development Territory" and jointly the "Development Territories"). The location of each Brand Hotel within the Development Territories must be expressly approved in writing by Franchisor. Approval of a location is in Franchisor's sole and absolute discretion.

- (B) Registration States. Notwithstanding the foregoing, Franchisor is not offering Franchisee the right to obtain franchises to establish and operate Brand Hotels in any of the states in which Franchisor is required to be registered to sell and offer franchises (the "Registration States") unless and until Franchisor registers. Franchisee understands and agrees that nothing herein obligates Franchisor to register in any of the Registration States and that Franchisee may not establish and operate a franchise without obtaining Franchisor's approval (which approval may be granted or withheld in Franchisor's sole and absolute discretion) and without signing the applicable Franchise Agreement.
- Agreement are expressly limited to the Development Territories and are expressly subject to the terms and conditions of this Agreement. Franchisor will not franchise, license, subfranchise, develop, own or operate ("develop") Brand Hotels in the Development Territories while this Agreement is in effect and Franchisee has not signed a Franchise Agreement for a location in such Development Territory. Once a Franchise Agreement is signed for a location in a Development Territory, Franchisee's rights to such Development Territory will cease (and such territory will no longer be considered a Development Territory for purposes of this Agreement), except for such rights therein as are set forth in the Franchise Agreement. Notwithstanding the foregoing, nothing herein restricts the Franchisor or any other party to advertise one or more Brand Hotels or the Business System within the Development Territory, or to open, operate, or grant others the right to open or operate Brand Hotels with protected areas that overlap the Development Territory, for as long as no other Brand Hotels are located in Franchisee's Development Territories.
- 1.3 FRANCHISOR'S RIGHTS. Franchisor, its affiliates, or related companies with common ownership have the right: (i) outside of the Development Territory and the Protected Area of any Brand Hotel subject to a then currently effective Franchise Agreement, to establish franchised or company-owned hotels (or other businesses) or to offer, sell or distribute products or services associated with the Business System under the Marks or any other trademarks or through any distribution channel or method, all without compensation to any franchisee, including Franchisee; and (ii) within and outside of Development Territory, to establish franchised or company owned hotels (or other businesses) or to offer, sell or distribute products or services under trademarks other than the Marks, or to offer, sell or distribute any products or services through channels of distribution other than franchised or company-owned hotels operating under the Marks. Any rights not expressly granted to Franchisee herein are reserved by the Franchisor.
- 1.4 <u>CONVERSION HOTELS</u>. If Franchisor or any of its affiliates acquire any chain or system of properties, or multiple units (more than 4) of a chain or system, and such purchase includes properties in the Development Territory which Franchisor or the acquiring affiliate desires to convert to Brand Hotels (hereinafter, whether one or more, "Conversion Hotels"), Franchisor shall provide written notice to Franchisee within a reasonable time of its

intent to convert the Conversion Hotels into Brand Hotels. Such notice shall provide Franchisee with a right of first refusal to acquire such Conversion Hotels from Franchisor on the terms provided below if the sale by Franchisor of such properties to Franchisee is allowed by contract and applicable law. Subject to the foregoing, Franchisee shall have the right and option, exercisable within 90 days after receipt of such written notification, which notification shall include the proposed purchase price to provide written notice to Franchisor that Franchisee desires to purchase the Conversion Hotels and to convert all of such properties to Brand Hotels under the Business System. In the event Franchisee elects to purchase and convert the Conversion Hotels Franchisee must close on such purchase and execute the then current franchise agreement (which shall require payment of the initial franchise due under such franchise agreement) within 60 days from the date of notice to Franchisor of Franchisee's election to purchase and convert. The purchase price to be paid by Franchisee for the Conversion Hotels shall be the cash equivalent of the fair market value for each of the Conversion Hotels, as determined by an independent appraiser selected and retained by Franchisor. In the event Franchisee does not elect to purchase and convert the Conversion Hotels as provided in this paragraph, Franchisee shall have no further right or option to acquire such Conversion Hotels and Franchisor may sell such Conversion Hotels to another franchisee under the Business System or Franchisor or its affiliates may own and operate the Conversion Hotel under the Business System.

1.5 <u>PERSONAL RIGHTS</u>. Franchisee will not be entitled to franchise, subfranchise, license or sublicense other persons or entities under this Agreement. The rights, privileges and franchise granted and conveyed to Franchisee in this Agreement may not be assigned, sold or transferred by Franchisee, except as specifically provided for in this Agreement.

ARTICLE 2 TERM

2.1 TERM. The term of this Agreement commences on the Effective Date and, unless sooner terminated in accordance with the terms of this Agreement, the term of this Agreement and all rights granted hereunder (except for any rights which, by their specific terms, are intended to survive termination of this Agreement) shall expire on the earlier of: (a) the date of Franchisor's acceptance and execution of a Franchise Agreement for the last of the Hotels to be established pursuant to the Development Schedule set forth in Exhibit A, and (b) the anniversary of the Effective Date (the "Term"). No renewal rights are granted hereunder.

ARTICLE 3 DEVELOPMENT TERRITORY FEE; INITIAL FEES; DEVELOPMENT SCHEDULE

3.1 <u>DEVELOPMENT TERRITORY FEE</u>. No later than on the date this Agreement is executed by Franchisee, Franchisee will pay Franchisor a nonrefundable Development Fee equal to \$_____ (i.e., \$40,000 multiplied by the number of Brand Hotels to be developed pursuant to this Agreement (the "Development Fee"). The Development Fee is fully earned when paid to Franchisor.

- **3.2** <u>INITIAL FRANCHISE FEES</u>. The Initial Franchise Fee of \$_____ for each Franchise Agreement to be executed pursuant to this Agreement will be off-set against the Development Fee.
- 3.3 <u>DEVELOPMENT SCHEDULE</u>. Franchisee acknowledges and agrees that time is of the essence and that Hotels to be opened under Franchise Agreements entered into pursuant to this Agreement must be opened by the dates set forth in the Development Schedule and must be continuously operating in the Development Territory during the Term in accordance with the Development Schedule. For purposes of determining compliance with the Development Schedule, only Hotels actually open and continuously operating for business as of the end of each period identified in the Development Schedule will be counted toward the number of Brand Hotels required to be open and continuously operating for business.
- **3.4** REASONABLENESS OF DEVELOPMENT SCHEDULE. Franchisee represents that it has conducted its own independent investigation and analysis of the prospects for the establishment of Brand Hotels within the Development Territory, approves of the Development Schedule as being reasonable and viable, and recognizes that failure to achieve the results described in the foregoing development schedule will constitute a material breach of this Agreement.

3.5 FAILURE TO COMPLY WITH DEVELOPMENT SCHEDULE.

Franchisee's failure to comply with the Development Schedule will constitute a material breach of this Agreement by Franchisee and, in that event, Franchisor will have the right to terminate this Agreement as provided herein. If so terminated, all Franchisee rights under this Agreement, including any rights to Development Territories will terminate. Termination of this Agreement as a result solely of Franchisee's failure to meet the Development Schedule set forth above will not affect the individual Franchise Agreements signed by Franchisee for the Hotels opened and operated in the Development Territories pursuant to this Agreement prior to termination. Upon such termination: (A) Franchisee will have no further rights to open and operate additional Hotels within the Development Territories remaining to be developed; (B) Franchisee will continue to pay all required fees and to operate its Hotels opened and operating pursuant to the terms of the applicable Franchise Agreements signed by Franchisee prior to the date of the termination of this Agreement; (C) Franchisor will have the absolute right to develop the Development Territories or to contract with other parties for future development of the Development Territories; and (D) Franchisee shall not be entitled to a refund of any Development Fees or any other fees paid hereunder.

3.6 <u>COMPLIANCE WITH FRANCHISE AGREEMENTS</u>. Each Hotel to be developed pursuant to this Agreement shall be governed by the terms of the applicable Franchise Agreement executed between Franchisor and Franchisee for such Hotel including, but not limited to, the requirement that the Hotel must be constructed or renovated to conform with Franchisor's standard prototype layout and specifications for exterior design and fixtures. Franchisee shall not undertake any development activity for any Hotel for which no Franchise Agreement has been executed between Franchisor and Franchisee. Franchisee is solely responsible for locating and securing acceptable sites.

ARTICLE 4 OTHER OBLIGATIONS OF FRANCHISEE

- 4.1 <u>COMPLIANCE WITH APPLICABLE LAWS.</u> Franchisee agrees to and will, at its expense, comply with all federal, state, city, local and municipal laws, ordinances, rules and regulations which apply to the establishment and operation of the Hotels. The laws include those affecting zoning and construction, public accommodations, accessibility by persons with disabilities, health and safety, and labor. Franchisee will, at its expense, be responsible for determining the licenses and permits required by law for the Hotels, obtaining and qualifying for all such licenses and permits, and for maintaining all such licenses and permits in full force and effect.
- **4.2** EXECUTION OF FRANCHISE AGREEMENTS. For each Hotel opened, owned, and operated for business by Franchisee in the Development Territory, Franchisee (and, if applicable, Franchisee's owners and Personal Guarantors) must execute Franchisor' then-current standard Franchise Agreement (the "Franchise Agreement"). If Franchisee fails to provide Franchisor with an executed Franchise Agreement on or before the date on which construction or remodeling on the Hotel premises commences, it will be deemed a material breach of this Agreement and Franchisor will have the right to terminate this Agreement as provided herein.
- 4.3 <u>MODIFICATIONS</u> TO <u>FRANCHISE</u> <u>AGREEMENT</u>. Franchisee acknowledges that the then-current Franchise Agreement may be modified from time to time by Franchisor and that reasonable modifications and amendments to the then-current Franchise Agreement will not alter Franchisee's obligations under this Agreement.

ARTICLE 5 FRANCHISOR'S RIGHT OF TERMINATION

5.1 GROUNDS FOR TERMINATION. In addition to the other rights of termination contained in this Agreement, Franchisor will have the right and privilege to terminate this Agreement if: (A) Franchisee violates any material provision, term or condition of this Agreement, including fails to comply with the Development Schedule; (B) Franchisee fails to conform to the Business System, the standards of uniformity and quality for the goods and services or the policies and procedures promulgated by Franchisor in connection with the Business System; (C) Franchisee fails to timely pay any of its uncontested obligations or liabilities due and owing Franchisor, suppliers, banks, vendors, other creditors or any federal, state and municipal government (including, if applicable, federal and state taxes); (D) any Franchise Agreement or other agreement executed by Franchisee or its affiliates is (1) terminated by Franchiser or its affiliates or (2) wrongfully terminated by Franchisee.

In lieu of terminating this Agreement for any of the above stated defaults, Franchisor may, in its sole discretion, without giving Franchisee prior notice or the right to cure any such default, do any one or more of the following:

1. Reduce the number of development rights granted Franchisee in Article 1.1 of this Agreement;

- 2. Reduce the Development Territory; or
- 3. Terminate the territorial exclusivity granted Franchisee in the Development Territory.
- 5.2 NOTICE OF BREACH. Except as provided for in Article 5.4 and Article 5.5 of this Agreement, Franchisor will not have the right to terminate this Agreement unless and until written notice setting forth the alleged breach in detail has been given to Franchisee by Franchisor and after having been given such written notice of breach Franchisee fails to correct the alleged breach within the period of time specified by applicable law. If applicable law does not specify a time period to correct an alleged breach, then Franchisee will have 60 days after having been given such written notice to correct the alleged breach. If Franchisee fails to correct an alleged breach set forth in the written notice as provided herein within the applicable period of time, then this Agreement may be terminated by Franchisor as provided in this Agreement. For the purposes of this Agreement, an alleged breach of this Agreement by Franchisee will be deemed to be "corrected" if both Franchisor and Franchisee agree in writing that the alleged breach has been corrected.
- 5.3 NOTICE OF TERMINATION. If Franchisor has complied with the notice provisions of this Article and Franchisee has not corrected the alleged breach set forth in the written notice within the time period specified in this Article, then Franchisor will have the absolute right to terminate this Agreement by giving Franchisee written notice stating to Franchisee that this Agreement is terminated, and in that event, unless applicable law provides to the contrary, the effective date of termination of this Agreement will be the day the written notice of termination is given to Franchisee. At Franchisor's option, Franchisor may include the date of termination in the notice of default, and if the Franchisee has not cured the defaults enumerated in such notice of default by the date of termination, this Agreement will terminate without further action by the Franchisor upon such date of termination.
- 5.4 GROUNDS FOR IMMEDIATE TERMINATION. Franchisee shall be deemed in default under this Agreement, and all rights granted herein shall automatically terminate without notice, if Franchise is adjudicated bankrupt, becomes insolvent, suffers temporary or permanent court-appointed receivership of substantially all of Franchisee's property, makes a general assignment for the benefit of creditors or suffers the filing of a voluntary or involuntary bankruptcy petition which is not dismissed within 90 days after filing. Franchisor will have the absolute right and privilege, unless prohibited by applicable law, to immediately terminate this Agreement if: (A) Franchisee is involved in any act or conduct which materially impairs the goodwill associated with the Marks or the Business System; (B) Franchisee willfully and materially falsifies any report, statement or other written data furnished to Franchisor; (C) a suit to foreclose any lien or mortgage against a Hotel's premises is instituted against Franchisee and not dismissed within 30 days; (D) Franchisee or any of its partners, Directors, officers or majority equity owners are convicted of, or plead guilty to or no contest to a charge of violating any law relating to Franchisee's business, or any felony; (E) Franchisee ceases to do business or otherwise abandons the business; (F) the nature of Franchisee's breach makes it not curable; or (G) Franchisee fails to comply with 1 or more requirements of this Agreement 3 times in any one 12-month period even if the defaults are cured.

- 5.5 NOTICE OF IMMEDIATE TERMINATION. If this Agreement is terminated by Franchisor pursuant to Article 5.4 above, other than in the event of automatic termination pursuant to Article 5.4, Franchisor will give Franchisee written notice that this Agreement is terminated, and in that event, unless applicable law provides to the contrary, the effective date of termination of this Agreement will be the day the written notice of termination is given to Franchisee.
- 5.6 <u>DAMAGES</u>. In the event this Agreement is terminated by Franchisor pursuant to this Article, or if Franchisee breaches this Agreement by a wrongful termination of this Agreement, then Franchisor will be entitled to seek recovery from Franchisee for all of the damages that Franchisor has sustained and will sustain in the future as a result of Franchisee's breach of this Agreement, which will include damages based upon the Initial Franchise Fees, Royalty Fees, Marketing Fees and other fees that would have been payable by Franchisee pursuant to this Agreement.
- **5.7** OTHER REMEDIES. Nothing in this Article or this Agreement will preclude Franchisor from seeking other damages or remedies under common law, state or federal laws or this Agreement against Franchisee including, but not limited to, attorneys' fees, punitive damages and injunctive relief.

ARTICLE 6 FRANCHISEE'S TERMINATION RIGHTS

- **6.1** CONDITIONS OF BREACH. Franchisee will have the right and privilege to terminate this Agreement, as provided herein, if Franchisor violates any material provision, term or condition of this Agreement.
- Agreement or to commence an action or lawsuit against Franchisor for breach of this Agreement, injunctive relief, violation of any state, federal or local law, violation of common law (including allegations of fraud and misrepresentation), rescission, general or punitive damages, or termination, unless and until written notice by personal service, prepaid registered or certified mail, or nationally recognized overnight delivery service setting forth the alleged breach or violation in detail has been delivered to Franchisor by Franchisee; and Franchisor fails to correct the alleged breach or violation within 60 days after receipt of the written notice by personal service or prepaid registered or certified mail or nationally recognized overnight delivery service. If Franchisor fails to correct the alleged breach or violation as provided herein within 60 days after receiving written notice, then this Agreement may be terminated by Franchisee provided that Franchisee complies with the post-term obligations described in Article 7.
- 6.3 NOTICE OF TERMINATION. If Franchisee has complied with the notice provisions of this Article and if Franchisor has not corrected the alleged breach set forth in the written notice within the time period specified in this Agreement, then Franchisee will have the right to terminate this Agreement by giving Franchisor written notice by personal service or prepaid registered or certified mail or nationally recognized overnight delivery service that this

Agreement is terminated, and in that event the effective date of termination of this Agreement will be the day the written notice is received by Franchisor.

6.4 <u>WAIVER</u>. Franchisee must give Franchisor immediate written notice of an alleged breach or violation of this Agreement after Franchisee has knowledge of, believes, determines, or is of the opinion that there has been an alleged breach or violation of this Agreement by Franchisor. If Franchisee fails to give written notice to Franchisor of an alleged breach or violation of this Agreement within 1 year from the date that Franchisee has knowledge of, believes, determines or is of the opinion that there has been an alleged breach or violation by Franchisor, then the alleged breach or violation will be deemed to be condoned, approved and waived by Franchisee, the alleged breach or violation will not be deemed to be a breach or violation of this Agreement by Franchisor and Franchisee will be barred from commencing any action against Franchisor for that alleged breach or violation.

ARTICLE 7 FRANCHISEE'S RIGHTS AND OBLIGATIONS UPON TERMINATION

- 7.1 OBLIGATIONS UPON TERMINATION. In the event this Agreement is terminated for any reason, then Franchisee will: (A) within 5 days after termination, pay all amounts due and owing to Franchisor under this Agreement or any other contract, promissory note or other obligation payable by Franchisee to Franchisor; and (B) comply with all other applicable provisions of this Agreement, including those provisions with obligations that continue beyond the termination of this Agreement.
- **7.2 REVERSION OF RIGHTS**. Upon termination of this Agreement for any reason, all rights to open and operate additional Brand Hotels in the Development Territory and all other rights granted to Franchisee pursuant to this Agreement will automatically revert to Franchisor, and Franchisor will have the right to develop the Development Territory or to contract with another franchisee for the future development of the Development Territory.
- 7.3 FRANCHISE AGREEMENTS NOT AFFECTED. Franchisee will continue to operate the Hotels owned and operated by Franchisee in the Development Territory pursuant to the terms of the applicable Franchise Agreements signed by Franchisee and Franchisor prior to the termination of this Agreement, unless otherwise terminated on their own grounds. The rights and obligations of Franchisee and Franchisor with respect to Franchisee's Brand Hotels in the Development Territory will be governed by the terms of the applicable Franchise Agreements.

ARTICLE 8 ASSIGNMENT

8.1 ASSIGNMENT BY FRANCHISOR. This Agreement may be unilaterally assigned and transferred by Franchisor without Franchisee's approval or consent, and will inure to the benefit of Franchisor' successors and assigns. The assignee will be required to fulfill Franchisor' obligations under this Agreement. Franchisee waives any and all rights that it may have or allege against Franchisor resulting from the assignment of this Agreement to an entity which owns, operates, franchises, or has an interest in a hotel business which includes hotels in the Development Territory. Franchisor and its owners also have the right to sell, pledge, assign,

transfer, trade or otherwise dispose of their ownership interest in Franchisor without the approval or consent of Franchisee.

- ENTITY. If Franchisee is an individual or a partnership, this Agreement may be transferred or assigned by Franchisee to an entity which is owned or controlled (ownership of at least 51% of the ownership interest) by Franchisee, provided that Franchisee and all of the Principal Owners of the assignee entity sign or have signed the Personal Guaranty in the form attached to this Agreement as Exhibit B, and further provided that Franchisee furnishes prior written proof to Franchisor substantiating that the entity will be financially able to perform all of the terms and conditions of this Agreement. Franchisee will give Franchisor 15 days' written notice prior to the proposed date of assignment or transfer of this Agreement to an owned or controlled entity of Franchisee; however, the transfer or assignment of this Agreement will not be valid or effective until Franchisor has received the legal documents which its legal counsel deems necessary to properly and legally document the transfer or assignment of this Agreement to the corporation as provided herein.
- 8.3 ASSIGNMENT BY INDIVIDUAL FRANCHISEE IN THE EVENT OF DEATH OR PERMANENT DISABILITY. If Franchisee is an individual, then this Agreement may be assigned, transferred or bequeathed by Franchisee to any designated person or beneficiary upon his or her death or permanent disability. However, the assignment of this Agreement to the transferee, assignee or beneficiary of Franchisee will be subject to the applicable provisions of Article 8.4, and will not be valid or effective until Franchisor has received the properly executed legal documents which its legal counsel deems necessary to properly and legally document the transfer, assignment or bequest of this Agreement, and until each of the Principal Owners of the transferee, assignee or beneficiary agrees to be unconditionally bound by the terms and conditions of this Agreement and to personally guarantee the performance of Franchisee's obligations under this Agreement, and signs the Personal Guaranty, as applicable, in the form attached as Exhibit B to this Agreement.
- 8.4 APPROVAL OF TRANSFER; CONDITIONS FOR APPROVAL. Franchisee (and its Principal Owners) will not (whether voluntary or involuntary), transfer, assign or otherwise dispose of, in one or more transactions, this Agreement or any interest in Franchisee unless Franchisee obtains Franchisor's prior written consent. Franchisor will not unreasonably withhold its consent to any transfer, assign or otherwise dispose of this Agreement, provided that Franchisee and the transferee franchisee comply with the following conditions: (A) all of Franchisee's monetary obligations due to Franchisor have been paid in full, and Franchisee is not otherwise in default under this Agreement; (B) Franchisee (and each Principal Owner, if applicable) has executed a written agreement in a form satisfactory to Franchisor in which Franchisee and each Principal Owner agrees to observe all applicable post-term obligations and covenants contained in this Agreement; (C) the transferee franchisee will have demonstrated to Franchisor's satisfaction that he, she or it meets Franchisor's managerial, financial, and business standards for new area development franchisees, possesses a good business reputation and credit rating, and possesses the aptitude and ability to operate the hotel business franchised hereunder (as may be evidenced by prior related business experience or otherwise); (D) Franchisee has paid the transfer fee required under Article 8.5; (E) the transferee franchisee has executed Franchisor's then-current standard Multi-Unit Development Agreement and each Principal

Owner of the transferee franchisee as applicable, has executed the Personal Guaranty attached to this Agreement, all as required by Franchisor; and (F) Franchisee (and each Principal Owner, if applicable) sign a general release, in a form and substance satisfactory to us, of any and all claims against us and our affiliates, officers, directors, employees and agents, except to the extent limited or prohibited by applicable law.

8.5 TRANSFER FEE. If, pursuant to the terms of this Article, this Agreement is assigned, transferred or bequeathed to another person or entity, or if Franchisee's owners transfer over 50% of their ownership interest to another person or entity, then Franchisee will pay Franchisor a transfer fee of \$5,000. This fee is to cover the costs incurred by Franchisor for attorneys' fees, accountants' fees, out-of-pocket expenses, administrative expenses, and the time of its employees and officers.

ARTICLE 9 DISPUTE RESOLUTION

- 9.1 MEDIATION. Except as provided in Article 9.3, all disputes and controversies between Franchisor, its officers, directors, partners, owners and affiliates, and Franchisee, its officers, directors, partners, owners and Personal Guarantors, arising under, as a result of, or in connection with this Agreement (each, a "Dispute") will be subject to non-binding mediation. The mediator will be appointed in accordance with the Commercial Mediation Rules of the American Arbitration Association unless Franchisor and Franchisee agree on a mediator in writing within 10 days after either party gives written notice of mediation. All mediation hearings will take place exclusively in such city and state where the Franchisor has its headquarters at the time, and will be held within 20 days after a request for mediator has been made. The mediation shall be private, voluntary, and nonbinding. The hearing will be informal and the mediator will have the right to hear and review all testimony and evidence presented by either Franchisor or Franchisee. Any party may withdraw from the mediation at any time before signing a settlement agreement upon written notice to each other party and to the mediator. The cost of the mediator will be paid equally by Franchisor and Franchisee. The mediator shall be disqualified as a witness, consultant, expert or counsel for either party with respect to the matters in Dispute and any related matters. Unless the parties agree otherwise, the entire mediation process shall be confidential and without prejudice. The parties and the mediator shall not disclose any information, documents, statements, positions, or terms of settlement. Nothing said or done or provided by the parties in the course of mediation shall be reported or recorded or, except as ordered by a court of competent jurisdiction, placed in any legal proceeding or construed for any purpose as an admission against interest. Nevertheless, evidence otherwise discoverable or admissible is not excluded from discovery or admission as a result of its use in mediation. The commencement of any mediation shall not act to prevent either party from instituting or proceeding with any action which may be the subject of the Dispute.
- **9.2** ARBITRATION. If a Dispute is not resolved in accordance with the provisions for mediation set forth in Article 9.1, then the Dispute shall be submitted to binding arbitration in accordance with the Commercial Rules and Regulations of the American Arbitration Association. All arbitration hearings will take place exclusively in such city and state where the Franchisor has its headquarters at the time. The authority of the arbitrator will be limited to making a finding, judgment, decision and award relating to the interpretation of or adherence to

the written provisions of this Agreement. The arbitrator will not have the authority or right to add to, delete, amend or modify in any manner the terms, conditions and provisions of this Agreement, nor have the right or authority to award punitive damages to Franchisor or Franchisee or their officers, directors, or owners, and Franchisee's Personal Guarantors. Franchisor and Franchisee and their respective officers, directors, owners, and Franchisee's Personal Guarantors expressly waive their rights to plead or seek punitive damages. All findings, judgments, decisions and awards by the arbitrator will be final and binding on Franchisor, Franchisee, and their respective officers, directors, owners, and Franchisee's Personal Guarantors. The written decision of the arbitrator will be deemed to be an order, judgment and decree and may be entered as such in any court of competent jurisdiction by either party.

- 9.3 <u>DISPUTES NOT SUBJECT TO MEDIATION OR ARBITRATION.</u> The following disputes between Franchisor and Franchisee will not be subject to mediation or arbitration:
 - (A) the payment of fees by Franchisee to Franchisor;
 - (B) any dispute involving the Marks; and
 - (C) any dispute involving immediate termination of this Agreement by Franchisor pursuant to Article 5.5 of this Agreement.
- **INJUNCTIVE RELIEF**. Notwithstanding any provisions to the contrary, Franchisee recognizes that its Hotels will be part of a larger number of hotels identified by the Marks and Business System and the failure on the part of a single franchisee to comply with the terms and conditions of this Agreement could cause irreparable damage to Franchisor and/or to other Brand Hotel franchisees. Therefore, it is mutually agreed that in the event of a breach of threatened breach of any of the terms of this Agreement by Franchisee, then Franchisor will be entitled to an injunction restraining such breach and/or to a decree of specific performance, without showing or proving actual damage, together with recovery of reasonable attorneys' fees and other costs incurred in obtaining said equitable relief, until such time as a final and binding determination is made by the arbitrator. Similarly, it is mutually agreed that in the event of a breach or threatened breach of any of the terms of this Agreement by Franchisor, Franchisee will be entitled to an injunction restraining such breach and/or to a decree of specific performance, without showing or proving any actual damage, together with recovery of reasonable attorneys' fees and other costs incurred in obtaining said equitable relief, until such time as a final and binding determination is made by the arbitrator. The foregoing equitable remedies shall be in addition to, and not in lieu of, all other remedies or rights which the parties might otherwise have by virtue of any breach of this Agreement by the other party.
- 9.5 <u>NO COLLATERAL ESTOPPEL OR CLASS ACTIONS</u>. Except as provided herein, all arbitration findings and awards expressly made by the arbitrator will be final and binding on Franchisor and Franchisee and their officers, Directors, or owners, and Franchisee's Personal Guarantors; however, such arbitration findings and awards may not be used to collaterally stop either party from raising any like or similar issues, claims or defenses in any other or subsequent arbitration, litigation, court hearing or other proceeding involving third parties or other franchisees. Any disagreement between Franchisee and Franchisor (and

Franchisor's affiliates, officers, directors, or owners) will be considered unique as to its facts and must not be brought as a class action, and Franchisee waives any right to proceed against Franchisor (and Franchisor's affiliates, officers, directors, or owners) by way of class action, or by way of a multi-plaintiff, consolidated or collective action. No party except Franchisor, Franchisee and their officers, Directors, owners or partners and Franchisee's Personal Guarantors will have the right to join in any arbitration proceeding arising under this Agreement, and, therefore, the arbitrator will not be authorized to permit or approve class actions or to permit any person or entity that is not a party to this Agreement to be involved in or to participate in any arbitration hearings conducted pursuant to this Agreement.

Any disagreement between Franchisee and Franchisor (and Franchisor's affiliates, officers, directors, or owners) will be considered unique as to its facts and must not be brought as a class action, and Franchisee waives any right to proceed against Franchisor (and Franchisor's affiliates, officers, directors, or owners) by way of class action, or by way of a multi-plaintiff, consolidated or collective action

- 9.6 <u>DAMAGE LIMITATIONS</u>. NO PUNITIVE OR EXEMPLARY DAMAGES SHALL BE AWARDED AGAINST EITHER FRANCHISOR OR FRANCHISEE OR ANY AFFILIATES OF EITHER OF THEM, IN ANY PROCEEDING, AND ALL CLAIMS TO SUCH DAMAGES ARE HEREBY WAIVED. FURTHER, IN NO EVENT WILL FRANCHISOR BE LIABLE TO FRANCHISEE FOR PROSPECTIVE PROFITS OR SPECIAL, INDIRECT, PUNITIVE OR CONSEQUENTIAL DAMAGES FOR ANY CONDUCT ARISING OUT OF THIS AGREEMENT OR FRANCHISOR'S RELATIONSHIP WITH FRANCHISEE.
- 9.7 <u>BUSINESS JUDGMENT</u>. The parties hereto recognize, and any mediator, arbitrator, or judge is affirmatively advised, that certain provisions of this Agreement describe the right of Franchisor to take (or refrain from taking) certain actions in the exercise of its discretion based on its assessment of the overall best interest of the network and/or franchise program. Where such discretion has been exercised, and is supported by the business judgment of Franchisor, neither a mediator, nor an arbitrator, nor a judge shall substitute his or her judgment for the judgment so exercised by Franchisor.
- 9.8 <u>CONFIDENTIALITY</u>. All evidence, testimony, records, documents, findings, decisions, judgments and awards pertaining to any arbitration hearing between Franchisor and Franchisee will be secret and confidential in all respects. Franchisor and Franchisee will not disclose the decision or award of the arbitrator and will not disclose any evidence, testimony, records, documents, findings, orders, or other matters from the arbitration hearing to any person or entity, except as required by law.
- 9.9 <u>ENFORCEMENT</u>. Any claim arising out of or relating to this Agreement, the relationship of the parties, Franchisor's operation of the stayAPT® Suites system, or Franchisee's development and/or operation of any Hotels will be barred unless filed before the expiration of the earlier of: (1) the time period for bringing an action under any applicable state or federal statute of limitations; (2) 1 year after the date upon which a party discovered, or should have discovered, the facts giving rise to an alleged claim; or (3) 2 years after the first act or omission giving rise to an alleged claim. Claims of Franchisor for the underreporting of Gross Room

Revenues, for indemnification, or for claims related to Franchisor's rights under any of the Marks shall be subject only to the applicable state or federal statute of limitations.

9.10 ENFORCEMENT COSTS. In the event Franchisor incurs legal fees or costs or other expenses to enforce any obligation of Franchisee hereunder, or to defend against any claim, demand, action or proceeding by reason of Franchisee's failure to perform or observe any obligation imposed upon Franchisee by this Agreement, then Franchisor shall be entitled to recover from Franchisee the amount of all legal fees, costs and expenses, including reasonable attorneys' fees, whether incurred prior to, or in preparation for or contemplation of the filing of any claim, demand, action, or proceeding to enforce any obligation of Franchisee hereunder or thereafter or otherwise.

ARTICLE 10 GENERAL PROVISIONS

- 10.1 GOVERNING LAW. Except to the extent governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. §1051 et seq.), or modified by the Federal Arbitration Act of the United States of America (9 U.S.C. § 1 et seq.), this Agreement and the relationship between Franchisor and Franchisee will be governed by the laws of the state of North Carolina, without reference to its choice of law provisions. The provisions of this Agreement which conflict with or are inconsistent with applicable governing law will be superseded and/or modified by such applicable law only to the extent such provisions are inconsistent. All other provisions of this Agreement will be enforceable as originally made and entered into upon the execution of this Agreement by Franchisee and Franchisor.
- Agreement will be interpreted and enforced as if all completely invalid or unenforceable provisions were not contained herein and partially valid and enforceable provisions will be enforced to the extent valid and enforceable. If any applicable law or rule of any jurisdiction requires a greater prior notice of the termination of or refusal to renew this Agreement than is required hereunder or the taking of some other action not required hereunder, or if under any applicable law or rule of any jurisdiction, any provision of this Agreement or any specification, standard or operating procedure prescribed by the Franchisor is invalid or unenforceable, the prior notice or other action required by such law or rule will be substituted for the notice requirements hereof, or such invalid or unenforceable provision, specification, standard or operating procedure will be modified to the extent required to be valid and enforceable. Such modifications to this Agreement will be effective only in such jurisdiction and will be enforced as originally made and entered into in all other jurisdictions.
- 10.3 <u>WAIVER</u>. Franchisor and Franchisee may, by written instrument signed by Franchisor and Franchisee, waive any obligation of or restriction upon the other under this Agreement. Acceptance by Franchisor of any payment by Franchisee and the failure, refusal or neglect of Franchisor to exercise any right under this Agreement or to insist upon full compliance by Franchisee of its obligations hereunder including, without limitation, any mandatory specification, standard or operating procedure, will not constitute a waiver by Franchisor of any provision of this Agreement.

- 10.4 <u>CUMULATIVE RIGHTS</u>. The rights of Franchisor hereunder are cumulative and no exercise or enforcement by Franchisor of any right or remedy hereunder will preclude the exercise or enforcement by Franchisor of any other right or remedy hereunder or which Franchisor is entitled by law to enforce.
- **10.5 BINDING AGREEMENT.** This Agreement is binding upon the parties hereto and their respective executors, administrators, heirs, assigns and successors in interest.
- entire agreement executed by the parties contemporaneously hereto, supersedes and terminates all prior agreements, either oral or in writing, between the parties involving the franchise relationship and therefore, representations, inducements, promises or agreements between the parties not contained in this Agreement or not in writing signed by Franchisor and Franchisee will not be enforceable. The preambles are a part of this Agreement, which, together with any other documents or agreement executed by the parties contemporaneously hereto, constitutes the entire agreement of the parties, and there are no other oral or written understandings or agreements between Franchisor and Franchisee relating to the subject matter of this Agreement. Nothing in the Agreement is intended to disclaim the representations Franchisor made in the Franchise Disclosure Document that Franchisor furnished to Franchisee. No change, modification, amendment or waiver of any of the provisions hereof, including by custom or usage of trade or course of dealing or performance, shall be effective and binding upon either party unless it is in writing, specifically identifies as an amendment hereto, and signed by the party to be charged.
- **10.7 JOINT AND SEVERAL LIABILITY.** If Franchisee consists of more than 1 individual, then the liability of all such individuals under this Agreement will be deemed to be joint and several.
- 10.8 <u>NO ORAL MODIFICATIONS</u>. No oral modification, change, addition, rescission, release, amendment or waiver of this Agreement and no approval, consent or authorization required by any provision of this Agreement may be made except by a written agreement signed by Franchisor.
- and do not define, limit or construe the contents of such Articles. The term "Franchisee" as used herein is applicable to one or more persons, a corporation or a partnership, as the case may be, and the singular usage includes the plural, the masculine usage includes the neuter and the feminine, and the neuter usage includes the masculine and the feminine. References to "Franchisee," "assignee" and "transferee" which are applicable to an individual or individuals will mean the principal owner or owners of the equity or operating control of Franchisee or any such assignee or transferee if Franchisee or such assignee or transferee is a corporation or partnership.
- 10.10 <u>NOTICES</u>. Unless expressly otherwise stated herein, all notices to required or permitted under this Agreement will be in writing and will be made by personal service, certified or registered mail, postage prepaid, return receipt requested, sent via a nationally recognized overnight delivery service, or sent by e-mail (with a confirming copy mailed or sent by overnight

delivery service) at the following addresses designated by the parties or such other address as has been designated by written notice to the other party:

Notices to Franchisor:	
Attn:10801 Monroe Road, Su Matthews, NC 28105	
Email:	_@stayAPT.com
Notices to Franchisee: Attn.:	
Fmail:	

Any notice shall be deemed to have been given at the earlier of actual receipt or 3 business days after mailing by certified or registered mail, or 1 business day after sending by e-mail or overnight delivery service.

- 10.11 PRINCIPAL OWNER. "Principal Owner" means any person or entity who directly or indirectly owns a 10% or greater interest in Franchisee. If any entity other than a partnership is a Principal Owner, a "Principal Owner" also will mean an owner of a 10% or greater interest in such entity. If a partnership is a Principal Owner, a "Principal Owner" also will mean each general partner of such partnership and, if such general partner is an entity, each owner of a 10% or greater interest in such general partner. If Franchisee is one or more individuals, each individual will be deemed a Principal Owner of Franchisee. Franchisee must have at least one Principal Owner. Franchisee's Principal Owner(s) are identified on the Management Addendum attached as Exhibit C. As used in this Agreement, any reference to Principal Owner includes all Principal Owners.
- 10.12 PERSONAL GUARANTY. All of Franchisee's Principal Owners (if Franchisee is an entity) will sign the Personal Guaranty in the form attached to this Agreement as Exhibit B (the "Personal Guaranty"). Any person or entity that at any time after the effective date of this Agreement becomes a Principal Owner of Franchisee will, as a condition of becoming a Principal Owner, sign the Personal Guaranty. In addition, any other person Franchisor designates, including without limitation any person who co-signed a loan or was involved in obtaining financing for this Agreement, must also sign the Personal Guaranty. Any person or entity that signs the Personal Guaranty is referred to herein as a "Personal Guarantor." In connection with the Personal Guaranty, Franchisor has the right to require Franchisee to provide Franchisor with additional information, including without limitation loan documents.
- **10.13** TERMS DEFINED IN FRANCHISE AGREEMENT. Capitalized terms used but not defined in this Agreement will, if defined in the Franchise Agreement, have the meanings ascribed to such terms in the Franchise Agreement.

10.14 PUBLICITY AND PRESS RELEASE. Franchisee agrees to use its best efforts to cooperate with and assist Franchisor in publicizing their relationship upon reasonable request of Franchisor. Franchisee hereby authorizes Franchisor to promote and market the existence of the relationship as Franchisor deems appropriate.

ARTICLE 11 ACKNOWLEDGMENTS

- **BUSINESS** RISK; NO FINANCIAL PROJECTIONS. acknowledges that it has conducted an independent investigation of the Hotel business licensed hereunder, and recognizes that the business venture contemplated by this Agreement involves business and economic risks and that the success of its Hotels will be primarily dependent upon the personal efforts of Franchisee, its management, and its employees. Franchisor expressly disclaims the making of, and Franchisee acknowledges that it has not received, any estimates, projections, warranties, or guaranties, expressed or implied, regarding potential Gross Room Revenues, income, profits, earnings, expenses or the financial or business success of the Hotel that were not contained in Item 19 of the Franchise Disclosure Document received by Franchisee, or if such financial information was received, then Franchisee acknowledges that the parties' entering into this Agreement is a transaction exempt under applicable state and federal franchise disclosure laws and regulations. Franchisee further acknowledges that if it had received any such representations and no exemption applied, as described in the preceding sentence, it would not have executed this Agreement and Franchisee would have: (A) promptly notified the President of Franchisor in writing of the person or persons making such representations; and (B) provided to Franchisor a specific written statement detailing the representations made.
- 11.2 NO INCOME OR REFUND WARRANTIES. Franchisee acknowledges that Franchisor does not warrant or guarantee that: (A) Franchisee will derive income or profits from the Hotels; or (B) Franchisor will refund all or any part of the Development Territory Fee, the Initial Franchise Fee or the price paid for the Hotels or repurchase any of the furniture, fixtures, products, equipment, supplies or chattels supplied by Franchisor or an approved supplier if Franchisee is unsatisfied with its Hotel.
- 11.3 RECOMMENDATION TO OBTAIN LEGAL COUNSEL. Franchisee acknowledges that Franchisor has strongly recommended that Franchisee should retain legal counsel to review this Agreement and Franchisor's Franchise Disclosure Document and to advise Franchisee as to the terms and conditions of this Agreement and the potential economic benefits and risks of loss relating to this Agreement and the Hotel business. Further, Franchisee acknowledges that Franchisee has had an opportunity to obtain legal advice regarding, and currently complies with, all applicable legal requirements that prohibit unfair, fraudulent or corrupt business practices, including U.S. and other legal requirements that are designed to combat terrorism and terrorist activities. In addition, Franchisee represents that neither Franchisee nor any holder of an ownership interest in Franchisee is named as a "specially designated national" or "blocked person" as designated by the United States Department of the Treasury's Office of Foreign Assets Control.
- 11.4 <u>OTHER FRANCHISEES</u>. Franchisee acknowledges that other area franchisees of Franchisor have or will be granted multi-unit or area development agreements at different

times and in different situations, and further acknowledges that the terms and conditions of such multi-unit or area development agreements may vary substantially in form and in substance from those contained in this Agreement.

FRANCHISE DISCLOSURE DOCUMENT. Unless exempt from applicable federal and state franchise disclosure laws, except for fill-in-the-blank provisions and changes made as a result of negotiations that Franchisee initiated, Franchisee acknowledges that it received a copy of this Agreement with all material blanks fully completed at least 7 calendar days prior to the date that this Agreement was executed by Franchisee. Franchisee further acknowledges that it received a copy of Franchisor's Franchise Disclosure Document at least 14 calendar days prior to the date on which this Agreement was executed by Franchisee.

11.6 POTENTIAL INCREASES IN INVESTMENT REQUIREMENTS. Franchisee recognizes and acknowledges that this Agreement requires it to open additional Hotels in the future pursuant to the Development Schedule set forth in Section 3. Franchisee further acknowledges that the estimated expenses and investment requirements set forth in Items 6 and 7 of Franchisor's Franchise Disclosure Document are subject to increase over time, and that future Hotels opened and operated by Franchisee may involve greater initial investment and operating capital requirements than those stated in the Franchise Disclosure Document provided to Franchisee prior to the execution of this Agreement.

{Signatures appear on the following page}

IN WITNESS WHEREOF, Franchisor and Franchisee have respectively signed and sealed this Franchise Agreement as of the day and year first above written.

	"Franchisor"		
	LG AS FRANCI	HISOR LLC	
	By:		_
	Name: Its:		
	"Franchisee"		
	By:		-
	Name: Its:		_
	OWNER'S ACKNOWLEDGI	FMENT	
Acknowledgment, not as	an owner of Franchisee. Eac a party, but only to the extent r nent to be bound by, the specific	h following party is signing necessary to indicate its, his or	her
NAME	SIGNATURE	PERCENTAGE OF OWNERSHIP	
	_	_	%
	_		%
			%
			%

EXHIBIT A

DEVELOPMENT TERRITORIES AND DEVELOPMENT SCHEDULE

INITIAL GRANT:

Each Development Territory consists of the USPS ZIP codes listed below:

Development Territory	USPS ZIP Codes

DEVELOPMENT SCHEDULE:

Period End Date	New/additiona I Brand Hotels for Which Franchise Agreements are Signed	New/additional Brand Hotels for Which Construction has Begun	New/additional Brand Hotels Required to be Opened During the Period	Cumulative Number Brand Hotels Required to be Open and Continuously Operating for Business in the Development Territory at the end of the Period
1 st anniversary of the Effective Date				
2 nd anniversary of the Effective Date				
3 rd anniversary of the Effective Date				

EXHIBIT B

PERSONAL GUARANTY AND AGREEMENT TO BE BOUND PERSONALLY BY THE TERMS AND CONDITIONS OF THE AREA DEVELOPMENT AGREEMENT

In consideration of the execution of this Multi-Unit Development Agreement by the Franchisor, and for other good and valuable consideration, the undersigned, for themselves, their heirs, successors, and assigns, do jointly, individually and severally hereby become surety and guaranty for the payment of all amounts and the performance of the covenants, terms and conditions in this Multi-Unit Development Agreement, to be paid, kept and performed by Franchisee.

Further, the undersigned, individually and jointly, hereby agree to be personally bound by each and every condition and term contained in this Multi-Unit Development Agreement and agree that this Personal Guaranty should be construed as though the undersigned and each of them executed a Multi-Unit Development Agreement containing the identical terms and conditions of this Multi-Unit Development Agreement, including without limitation the dispute resolution, confidentiality, and noncompete provisions.

If any default should at any time be made therein by Franchisee, then the undersigned, their heirs, successors and assigns, do hereby, individually, jointly and severally, promise and agree to pay to Franchisor all monies due and payable to the Franchisor under the terms and conditions of this Multi-Unit Development Agreement.

In addition, if Franchisee fails to comply with any other terms and conditions of this Multi-Unit Development Agreement, then the undersigned, their heirs, successors and assigns, do hereby, individually, jointly and severally, promise and agree to comply with the terms and conditions of this Multi-Unit Development Agreement for and on behalf of Franchisee.

In addition, should Franchisee at any time be in default on any obligation to pay monies to the Franchisor or any subsidiary or related company of the Franchisor, whether for products, supplies, furniture, fixtures, equipment or other goods purchased by Franchisee from the Franchisor, or any subsidiary or affiliate of the Franchisor, or for any other indebtedness of Franchisee to the Franchisor, or any subsidiary or affiliate of the Franchisor, then the undersigned, their heirs, successors and assigns, do hereby individually, jointly and severally promise and agree to pay all such monies due and payable from Franchisee to the Franchisor, or any subsidiary or affiliate of the Franchisor.

Each of the undersigned waives: (i) notice of demand for payment of any indebtedness or nonperformance of any obligations hereby guaranteed; (ii) protest and notice of default to any party respecting the indebtedness or nonperformance of any obligations hereby guaranteed; and (iii) any right the undersigned has to require that an action be brought against Franchisee or any other person as a condition of liability.

It is further understood and agreed by the undersig conditions of this Personal Guaranty will inure to the bene Franchisor.	<u> </u>	
In the Presence of:	PERSONAL GUARANTORS:	
	Individually	
	Address	
	City State Zip Code	
	Individually	
	Address	
	City State Zip Code	
To be signed by spouses of guarantors residing in Alaska, Nevada, New Mexico, Texas, Washington, and Wisconsin.	Arizona, California, Idaho, Louisiana,	
CONSENT OF SPOUSE OF G	UARANTOR	
The undersigned spouse of hereby consents to the execution of the foregoing Guaranty by their spouse and agrees to be bound thereby to the extent of his/her interest in any assets or property of their spouse and further agrees that their community assets shall be bound thereby. This consent shall not be construed as an agreement by the undersigned spouse that such spouse's separate property is subject to the claims of franchisor arising out of enforcement of this Guaranty.		
By:		

EXHIBIT C

MANAGEMENT ADDENDUM

is effective as of this day of	Name	Percentage of Interest	Home Address
diately notify Franchisor in writing of any charn and, at Franchisor's request, prepare and signation.			
diately notify Franchisor in writing of any charn and, at Franchisor's request, prepare and significant. The is effective as of this day of		<u> </u>	
n and, at Franchisor's request, prepare and signation. It is effective as of this day of			
"Franchisor"		. 1 11 . 1: 4 . 4.0	Franchisor in writing of any chan
1 fanemsor	the information contained in ew Addendum containing the 3. Effective Date.	n this Addendum and, at Francorrect information.	nchisor's request, prepare and sign
LG AS FRANCHISOR LLC	the information contained in ew Addendum containing the 3. Effective Date.	n this Addendum and, at Francorrect information.	nchisor's request, prepare and signals as of this day of
LG .	n the information contained in new Addendum containing the	n this Addendum and, a correct information.	nt Fran
B;	e information contained in Addendum containing the 3. Effective Date	n this Addendum and, at Franchiscon this Addendum is effective a "Franchiscon LG AS FF	as of this day of

EXHIBIT D STATE ADDENDA

[Insert, if applicable]

EXHIBIT J

Data Transfer Agreement



Data Transfer Agreement and Disclaimer

In consideration of the transfer of certain data, including electronic and/or digital information, by Woolpert, Inc. to the Recipient identified below, for the Project and Purpose identified below, the Recipient agrees and acknowledges the following:

Recipient:	
Project:	
Limited Purpose	

- 1. Any and all plans, drawings, other documents, including electronic/digital files, data, meta data prepared or furnished at any time by Woolpert (referred to herein as "data") are instruments of service, and Woolpert shall retain all ownership and property interests.
- 2. The transfer of data by Woolpert for the express and limited Purpose described above does not grant to Recipient an assignment of any right or interest in these data, or in any software or process utilized to develop these data.
- 3. Woolpert makes no representation as to the compatibility of these data with any hardware or software.
- 4. These data and any information contained in any medium including any electronic or digital file are proprietary, and Recipient will not use or copy the information for use on any other project or for any other purpose. Though the information may not bear a copyright symbol, Recipient understands that Woolpert is the author and/or owner of these data and agrees that Woolpert will be entitled to statutory and other damages, including attorneys' fees for any use of these data, except for the limited Purpose described herein above.
- 5. The Recipient shall not make or permit to be made any copies or any modification to these data, and shall not be permitted to assign or transfer any rights Recipient may have under this Agreement, including transfer or assignment of notes, payments or damages related to this Agreement or the data.
- 6. The Recipient shall not rely upon, attribute, reference and/or publish, in any form or manner, Woolpert's name, seal, or any identifying indicia thereof, which may confuse the market, channel, industry, and/or jurisdiction without Woolpert's express written consent.
- 7. The transfer of these data shall not in any way obviate the architect, engineer or designer of record's requirement, or any contractor's responsibility, to properly review, administer and exercise responsible control and charge over these data and to fully coordinate these data, as may be required for Recipient's intended or actual use or obligation.
- 8. The Recipient may make and retain copies for information and reference in connection with the Project identified herein above for the limited Purpose identified herein above; however, such documents are not intended or represented to be suitable for use by the Recipient.
- 9. These data have been developed based on Woolpert's standard formatting, layering, drafting and design practice using technical processes and/or software maintained by Woolpert, which may or may not be compatible with standards or software used by Recipient. These data may not always show information to scale.

- 10. The data is being provided as a "convenience" to the Recipient and is provided "as-is" without warranty. In the event of a conflict between Woolpert's sealed contract drawings and these data, the sealed contract drawings shall govern. It is the Recipient's sole responsibility to determine whether a conflicts exists.
- 11. Due to the easily alterable nature of electronic/digital files, Woolpert makes no warranties, either express or implied, with respect to the accuracy, completeness, merchantability, or fitness for any particular purpose, including, but not limited to, performance of electronic/digital files in cost estimating, quantity calculating, survey layout, or other software used by the Recipient, its Client or any other consultant or contractor.
- 12. Use of any data will be at the Recipient's sole risk, and the Recipient releases and waives any and all claims against Woolpert and StayAPT, its successors and assigns, reasonably derivative of such data, and further agrees to indemnify, defend and hold Woolpert, its parents, subsidiaries, affiliates, officers, partners, employees, agents, and lower-tier consultants as well as StayAPT, its parents, subsidiaries, affiliates, officers, partners, employees, agents, and lower tier consultants, harmless from all claims, damages, losses, and expenses including reasonable attorneys' fees and costs of defense actually or allegedly arising out of or resulting from any use of these data.
- 13. This constitutes the entire agreement and understanding of the parties and all other writings, expressions and/or representations, whether orally or in writing, shall be superseded by this Agreement.
- 14. This Agreement shall be governed by and construed in accordance with the laws of Ohio, without regard to conflict of laws principles. Any action brought under this Agreement shall be brought only in a court of competent jurisdiction located in Greene County, Ohio. The parties consent to the exclusive jurisdiction of such courts, and agree to accept service of process by mail.
- 15. Recipient, intending to be bound and obligated as described herein, has designated and authorized its representative, identified below, to execute this Agreement, as of_____, which shall be the effective date of this Agreement.

RECIPIENT:
Firm Name:
Ву:
Printed Name:
Title:

EXHIBIT K

Franchise Application



FRANCHISE APPLICATION

Thank you for your interest in a franchise for stayAPT® Suites. Please complete this application form in its entirety and provide all the additional information requested. A checklist is provided below for your assistance.

CHECKLIST

	Franchise Disclosure Document (FDD) Acknowledgment of Receipt – signed & dated as of the date you received the FDD.		
		cable, a copy of the formation an equested, its parent or sub-entit	d operating documents for the Applicant's legal business entity ies:
	Entity	Туре	<u>Documents</u>
	Corpo	ration:	EIN Number, Articles of Incorporation, Bylaws
	Gener	al Partnership:	EIN Number, Partnership Agreement
	Limite	d Partnership:	EIN Number, Partnership Agreement
	Limited Liability Company:		Certificate of Formation, EIN Number, Articles of Organization, Operating Agreement
	Bank:		Bank Formation documents, Bank Management, and Structure documents
	10% or each ed	greater interest in the Applicant Juity partner owning a 10% or gro	pplicant or, if Applicant is an entity, each equity partner owning a entity. In addition, each Applicant or, if Applicant is an entity, eater interest in the Applicant entity must complete our process (See page 5 of the Application)
	Site dod	cuments, including:	
	 Proof of land/site control (purchase contract, deed, option, or lease) in the name of the Applicant Plat, survey, or site plan showing the legal boundaries of the site Any codes, covenants, or restrictions for development, if available 		
		•	
	Ownership information and organizational chart, including:		
	 Information for each individual franchisee or, if the proposed franchisee is a legal entity, each sole proprietor, partner, general manager, manager, managing tenant in common, and major owner/shareholder (those owning beneficially 10% or more) of the proposed franchisee and its ownership, and any individual that will serve as a guarantor on the Franchise Agreement 		
	0	If the proposed franchisee is	a legal entity, a current organizational chart showing the full

ownership structure for the proposed franchisee

APPLICATION LETTER

LG AS Franchisor LLC
Attn: Franchising

10801 Monroe Road, Suite 200

Matthews, NC 28105



Applicant:	Proposed Location:

This franchise application letter is being provided by the individual(s) or entity listed above ("Applicant") to LG AS Franchisor LLC ("Franchisor") in connection with the consideration and processing of an application for a franchise license to operate a hotel under the stayAPT® Suites brand at the proposed location listed above (the "Hotel").

Applicant understands and acknowledges that Franchisor will rely on the information provided in the franchise application and all documents submitted by Applicant in connection with or in support thereof, including all financial statements and this application letter (collectively, the "Application"). Providing false or misleading information may result in denial of the Application or immediate termination of any resulting agreement(s).

Applicant represents that:

- (a) All information contained in the Application is true, correct and complete as of the date hereof, and no material information or facts have been omitted that would cause the Application to be misleading. Applicant will promptly notify Franchisor of any change in the information provided in the Application.
- (b) Applicant has authority to submit the Application and to enter into a franchise agreement with Franchisor for the Hotel (the "Franchise Agreement") and any such other documents as are contemplated by the Application.
- (c) Neither the Application nor, if approved, the execution of a Franchise Agreement will conflict with any obligations of Applicant, its parent, affiliates, subsidiaries, or owners to other parties. Applicant has not been induced by Franchisor to terminate or breach any agreement with respect to the Hotel or the site.
- (d) Applicant has received information concerning the brand and the system (including the appropriate franchise agreement form), and Applicant is familiar with the requirements. If the Application is approved, Applicant will execute and comply with the terms of the Franchise Agreement.
- (e) Applicant has or will have legal control over the Hotel site, through fee ownership, leasehold, or purchase contract.

Applicant further understands and acknowledges that:

- (a) Franchisor does not enter into oral agreements or understandings with respect to franchises or matters pertaining to the granting of a franchise.
- (b) There are no oral agreements or understandings whatsoever between Applicant and Franchisor with respect to the Application or any proposed franchise.
- (c) A contract or agreement with respect to a proposed franchise will be effective only upon the mutual execution of the Franchise Agreement.
- (d) By submitting the Application, Applicant is authorizing Franchisor to verify the credit history, references, and other financial and background information of the Applicant, the proposed franchisee entity (if different than Applicant), and any co-owners or equity partners holding a 10% or greater interest in Applicant or the proposed franchisee entity, including background checks for U.S. OFAC compliance. Applicant may be asked to answer questions about their credit history.

- (e) If the Application is approved, Applicant will pay an Initial Franchise Fee of \$40,000 at the time of execution of the Franchise Agreement. The Initial Franchise Fee will be fully earned upon receipt and nonrefundable and may be invested, commingled with other funds of Franchisor, or otherwise used by Franchisor as it deems appropriate in its discretion.
- (f) Franchisor reserves the sole right to approve or deny the Application for any reason it may determine. Applicant will not acquire any rights by virtue of its submission of the Application, whether or not approved. In the event Franchisor denies the Application, Franchisor will have no liability to Applicant. Any expenses incurred by or on behalf of Applicant in connection with the Application or approval of the Application, including costs for acquiring property or designing, constructing, or operating a hotel at the proposed location, are at Applicant's sole risk and are not being made in reliance on any action of Franchisor.

Applicant, jointly and severally (if applicable), agrees to indemnify Franchisor and its affiliates, directors, officers, employees, agents, and representatives and to hold them harmless from all losses, consequently, directly or indirectly incurred (including legal and accounting fees and expenses) or arising from, as a result of, or in connection with the Application or Franchisor's reliance on any representation or information therein. Franchisor may independently take any action it deems necessary in its sole discretion, to protect and defend itself against any threatened action, subject to the undersigned's indemnification, without regard to the expense, forum or other parties that may be involved. Franchisor may, in its discretion, have sole and exclusive control over the defense of any such action (including the right to be represented by counsel of its choosing) and over the settlement, compromise or other disposition thereof.

This application letter will be governed by the laws of the state of North Carolina, without reference to its choice of law provisions.

For Washington Prospective Franchisees: The Application does not waive any liability the franchisor may have under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted thereunder.

For Individual Applicant(s):	For Business Entity Applicant:
Signature:	Signature:
Print Name:	Print Name:
Date:	Title:
Signature:	Date:
Print Name:	
Date:	
Signature:	
Print Name:	
Date:	

Section I – The Applicant

APPLICANT INFORMATION

For individual Applicant(s):		
· or maintagar, ppinoam(o)		
(First Name)	(Middle Name)	(Last Name)
(First Name)	(Middle Name)	(Last Name)
(First Name)	(Middle Name)	(Last Name)
For Business Entity Applica	nt:	
Name of Entity		
Entity Type		State of Formation
If approved, the "franch	vill be the "franchisee" on the F hisee" will be an entity related the Franchise Agreement	ranchise Agreement to Applicant (i.e. SPE) that will be legally
	ng the word "stayAPT", or any com	ity name cannot include any of the stayAPT® Suites abination, variation, or abbreviation thereof, or an
Authorized Signer(s) for Ap	pplicant/Franchisee Entity:	
Name	Title	
Name (if more than one)	Title	
Name	Title	
Applicant's Principal Conta	ct (for notices):	
Name		
Street Address		
City	State	7in/Postal Code

Mobile Phone

Fax

Business Phone

Email

Franchise/Business Experience (attach additional pages, if needed): Do you expect to have equity partners? Yes \square Have you previously owned, operated, or franchised a hotel? Yes \square If yes, for each please list the hotel name, brand, whether it was owned, operated or franchised by you, your percentage of ownership (if applicable), and the current status. Have any of your partners previously owned, operated, or franchised a hotel? ☐ Yes☐ No If yes, for each please list the hotel name, brand, whether it was owned, operated or franchised by an equity partner, applicable partner and his/her percentage of ownership (if applicable), and the current status. Have you previously owned or operated any other type of franchised business? If yes, for each please list the brand/concept, your position, and your percentage of ownership. Have any of your equity partners previously owned or operated any other type of Yes franchised business? If yes, for each please list the brand/concept, applicable partner, his/her position and percentage of ownership. Have you or your partners ever developed a new construction property? Yes If yes, for each please describe: Is or has (i) Applicant, (ii) any direct or indirect owner of the hotel franchise for ☐ Yes☐ No which this Application is being submitted, or (iii) the proposed hotel site been the subject of a voluntary or involuntary bankruptcy, receivership, foreclosure or other insolvency proceeding either currently pending or filed within the 3-year period immediately preceding this Application? If yes, please describe:

MANAGEMENT

Anticipated Management of the Hotel:	
☐ Self-managed ☐ Third-Party Management ☐ AS Manager LLC (Franchisor's management entity)	
If Third-Party, Management Name:	
Management Address:	
Management Phone:	
☐ We have an executed Hotel Management Agreement dated	
FINANCING INFORMATION	
Do you have a loan or loan commitment for this project? Yes No	
Name of lender(s):	
Please describe the existing or anticipated financing for this project:	
When do you expect to break ground? Capital available to invest:	

CREDIT REVIEW/BACKGROUND CHECK

Franchisor currently uses a third-party company, BoeFly, Inc., to qualify prospective franchisees. After submitting this Application, Applicant, the proposed franchisee entity (if different than Applicant), and any co-owners or equity partners holding a 10% or greater interest, will receive a link to BoeFly's online bVerify tool. The failure of all required entities and individuals to complete Franchisor's diligence process may result in denial of this Application.

Section II – The Site

PROPERTY DETAILS

Property Address/Interse	ection:						
City:	State:	Zip Code:		Coun	ty:		
Phone (if available):							
Acreage:	Site Cos	st:		Site	e Cost	per S	q. Ft.:
Current Status of Land: [☐ Owned ☐ Leased	☐ Optione	d 🔲 Und	er Contrac	t 🔲 Othe	er:	
If leased, name of lease h	nolder:						
Is the lease holder relate							
If optioned, when does t	he option expire?						
Loan to Value:	Amortization and	Term:		Inte	rest Rate o	on Loan	n:
Is the site currently zone	d for hotel use? ☐ Y	∕es					
Maximum hieght allowed	d by zoning code: Fee	et:	Flooi	rs:			
Your submission of this App acceptance. Franchisor will Franchisor makes no repr representation that you will	convey its acceptance esentation that your s	of the site in site will be a	writing vio	a fully exe	cuted Fran epted, Frai	chise Ag nchisor	greement. makes no
LOCATION DETAILS							
City Population:	Dista	ance to Se	ervices (grocery,	gas, rest	aurant,	etc.):
Local attractions (i.e	e. military bases, –	, healthca	re facil	ities, bu	siness l	hubs,	etc.):
Local Competitors:							
Ho	tel Name		Distance rom Site	Room Count	Rate Rar	nge l	Property Age

	(miles)		
(1)			
(2)			
(3)			
(4)			
(5)			
(6)			
(7)			

BUILD DETAILS

Anticipated Construction Start Date:
Anticipated Construction Completion Date:
Estimated Hotel Opening Date:
Number of Guest Rooms: Number of Floors:
Street Frontage (Sq. Ft.):

Section III – The Ownership and Organization

OWNERSHIP STRUCTURE

Using the example below, please provide information for each Principal Owner, meaning any individual or entity who directly or indirectly owns or will own a 10% or greater interest in the franchise or franchisee entity. For each entity listed, please provide information for each sole proprietor, partner, manager, managing tenant in common, and major owner/shareholder (those owning beneficially 10% or more of the stock), as applicable. Attach additional pages, as needed.

If the franchisee will be a legal entity, please also provide a current organizational chart showing the full ownership structure.

Individual/Entity Name	Type of Interest	% Ownership	Business Address	Phone Number	Email
FRANCHISEE ENTITY: Ha					
Doe Hotels, LLC	General	1%	14 Peachtree Street,	(111) 111-1111	john.doe@xxxxxx.com
	Partner		Suite 400		
			Atlanta, GA 30303		
Hotel Investments, GP	Limited	99%	48 Merryville Road	(222) 222-2222	sjones@xxxxxx.com
	Partner		Raleigh, NC 27560		
		Tota	l: 100%		
OWNER ENTITY: Doe Ho	tels, LLC				
John Doe	Managing	45%	105 Sweet Lane	(333) 333-3333	john.doe@xxxxxx.com
	Member		Sandy Springs, GA		
			30339		
Jane Doe	Member	35%	16 Masters Court	(444) 444-4444	jane.doe@xxxxxx.com
			Decatur, GA 30334		
Julian Doe	Member	20%	73 Robinson Avenue	(555) 555-5555	julian.doe@xxxxxx.co
			Raleigh, NC 27560		m
		Tota	l: 100%		
OWNER ENTITY: Hotel In	nvestments, GF				
Sally Jones	General	60%	9 Joyful Circle	(777) 777-7777	sjones@xxxxxx.com
	Partner		Charlotte, NC 28226		
Tom Warren	General	40%	73 Palmer Road	(888) 888-8888	twarren@xxxxxx.com
	Partner		Durham, NC 27703		
		Tota	l: 100%		

Individual/Entity Name	Type of Interest	% Ownership	Business Address	Phone Number	Email
FRANCHISEE ENTITY:					
		Total	: 100%		
OWNER ENTITY:					
		Total	: 100%		
OWNER ENTITY:					
Total: 100%					

STATE EFFECTIVE DATES

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

California	[Pending]
Hawaii	[Pending]
Illinois	[Pending]
Indiana	[Pending]
Maryland	[Pending]
Michigan	[Pending]
Minnesota	[Pending]
New York	[Pending]
North Dakota	[Pending]
Rhode Island	[Pending]
South Dakota	[Pending]
Virginia	[Pending]
Washington	[Pending]
Wisconsin	[Pending]

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

EXHIBIT L

Receipts

Receipt

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If LG AS Franchisor LLC ("stayAPT") offers you a franchise, stayAPT must provide this disclosure document to you 14 calendar days before you sign a binding agreement with, or make a payment to, stayAPT or its affiliate in connection with the proposed franchise sale. Iowa and New York require that we give you this disclosure document at the earlier of the first personal meeting or 10 business days (or 14 calendar days in Iowa) before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

If stayAPT does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and those state administrators listed on Exhibit A.

The franchisor is LG AS Franchisor LLC located at 10801 Monroe Road, Suite 200, Matthews, North Carolina 28105. Its telephone number is (980) 368-8100.

Issuance Date: May 7, 2025.

StayAPT's franchise sellers involved in offering and selling the franchise to you are listed on the following page (with address and telephone number).

stayAPT authorizes the respective state agencies identified on Exhibit A to receive service of process for us in the particular state.

I have received a disclosure document with an issuance date of May 7, 2025 that included the following Exhibits:

Α.	List of Agents to Receive Service of	r.	State Addenda
Proce	ess	G.	Form of Release of Claims
B.	List of Franchises	H.	Comfort Letter
C.	Financial Statements	I.	Multi-Unit Development Agreement
D.	Franchise Agreement	J.	Data Transfer Agreement
E.	Franchise Standards Manual Table of	K.	Franchise Application
Conte	ents	L.	Receipts
Date:	·		
(Do not leave blank)		Signa	ature of Prospective Franchisee
		Print	Name

Copy for Franchisee

Franchise Sellers:

Please check the box by each franchise seller listed below with whom you have had significant contacts in connection with the offer or sale of a stayAPT franchise to you:

□ Gary DeLapp	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Jennifer Kearney	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Sharon Zinser	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Brad Dew	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Chris Rutherford	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Adam Binder	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Joe Bickelmann	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Will Wildes	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100

Receipt

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If LG AS Franchisor LLC ("stayAPT") offers you a franchise, stayAPT must provide this disclosure document to you 14 calendar days before you sign a binding agreement with, or make a payment to, stayAPT or its affiliate in connection with the proposed franchise sale. Iowa and New York require that we give you this disclosure document at the earlier of the first personal meeting or 10 business days (or 14 calendar days in Iowa) before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

If stayAPT does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and those state administrators listed on Exhibit A.

The franchisor is LG AS Franchisor LLC located at 10801 Monroe Road, Suite 200, Matthews, North Carolina 28105. Its telephone number is (980) 368-8100.

Issuance Date: May 7, 2025.

stayAPT's franchise sellers involved in offering and selling the franchise to you are listed on the following page (with address and telephone number).

stayAPT authorizes the respective state agencies identified on <u>Exhibit A</u> to receive service of process for us in the particular state.

I have received a disclosure document with an issuance date of May 7, 2025 that included the following Exhibits:

A.	List of Agents to Receive Service of	F.	State Addenda
Proce	ss	G.	Form of Release of Claims
B.	List of Franchises	Н.	Comfort Letter
C.	Financial Statements	I.	Multi-Unit Development Agreement
D.	Franchise Agreement	J.	Data Transfer Agreement
E.	Franchise Standards Manual Table of	K.	Franchise Application
Contents		L.	Receipts
Date:			
Do no	et leave blank)	Signa	ature of Prospective Franchisee
		Print	Name

Copy for LG AS Franchisor LLC

Please sign and date both copies of this receipt, keep one copy (the previous page) for your records, and return one to LG AS Franchisor LLC

Franchise Sellers:

Please check the box by each franchise seller listed below with whom you have had significant contacts in connection with the offer or sale of a stayAPT franchise to you:

□ Gary DeLapp	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Jennifer Kearney	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Sharon Zinser	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Brad Dew	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Adam Binder	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Chris Rutherford	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Joe Bickelmann	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100
□ Will Wildes	10801 Monroe, Road, Suite 200, Matthews, North Carolina 28105, 980-368-8100